NOAA Manual NOS NGS 3



# **Geodetic Leveling**

National Geodetic Survey Rockville, Md. August 1981 Reprinted June 2001

U.S. DEPARTMENT OF COMMERCE National Oceanic and Atmospheric Administration National Ocean Survey

( ( 5.2 NOAA Manual NOS NGS 3



# **Geodetic Leveling**

M. Christine Schomaker, Lt., NOAA Ralph Moore Berry

National Geodetic Survey Rockville, Md. August 1981 Reprinted June 20001

Supersedes Coast and Geodetic Survey Special Publication 239, Manual of Geodetic Leveling, 1948.

U.S. DEPARTMENT OF COMMERCE Malcolm Baldrige, Secretary

National Oceanic and Atmospheric Administration John V. Byrne. Administrator

National Ocean Survey H. R. Lippold, Jr., Director

(

.

### NOAA Manual NOS NGS 3

## **GEODETIC LEVELING**

ABSTRACT: This manual furnishes instructions and background information suitable for establishing and maintaining vertical control. It presents general specifications and detailed instructions for (1) reconnaissance and bench mark setting, (2) geodetic leveling, (3) water and valley crossings, and (4) data processing in the field. These instructions should be used by field surveyors engaged in collecting elevation data for inclusion in the National Geodetic Vertical Network.

Mention of a commercial company or product does not constitute an endorsement by the National Oceanic and Atmospheric Administration. Use for publicity or advertising purposes of information from this publication concerning proprietary products or the tests of such products is not authorized

(

Ć

## CONTENTS

Chapter 1.	VERTICAL CONTROL NETWORKS
	1.2 Development of Geodetic Leveling
	1.3 Height Systems
	1.4 National Geodetic Vertical Control Network
hapter 2.	RECONNAISSANCE AND BENCH MARK SETTING
	2.2 Organizing a Mark setting Unit
	2.2.1 TOISOUMCI VALUEL LEADER CALENCE ALCONCEAL CONCEAL CO
	2.2.2 Equipinent excension over here and a conservation of the con
	2.2.3 Salely statistic measure of a statistic statistic strategies and the statistic statistic strategies and the statistic strategi
	2.2 Tropolds
	2.5 Rolling a Lille for Lovening and an and a service of the servi
	2.3.2.1.1.10mmillary Data - a conversion of the
	2.2.2 Edulouting Specifications
	2.3.4 Connections
	2354 Comilections. All references and and an and
	2.3.3 ACCOLD STREAM AND THE STREAM AND A CONTRACT AND A
	2.4 1 Control Doint Decovery
	2.4.1 Control Funit Recovery interesting
	2.4.2 Deficit Mark Setting
	2.4.5 Assigning Ocographic Positions
	2.4.4 Descriptions control Dainta Dainta
Chapter 3.	GEODETIC LEVELING
•	3.1 Introduction
	3.1.1 General Procedures
	3.1.2 Sources of Error
	3.1.3 Tolerances
	3.2 Organizing a Leveling Unit
	3.2.1 Personnel
	3.2.2 Equipment
	3 7 3 Safety
	3) 4 Renorts
	3 Leveling Instruments
	3.3.1 Cloceification
	J.J. Telecone Suctor
	3.3.2.1 Closuppe Dystelling and an
	J.J.J. Compensation Systemi and concerned and concerned and concerned and concerned and and an experimental and the concerned and an experimental and the concerned and and the conc
	5.5.4 Optical information interaction of the second and the second and the second second second and the second secon
	3.5.5 Use and Maintenance automatic automatic automatic automatical automatic
	3.4 Leveling Rods
	3.4.2 Calibration at a reason of the second statement of t
	3.4.3 Use and Maintenance
	3.5 Turning Points
	3.5.1 Selection and Use contraction of the second s
	3.5.2 Pacing Balanced Setups
	3.6 Atmospheric Conditions
	261 Air Tomporeture Sun and Wind
	5.0.1 AIT Temperature, Sun, and wind seven-constructions and seven construction and the seven construction of the seven co

### **CONTENTS**—Continued

INDEX		I-1	
APPENDIX C.	STANDARD GEODETIC LEVELING FORMS	C-1 (	
APPENDIX B.	INDEX OF GEODETIC CONTROL DIAGRAMS	B-1	<i>,</i> .
APPENDIX A.	LISTS OF EQUIPMENT	A-1	
	5.6 References	5-27	
	5.5.2 Computing Corrections	5-24 5-26	
	5.5.1 Quality Control Review	5-23	
	5.5 Verification and Adjustment	5-23	
	5 4.4 Statistics	5-20	
	5.4.2 Checking the Field Abstract	5-15 5-20	
	5.4.1 Checking Observation Records	5-13	
	5.4 Processing Leveling Data	5-13	
	5.3.3 Checking the Entire File	5-10	
	5.3.2 Checking Individual Descriptions	5-7	
	5.3.1 Creating the Description File	5-7 5-7	
	5.2.4 Submitting Data to the National Geodetic Survey	5-3 5_7	
	5.2.3 Organizing the Data	5-2	
	5.2.2 Equipment for Data Processing	5-1	
	5.2.1 Personnel	5-1	
	5.2 Organizing a Project Office	5-1	
Unapter 5.	DATA FRUCESSING	D-1 5_1	
<b>C1</b>	DATA DROCECCINC	<i>E</i> 1	
	4.5 Reference	<b>4-22</b> (	r
	4.4.2 Observing Routine for Two Targets (Three-wire Leveling)	4-18	
	4.4.1 Equipment and Setups	4-17	
	4.4 Instructions for Ordinary Leveling Instruments	4-17	
	4.5.5 Observing Kouline	4-10 4.16	
	4.3.2 Setups	4-7	
	4.3.1 Description of Equipment	4-5	
	4.3 Instructions for Zeiss River Crossing Equipment .	4-5	
	4.2.4 Environmental Conditions	4-5	
	4.2.3 Equipment	4-5	
	4.2.1 Reconnel	4-4	
	4,2 Preparations	4-3	
	4.1 Introduction	4-1	
Chapter 4.	RIVER OR VALLEY CROSSING	4-1	
	3.9 References	3-72	
	3.8.4 Submitting Records to the Project Office	3-72	
	3.8.3 Field Abstract	3-69	
	3.8.2 Computer Recording Equipment	3-68	
	3.8 Field Records and the internet of the inte	3-00 3.66	
	3.7.7 Releveling and Closing Sections	3-65	
	3.7.6 Tide and Water-level Stations	3-64	
	3.7.5 Leveling to Awkward Control Points	3-62	
	3.7.4 Precautions to Take at Control Points	3-59	
	3.7.3 Three-wire I eveling	3-57	
	3.7.1 General Instructions	3-31	
	3.7 Observing Routine	3-51	
		<b>.</b>	

(

### FIGURES

Chapter 1		
Figure 1-1	Orthometric height, H, on a surface not parallel to the reference surface	-3
Figure 1-2	Observed difference in orthometric height, $\Delta H$ , depends on the leveling route 1.	-3
Figure 1-3.	Basic net A of the National Geodetic Vertical Network	-6
Chapter 2		
Figure 2-1.	Mark setting monthly report and a set of the	-3
Figure 2-2.	Scale relationships of preliminary data 2	-5
Figure 2-3.	Example of line assignments in project instructions and a second	-5
Figure 2-4.	State Index Map of Control Leveling Constrained and the second se	-6
Figure 2-5.	Geodetic Control Diagram	-7
Figure 2-6	Index to I opographic Maps, project lines plotted	-8
Figure 2-/a	Index plot for 30' quadrant of Vertical Control Data	10
Figure 2-76.	Vertical Control Data sheet	i I
Figure 2-7C.	Alphanumeric indexes for 30' quadrant of Vertical Control Data	12
Figure 2-/d	Line order index for 30° quadrant and the second se	13
Figure 2-8a		14
Figure 2-80.	Listing of synoptic file when the treatment of the second se	15
Figure 2-9a	Form letter distributed to property owners	17
Figure 2-90	2-1	18
Figure 2-10.	Network junctions weather failed and the feature of the second se	19
Figure 2-11.	Locations for monuments of the best possible quality	20
Figure 2-12.	Connections to parallel surveys and a second	22
Figure 2-13	Connections to intersected surveys	23
Figure 2-14.	Connections to an area survey and a construction of the second se	23
Figure 2-15	Map prepared on a working plot	<u>14</u>
Figure 2-16.	Example of bench mark designations assigned by mark setter assessment and the setter assigned by mark setter as a	20
Figure 2-17	Foundation types of massive orloges	11
Figure 2-18.		28
Figure 2-19.	Stainiess steel cap for a rod mark and a second second and a second seco	28
Figure 2-20.	Souling latitude D 50 LISES in 42°01/22/2 NI	5U 51
Figure 2-21.	Scaling landinde: D 59 USLS IS 42 UI 33 IN DEPARTMENT AVERAGE FOR THE DEPARTMENT 2-5	51 55
Figure 2-22.	Scaling longitude: D 39 USLS is 83 13 32 W	52
Figure 2-25		5.5 1 <i>5</i>
Figure 2-24.		10
Figure 2-25a		10 10
Figure 2-200.	Recovery note and a contraction of the former U.S. Coast and Goodetic Survey distribution of the former U.S. Coast and Goodetic Survey distribution of the former U.S. Coast and Goodetic Survey distribution of the former U.S. Coast and Goodetic Survey distribution of the former U.S. Coast and Goodetic Survey distribution of the former U.S. Coast and Goodetic Survey distribution of the former U.S. Coast and Goodetic Survey distribution of the former U.S. Coast and Goodetic Survey distribution of the former survey	)/ 47
Figure 2-20.	Survey disks of the National Ocean Survey and National Conducto Survey	+2. 4 7
Figure 2.22	Benort on the condition of a survey point 2-4	+3 45
Figure 7.70	Report of a relocated heach mark	1) 14
Tigute 2-29.		10
Chapter 3		
Figure 3-1.	Setup of leveling, $\Delta h = B - F$ and $s = s_p + s_r$ .	-1
Figure 3-2	Section of leveling, $\Delta H = \Delta h_1 + \Delta h_2 + \dots + \Delta h_n$ and $S = s_1 + s_2 + \dots + s_n$ .	-2
Figure 3-3.	Effect of collimation error, $\alpha$ 3-	-2
Figure 3-4	Consistent collimation error cancels in a balanced setup since $s_p = s_r$ .	-2
Figure 3-5.	Inconsistent collimation error does not cancel in a balanced setup since $\alpha_{p} \neq \alpha_{p}$	
U	even if $s_p = s_r$ , so the transformer product of the second construction of the second construc	-3
Figure 3-6.	Shimmer	-3
Figure 3-7.	Refraction error, r, does not cancel on sloping terrain since $r_p \neq r_p$ , even if $s_p = s_p$ and $3$ -	-3
Figure 3-8	Curvature error, c, where the line of sight is not parallel to an equipotential surface,	
-	cancels if $s_p = s_F$ and a submatrix constraints of the second state of the second	-4
Figure 3-9.	Effect, a, of tidal deflection, $\epsilon$ , on a section of length and direction, $\vec{S}$	-4
Figure 3-10.	Effect of rod plumbing error, $\rho$ , on a height observation, <b>B</b>	-4
Figure 3-11	Systematic effect of plumbing error (and scale errors) is small on flat terrain,	
	since $B_1 \simeq F_2$ and $F_1 \simeq B_2$	-5
Figure 3-12.	Effect of plumbing error (and scale errors) accumulates on sloping terrain,	
	since $B_1 \neq F_2$ and $F_1 \neq B_2$ .	-5
Figure 3-13.	Controlling traffic	0
Figure 3-14	Leveling unit weekly report	1

(

 $\left( \right)$ 

Ć

Figure 3-15.	Leveling unit monthly report	3-12
Figure 3-16	Fischer level	3-13
Figure 3-17.	Jenoptik Ni004	3-13
Figure 3-18.	Zeiss Ni2, with micrometer attachment	3-14
Figure 3-19	Zeiss Nil	3-14
Figure 3-20.	MOM NIA31	3-14
Figure 3-21	Jenoptik NI 002	3-15
Figure 3-22.	External-focusing telescope	3-16
Figure 3-23.	Internal-focusing telescope	3-16
Figure 3-24.	Ni2 compensator system, telescope axis horizontal	3-17
Figure 3-25.	Ni2 compensator system, telescope axis tilted	3-18
Figure 3-26.	Deflection of the line of collimation by the compensator	3-18
Figure 3-27.	Condition for dynamic stability of the Ni2	3-19
Figure 3-28.	Nil compensator system, telescope axis horizontal	3-19
Figure 3-29.	Nil compensator system, telescope axis tilted	3-20
Figure 3-30	Diagram of the NI 002	3-21
Figure 3-31.	Lines of sight provided by the NI 002	3-22
Figure 3-32.	Snell's law applied to a plane-parallel plate micrometer	3-23
Figure 3-33.	Plane-parallel plate unrotated	3-23
Figure 3-34.	Plane-parallel plate rotated to intercept a graduation	3-24
Figure 3-35	Plane-parallel plate micrometer scale	3-25
Figure 3-36.	Carrying the instrument-tripod assembly	3-25
Figure 3-37	Centering the bubble in a circular level	3-25
Figure 3-38.	Wedge reticle, graduation not centered	3-26
Figure 3-39a/b.	Wedge reticle, graduation centered	3-27
Figure 3-40.	Relationship of stadia interval to sighting distance	3-28
Figure 3-41	Stadia factor determination, micrometer procedure, half-centimeter rods	3-30
Figure 3-42	Stadia factor determination, three-wire procedure, centimeter rods	3-30
Figure 3-43	Collimation check, Kukkamaki's method	3-32
Figure 3-44.	Collimation check, Kukkamaki's method, micrometer procedure, half-centimeter rods,	
-	reversible compensator	3-33
Figure 3-45.	Collimation check, Kukkamaki's method, micrometer procedure, centimeter rods,	
-	nonreversible compensator	3-34
Figure 3-46.	Collimation check Kukkamaki method three-wire procedure	
+	commation check, Rukkamaki methou, mice who procedure,	
	half-centimeter rods	3-35
Figure 3-47.	half-centimeter rods	3-35 3-36
Figure 3-47. Figure 3-48	Collimation check, 10-40 method, micrometer procedure, half-centimeter rods,	3-35 3-36
Figure 3-47. Figure 3-48	Collimation check, 10-40 method Collimation check, 10-40 method Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator	3-35 3-36 3-37
Figure 3-47. Figure 3-48. Figure 3-49.	Collimation check, 10-40 method Collimation check, 10-40 method Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator Collimation check, 10-40 method, micrometer procedure, centimeter rods,	3-35 3-36 3-37
Figure 3-47. Figure 3-48. Figure 3-49.	<ul> <li>Collimation check, 10-40 method</li> <li>Collimation check, 10-40 method</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> </ul>	3-35 3-36 3-37 3-37
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50.	<ul> <li>Collimation check, 10-40 method</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods, nonreversible compensator</li> </ul>	3-35 3-36 3-37 3-37 3-38
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51.	<ul> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator.</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator.</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods, nonreversible compensator.</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods.</li> </ul>	3-35 3-36 3-37 3-37 3-38 3-39
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52.	<ul> <li>half-centimeter rods</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods, compensation check for the Ni1 and half-centimeter rods</li> <li>Compensation check, micrometer leveling, half-centimeter rods</li> </ul>	3-35 3-36 3-37 3-37 3-38 3-39 3-40
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53.	<ul> <li>collimation check, 10-40 method, micrometer procedure, half-centimeter rods,</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods,</li> <li>reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods,</li> <li>nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods,</li> <li>compensation check for the Ni1 and half-centimeter rods</li> <li>Compensation check, micrometer leveling, half-centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> </ul>	3-35 3-36 3-37 3-38 3-39 3-40 3-40
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54.	<ul> <li>half-centimeter rods</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods, compensation check for the Ni1 and half-centimeter rods</li> <li>Compensation check, micrometer leveling, half-centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> <li>Inverted scale for a leveling rod</li> </ul>	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55.	<ul> <li>half-centimeter rods</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods.</li> <li>Compensation check for the Nil and half-centimeter rods.</li> <li>Compensation check, micrometer leveling, half-centimeter rods.</li> <li>Compensation check, micrometer leveling, centimeter rods.</li> <li>Inverted scale for a leveling rod.</li> <li>Double scale with half-centimeter line graduations</li> </ul>	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56.	Collimation check, 10-40 method         Collimation check, 10-40 method         Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, three-wire procedure, centimeter rods, compensation check for the Nil and half-centimeter rods         Compensation check, micrometer leveling, half-centimeter rods         Compensation check, micrometer leveling, centimeter rods         Inverted scale for a leveling rod         Double scale with half-centimeter line graduations	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42 3-42
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-57.	<ul> <li>half-centimeter rods</li> <li>Collimation check, 10-40 method</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods</li> <li>Compensation check for the Nil and half-centimeter rods</li> <li>Compensation check, micrometer leveling, half-centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> <li>Inverted scale for a leveling rod</li> <li>Double scale with half-centimeter line graduations</li> <li>Brace poles for a leveling rod</li> </ul>	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42 3-42 3-42 3-43
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-57. Figure 3-58.	<ul> <li>half-centimeter rods</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods</li> <li>Compensation check for the Ni1 and half-centimeter rods</li> <li>Compensation check, micrometer leveling, half-centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> <li>Inverted scale for a leveling rod</li> <li>Double scale with half-centimeter line graduations</li> <li>Brace poles for a leveling rod</li> <li>Rod calibration data</li> </ul>	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42 3-42 3-42 3-43 3-43
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59.	<ul> <li>half-centimeter rods</li> <li>Collimation check, 10-40 method</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods</li> <li>Compensation check for the Ni1 and half-centimeter rods</li> <li>Compensation check, micrometer leveling, half-centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> <li>Inverted scale for a leveling rod</li> <li>Double scale with half-centimeter line graduations</li> <li>Brace poles for a leveling rod</li> <li>Rod calibration data</li> <li>Carrying the leveling rod</li> </ul>	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-40 3-41 3-42 3-42 3-42 3-43 3-43 3-44
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59. Figure 3-60.	collimation check, 10-40 method	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-40 3-41 3-42 3-42 3-42 3-43 3-43 3-44 3-44
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59. Figure 3-60. Figure 3-61.	collimation check, 10-40 method         Collimation check, 10-40 method         Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, three-wire procedure, centimeter rods         Compensation check, 10-40 method, three-wire procedure, centimeter rods         Compensation check, norrometer leveling, half-centimeter rods         Compensation check, micrometer leveling, centimeter rods         Compensation check, micrometer leveling, centimeter rods         Inverted scale for a leveling rod         Double scale with half-centimeter line graduations         Brace poles for a leveling rod         Rod calibration data         Carrying the leveling rod         Turning-point guide	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-40 3-41 3-42 3-42 3-42 3-43 3-43 3-44 3-44 3-45
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59. Figure 3-60. Figure 3-61. Figure 3-62.	collimation check, 10-40 method         Collimation check, 10-40 method         Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, three-wire procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, three-wire procedure, centimeter rods.         Compensation check, normeter leveling, half-centimeter rods.         Compensation check, micrometer leveling, centimeter rods.         Compensation check, micrometer leveling, centimeter rods.         Compensation check, micrometer leveling, centimeter rods.         Inverted scale for a leveling rod         Double scale with half-centimeter line graduations         Leveling rods with block graduations         Brace poles for a leveling rod         Rod calibration data         Carrying the leveling rod         Turning-point guide.         Setup of leveling rod         Tracings for checking the adjustment of the circular level	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42 3-42 3-42 3-43 3-43 3-44 3-44 3-45 3-46
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59. Figure 3-60. Figure 3-61. Figure 3-62. Figure 3-63.	collimation oncerk, funktional method, three whe procedure, half-centimeter rods.         Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator.         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator.         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator.         Collimation check, 10-40 method, three-wire procedure, centimeter rods.         Collimation check, 10-40 method, three-wire procedure, centimeter rods.         Compensation check, norrometer leveling, half-centimeter rods.         Compensation check, micrometer leveling, centimeter rods.         Compensation check, micrometer leveling, centimeter rods.         Compensation check, micrometer leveling, centimeter rods.         Inverted scale for a leveling rod         Double scale with half-centimeter line graduations         Leveling rods with block graduations         Brace poles for a leveling rod         Carrying the leveling rod         Turning-point guide.         Setup of leveling rod         Tracings for checking the adjustment of the circular level         Turning pin (a) and hammer (b).	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42 3-42 3-42 3-43 3-43 3-44 3-44 3-45 3-46 3-47
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59. Figure 3-60. Figure 3-61. Figure 3-62. Figure 3-63. Figure 3-64.	<ul> <li>Collimation check, 10-40 method</li> <li>Collimation check, 10-40 method</li> <li>Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator</li> <li>Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator</li> <li>Collimation check, 10-40 method, three-wire procedure, centimeter rods.</li> <li>Compensation check for the Ni1 and half-centimeter rods</li> <li>Compensation check, micrometer leveling, half-centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> <li>Compensation check, micrometer leveling, centimeter rods</li> <li>Inverted scale for a leveling rod</li> <li>Double scale with half-centimeter line graduations</li> <li>Leveling rods with block graduations</li> <li>Brace poles for a leveling rod</li> <li>Rod calibration data</li> <li>Carrying the leveling rod</li> <li>Turning-point guide.</li> <li>Setup of leveling rod</li> <li>Tracings for checking the adjustment of the circular level</li> <li>Turning pin (a) and hammer (b).</li> </ul>	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42 3-42 3-43 3-43 3-44 3-44 3-45 3-46 3-47 3-47
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59. Figure 3-60. Figure 3-61. Figure 3-62. Figure 3-63. Figure 3-64. Figure 3-65.	communities       continue of the control of the circular level         collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator       collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator       collimation check, 10-40 method, three-wire procedure, centimeter rods, nonreversible compensator         Collimation check, 10-40 method, three-wire procedure, centimeter rods       compensation check, nicrometer leveling, half-centimeter rods         Compensation check, micrometer leveling, centimeter rods       compensation check, micrometer leveling, centimeter rods         Compensation check, micrometer leveling, centimeter rods       compensation check, micrometer leveling, centimeter rods         Leveling rod south block graduations       leveling rod         Double scale with half-centimeter line graduations       leveling rod         Rod calibration data       carrying the leveling rod         Turning-point guide       carrying the adjustment of the circular level         Turning pin (a) and hammer (b).       truning pin dimensions.         Turning pin dimensions       the circular level	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-41 3-42 3-43 3-43 3-43 3-43 3-44 3-44 3-45 3-46 3-47 3-47 3-48
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-58. Figure 3-59. Figure 3-60. Figure 3-61. Figure 3-63. Figure 3-64. Figure 3-65. Figure 3-66.	Collimation check, 10-40 method, micrometer procedure, half-centimeter rods,         Collimation check, 10-40 method, micrometer procedure, half-centimeter rods,         reversible compensator         Collimation check, 10-40 method, micrometer procedure, centimeter rods,         nonreversible compensator         Collimation check, 10-40 method, three-wire procedure, centimeter rods,         nonreversible compensator         Collimation check, 10-40 method, three-wire procedure, centimeter rods.         Compensation check for the Ni1 and half-centimeter rods         Compensation check, micrometer leveling, half-centimeter rods         Compensation check, micrometer leveling, centimeter rods         Compensation check, micrometer leveling, centimeter rods         Double scale for a leveling rod         Double scale with half-centimeter line graduations         Leveling rods with block graduations         Brace poles for a leveling rod         Rod calibration data         Carrying the leveling rod         Turning-point guide.         Setup of leveling rod         Turning pin (a) and hammer (b).         Turning pin dimensions.         Turning pin dimensions.         Turning pin dimensions.	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-40 3-41 3-42 3-43 3-43 3-43 3-43 3-44 3-44 3-45 3-46 3-47 3-48 3-49
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-56. Figure 3-56. Figure 3-59. Figure 3-60. Figure 3-61. Figure 3-62. Figure 3-63. Figure 3-64. Figure 3-65. Figure 3-66. Figure 3-67.	Commutation intext, italication intervention, interventinterventervention, interventinterventervention, interve	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-40 3-40 3-41 3-42 3-42 3-43 3-44 3-44 3-44 3-45 3-46 3-47 3-46 3-47 3-48 3-49 3-49
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-56. Figure 3-56. Figure 3-60. Figure 3-60. Figure 3-61. Figure 3-62. Figure 3-63. Figure 3-64. Figure 3-65. Figure 3-66. Figure 3-67. Figure 3-68.	Collimation check, 10-40 method, micrometer procedure, half-centimeter rods. Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator. Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator Collimation check, 10-40 method, three-wire procedure, centimeter rods. Compensation check for the Ni1 and half-centimeter rods. Compensation check, micrometer leveling, half-centimeter rods. Compensation check, micrometer leveling, centimeter rods. Compensation check, micrometer leveling, centimeter rods. Inverted scale for a leveling rod Double scale with half-centimeter line graduations Leveling rods with block graduations Brace poles for a leveling rod Turning-point guide. Setup of leveling rod Turning pin (a) and hammer (b). Turning pint (a) and hammer (b). Turning plate Hand level Pacing a balanced setup on a slope . Thermometer, mounted on tripod.	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-40 3-40 3-42 3-42 3-42 3-43 3-44 3-44 3-44 3-45 3-46 3-47 3-47 3-48 3-49 3-49 3-49 3-49
Figure 3-47. Figure 3-48. Figure 3-49. Figure 3-50. Figure 3-51. Figure 3-52. Figure 3-53. Figure 3-54. Figure 3-55. Figure 3-56. Figure 3-59. Figure 3-60. Figure 3-61. Figure 3-62. Figure 3-63. Figure 3-64. Figure 3-65. Figure 3-66. Figure 3-68. Figure 3-69.	Collimation check, 10-40 method, micrometer procedure, half-centimeter rods. Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator. Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator Collimation check, 10-40 method, three-wire procedure, centimeter rods. Compensation check for the Ni1 and half-centimeter rods. Compensation check, micrometer leveling, half-centimeter rods. Compensation check, micrometer leveling, half-centimeter rods. Compensation check, micrometer leveling, centimeter rods. Inverted scale for a leveling rod Double scale with half-centimeter line graduations Leveling rods with block graduations Brace poles for a leveling rod Turning-point guide. Setup of leveling rod Tracings for checking the adjustment of the circular level Turning pin (a) and hammer (b). Turning pint (a) and hammer (b). Turning plate Hand level Pacing a balanced setup on a slope Thermometer, mounted on tripod Aspirated thermistors for measuring air-temperature differences	3-35 3-36 3-37 3-38 3-39 3-40 3-40 3-40 3-41 3-42 3-42 3-42 3-42 3-43 3-44 3-44 3-44

### **FIGURES-Continued**

Figure 3-71.	Micrometer leveling, half-centimeter rods	3-55
Figure 3-72.	Micrometer leveling, centimeter rods	3-55
Figure 3-73.	Three-wire leveling, half-centimeter rods	3-58
Figure 3-74	Three-wire leveling, centimeter rods	3-58
Figure 3-75.	Leveling routes through a cluster of control points at a triangulation station (a) and a tide station (b)	3-60
Figure 3-76	Control points on horizontal surfaces (a) and vertical surfaces (b)	3-61
Figure 3-77	When using spacers the height e of each spacer cancels when the elevation difference	5.01
1.84.0 3 11/	At is computed, $\Delta h - (R+q) - (F+q) = R-F$	3-62
Figure 3-78	Leveling to awkward control points	3.62
Figure 3.79	Leveling directly to a control point	3-62
Figure 3-80	Short scale	3-64
Figure 3-81	Centering the index line of a short scale on a control point	3-64
Figure 3-87	Leveling to an uncide-down scale	3.65
Figure 3-83	Leveling line identification	3-70
Figure 3-84	Sketch of cingle-run leveling	3-70
Figure 3.85	Abstract of single-run leveling	3-71
Figure 3-86	Sketch of double-run leveling	3_71
Figure 3-87	Abstract of double_run leveling	3_71
Tigure 5-07.		.J-11
Chapter 4		
Figure 4-1	Curvature, c, refraction, r, and the effect, $C \times s$ , of collimation error, $\alpha$ ,	
-	are magnified by a long sighting distance	4-1
Figure 4-2	Intercepting two targets to reduce pointing error and the second se	4-2
Figure 4-3.	The effects of curvature, c, cancel if a pair of reciprocal setups are observed	
-	and the results, $\Delta h_A$ and $\Delta h_B$ , are meaned	4-3
Figure 4-4.	The effects of refraction, r, cancel when the results of a set of simultaneous, reciprocal	
-	observations are meaned; however, the collimation error of one instrument, $\alpha_1$ ,	
	is only partly offset by the collimation error of the other instrument, $\alpha_2$	4-3
Figure 4-5.	Simultaneous reciprocal observations with four leveling instruments,	
_	the two instruments on each side having reciprocal collimation errors	4-4
Figure 4-6.	Rotary wedge, unrotated (a) and rotated (b)	4-6
Figure 4-7.	Rotary wedge scale	4-7
Figure 4-8.	Target column	4-7
Figure 4-9	Targets mounted on target column	4-8
Figure 4-10.	Using the auxiliary scale	4-8
Figure 4-11.	Setups for Zeiss River Crossing Equipment, top view (a) and side view (b).	
	$\Delta h = \frac{1}{2} \left[ \left( \Delta h_{\text{Series A}} + \Delta h_{\text{Leveling B}} \right) - \left( \Delta h_{\text{Series B}} + \Delta h_{\text{Leveling A}} \right) \right]$	4-9
Figure 4-12.	Instruments mounted on base plate	4-10
Figure 4-13.	Leveling from the nearby target station to the nearby bench mark,	
	forward (a) and backward (b)	4-12
Figure 4-14	Annotated recording form for the series of observations from the nearby bench mark	
	to the opposite target station and the presence of the presenc	4-13
Figure 4-15.	Series from the nearby bench mark to the opposite target station construction and the series of the	4-14
Figure 4-16.	Reciprocal collimation, before (a) and alter (b)	4-15
Figure 4-17.	Backsight, wedge settings on rod graduations	4-15
Figure 4-18.	Foresight, wedge settings on targets and an and a setting of the s	4-16
Figure 4-19.	Computation of river crossing with Zeiss River Crossing Equipment	4-18
Figure 4-20.	Setups for simultaneous reciprocal observations with ordinary leveling instruments,	
	top view (a) and side view (b). $\Delta H = \frac{1}{2} (\Delta h_{\text{Set I}} + \Delta h_{\text{Set 2}})$	4-19
Figure 4-21.	Fischer level, with micrometer tilting screw	4-20
Figure 4-22.	One side of a set of simultaneous reciprocal observations, three-wire leveling	4-21
Figure 4-23.	Cover sheet for a running with ordinary equipment	4-22
Chapter 5		
Figure 5-1	Sample label for project records	5-3
Figure 5-2.	Sample memorandum releasing data	5-4
Figure 5-3a	Sample form (NOAA 61-29) accompanying transmitted data	5-5
Figure 5-3b.	Sample form (NOAA 61-29) accompanying transmitted data	5-6
Figure 5-4	Flow chart of project data processing	5-8
Figure 5-5.	Flow chart of data files for a line of leveling	5-9

### **FIGURES-Continued**

Figure 5-6.	Sample check-off list for station-description editing	5-10
Figure 5-7.	Example of a proofread station description and an	5-10
Figure 5-8.	Example of description of file status sheet	5-11
Figure 5-9	Examples of apparent position errors	5-12
Figure 5-10	Example of observation file status sheet	5-14
Figure 5-11.	Example of abstract status sheet	5-16
Figure 5-12.	Sample of computer-produced field abstract	5-18
Figure 5-13.	Samples of incorrect survey-point serial numbers.	
-	Mismatched survey-point serial numbers are underlined	5-19
Figure 5-14	Example of comparisons between new-minus-old observed elevations	5-21
Figure 5-15	Sample plot of new-minus-old values for observed elevations	5-22
Figure 5-16	Sample sketch for checking a loop	5-22
Figure 5-17.	Combining loops	5-23
Figure 5-18.	Sample corrections applied in phase 1 of office abstract	5-25
Figure 5-19	Sample profile of a leveling line	5-27

Ć

Ć

### TABLES

Chapter 1 Table 1-1.	Classification of vertical control networks	1-5
Chapter 2		
Table 2-1	Preliminary data for reconnaissance and mark setting	2-4
Table 2-2.	Location codes for tide and water-level stations	2-16
Table 2-3.	Spacing control points along slopes	2-20
Table 2-4	Requirements for connections.	2-21
Table 2-5.	Standard entries for descriptions	2-34
Table 2-6.	Examples of bench mark designations.	2-39
Table 2-7	Agency classification codes	2-40
Table 2-8	Control classification codes	2-41
Table 2-9	Monument classification codes	2-41
Table 2-10	Bench mark classification codes	2-42
Chapter 3		
Table 3-1.	Tolerances for geodetic leveling	3-7
Table 3-2.	Features of some geodetic leveling instruments	3-15
Table 3-3	Refraction and curvature errors in a single sight	3-31
Table 3-4.	Checklist for leveling	3-52
Table 3-5	Summary of the micrometer leveling procedure	3-56
Table 3-6	Summary of the three-wire leveling procedure	3-59
Table 3-7.	Example of rejection procedure for first-order, class II leveling	3-67
Table 3-8.	U.S. Navy time zone designations	3-67
Chapter 4		
Table 4-1.	Summary of the crossing routine for Zeiss River Crossing Equipment	4-11
Table 4-2.	Data set for a complete crossing with Zeiss River Crossing Equipment	4-17
Table 4-3.	Summary of the crossing routine for ordinary leveling instruments	4-21
Appendix B		
Table B-1.	Codes for NGS Geodetic Control Diagrams.	B-1

## Chapter 1 VERTICAL CONTROL NETWORKS

### **1.1 Introduction**

Since ancient times, the ability to locate widely separated points on the Earth's surface has been of vital interest to public commerce and defense. Accurate positioning has become increasingly important to construction, boundary definition, navigation, and the prediction and monitoring of changes in the topography. To provide accurate positions over the entire globe, a detailed knowledge about the size and shape of the Earth is critical. The pursuit of this knowledge is the science of geodesy.

Traditionally, points on the surface of the Earth have been located by assigning geographic positions and elevations. Geographic position (latitude and longitude) is determined by surveying techniques such as triangulation, trilateration, and traverse, which determine horizontal distances and directions between points. Elevations are obtained by techniques that determine vertical differences between points. These include differential leveling, trigonometric leveling, and observing changes in atmospheric pressure. Modern satellite and inertial systems determine geometric distances, in three dimensions, that can be converted to both geographic positions and elevations above an ellipsoid.

Because of the differences in surveying techniques, separate networks of horizontal and vertical control have evolved. However, all geodetic networks have in common the fact that measurements are made with instruments oriented to the Earth's gravity field.

An interconnected system of points, each of which is assigned an elevation referred to a common surface, is called a vertical control network. Since geodetic leveling has in the past provided, and continues to provide, the most accurate means for measuring precise elevation differences, a vertical network typically consists of lines of control points, reflecting the progression of leveling from point to point.

Vertical control networks provide elevations for many purposes: from localized construction projects to studies of widespread motions of the Earth's crust. To establish a sufficiently accurate national network, survey techniques must be of geodetic quality: they must employ accurate equipment, maintain a high degree of precision when extended over large areas, and adequately define the gravity field.

### **1.2 Development of Geodetic Leveling**

Leveling was successfully employed by the ancient Egyptians who attempted to connect the Nile River with the Red Sea, by the Babylonians who constructed an extensive irrigation system in the Euphrates Valley, and by the Romans who constructed extensive networks of aqueducts, both in Rome and such regions as Spain and the Middle East. In these cases, however, leveling depended on crude instruments that had provision for sighting along a water surface or that operated by some mechanical application of a plumb line.

The development of geodetic leveling as it is conducted today depended on the invention of three important items: the telescope, the reticle, and the level

The invention of the telescope in 1608 is attributed to Lippershey, a Dutch spectacle maker. It was first used for scientific purposes by Galileo Galilei in 1609 for magnifying the image of a distant object, but it was not very useful as a pointing device until the introduction of the reticle.

The invention of the reticle, which provides "cross hairs" at the common focus of the objective lens and the ocular lens, was not possible until the invention of the "positive" ocular lens by Johann Kepler in 1611 and the actual placement of a measuring device at the common focus by the English astronomer William Gascoigne in 1639 In 1669, while working on a project to measure the length of a degree of latitude for the Royal Academy of France, Jean Picard first placed a reticle in a surveying instrument.

The invention of the level—a tube of glass with fluid sealed inside in such a way that a bubble is formed—is credited to Melchisedech Thevenot, who published information on instrumental details and manufacturing methods in 1666. Nearly a hundred years later, after procedures were perfected to manufacture level vials with uniform curvature, the telescope, reticle, and level were assembled in a spirit-level instrument similar to the one still used today in construction work. It is believed that such instruments were devised independently by Antoine de Chezy, a road and bridge engineer in France, and by Jesse Ramsden, a mathematical instrument maker in England.

Further developments in leveling are evident in early texts, such as Love's classical *Geodaesia* published in 1760, which recommends, under the heading "How to know whether water may be made to run from a Spring head to any appointed place," that: "It is better to get a water-level, such as you may buy at the Instrument-Makers." Later texts, such as the *Complete Manual of Leveling* by Simms, published in England in 1836, treat subjects such as corrections for curvature and refraction, and the calculation of earthwork from cross sections observed with a spirit-level instrument. References are made to both the "Y" level and the dumpy level, instrumental designs which are familiar to twentiethcentury engineers. The "Y" level is believed to have been invented in 1740 by Jonathan Simmons of London, and the dumpy level in 1845 by William Gavatt in England

The first leveling in Europe that is considered of geodetic quality was conducted in France under the direction of M Bourdaloue between 1857 and 1860. His results were published in 1864. The complex observing procedure was designed to produce results with a high degree of accuracy by eliminating systematic errors and detecting and eliminating blunders. This work is said to have required that every two measurements agree to within  $2\sqrt{K}$  millimeters, where K was the length of the leveling line in kilometers.

The French work inspired the Swiss to engage in a similar effort. In 1864, a Swiss recommendation for the execution of a precise leveling network over a large part of Europe was adopted by the International Geodetic Conference. The methods of observation and the use of a mean sea-level datum were prescribed in the resolution. For the observations, a precise spirit-level instrument was designed by Kern of Aarau, Switzerland. These instruments were widely used in Europe and later several were used by the U.S. Corps of Engineers.

Although some leveling was undoubtedly conducted locally in the United States (primarily to tidal bench marks) prior to the Revolutionary War and by the U.S. Coast Survey upon its establishment in 1807, the first recorded effort was a geodetic leveling project by the U.S. Coast Survey in 1856-57. To support detailed studies of the tides and currents in New York Bay and the Hudson River, a series of tide gages was established along the Hudson River which connected with a line of leveling established by G. B. Vose. Vose states in the 1857 Report of the Superintendent of the Coast Survey: "As you directed, a double series of levelings were made throughout the whole route and every doubtful step was retraced... It appears that the probable error for the entire distance from New York to Greenbush does not exceed two-tenths of a foot." A bench mark in this line provided the sea-level datum to which subsequent levelings by the U.S. Lake Survey were referred in determining elevations of the water surfaces in the Great Lakes.

In 1871 the Coast Survey formulated plans for a transcontinental arc of triangulation along the 39th Parallel. It quickly became apparent that accurate elevations would be required to reduce the triangulation data along the route After Congress authorized the survey in 1876, new leveling instruments were designed and fabricated by the Coast Survey. These were used on the first transcontinental leveling in 1875. The first bench mark was set in the courthouse at Hagerstown, Md.

The line of "geodesic" leveling proceeded west with interruptions, reaching St. Louis, Mo., in 1882. A line leveled in 1881 from a tide gage at Sandy Hook, N. J., to Hagerstown, Md., provided a connection to mean sea level In 1899 the transcontinental line reached within a few miles of Cheyenne, Wyo, and it was completed to the tide gage at Seattle, Wash, in 1905. Other lines were leveled, in cooperation with the Corps of Engineers, along portions of the Mississippi River and its major tributaries.

(

In 1875 the US Lake Survey, requiring accurate elevations above mean sea level for the water levels in the Great Lakes and for bench marks in the adjacent harbor areas, extended geodetic leveling into the Great Lakes area. A line of control points was leveled in New York along the Erie Canal, various wagon roads, and the New York and Oswego Midland Railroad. The line extended from bench mark GRISTMILL, at the town of Greenbush, to bench mark A at the harbor in Oswego. During the same year, lines were leveled between Lake Ontario and Lake Erie, and a single line of leveling connected Lake Erie with Lake Huron. Thus, connections were made across the land between the Hudson River, Lake Ontario, Lake Erie, Lake Huron, and Lake Michigan.

An important procedure introduced in 1875 was waterlevel transfer in which the mean surface of each lake, averaged from data obtained at water-level gages during a 3- to 4-month period, was assumed to define an equipotential surface. An elevation relative to mean sea level, determined by leveling to a gage on a lake, was thus transferable to other gages on the lake. By this step-by-step process an elevation relative to mean sea level at Greenbush, N.Y., was assigned to the gage at Escanaba, Mich. For the first time, reasonably accurate elevations were available for all of the Great Lakes, except Lake Superior, based on the results of geodetic leveling and water-level transfers.

In 1877 the principle of "double-simultaneous" leveling was introduced along the Mississippi River by the Corps of Engineers. Two pairs of rods were used with one leveling instrument. The line of leveling was carried forward at each instrument setup with two independent observations of the backsight and foresight rods, on separate turning points. Thus, two separate levelings of the route were made, although only one observer and one leveling instrument were required. This provided continuous and effective checks against blunders as the work progressed. In 1882-83, J. B. Johnson of the Corps of Engineers introduced the observing procedure known as three-wire leveling.

In 1896 Congress authorized the U.S. Geological Survey to determine elevations in support of topographic mapping "above a base level located in each area under survey." Leveling was conducted in many States under this authorization. The first, and very important, line started at a tide gage at Morehead City, N.C., continued inland across the State through Raleigh, Greensboro, and Asheville; ran across Tennessee through Knoxville and Cleveland; extended southward into Georgia through Rome, Atlanta, and Macon, and ended on a tide gage at Brunswick, Ga.

Important contributions to the growing network of geodetic leveling were also made by railroad companies, which relied on leveling to support the construction and maintenance of their extensive track systems. Principal contributors were the Pennsylvania Railroad and the Baltimore and Ohio Railroad. In addition, almost all of the early leveling by the Federal Government was performed along railroad routes because they provided almost the only available cleared routes without excessively steep grades.

In 1898 an *ad hoc* committee of the Coast and Geodetic Survey (C&GS) was appointed to investigate the instruments, observational procedures, and results of geodetic leveling. After extensive analysis, the committee recommended a new design for the leveling instrument that had been used since 1877 and the adoption of the three-wire leveling procedures used by the Corps of Engineers. In 1899 three prototypes, modifications of the 1877 instruments, were produced in the C&GS shops and tested in the field.

The new spirit-level instrument was first used in 1900. It became known as the Fischer level, named for E. G. Fischer, chief of the C&GS Instrument Division, who was responsible for its design and manufacture. The instrument was designed to be very rigid and to change little in adjustment during temperature variations. Its freedom from mechanical failures and its high degree of accuracy soon became apparent.

The Fischer level was the standard leveling instrument of the Coast and Geodetic Survey until the mid-1960's when it was replaced by instruments equipped with optical micrometers.

Since 1900 further improvements in accuracy have been possible. These resulted from the introduction of Invar scales for leveling rods in 1917, instruments with optical micrometers in 1910, and compensator instruments in 1950. These features are still important today

### 1.3 Height Systems

Before a network of geodetic leveling may be considered a network of vertical control, the observed elevation differences must be placed within a common frame of reference. The first step is taken when the differences are measured after leveling an instrument and a pair of rods; all measurements are then relative to the direction of gravity. The second step is to minimize discrepancies in the results obtained by leveling along different routes between the same two points. For this process, termed adjustment, an appropriate height system must be selected to account for the nonspherical and irregular nature of the gravity field. The third step is to define the surface datum to which heights may be referred.

Orthometric height. The gravity field can be represented as a series of surfaces of equal potential, termed "level" or equipotential surfaces, one of which is specified as the reference surface from which orthometric heights are measured. The orthometric height of a point is the distance from the projection of the point on the reference surface to the point itself, along a line perpendicular to every equipotential surface in between

If an equipotential surface were consistently parallel to the reference surface, the orthometric height of every point on the surface would be the same. However, because of the Earth's rotation and anomalies in the gravity field itself, the equipotential surfaces defined by the gravity field are not parallel. The orthometric heights of points on a common equipotential surface vary, increasing most noticeably as a person moves towards the equator (fig. 1-1). For example, a level surface 1000 m above mean sea level at the Equator would be only 995 m above mean sea level at the poles.



Figure 1-1.—Orthometric height, *H*, on a surface not parallel to the reference surface.

As a result, even though no elevation difference is measured when leveling between two points on an equipotential surface, the orthometric heights of the points are different if the points are at different latitudes. The elevation computed for the second point, by adding the observed elevation difference to the elevation of the first point, is not the orthometric elevation of the second point.

Furthermore, if the leveling follows a different route between the same two points, along a series of different equipotential surfaces, yet another elevation will be computed for the second point (fig. 1-2).

To obtain orthometric elevations that are consistent with respect to a single reference surface, this effect



Figure 1-2.—Observed difference in orthometric height,  $\Delta H$ , depends on the leveling route.

must be removed. One way is to correct the elevation difference for each section of leveling, subtracting the error predicted by using a model of the Earth's normal gravity field. (See end of sec. 5.5.2.) The elevations obtained in this manner are termed normal orthometric heights.

Geopotential number. A more accurate way to account for the effect of the gravity field is to compute geopotential numbers rather than orthometric heights. The geopotential number of a point is a measure of the difference in potential from the reference surface to the equipotential surface passing through the point. It is numerically equivalent to the work required to raise a mass of 1 kg against gravity, g, through the orthometric height, H, to the point:

Geopotential number 
$$= \int_{0}^{H} g dh$$

The difference in height, dh, measured during each setup of leveling, may be converted to a difference in potential by multiplying it by the mean value of gravity for the setup. To compute accurate geopotential numbers, gravity must be determined with sufficient accuracy and density throughout the vertical network. Only since the development of the modern gravimeter has this become practical.

Although geopotential numbers are useful for the adjustment of vertical networks, for many purposes orthometric heights above a physically defined reference surface are still necessary. A geopotential number may be converted to an orthometric height by dividing it by the mean value of gravity along the plumb line between the point and the reference surface. Since such a value cannot be directly measured (because the reference surface lies within the Earth beneath the point), a model must be used to derive the value as a function of the geographic position, measured value of gravity at the point, and other variables.

Datums. Traditionally, mean sea level has been selected as the reference surface for computing heights because a water surface conforms to the gravity field by approximating an equipotential surface. A point on mean sea level is determined at a tide gage, where the fluctuations of water level are observed for many years to assess tidal effects adequately. If the elevation at such a gage is connected to a leveling network, and zero is fixed as the height of mean sea level, the reference surface thus defined is called a sea-level datum.

However, a variety of physical factors—changes in volume, currents, temperature and salinity gradients, atmospheric pressure, wind, and sea-floor topography affect the mean water level determined at each tide gage. As a result, the points assigned to be on mean sea level at different tide gages do not necessarily lie on the same equipotential surface.

To avoid the implication that an elevation difference of zero will be obtained when leveling between two widely separated tide gages, the reference surface is simply referred to as the datum. It can be defined in many different ways that may or may not refer to mean sea level As described in the following paragraphs, the datum of the National Geodetic Vertical Network has always been defined by a mean sea level. It has changed with each adjustment as additional tide gages have been connected.

### **1.4 National Geodetic Vertical Network**

By 1900 geodetic leveling by the Coast and Geodetic Survey and other agencies had become so extensive that a general adjustment of the results became necessary to obtain consistent and accurate elevations for all control points. Data from 21,095 km of leveling were obtained by the Coast and Geodetic Survey from the U.S. Geological Survey, the Corps of Engineers (U.S. Lake Survey, Mississippi River Commission, Missouri River Commission, Deep Waterways Commission, and others), and the Pennsylvania Railroad.

The adjustment of this, the first national network, produced elevations for about 4,200 control points that were referred to mean sea level as determined at the following tide gages: Boston, Mass., New York, N. Y., Sandy Hook, N.J., Washington, D.C., and Biloxi, Miss. A connection to sea level on the Pacific coast had not yet been obtained.

Because of an expanded leveling program, more than 10,000 km of new leveling were completed by 1903, including the releveling of some previously unsatisfactory lines. A second adjustment was made, including approximately 6,900 control points in a network of 31,789 km.

In 1907 another 6,500 km of new work had been completed, including the transcontinental leveling through Wyoming, Utah, Idaho, Oregon, and Washington to connect with the tide gage at Seattle, Wash. To utilize the new data the adjustment of 1907 was made. At that time, the network included a total of 38,359 km of leveling and about 9,100 control points. As a matter of expediency, the elevations of most points in the eastern United States were not changed.

New connections to mean sea level on the Gulf coast and the Pacific coast, totaling 9100 km, became available in 1912, including a line across the southern United States from Louisiana to San Diego, Calif., and a north-south line. For the fourth general adjustment of the network, mean sea level was fixed at five tide gages on the Atlantic coast, two on the Gulf coast, and two on the Pacific coast. Orthometric corrections were applied to leveling data in the western States. The 1912 network included 46,462 km of leveling and 11,100 control points.

General Adjustment of 1929. After the previous period of comparatively short intervals between adjustments, 17 years elapsed before the network was adjusted again. In the meantime, it had become more extensive and complex, and included many more sea-level connections. The General Adjustment of 1929 incorporated 75,159 km of leveling in the United States and, for the first time, 31,565 km of leveling in Canada. The U.S. and Canadian networks were connected by 24 ties between Calais, Me./Brunswick, New Brunswick; and Blaine, Wash./Colebrook, British Columbia. A fixed elevation of zero was assigned to the points on mean sea level determined at the following 26 tide stations:

Father Point. Quebec	St Augustine, Fla
Halifax, Nova Scotia	Cedar Keys, Fla
Yarmouth. Nova Scotia	Pensacola, Fla
Portland. Me	Biloxi, Miss
Boston, Mass	Galveston, Tex
Perth Amboy, N J.	San Diego, Calif
Atlantic City, N J	San Pedro, Calif
Baltimore, Md	San Francisco, Calif
Annapolis, Md	Fort Stevens, Ore
Old Point Comfort, Va	Seattle, Wash
Norfolk, Va	Anacortes, Wash
Brunswick. Ga	Vancouver, British Columbia
Fernandina, Fla	Prince Rupert, British Columbia

'There was no tide station at Perth Amboy, but the elevation of a bench mark at Perth Amboy was established by leveling from the tide station at Sandy Hook

The 1929 adjustment provided the basis for the definition of elevations throughout the national network as it existed in 1929, and the resulting datum is still used today. The elevations were referred to the "Sea Level Datum of 1929" until 1973, when the more appropriate name "National Geodetic Vertical Datum of 1929" (NGVD29) was adopted for the same reference surface.

The modern network. Since 1929, more than 625,000 km of leveling throughout the United States have been added to the national network. New results have been adjusted to fit into the network of 1929. If both old and new levelings were of similar accuracy and if the Earth's

crust were perfectly stable, this procedure could have continued indefinitely. However, instrumental and procedural improvements make possible a higher degree of accuracy in modern leveling, and, increasingly, the national network has come to play an important role in monitoring crustal movements.

In many regions, withdrawal of underground fluids has led to land subsidence at the surface, which has resulted in significant economic losses In tectonically active areas earthquakes cause sudden changes in elevation. Elevations are not fixed in time, and, therefore, elevation differences observed at widely different times cannot be expected to agree exactly. Leveling results must be adjusted by epoch and then compared to assess topographic change.

These considerations have led to the establishment of local programs in which regions of known crustal motion have been releveled periodically to assess the rate and extent of subsidence or uplift. To relate these surveys adequately and to determine where additional work is necessary, the national network itself must be resurveyed at regular intervals.

At present, a minimum network of 100,000 km has been targeted for releveling. This network, known as basic net A (fig. 1-3), is designed to satisfy the standards established by the Federal Geodetic Control Committee for first-order vertical networks (table 1-1).

Vertical control surveys (projects) are conducted continually by field parties of the National Geodetic Survey (NGS) to maintain and update the national network. The procedures by which such projects are accomplished—reconnaissance and bench mark setting, geodetic leveling, river crossing, and data processing—are the subjects of the remaining chapters of this manual.

	First order class I	First order class II	Second order class I	Second order class II	Third order	
Principle use	Basic	net A	Area co	ontrol	Local contro	
Line spacing,						
National net	100-300 km (60-190 mi)	50-100 km (30-60 mi)	20-50 km (10-30 mi)	10-25 km (5-15 mi)	As needed	
Metropolitan net	2-8 km (1-5 mi)	2-8 km (1-5 mi)	0.5-1 km (0 3-0 6 mi)	As needed	As needed	
Maximum length of leveling line						
between junctions	. 300 km (190 mi)	100 km (60 mi)	50 km (30 mi)	50 km (30 mi) double run, 25 km (15 mi) single run	25 km (15 mi) double run, 10 km (5 mi) single run	
Control point spacing	Average 1 6 km maximum 3 km	(1 0 mi). (2 mi)	Maxin	num 3 km (2 mi).		

#### Table 1-1.—Classification of vertical control networks



Figure 1-3.—Basic net A of the National Geodetic Vertical Network.

### Chapter 2

### **RECONNAISSANCE AND BENCH MARK SETTING**

### 2.1 Introduction

Before leveling proceeds in a network of vertical control, lines of control points must be defined. This is best accomplished by making a reconnaissance and setting the necessary bench marks a month or more before leveling begins. Detailed instructions for reconnaissance and mark setting are presented in this chapter.

The task is normally performed by personnel organized into mark setting units Leveling routes are planned in detail to satisfy network requirements. Maps with written instructions (referred to as "logs") are prepared to illustrate the routes for the leveling units. Monuments are recovered, established, and described to mark permanently the lines of vertical control.

Sometimes a single monument must be removed and set elsewhere by a mark maintenance engineer or local surveyor before it is destroyed by highway construction or other activity. The chapter concludes with instructions for relocating such control points.

### 2.2 Organizing a Mark Setting Unit

At the start of a project, the mark setting personnel should be organized into units and each unit should be assigned a line to reconnoiter and prepare for leveling. Since a typical project encompasses a large area, each unit must be capable of working self-sufficiently, often several hundred kilometers from the project office. This section presents guidelines for organizing and maintaining such units.

#### 2.2.1 Personnel

A foreman should be assigned to supervise the mark setting units and to manage their activities, particularly when many units are operating. Two persons are normally required on each mark setting unit, a mark setter and an assistant.

Mark setting foreman. The mark setting foreman coordinates, supports, and monitors the mark setting units. The foreman's skills should include technical competence in both mark setting and leveling, as well as proficiency in organizing unit activities and maintaining good relations with local agencies and the public. The foreman should set an example which ensures correct, safe, and efficient mark setting.

The foreman is responsible for coordinating the mark setting effort on all the lines in a project. From a study of the preliminary data, he or she should specify routes for each line and determine areas requiring special attention so that data can be organized and distributed to the appropriate units. Efforts of a specialized unit, such as a drilling-rig team, should be smoothly merged with those of the unit responsible for the line.

The foreman supports the units by providing liaison, training, and supplies. He or she should establish initial contacts with local agencies, and informs the mark setting units if follow-up contacts are necessary. Throughout the project, the foreman should provide technical guidance and training, and, in addition, inventory supplies regularly and place orders to meet future needs.

The foreman should ensure that the specifications and guidelines stated in this manual and in NOAA Manual NOS NGS 1, Geodetic bench marks (Floyd 1978), hereinafter referred to as the mark setting manual, are strictly followed. A monthly inspection should be made of each unit. Equipment should be examined to ensure that it is properly maintained. Mark setting techniques should be checked to ensure that safe and correct procedures are followed by the units. After receiving descriptions and logs from a unit, the foreman should routinely inspect the line, looking for poorly set marks, plotting mistakes, and description mistakes. The unit should be made aware of any deficiencies and should correct them immediately. Whenever there is substantial confusion or doubt about a route or mark setting procedure, the project director should be consulted.

The foreman should prepare a report of activities for the project director each month.

Often, if a project is small or the number of available personnel is limited, the foreman may be required to serve as chief of a mark setting unit or leveling unit. When this occurs, the project director should assume most of the responsibilities of the foreman.

Mark setter. As head of a mark setting unit, this individual is normally responsible for the correct reconnaissance and mark setting of an entire line. In addition, he or she must properly maintain and protect a large and varied selection of mechanical equipment. The mark setter who is in charge of a drilling rig has additional maintenance responsibilities. Training and experience in reconnaissance and mark setting as well as strong mechanical ability are necessary qualifications. Leveling experience is highly desirable.

The mark setter should follow the instructions stated in this manual and in the mark setting manual. The project instructions and preliminary data provided by the project director or foreman should be studied and understood. The mark setter should not hesitate to seek guidance when questions arise. The mark setter should establish an organized work routine with the assistant, incorporating safety precautions into every procedure. Equipment failures should be reported promptly to the foreman or project director. The mark setter should submit monthly reports to the project office.

Assistant mark setter. The assistant mark setter is often a new employee or a member of a leveling unit who is receiving training. The assistant helps the mark setter in all phases of the work, in addition to being assigned certain routine responsibilities, such as truck maintenance and checking control point descriptions.

#### 2.2.2 Equipment

Each unit should be assigned a truck, tools, and other equipment (as itemized in appendix A). When the equipment is issued, the mark setter should conduct an inventory and repeat it at least once every year thereafter. All equipment should be secured in an organized fashion in the assigned vehicle.

The mark setter is responsible for maintaining the truck and tools in good condition. In addition, he or she should study and use the equipment manuals. Equipment losses, damage, or defects should be reported immediately to the foreman or project director. Replacements and repairs should be noted on the monthly report.

Drilling rigs and gasoline-powered portable drills require especially rigorous maintenance if they are to function properly. They should be assigned only to responsible individuals who have demonstrated competence in operating such equipment.

### 2.2.3 Safety

Mark setting presents numerous situations where personal injury may occur. For this reason, a mark setting unit should always include two persons. All mark setters and assistants should be trained in basic first aid and cardiopulmonary resuscitation. They should know the location and telephone number of an emergency treatment center in the work area, so prompt assistance can be obtained if an accident occurs.

Accidents usually result from hazards created by the environment or the equipment. Anticipate potentially hazardous situations and be prepared to handle them.

The mark setting environment is normally the rightof-way of a highway or railroad. Obtain permission to work in these zones and find out from local officials what safety procedures are required. When working alongside a highway, stay well off the road, wear orange vests, and turn on the truck's warning light. If necessary, and only after consulting with local authorities, use traffic cones, warning signs, or flagmen to divert traffic while setting a mark. When working along a railroad, do not wear orange vests or use flashing lights because these may cause a train to stop unnecessarily. Cross the tracks only at designated crossings; otherwise keep clear of them. Right-of-way zones are often used as routes for utility pipelines and cables. Select sites for marks with this in mind. Before using an auger or driving a rod, check the site with a pipe and cable locator. Similarly, be alert to the presence of wires *above* the work location, especially when maneuvering a drilling rig or erecting lengths of steel rod.

Mark setting equipment may present a hazard if used improperly. Learn the correct techniques for handling and operating all equipment. Use the right tools for the job and keep them clean. To prevent back injury, lift heavy tools and supplies from a squatting position; use the leg muscles, not the back. Get help if an object is too heavy to lift safely. Above all, stay alert. Do not allow a steady routine to make you complacent.

Two items of equipment deserve special mention: the gasoline-powered rock drill and the drilling rig. When driving rods or drilling with these tools, prevent eye injury by wearing goggles or shatter-proof glasses. Prevent hearing loss by wearing ear plugs or mufflers. Wear hard hats and steel-toed boots.

Mark setters operating a drilling rig must be thoroughly trained in the proper routines for starting and stopping the drill, leveling and stabilizing the truck, and assembling and disassembling the augers. These instructions are outlined in the maintenance manual accompanying the rig. Test the emergency stopping system each day. While drilling, do not wear gloves near the turning augers.

#### 2.2.4 Reports

The project director or foreman should visit each mark setting unit at least once each month. At all other times the mark setters should maintain regular communication, either in person or by telephone, with the project director. In addition, the mark setter should submit the reports described in the following paragraphs.

Monthly report. Mail or submit in person, a report to the project office at the end of work each month. (See fig 2-1.) If work involved more than one State or project, prepare a separate report for each activity. The report is normally forwarded to the headquarters of the National Geodetic Survey, where it provides statistical information about the progress of mark setting in the national network. Include the following information:

- 1. Number of points recovered in good condition.
- 2. Number of points recovered in poor condition.
- 3. Number of points recovered destroyed.
- 4. Number of points not recovered.
- 5. Number of bench marks set; types of bench marks set.
- 6. Notes about condition of equipment and main-
- 7. Inventory of existing supplies.
- 8. Notes on project activities.

HOA) (3 - 8	1 FORI	4 74-1	76											нати	DNAL C	CEANIC	U.5 A ND A	DEPARTMENT OF CONNERCE
				Ņ	ONT	HLY	REP	ORT	~ BE	NCH	MAF	K AI	ID R	ECO	INAR	SSANC	E TE/	AMS
TÉA	ZAM LEADER											PROJECT DIRECTOR						
TEAR	TEAN ASSISTANT							PART	Y					NONTH/YEAR 19				
			MAR	1K5 51	EARC	EO F	0R								TASE	NUMB	E8	
	VER	TICAL	СОНТ	ROL	нс		TAL C	ONTR	oL	MA	RKSI	ESTAR	LISH	50	STAT	۴ <b>E</b> :		
			_					· · · ]				-						
Date	and Con	acoveras oor Cons	0T ecoveres	estroyed of inusable	talion\$	elemnos ates	Z Marks	ther	TO	ock otka	uality A ads [seved	uality A ads na leere	uality B ada	Inucture	kine Kumber	i lumeter	raff-Dey	Remarks
1	<b>¤</b> 0	æ 0.	z∝	<u>a</u> 3		<u> </u>	-	0	2 @	<u></u>	00.0	98.3	04	<u> </u>	52	*	5	
2									1		1							
3	ļ –			<u> </u>				-	<u> </u>		t			<u> </u>				
4					İ				<b></b>		<b>†</b>							
5				i i														
6														Ĺ				
7		L.							L				L					
8	<u> </u>				<u> </u>			ļ	<u> </u>				ļ					
9	L	ļ	<u> </u>	ļ	ļ		L	ļ	ļ		ļ	ļ	ļ	ļ	Ļ			
10	<b> </b>	<u> </u>	ļ	L	L					1			ļ	ļ	Ļ	ļ		
11	<u> </u>		<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>			L_	ļ	<u> </u>	<u> </u>			
12	<u> </u>	Ļ	<b> </b>	<b> </b>	ļ	<u> </u>	<u> </u>		<u> </u>		<u> </u>	ļ			<u> </u>		$\vdash$	
13	<b>_</b>	ļ	ļ	ļ	ļ	ļ		ļ	ļ	ļ	Į	ļ			<u> </u>	<b></b>		
14	<b> </b>		┣				ļ			<b> </b>	<b> </b>	<u> </u>			<u> </u>			
61 26	<u>+</u>	ļ	-	-	-	-		-	-	-			$\vdash$	-	-			
17		-	-	-			<u> </u>	ļ	<u> </u>	<u> </u>	<u> </u>		┣		+			
18	<u> </u>		<u> </u>				<b> </b>	╂				–−−	<u> </u>	-				
19					<u> </u>			+		+		┣─	-	$\vdash$	╞──			
20	+		+	+	-	-	-	<u> </u>	$\vdash$	+	<u> </u>		┼──		┼──	<u> </u>	<b> </b>	<u> </u>
21	+	†	1	+	╈	+	<del> </del>	+	†	┢┈	+	+-			1	<u> </u>	<u>}</u>	<u>}</u>
22	+	+	+	+	+		<u>†</u>	+	+	1	+		†	+	$\vdash$	<u> </u>	<u> </u>	
23	1	$\square$	+	†	1	1	1	$\mathbf{t}$	<b>†</b>	$\vdash$	1	$\uparrow$	1	1	†	1		1
24	$\top$	1	1	1	+	1	†	1	1	t	1		1	1	1	1	1	
25	1	1	1	1		$\mathbf{T}$	<u> </u>	1	1	1	1	1	1	1		1		
26	1	t	1	1		1	ľ	1		Γ	T	_	Ι	Γ				
27					T											<u> </u>		
28												Ι				}		
29																1		
30																		
31																		
Tera													L	l	1	<u> </u>	L	

		INVENTORY OF	MARK SUPPLIES			
PARTY		TRUCK NUMBER		MONTH/YEAR		
INVENTORIED BY			CONDITION OF TRUCK	I		
	BEGINWING INVENTORY		ENDING INVENTORY	SUPPLIES USED	SUPPLIES NEEDED	
Rods						
Studs						
Rod Entry pts. (fluted)						
Rod Entry pts. (smooth)			1			
Datum pts.						
Logo Cops					1	
5" PVC						
1" PVC						
Disk (stem)			1		1	
Disk (rod)						
Wilness Post					1	
Witness Sign			1			
Greose	• •					
"O" Ring						
PVC Reducer						
PVC Cement						
PVC Schedule						
Pop Rivets					1	
BM Description			1		1	
Flagging				-		
Ctayon						
	••					
			<u> </u>		1	
		-				
		-				
			<u> </u>			
		-				
	12	. 1	<u> </u>	<u> </u>	1	

HOAA FORM /L+188 J-82

TO U.S. GOVERNMENT PROVING DIFFICE 1982-554 CD8 1146



Chapter 2-Reconnaissance and Bench Mark Setting

Maintain the report each day, recording the number of points searched for or set and a general statement about the day's activities. As appropriate, include remarks about weather, equipment malfunction, equipment repair, geological conditions, and suggestions. Be sure to list names, telephone numbers, and addresses of officials and individuals who were contacted.

At the end of the month, summarize progress along the assigned lines by totaling the distance of line reconnoitered and prepared for leveling, the number of 7.5 position plots prepared (a 15' plot counts as four), and the number of descriptions written. Do not count plots or descriptions that are not yet finished. Also, inventory specialized mark setting supplies, so their usage can be assessed and future stocks can be procured. List any supplies needed.

*Truck report.* Keep careful records of all expenses incurred for fuel, oil, repair, and maintenance of mark setting vehicles. At the end of the month, submit a report to the project office summarizing these expenses and include the receipts.

Miscellaneous reports. Be familiar with current organizational procedures for submitting per diem vouchers, accident reports, personnel reports, and expense receipts

### 2.3 Routing a Line for Leveling

The reconnaissance and mark setting effort begins with the detailed planning of the leveling routes. The foreman (or project director) should organize and consolidate the preliminary data for a project into packages corresponding to each line or group of lines. A general reconnaissance of the area should be made and liaison should be established with local agencies. The mark setting units then proceed to route each line according to the specifications given in this section, recovering and establishing control points and preparing instructions for the leveling units.

#### 2.3.1 Preliminary Data

For each line or group of lines in a project, a mark setting unit should possess the data listed in table 2-1.

Figure 2-2 shows the areas covered by some of these items. Most of the items are forwarded to the project director along with project instructions. Some items must be obtained in the field. Instructions pertaining to their use are described in the following paragraphs.

*Project instructions.* Project instructions are the starting point for any survey. They include information and instructions specific to the project which supplement the standard operating procedures given in this manual Before beginning work, read the instructions thoroughly and consult them whenever in doubt about the project requirements.

#### Table 2-1.—Preliminary data for reconnaissance and mark setting

#### Requirements

Project instructions:
Project listi defiolis.
Maps for planning and organization:
NGS State index map for control leveling
NGS Geodetic Control Diagrams
USGS State index to topographic maps
State and county highway maps
Geological and soil maps, as available
Descriptions of control points:
NGS vertical control data, including a separate listing of archiva cross-reference numbers if necessary
NGS horizontal control data
NGS unpublished descriptions for points recently leveled or reset
NOS tide and water-level station reports
Descriptions of specially requested points
Maps for logs and position plots:
USGS topographic maps
City and county street maps, as necessary

Refer to the project instructions to obtain the following information:

- 1. A rough sketch of the proposed leveling routes.
- 2. Line numbers and titles.
- Accession numbers ("L-numbers") of previous survey lines that coincide or connect with the proposed routes.
- 4. Specific control points to be leveled.
- 5. Survey-point serial numbers to be assigned to the points in each line.

A table similar to that in figure 2-3 is normally included in the project instructions.

Maps for planning and organization. Several types of maps, covering large parts or all of the project area, are useful for planning and monitoring progress along each line. These maps are most useful if the project lines have been highlighted and labeled in advance.

The Index Map of Control Leveling (fig. 2-4) for the State in which the project is located is available from NGS. It provides an overview of the project area, showing existing first- and second-order lines of leveling. On the index map are plotted and labeled the 30' quadrants into which vertical control data are grouped.

Geodetic Control Diagrams (fig. 2-5) are also available from NGS. They are plots of existing geodetic control at a scale of 1:250,000 on the 1° by 2° quadrangle series of the U.S. Geological Survey (USGS). Control for Alaska is plotted at a scale of 1:500,000 on sectional aeronautical charts. Diagrams are designated by name, number, and two-letter area code. They show the locations of both vertical and horizontal control points and the boundaries of 30' quadrants, 15' topographic maps, and counties. The diagrams are particularly useful when planning connections between leveling lines and other types of control networks. An index of control diagrams for the United States is given in appendix B.



Figure 2-2.—Scale relationships of preliminary data.

TABLE	1
-------	---

*JC*	LINE NO.	PROJECT TITLE/LINE TITLE
*W2*		HARRIS-GALVESTON COASTAL SUBSIDENCE AREA TX
*W2*	1	ADDICKS VIA HOUSTON TO CROSBY
*W2*	2	ADDICKS VIA BELLAIRE TO HOUSTON
*W2*	3	HOUSTON VIA LEAGUE CITY TO GALVESTON - PIER 21 TIDE STATION
*W2*	4	CROSBY VIA BAYTOWN AND TEXAS CITY TO LA MARQUE
*W2*	5	3 KM W QF PASADENA TO LA PORTE
*W2*	6	WEBSTER TO SEABROOK

		ТА	BLE 2	
*JC*	LINE NO.	LENGTH KM.	OLD HGZ NO.	USE SPSN'S
*W2*	1	80	L24406-12,-19,-25	1-200
*W2*	2	60	L24406-10,-11	201-300
*W2*	3	80	L24406-7,-2	301-500
*W2*	4	100	L24406-19,-20,-4	501-700
*W2*	5	30	L24406-21,-22	701-800
*W2*	6	10	L24406-6	801-900

Figure 2-3.—Example of line assignments in project instructions.

NOAA Manual NOS NGS 3, Geodetic Leveling



Figure 2-4.—State Index Map of Control Leveling.



2-7

The Index to Topographic Maps for the State in which the project is located is available from USGS. Although the index does not show all geographical features, the project lines may be plotted approximately (fig. 2-6). It is extremely useful as an index and checkoff list for logs and smooth plots.

State and county highway maps are usually available from district offices of a State highway or State transportation department. Since topographic maps do not always show the latest highway construction and route numbers, current State highway maps should be the final source for this information.

Consult geological and soil maps when selecting sites for bench marks. Basic information about the location and type of expansive soils throughout the United States is provided in the mark setting manual. Additional information, especially for locating bedrock and fault zones, can be obtained from geological maps published by the USGS or State agencies, such as departments of mines and natural resources. Information concerning soil types and depths is often available from local offices of the U.S. Soil Conservation Service.

> INDEX TO TOPOGRAPHIC MAPS OF MISSISSIPPI ORDER MAPS BY NAMES PRINTED IN BLACK AND BY SERIES DESIGNATION ALL MAPS SHOWN ON THIS INDEX ARE DISTRIBUTED BY THE GEOLOGICAL SURVEY

Other sources that may be useful for certain projects include aeronautical charts, storm evacuation maps, and nautical charts. These are available from the National Ocean Survey (NOS).

Descriptions of control points. To recover existing vertical and other control points, published descriptions should be consulted. Because they serve different purposes, descriptions for vertical control, horizontal control, and tide or water-level stations are prepared, stored, and published separately.

Vertical Control Data, available from NGS, are published in sets. Each set ("quad" or "litho list") includes data for all the vertical control points in a 30' quadrant. The quadrant is identified by hemisphere, latitude, longitude, and quarter. After the hemisphere symbol, "N" or "S", a 1° by 1° area is identified by the degrees of latitude bounding it on the south (on the north in the southern hemisphere) and by the degrees of longitude bounding it on the east. The quadrants are numbered clockwise within the 1° by 1° area, beginning with the northeast: "1" signifies the northeast quarter, "2" the southeast, "3" the southwest, and "4" the northwest.

MISSISSIPPI



Figure 2-6.—Index to Topographic Maps, project lines plotted.

For example, "N340901" signifies a 30' quadrant which is the northeast quarter of the 1° by 1° area bounded by  $34^{\circ}$ N on the south and 090"W on the east.

Each quadrant includes a plot showing the lines of control points described within the quad (fig. 2-7a). The line numbers given in the plot are unique only within the quadrant; they do not necessarily correspond to those of adjacent quadrants. Arrows indicate the order in which control points are described along each line. The plot is followed by two indexes that give the designations of points in alphanumeric order (fig. 2-7b) and line order (fig. 2-7c). Descriptions and other data (fig. 2-7d) are then given in line order

Each control point is assigned a unique number (the archival cross-reference number) when it is initially included in the NGS data base. If the published descriptions do not include these numbers, obtain special listings of the synoptic file ("archive lists") from the NGS Vertical Network Division. Identified by two-letter area codes, these listings are available for each Geodetic Control Diagram (See fig. 2-8a,b.)

Data for horizontal control points are also available in 30' quadrants from NGS. Descriptions are usually arranged chronologically within each quadrant. An alphanumeric index of designations is included. After finding the designation of the desired point on a Geodetic Control Diagram, look it up in the appropriate index to find the page number of the description. Further information about horizontal control data is available in NOAA Technical Memorandum NOS NGS 5, National Geodetic Survey data: availability, explanation, and application (Dracup 1979).

For up-to-date information on points that have recently been recovered, established, or relocated, obtain descriptions from the NGS Vertical Network Division. It is possible that these data may not yet be included in the published descriptions. File the recent descriptions with the appropriate 30' quadrant, and remember to use them. A reset mark may be mistaken for the original one if the mark setter is unaware of the complete history of the control point

Descriptions for tide and water-level stations are available from the NOS Tides and Water Levels Division. Each station is identified by a unique seven-digit number. The first three digits identify the State or body of water. A list of these codes is shown in table 2-2. Station listings are published in NOAA Tide Tables: High and Low Water Levels (National Ocean Survey, annual publication) Request the most recent station reports for all tide and water-level stations in a project area. Station reports include the name, address, and telephone number of the station observer, bench mark descriptions, and a sketch of the station showing bench mark locations.

Leveling to specific control points may be specially requested either in the project instructions or by local agencies. The points may include junction points, international boundary monuments, airport monuments, compaction meters, deep-well casings, and monuments of USGS, State highway departments, counties, and cities. If the necessary descriptions are not provided with project instructions, the descriptions should be obtained from the appropriate agency.

Maps for position plots and logs. Positions of all points recovered or established must be plotted on the standard 7'.5 series of topographic maps available from USGS. If 7'.5 maps are not available, the 15' series should be used. Because plots are difficult to see on orthophotographic maps, maps of this type should not be used.

The U.S. Geological Survey may be revising or compiling new 7'.5 maps for the project area. Check with USGS and request State progress maps, available from the USGS regional topographic division. Up-to-date 7'.5 "bluelines," which are preliminary compilations of topographic maps, can then be ordered.

In major metropolitan areas, logs should be prepared on up-to-date city or county street maps, available from the Chamber of Commerce, the city government, or local gasoline stations. However, plots must still be made on 7'.5 maps to ensure consistent and accurate scaling of positions.

### 2.3.2 Liaison

While routing the leveling line, a mark setting unit must communicate frequently with the owners, managers, and users of both public and private property. The purpose of these contacts is to obtain permission and make detailed arrangements for both mark setters and leveling units to work in the areas necessary for completion of the project. In addition, a systematic liaison effort may bring to light unusual environmental conditions or ongoing projects of other agencies which affect the choice of leveling routes. Maintaining good public relations is vital to the project.

The project director lays the groundwork for future cooperation and coordination by notifying the appropriate State highway or transportation department in advance of the project. The project director should send a letter requesting permission to operate in the right-ofway zones of the necessary State highways, plus a courtesy copy of the project instructions. If routes include railroad right-of-way zones, follow a similar procedure. The project instructions may list other agencies or individuals to contact.

Immediately after arrival in the project area, the mark setting foreman should visit officials of the agencies with which liaison is necessary. To set bench marks of the best possible quality, visit agencies and individuals who may provide maps or other specific information about the geology along the proposed routes. Determine if surveys exist that should be connected to the national network and inform the appropriate mark setting units In addition, make arrangments to obtain sufficient supplies during the project. U. S. DEPARTMENT OF COMMERCE NATIONAL DEEANIC AND ATMOSPHERIC ADMINISTRATION NATIONAL OCEAN SURVEY

and a second of the second of

 $\sim$ 

BY THE		
NATIONAL GEODETIC	SURVEY	
NATIONAL GEODETIC VERTICAL	DATUR OF	1929

OUAD	340901			
STATE	AR-MS-	TH .		
LATITUDE	34 1	30	TD	35 0
LONGITUDE	90 °	٥`	TO	90 30
DIACDAM	UCICEA	t u	15-	



Figure 2-7a.-Index plot for 30' quadrant of Vertical Control Data.

SEON--051 011AD--N34090100 | [NE--10] US DEPARTMENT OF COMMERCE - NOAA VERTICAL CONTROL DATA STATE--HS DIAGRAM--NI 15-6 NATIONAL GEODETIC VERTICAL DATUM 1929 NOS - NATIONAL GEODETIC SURVEY ROCKVILLE MD Z0852 - APR 1980 ADJUSTED BY--CGS YEAR--1957 COUNTY -- TATE SOURCE--115801 ORDER--1ST MONUMENTATION QUALITY--B APPROX LAT 34-37-47N RENCH MARK POSITION--LON 090-13-22W DESIGNATION--V 194 ESTARI ISHED RY--FGS YEAR--1956 H - ELEVATION ΝΩΩΜΑΙ AROVE NGVD 1979 NORMAL GRAVITY GEOPOTENTIAL NUMBER MODELED BOUGUER MODELED (NORMAL ORTHOMETRIC HEIGHT) ANOMALY SIGMA SURFACE GRAVITY (1967 FORMULA) (GPU=KILOGALMETER) 58.238 METERS ( 191.069 FEET) -7.7 MGALS 0.9 979 682 GALS 979.702 GALS 57.056 GPUS STATE--MS COUNTY--TATE DUAD--N340901 XRN--EH0051 DESIGNATION--V 194 TANATANA MANUMENT BY--CGS TATATANATANATANA YR--1956 COP--UNK MARK TYPE--BENCH MARK DISK TATATANATANATANA TATALLER RECOVERY BY--MSHD TATALLER YR--1966 COP--UNK CONDITION--GOOD TATALLER TATALLER TATALLER TATALLER STAMPING--V 194 (1956) SETTING--PIER INCATED--0.3 MI SE EROM THE CITY OR TOWN DE--SAVAGE TO REACH FROM THE U.S. POST OFFICE IN CRENSHAW GO NORTH ON MISSISSIPPI HIGHWAY 3 FOR 7.55 MILES TO MISSISSIPPI HIGHWAY 4 ON THE RIGHT, CONTINUE NORTHWEST ON HIGHWAY 3 FOR Z.1 MILES TO THE BRIDGE OVER THE RAILROAD TRACK AND THE MARK SET VERTICALLY IN THE SOUTH FACE OF THE CONCRETE BASE OF THE FIRST PIER WEST OF THE TRACK. IT IS 35.7 FEET WEST OF THE WEST RAIL OF THE MAIN TRACK AND 22 FEET WEST OF THE WEST RALL OF THE SIDE TRACK. SEON--052 QUAD--N34090100 LINE--101 US DEPARTMENT OF COMMERCE - NOAA VERTICAL CONTROL DATA NOS - NATIONAL GEODETIC SURVEY NATIONAL GEODETIC VERTICAL DATUM 1929 STATE--MS DIAGRAM--NI 15-6 ROCKVILLE MD 20852 - APR 1980 ADJUSTED BY--CGS YEAR--1957 COUNTY--TATE SOURCE--L15801 ORDER--1ST MONUMENTATION QUALITY--D APPROX LAT 34-36-50N BENCH MARK DESIGNATION--W 194 ESTABLISHED BY--CGS YEAR--1956 POSITION--LON 090-13-11W H - ELEVATION NORMAL ABOVE NGVD 1929 MODELED BOUGUER NORMAL GRAVITY GEOPOTENTIAL NUMBER MODELED (NORMAL ORTHOMETRIC HEIGHT) ANOMALY SIGMA SURFACE GRAVITY (1967 FORMULA) (GPU=KILOGALMETER) 55.995 METERS ( 183.710 FEET) -6.5 MGALS 0.8 979.683 GALS 979.700 GALS 54.858 GPUS STATE--MS COUNTY--TATE DESIGNATION--W 194 QUAD--N340901 XRN--EH0052 STAMPING--W 194 (1956) SETTING--SHALLOW-SET PIPE LOCATED--- 4 MI SOUTH FROM THE CITY OR TOWN OF--SAVAGE TO REACH FROM THE U.S. POST OFFICE IN CRENSHAW GO NORTH ON MISSISSIPPI HIGHWAY 3 FOR 7.55 MILES TO MISSISSIPPI HIGHWAY 4 ON THE RIGHT. CONTINUE NORTHWEST ON HIGHWAY 3 FOR 1.0 TO A SIDE ROAD LEFT ALL-WEATHER ROAD FOR 0.15 MILE TO A TENANT HOUSE ON THE RIGHT, CONTINUE WEST ON A FIELD ROAD FOR 0.25 MILE TO THE RAILROAD TRACK, TURN LEFT SOUTH AND FOLLOW THE WEST EDGE OF A FIELD FOR 0.35 MILE TO THE MARK ON THE RIGHT AS DESCRIBED. IT IS APPROXIMATELY 185 YARDS NORTH OF MILEPOST 41, 110 FEET SOUTHWEST OF POLE NO. 725, 57.5 FEET WEST-NORTHWEST OF POLE NO. 725 1/2, 23 FEET WEST OF THE WEST RAIL. 4.3 FEET SOUTHEAST OF A TELEPHONE POLE, AND 1.6 FEET SOUTH OF A METAL WITNESS POST

Figure 2-7b.—Vertical Control Data sheet.

US DE NOS - ROCKV	PARTMENT OF COMMERCE NATIONAL GEODETIC S ILLE MD 20852 - APR	- NOAA URVEY 1980	NATI	VI ONAL (	ALPHABET ERTICAL GEODETIC	IC INDEX CONTROL I VERTICAI	TO DATA L DATUM 1929	9	LAT SPAN LON SPAN DIAGRAM	34-30.1N 90-00.1W NI 15-6	TO 3 TO 9 INDEX	4-59. 0-29. PG	9N 8H 06
SEON	DESIGNATION	YEAR ESTABLISHED	LAST RECOV	ELI ORD (I	EVATION METERS)	SOURCE	APPROXIMATE LATITUDE	E POSITION LONGITUDE	SURFACE GRAVITY	OTHER CONTROL	QUAD	OSN	ST
48 33 44 66 66	T 117 USE T 193 T 194 T 195 T' 195 USGS	UNK D 1956D 1956D 1956D 1956D	1966G 1966G 1966G 1966G 1966G	15T 1ST 1ST 1ST 1ST	57.354 58.406 56.160 56.793 56.793	L 15801 L 15801 L 15801 L 15801 L 15801	34-38-03N 34-45-09N 34-39-26N 34-30-17N 34-30-17N 34-30-17N	90-13-35W 90-14-38W 90-14-08W 90-11-42W 90-11-42W	979.682 979.686 979.683 979.679 979.679	N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100		MS MS MS MS MS
164 211 226 131 126	T 196 T 218 T 219 T 238 T 243	1956D 1976B UNK B 1974C 1974C	1966G 1976G 1976G	1ST 1ST 1ST 1ST 1ST	57.771 67.481 58.989 61.147 89.054	L 15798 L 24103 L 24103 L 23280 L 23280	34-38-28N 34-51-47N 34-49-01N 34-50-24N 34-46-57N	90-26-22W 90-21-22W 90-28-24W 90-16-23W 90-07-08W	979.687 979.704 979.709 979.696 979.681	N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100 0100		MS AR AR MS MS
146 54 55 55	T 260 TBM A STA 146 TBM A STA 146 USE TBM B STA 146 TBM B STA 146 USE	1974C 1953D 1953D 1953D 1953D	1966G 1966G 1966G 1966G	1ST 1ST 1ST 1ST 1ST	74.821 57.262 57.262 57.319 57.319	L23280 L15801 L15801 L15801 L15801 L15801	34-59-42N 34-35-56N 34-35-56N 34-35-56N 34-35-56N 34-35-56N	90-07-30W 90-13-00W 90-13-00W 90-13-01W 90-13-01W 90-13-01W	979.712 979.682 979.682 979.682 979.682 979.682	N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100		TN MS MS MS MS
174 174 183 183 35	TBM 44/45 TBM 44/45 USE TBM 46/47 TBM 46/47 USE U 193	UNK D UNK D UNK D UNK D 1956D	1966N 1966N 1966N 1966N 1966G	1ST 1ST 1ST 1ST 1ST 1ST	58.574 58.574 57.356 57.356 57.642	L00422 L00422 L00422 L00422 L00422 L15801	34-37-32N 34-37-32N 34-36-02N 34-36-02N 34-36-02N 34-44-25N	90-28-08W 90-28-08W 90-29-05W 90-29-05W 90-29-05W 90-14-36W	979.684 979.684 979.681 979.681 979.686	N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100 0100		MS MS MS MS MS
45 65 165 212 227	U 194 U 195 U 196 U 218 U 219	1956D 1956D 1956C 1976B 1976B	1966G 1966G 1974G	1ST 1ST 1ST 1ST 1ST	55.362 54.124 59.402 67.214 66.326	L15801 L15801 L23280 L24103 L24103	34-38-40N 34-31-04N 34-38-27N 34-51-31N 34-48-19N	90-14-02W 90-11-41W 90-26-57W 90-22-22W 90-28-57W	979.683 979.680 979.686 979.704 979.706	N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100		MS MS AR AR
132 139 1 36 51	U 238 U 243 U 75 V 193 V 194	1974C 1974C 1956D 1956D 1956B	1976G 1966G 1966G	1ST 1ST 1ST 1ST 1ST	61,488 61,652 71,125 58,295 58,238	L23280 L23280 L15801 L15801 L15801 L15801	34 - 50 - 45N 34 - 54 - 41N 34 - 59 - 55N 34 - 43 - 29N 34 - 37 - 47N	90-16-54W 90-11-21W 90-08-06W 90-14-35W 90-13-22W	979.697 979.702 979.714 979.685 979.682	N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100		MS MS TN MS MS
64 213 133 140 5	V 195 V 218 V 238 V 243 W 193	1956D 1976B 1974B 1974C 1956D	1966G 1976G 1974N	1ST 1ST 1ST 1ST 1ST	57.061 67.183 62.172 61.631 62.927	L 15801 L 24103 L 24103 L 23280 L 15801	34-31-58N 34-51-22N 34-50-46N 34-55-18N 34-57-49N	90-11-41W 90-23-23W 90-17-48W 90-10-47W 90-09-01W	979.680 979.705 979.698 979.703 979.710	N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100 0100		MS AR MS MS MS
52 62 82 135 141	W 194 W 195 W 237 W 238 W 243	1956D 1956D 1974B 1974C 1974C	1966G 1966G 1976G	15T 15T 15T 15T 15T	55.995 55.713 57.969 60.369 64.680	L15801 L15801 L23280 L23280 L23280 L23280	34~36-50N 34-32-34N 34-38-26N 34-51-01N 34-55-51N	90-13-11W 90-11-49W 90-22-33W 90-15-36W 90-09-59W	979.682 979.681 979.687 979.696 979.703	N 3 4 0 9 N 3 4 0 9	0100 0100 0100 0100 0100 0100		MS MS MS MS MS

Figure 2-7c.—Alphanumeric indexes for 30' quadrant of Vertical Control Data.

~.

NOAA Manual NOS NGS 3, Geodetic Leveling

2-12

US DEP NOS - ROCKV	PARTMENT OF COMMERCE NATIONAL GEODETIC S ILLE MD 20852 - APR	- NOAA URVEY 1980	NAT	V IONAL	LIST O /ERTICAL GEODETIC	F CONTENT CONTROL [ VERTICAL	FS DATA . DATUM 1929	)	LAT SPAN LON SPAN DIAGRAMN	34-30.1N 90-00.1W NI 15-6	TO 34- TO 90- INDEX F	59.9N 29.8W G02
SEON	DESIGNATION	YEAR ESTABLISHED	LAST RECOV	EL ORD (	EVATION	SOURCE	APPROXIMATE LATITUDE	POSITION LONGITUDE	SURFACE GRAVITY	OTHER CONTROL	QUAD OS	N ST
41 42 43 44 45	E 194 R 194 S 194 T 194 U 194	1956B 1956D 1956D 1956D 1956D	1966G 1966G 1966G 1966G 1966G	1ST 1ST 1ST 1ST 1ST	57.239 56.688 56.547 56.160 55.362	L15801 L15801 L15801 L15801 L15801	34-41-56N 34-41-06N 34-40-16N 34-39-26N 34-38-40N	90-14-12W 90-14-13W 90-14-10W 90-14-08W 90-14-02W	979.685 979.684 979.684 979.683 979.683 979.683	N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0	90100 90100 90100 90100 90100 90100	MS MS MS MS
46 47 48 49 50	186 T 116 T 117 F 194 PTS 30	UNK D UNK D UNK D 1956B 1908D	1966X 1966G 1966G 1966G 1966G	1ST 1ST 1ST 1ST 1ST	56.668 58.793 57.354 56.599 56.854	00726 L15801 L15801 L15801 L15801	34 - 38 - 25N 34 - 38 - 10N 34 - 38 - 03N 34 - 38 - 04N 34 - 38 - 01N	90-13-51W 90-13-38W 90-13-35W 90-13-32W 90-13-34W	979.683 979.682 979.682 979.682 979.682 979.682	N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0	90100 90100 90100 90100 90100 90100	MS MS MS MS
51 52 53 54 55	V 194 H 194 M 194 TBM A STA 146 TBM B STA 146	1956B 1956D 1956D 1953D 1953D	1966G 1966G 1966G 1966G 1966G	1ST 1ST 1ST 1ST 1ST	58.238 55.995 56.987 57.262 57.319	L 15801 L 15801 L 15801 L 15801 L 15801 L 15801	34 - 37 - 47N 34 - 36 - 50N 34 - 35 - 59N 34 - 35 - 56N 34 - 35 - 56N 34 - 35 - 56N	90-13-22W 90-13-11W 90-13-03W 90-13-00W 90-13-01W	979.682 979.682 979.682 979.682 979.682 979.682	N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0	90100 90100 90100 90100 90100 90100	MS MS MS MS
56 57 58 59 60	183 Z 195 Y 195 P-14-13 H 194	UNK C 1956D 1956D 1940D 1954B	1956N 1966G 1966G 1966G 1966G	1ST 1ST 1ST 1ST 1ST	55.768 56.316 56.819 61.029 60.886	00726 L15801 L15801 L15801 L15801	34 - 35 - 39N 34 - 35 - 00N 34 - 34 - 43N 34 - 34 - 08N 34 - 34 - 09N	90-12-59W 90-12-51W 90-12-49W 90-12-30W 90-12-30W	979.682 979.682 979.682 979.681 979.681 979.681	N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0	90100 90100 90100 90100 90100 90100	MS MS MS MS
61 62 63 64 65	X 195 W 195 PTS 32 V 195 U 195	1956D 1956D UNK D 1956D 1956D	1966G 1966G 1966X 1966G 1966G	1ST 1ST 1ST 1ST 1ST	56.696 55.713 55.306 57.061 54.124	L15801 L15801 00726 L15801 L15801	34-33-28N 34-32-34N 34-32-09N 34-31-58N 34-31-04N	90-12-10W 90-11-49W 90-11-42W 90-11-41W 90-11-41W	979.681 979.681 979.680 979.680 979.680 979.680	N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0	90100 90100 90100 90100 90100 90100	MS MS MS MS
66 67 68 69 70	T 195 S 195 PTS 33 CONCRETE MON 2 H 245	1956D 1956D 1908D UNK D 1974B	1966G 1966G 1967G 1956N	1ST 1ST 1ST 1ST 1ST	56.793 56.381 56.891 57.619 54.296	L15801 L15801 L15801 00726 L23280	34 - 30 - 17N 34 - 30 - 11N 34 - 30 - 10N 34 - 30 - 07N 34 - 30 - 30N	90-11-42W 90-11-51W 90-11-39W 90-11-47W 90-25-53W	979.679 979.679 979.679 979.679 979.679 979.677	N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0	90100 90100 90100 90100 90100	MS MS MS MS
71 72 73 74 75	C 245 D 245 F 245 G 245 J 245	1974D 1974C 1974C 1974D 1974B		1ST 1ST 1ST 1ST 1ST	54.362 55.091 56.735 55.160 56.363	L23280 L23280 L23280 L23280 L23280 L23280	34 - 31 - 33N 34 - 32 - 22N 34 - 33 - 13N 34 - 34 - 13N 34 - 34 - 13N 34 - 34 - 59N	90-25-30W 90-25-13W 90-24-54W 90-24-32W 90-24-32W 90-24-15W	979.679 979.680 979.681 979.683 979.683 979.683	N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0 N 3 4 0	90100 90100 90100 90100 90100 90100	MS MS MS MS
76 77 78 79 80	CLAYTON RM 1 CLAYTON CLAYTON RM 2 CLAYTON AZ K 245	1956C 1967C 1956C 1956C 1956C 1974C	1974G 1974G 1974G 1974G 1974G	1ST 1ST 1ST 1ST 1ST	58.021 57.727 57.790 57.555 57.281	L23280 L23280 L23280 L23280 L23280 L23280	34-35-56N 34-35-51N 34-35-56N 34-36-18N 34-36-18N 34-37-10N	90-23-53W 90-23-55W 90-23-55W 90-23-55W 90-23-46W 90-23-27W	979.684 979.684 979.684 979.685 979.685 979.686	N 340 H N 340 N 340 N 340 N 340 N 340	90100 90133 00 90100 90100 90100 90100	MS 102 MS MS MS MS

Figure 2-7d.-Line order index for 30' quadrant.

Ç

Ć

(

DESIGNATION	ACRN #	POSITION
A 10	BG0922	0300864
A 10 AZ MK	BG0921	0300864
A 10 BM	BG0524	0300864
A 10 RM 1	BG0923	0300864
A 10 RM 2	BG0924	0300864
A 11	BG2640	0300874
A 113	BG0222	0300871
A 115	BG0097	0300871
A 124	BG1001	0300864
A 125	BG1035	0300864
A 136	BG1036	0300864
A 137	BG2035	0300863
A 138	BG1115	0300864
A 139	BG1636	0300861
A 140	BG1458	0300861
A 141	BG1434	0300861
A 156	BG0907	0300864
A 158	BG0002	0300871
A 16 RM 1 A VITRO	BG1021	0300864
A 16 RM 1 B VITRO	BG1022	0300864
A 16 RM 2 A VITRO	BG1019	0300864
A 16 RM 2 B VITRO	BG1020	0300864
A 16 VITRO	BG1018	0300864
A 161	BG1817	0300872
A 162	BG0042	0300871
A 163	BG1706	0300872
A 163 RESET 1964	BG1707	0300872
A 165	BG0257	0300871
A 168	BG0498	0300864
A 169	BG2615	0300864
A 178	BG2255	0300802
A 179	BG2073	0300863
A 180	BG2081	0300863
A 181	BG1466	0300861
A 182	BG0207	0300862
A 182 RESET 1962	BG2275	0300862
A 25	BG1711	0300872
A 26	BG1772	0300872
A 27	BG2156	0300862
A 28	BG0800	0300864
A 29	BG1120	0300864
A 297	BG1974	0300863
A 299	BG1921	0300863
A 302	BG0137	0300871
A 303	BG0125	0300871
A 380	BG3435	030087123
A 395	BG3462	030086341
A 409	862644	0300874
Δ 436	BG2495	0300874
Δ 437	BG2528	0300874
A 86	860025	0300871
A 07 A 85	BC0387	0300864
X UJ	200307	000004

Figure 2-8a.—Listing of synoptic file.

ACRN #	DESIGNATION	POSITION
BG0001	I 8	0300871
BG0002	A 158	0300871
BG0003	B 158	0300871
BG0004	J 8	0300871
BG0005	C 158	0300871
BG0006	КВ	0300871
BG0007	D 158	0300871
BG0008	E 158	0300871
BG0009	L 8	0300871
BG0010	F 158	0300871
BG0011	G 158	0300871
BG0012	M 8	0300871
BG0013	H 158	0300871
BG0014	N 8	0300871
BG0015	K 158	0300871
BG0016	J 158	0300871
BG0017		0300871
BG0018	L 158	0300871
BG0019	P 8	0300871
BGUUZU	M 158	0300871
BGUUZI	N 158	0300871
B60022		03000/1
B60025 BC0026	P 150	0300071
	A 87	0300871
B60025	A 04 B 84 PECET 1939	0300071
BC0027	D 04 KCOCT 1737	0300871
BG0028	C 84 PESET 1939	0300071
860029	C 84	0300871
860030	D 84	0300871
BG0031	D 84 RESET 1939	0300871
BG0032	E 84 RESET 1939	0300871
BG0033	E 84	0300871
BG0034	F 84	0300871
BG0035	G 84	0300871
BG0036	H 84	0300871
BG0037	J 84	0300871
BG0038	К 84	0300871
BG0039	L 84	0300871
BG0040	R 158	0300871
BG0041	S 158	0300871
BG0042	A 162	0300871
BG0043	B 162	0300871
BG0044	Т8	0300871
BG0045	C 162	0300871
BG0046	U 8	0300871
BG0047	D 162	0300871
BGUU48	QUINIEITE AZ MK	0300871
BG0049	QUINIEITE 2 AZ MK	0300871
860050	QUINIEIIE KM I	0300871
860051	QUINIEIIE KM A	0300871
BGUU52	QUINIEIIE KM Z	0300871
860053	E 162	03008/1

Figure 2-8b.—Listing of synoptic file.

States	Codes	States	Codes		
Alabama (AL)	873	Montana (MT)	931		
Alaska (AK)	945	Nebraska (NE)	924		
Arizona (AZ)	938	Nevada (NV)	936		
Arkansas (AR)	881	New Hampshire (NH)	842		
California (CA)	941	New Jersey (NJ)	853		
Colorado (CO)	934	New Mexico (NM)	937		
Connecticut (CT)	846	New York (NY)	851		
Delaware (DE)	855	North Carolina (NC)	865		
District of Columbia (DC)	859	North Dakota (ND)	922		
Florida (FL)	872	Ohio (OH)	892		
Georgia (GA)	867	Oklahoma (OK)	927		
Hawaii (HI)	161	Oregon (OR)	943		
Idaho (ID)	932	Pennsylvania (PA)	854		
Illinois (IL)	896	Rhode Island (RI)	845		
Indiana (IN)	895	South Carolina (SC)	866		
Iowa (IA)	912	South Dakota (SD)	923		
Kansas (KS)	925	Tennessee (TN)	882		
Kentucky (KY)	883	Texas (TX)	877		
Louisiana (LA)	876	Utah (UT)	935		
Maine (ME)	841	Vermont (VT)	843		
Maryland (MD)	857	Virginia (VA)	863		
Massachusetts (MA)	844	Washington (WA)	944		
Michigan (MI)	901	West Virginia (WV)	864		
Minnesota (MN)	911	Wisconsin (WI)	902		
Mississippi (MS)	874	Wyoming (WY)	933		
Missouri (MO)	884	United States	83x		
Channels	Codes	Lakes	Codes		
St. Lawrance Diver	831.1977	Lake Champlain			
Niagara Diver	004 June	Lake Ontario	043-1222		
Detroit Piver	200-2XXX 904-4xxx	Lake Offanio Labe Eria	203-2444		
St Clair Diver	204-4AX 001 4888	Land Elle	200°3XXX		
	701-4XXX 001 0000	Lake Upon	202-4222		
Grand Diver	271-7777 006 4444	Lake Michigan	201-2888		
OTABLE KIVER	9US-XXXX	саке мисиндан	700-/XX)		

896-1111

902-01xx

902-03xx

907-6xxx

911-05xx

Lake Winebago

Lake Superior

Lake of the Woods

Rainy Lake

Table 2-2.-Location codes for tide and water-level stations

A mark setting unit should also contact officials of the local highway districts, county and city transportation departments, Indian reservations, and other public and private agencies as required when traversing their property. Ask district officials about current and proposed highway widening projects. Enlist their assistance to avoid the effects of such projects when routing the line. Learn the construction details of culverts and bridges, and the locations of bedrock. Often, local officials can point out or specify existing monuments that will be destroyed, assist in recovery work, and recommend suitable locations for new bench marks.

Since highway right-of-way zones are often occupied by gas, water, or sewer pipelines, and electric or telephone cables, determine the location of such utilities before installing a monument. Look for signs marking pipelines or cables and obtain the assistance of a representative from the local servicing office if a rod must be driven near a pipe or cable.

902-02xx

909-9xxx

911-01xx

911-09xx

ŕ

Although most vertical control points are located on public land, occasional points may require that a private property owner or manager be approached as well. Sometimes, in addition to the owner's permission, special information must be obtained, such as the age and construction details of a building or the presence of active irrigation wells adjacent to a proposed monument site. When ownership cannot be readily determined, consult the county Recorder of Deeds. Never search for or establish a control point without first obtaining the property owner's permission.

When requesting permission from property owners to use their property, give them a fact sheet that de-

Chicago River

Milwaukee River

St. Marys River

Rainy River

Fox River

scribes the purpose of the survey. The fact sheet should also explain how the monument will be constructed and the importance of the monument to the vertical control network. An example is shown in figure 2-9. Always behave in a respectful and tactful manner and obey the homeowners' wishes concerning the use of their property.

Maintain a record of each contact made, including the designations of the relevant control points and the



UNITED STATES DEPARTMENT OF COMMERCE National Oceanic and Atmospheric Administration NATIONAL OCEAN SURVEY Rockville. Md 20852

#### NATIONAL GEODETIC SURVEY - INFORMATION NOTICE

The National Geodetic Survey (NGS), an office of the National Ocean Survey, NOAA, Department of Commerce, will be conducting a Geodetic Survey in your area in the near future. In preparation for this work, NGS field units are now installing survey points, known as bench marks, at intervals of approximately one mile along major highways and other thoroughfares. These bench marks will be tied to the national network of thousands of similar points throughout the United States which form the framework for a precise network of vertical control stations (points of precisely known elevations). This survey work is undertaken at the request of Federal, state, and local agencies for use in such projects as mapping, resource management, flood control, crustal motion studies (geodynamics), space exploration and large-scale engineering projects.

At selected locations throughout the country, special high quality bench marks are being established as part of a seven-year program to redefine the entire national geodetic vertical control network. These marks often take the form of brass disks cemented into bedrock outcrops or large buildings. Where this kind of mark is not feasible, a special truck-mounted drill is used to install a sleeved rod bench mark (as described in Figure 1.) Due to the stringent accuracy requirements placed on the elevations of these bench marks, particular care is taken when they are established. The mark must be well constructed, unlikely to be disturbed, and accessible to the general public.

Because of the above considerations, selection of an acceptable site is crucial. Thus, it is often necessary to rely on the public spirit of land owners for permission to establish survey stations on private property. It is the policy of the National Geodetic Survey to install bench marks <u>only</u> upon the informed consent of the property owners. An NGS bench mark can provide a valuable public service, and enhance property values by serving as a reference for any surveying done in the immediate area.

Once it is installed the elevation is precisely determined by a National Geodetic Survey vertical control party. This elevation is then published along with the description of the physical location of the bench mark and its position (latitude and longitude). The bench mark then takes its place in the national network as a basic reference point for a variety of surveying, mapping, planning, engineering and scientific applications.

Enclosure



# NATIONAL GEODETIC SURVEY VERTICAL CONTROL MARKER

In 1978 the National Geodetic Survey (NGS) introduced a new, improved bench mark into the National Vertical Control Network. The reference point for the elevation is the top of a stainless steel rod. The rod is located inside a protective aluminum casement that bears the NGS logo and the stamped bench mark designation. Users can obtain access to the rod by lifting a hinged access cover.

The bench mark is designed to prevent near-surface soil movements, e.g., frost heave, shrinkage, swelling, from disturbing the mark. This is accomplished by encasing the rod in a lubricated sleeve to the depth of

expected soil movement and by anchoring the rod in the soil below.

Top and side views of the bench mark are depicted below. Additional information about this mark can be obtained by writing to the Director, National Geodetic Survey, National Ocean Survey, NOAA, Rockville, Md. 20852



SIDE VIEW



**U.S. DEPARTMENT OF COMMERCE** National Oceanic and Atmospheric Administration National Ocean Survey



name, address, and telephone number of the individual contacted. Upon completion of the project, this record is used to send published descriptions of the points to

private owners and public managers as a courtesy. Requests for survey results and elevation data should be referred to the National Geodetic Information Center.
## 2.3.3 Routing Specifications

The actual routes to be followed by the lines of a geodetic leveling project depend primarily on the purpose of the project, as do the types and spacing of control points along each route. Given in this section are routing specifications appropriate to lines of the National Geodetic Vertical Network.

Junctions. Lines in a network begin and end at clusters of control points called junctions. In a large network, considerable time (sometimes several years) may elapse between the completion of one line and the start of another; therefore, the monuments at junctions must be permanent and stable The most desirable locations for junctions are areas without significant subsidence or uplift relative to a majority of the area covered by the network. Though cities often serve as the logical intersections for lines routed along highways, cities may not be the best locations for junctions. Locate a junction in the most stable zone possible, even if it is necessary to detour as much as 10 km (6 mi) from the logical intersection of the lines. (See the examples in fig. 2-10.)

To establish the junction, recover or set three control points of the best possible quality. Locate them at intervals of at least 0.5 km (0.3 mi). This limits the possibility that they might all be affected by the same local disturbance. At an international boundary, locate the points along the border, if possible, so they are accessible to surveyors from either country. All three points must be leveled in every line beginning or ending at the junction.

At the end of the field season or project, if the leveling along a line must stop at some location short of the ending junction, the location should be chosen carefully and mommented with three control points of the best possible quality, as described in the previous paragraph. These are not strictly considered to be junction points, but they should be releveled for a tie when the line is resumed.



Figure 2-10.-Network junctions.

*Routes.* The mark setter must weigh many factors when routing the line Paramount among these are the permanence, stability, and accessibility of the control points. Choose a route which permits the greatest proportion of control points to meet these standards. Since the lines of the national network are, for the most part, already in place, routing typically involves recovering or relocating an existing line.

In the past, lines of leveling usually followed railroad right-of-way zones, which often provided the only negotiable path through undeveloped areas. Today, these zones are not always accessible or convenient to the person who will be using local vertical control. The improvements in mark setting and leveling efficiency made possible by the use of vehicles depend on greater accessibility. Therefore, whenever a railroad does not have a continuous access road available to both public and private surveyors within five leveling setups (at the most 0.5 km or 0.3 mi), relocate the line to the edge of a highway right-of-way zone which parallels the railroad. Special connections should be made to the existing line. (See sec. 2.3.4, "Previous surveys.")

In the vicinity of an interstate highway or other controlled access roads, whenever possible, route the line so it is accessible from parallel roads outside the rightof-way fence. Avoid routing the line along the proposed route of a new highway, but if this is unavoidable, route the line as close as possible to the edges of the proposed right-of-way zones.

Leveling experience is helpful when selecting a route across rough terrain or through congested urban areas. Unlike a control point for triangulation, a control point for leveling must be readily accessible by land from the points that exist before and after it in the line. Avoid routing the line through short, extreme changes in elevation, such as down gullies or up and over embankments. When a monument must be set in such an inconvenient location to ensure permanence and stability, note the most efficient leveling route on the log.

If an obstacle is encountered that cannot be crossed by ordinary leveling—such as a river, bay, mudflat, marsh, or deep ravine—a "river crossing" technique may be required. This procedure may also be required if a bridge cannot be leveled across, either because it is subject to intense vibration or because traffic cannot be stopped for the length of time necessary. See sec. 4.2.1 for selecting and preparing river crossing sites.

Spurs. Portions of line that branch to control points not in the main line are called spurs. Spurs end on points not otherwise connected to the network, unless they are routed to make connections with previous surveys. Avoid creating spurs of less than 1.0 km (0.6 mi). Control points located closer than 1.0 km should be leveled as part of the line.

If routing a new line along an existing line that includes spurs, the spurs may be excluded unless they provide access to control points of unusually high quality or they are specified for inclusion by the project instructions. Control points. The spacing between control points in the national network should average 1.6 km (1.0 mi). In dense networks, designed to measure such phenomena as movement near faults or local subsidence, spacing should average 1.0 km (0.6 mi) or less

The distance between any pair of control points may normally range between 1.0 and 3.0 km (0.6 and 1.9 mi). In municipalities, housing or commercial zones, and other areas of intensive development, shorten the spacing to 1.0 km. In rural areas, with gentle terrain and only infrequent periods of windy or stormy weather, spacing may be increased.

If a portion of the line must follow a steep grade, space the control points to limit the number of leveling setups required to 40 or less. If the steep grade extends over several kilometers or the entire line, adopt an average spacing which corresponds to the slope. Estimate the slope (elevation difference divided by distance) from a topographic map and consult table 2-3 for a suitable spacing.

Control points should be marked with monuments of the best expected quality. However, new monuments of the best quality may be the most expensive and timeconsuming to set Furthermore, existing monuments of lower expected quality historically have often performed well and should not be removed from the line unless they are in unsatisfactory condition. Consequently, bestquality monuments need not be installed at every control point. Unless project instructions specify otherwise, recover or set a monument of the best possible quality at the following locations (fig. 2-11):

- 1. At network junctions, including international boundaries (set three monuments).
- 2. At one or both ends of a portion of line not to be leveled to both junctions during one field season (set three).
- 3. At the base of a spur or spurs connecting a previous survey that intersects the line (set one).



Figure 2-11.—Locations for monuments of the best possible quality.

- 4. At the base of a spur connecting a primary tide station (set one).
- 5. At a primary tide station (set one).
- 6. At least every 16 km (10 mi) along the line (set one).

## 2.3.4 Connections

To provide a common frame of reference in which elevations may be computed, two or more ties should be established between the line or lines of every level-

Γa	ıble	2-3	.—Sr	acing	control	points	along	slopes	
					www	b conserved.		0102000	

This table is based on a change in elevation of 2 m (6 ft) during each setup, the normal maximum when following the specifications for geodetic leveling given in chapter 3.

Slope (percent)	Maximum leveling sighting distance m (ft)	Maximum distance between control points km (mi)	Recommended average spacing km (mi)
1	Maximum tolerance	3 0 (1.9)	1.6 (1.0)
2	50 (164)	30(19)	1.6 (1.0)
3	33 (108)	27(17)	1 1 (0 7)
4	25 (82)	20(12)	0.8 (0.5)
5	20 (66)	16(1.0)	06(04)
6	16 (52)	1 3 (0.8)	0 5 (0.3)
8	12 (39)	10(0.6)	04(0.25)
10	10 (33)	0 8 (0.5)	0.3 (0.2)
15	6 (20)	0 5 (0 3)	02(01)
30	3 (10)	0 3 (0 2)	0 1 (0 05)

ing project and the National Geodetic Vertical Network. A tie consists of two or more control points that, after they have been leveled and checked, connect a line to a previous survey

When routing the lines of a project, connecting points must be recovered to permit ties to be made that span the area of the project. To the extent that they serve the purpose of the project, connections should also be routed to other types of control networks. Given next are specifications for connections to be made when routing lines of the national network. The requirements are summarized in table 2-4.

*Previous surveys* When lines of a previous survey are releveled or intersected as part of a project, the ties established permit the previous survey to be readjusted and brought up-to-date with the current work.

If all or part of an existing line is to be releveled, route it by recovering the control points as they are described in the vertical control data. If the existing route is no longer accessible (as is the case with many lines following railroads) and the line must be relocated, recover the points of the best quality and include them in the new, parallel route. If possible, include such points at the following intervals along the line: 5 km (3 mi) if the distance separating the new and old routes is less than 0.5 km (0.3 mi), 10 km (6 mi) if the separation is between 0.5 and 2.0 km (0.3 and 1.2 mi), and 20 km (12 mi) if the separation is greater than 2.0 km(1.2 mi). (See fig. 2-12.)

Make a connection whenever the route intersects a line of a previous survey. If both current and previous surveys are of first-order precision, recover at least three control points that were leveled during the previous survey. Connect them with a spur (or spurs), recovering or setting a monument of the best possible quality at or near the point where the spur joins the main line (fig. 2-13). All three points must be leveled to establish a tie. If either or both of the surveys are of less than first-order precision, recover at least two control points for the connection.

Often "area surveys" are encountered, where two or more lines of a previous survey are intersected. In this case, make connections by recovering at least one control point from the previous survey at each intersection (fig. 2-14). At least three such points, spanning the area of the survey, must be leveled to establish a tie.

A local survey of good quality that has not previously been included in the national network may be connected by following these procedures. Consult the appropriate local officials to obtain descriptions of monuments that might serve as connecting points.

Do not extend a spur more than 8 km (5 mi) to make a connection to a previous survey, unless special project instructions apply.

Tide and water-level stations. Connections between the national network and primary tide stations are essential to compare and relate elevations to the sea-level surface at various points along the coasts of the United States. Similiarly, connections to water-level stations permit elevations to be obtained for the mean lake level at various points along the shores of the Great Lakes. Project instructions should provide a list of tide and waterlevel stations to be connected.

Numerous secondary and tertiary tide and water-level stations may be located near a line routed along a shore. Unless project instructions specify otherwise, connect these only if they are no more than 4 km (2.5 mi) from the line.

Connection	Spacing	Minimum number of connecting points
Previous surveys:		
Parallel, 0.5 km (0.3 mi) or less away	Every 5 km (3 mi)	1
Parallel, 0 5 to 2.0 km (0 3 to 1 2 mi) away	Every 10 km (6 mi)	1
Parallel, more than 20 km (1.2 mi) away	Every 20 km (12 mi)	1
Intersected, first-order line	At every intersection	3
Intersected, second-order line	At every intersection	2
Intersected, area survey	At two intersections that span the area	3
Tide and water-level stations:		
Primary	At every station	5
Secondary	At stations less than 4 km (2.5 mi) away	3
Tertiary	At stations less than 4 km (2.5 mi) away	3
Active airports	At airports not leveled	
	before and less than 4 km (2 5 mi) away	1
Gravity stations	At every station	1
Horizontal control and three-dimensional stations	At stations not leveled	
	before and less than 4 km (2 5 mi) away	1

Table 2-4.—Requirements for connections

0

**Connecting point** 



Figure 2-12.—Connections to parallel surveys.

Each primary or secondary station should have a cluster of at least five control points, a staff or electric-tape gage for measuring the height of the surface of the water relative to the points, and a water-level recorder that operates continuously. Obtain descriptions from the most recent station report. Contact the station observer to make arrangements for the leveling unit to level to the staff or gage. If no monument of quality A (sec 2.4.4) is present at a primary station, set one. Connect the points to the main line with one spur, recovering or setting another mark of quality A at the point where the spur joins the line.

Airports. Connections to control points at airports provide elevations for aeronautical charting. If such points have already been leveled as part of the national network, they need not be releveled. Connect an active airport not previously leveled only if it is within 4 km (2.5 mi) of the main line, unless the connection is specifically requested in project instructions. Recover or establish only one control point, near the terminal building, if possible, or at the entrance to the airport. To locate an existing control point or to establish a new one, check with a responsible official at the airport.

Other control. Leveling connections to control points for gravity, three-dimensional, and horizontal networks

are important for monitoring crustal motion and for continual development of a unified, precise, global positioning system.

Relative measurements of gravity are made at many control points of the vertical network, so special connections to gravity stations are usually not necessary. However, connect stations where repeated measurements of absolute gravity are made, when they are listed in project instructions.

Instructions should also list any three-dimensional ("satellite") and horizontal stations that are to be connected. Otherwise, connect these stations only if they have not previously been leveled, are within 4 km (2.5 mi) of the line, and are readily accessible to the leveling unit. Consult the appropriate Geodetic Control Diagrams to determine if such stations are near the line.

At these types of stations, more than one monument may have been established. Consult the horizontal control data or other descriptions that have been provided and, if practicable, include all the monuments in the line. For example, horizontal control stations normally include a station mark, two or more nearby reference marks, and an azimuth mark a few tenths of a kilometer away If a spur is necessary, route it over all the monuments; avoid creating a "spur-on-a-spur."



Figure 2-13.—Connections to intersected surveys.

### 2.3.5 Records

While routing a line, the mark setter must prepare numerous plots, descriptions, and logs, all critical to the future recovery and leveling of the control points. Plots and descriptions identify and provide permanent instructions for reaching the individual points; these are discussed in sections 2.4.3 and 2.4.4. Logs provide specific instructions to the leveling unit for leveling the line.

Logs. The logs must include clear and concise instructions that enable a unit to level efficiently to the control points along the route. Sometimes simple handwritten lists are sufficient, which briefly describe the locations of the points by providing driving instructions and measurements from the most obvious reference objects. Working position plots should be supplied with such logs. More convenient as logs are the working plots themselves, on which the plotted points and routes between them are briefly described (fig. 2-15). Prepare logs on working plots as explained here.

As each control point is recovered or established along the route, plot the control point on the appropriate topographic map. Provide a brief description by giving measurements to a witness post and other landmarks, the height of the point relative to the ground, any special leveling procedures required (such as for a vertically mounted disk or one requiring the use of spacers) and any special permission or contacts needed. If the monument site is cleared and marked with flagging, only a few of the most useful reference objects and distances supplied in the description need be noted on the log. Use a broad, light-colored pen to connect the points in leveling order on the topographic map. Note the distances and the most efficient routes between them. Label a spur intended to tie a previous survey with the accession number of the survey. Whenever points are grouped in clusters that may cause confusion, attach a largescale sketch. Tide and water-level stations nearly always require such a sketch; attach the station reports if possible. In an urban area, a large-scale city map may provide better information than a topographic map; however, a city map should not be used to scale positions.

After each map is completed, scale the positions of the points and assign survey-point serial numbers, taking care to match them with the correct designations. (See sec. 2.4.4, "Survey-point serial number.") Label the lower right-hand corner with the line number and a number corresponding to the location of the map in the line. Fold the map, printed side out, to a convenient size.

Submitting records to the project office. Keep all mark setting records up to date as work proceeds along a line. Use the Index to Topographic Maps as a check-off list of the maps completed (fig. 2-6).

Collect and bind, in line order, descriptions of all points to be leveled, and label the binder with the line number. Bind descriptions of points not to be leveled as



Figure 2-14.—Connections to an area survey.



Figure 2-15.-Map prepared on a working plot.

each 30' quadrant of vertical control data is completed. Label these binders with the quadrant numbers. Submit the binders to the project office together with logs and smooth plots covering the same area

## **2.4 Vertical Control Points**

More than 500,000 points define the lines of the National Geodetic Vertical Network. To mark these points, monuments of many types have been set. Most of these are bench marks—monuments designed to provide permanent and stable points within the vertical frame of reference. The remaining points are marked by monuments originally designed to provide other types of control.

When routing a line of a leveling project, after existing control points are recovered, bench marks must be set to complete the routing requirements described in the previous pages. Properly recovered and set control points are the most important product of the mark setting unit. They must be permanent, stable, accessible, and properly described.

### 2.4.1 Control Point Recovery

An efficient recovery effort depends on adequate preparation and organization, as well as skill and experience at interpreting previous descriptions and preparing new ones. In addition to recovering or setting control points for the lines assigned by project instructions, NGS mark setting units are sometimes expected to recover and prepare descriptions for all points listed in the NGS data base and located in the area covered by the 7.5 maps that include the lines.

*Preparation.* First, collect and organize the data for existing control points along the line to be routed. Determine the topographic maps to be used and, if possible, plot the positions of the points to be recovered. As reminders, note on the maps the locations of line junctions and connections to be made with previous surveys, tide and waterlevel stations, active airports, and other control. Make arrangements to meet with appropriate officials to obtain permission and assistance. While recovering the points, keep track of their spacing, condition, and quality, noting where new bench marks should be set.

*Recovering a control point.* Follow the instructions given in the description for reaching each control point. Often a witness post and sometimes the monument can be seen from the vehicle after driving to the described location. However, this type of sighting is not an adequate recovery.

Before reporting a point recovered as described, be sure that the monument found is in fact the one previously described. Observe the stamping on the mark and check that it agrees exactly with that given in the description. Check the measurements and directions from nearby reference objects. If necessary, install a new witness post and make measurements to additional reference objects. If the point is to be leveled, clear away brush or obstructions, install footholds for the rodman, and do whatever else may be necessary to make it accessible to the leveling unit. A disk recessed in concrete may not have sufficient clearance to permit a standard leveling rod to be set on the highest point. Note in the log that spacers will be required to level to such a point.

To assist the unit in finding the point, tie a single piece of brightly colored flagging at eye level around a nearby witness post, utility pole, tree, fence, or guardrail. Chalk may be used to draw attention to a mark set in stone or concrete, but only with the property owner's permission.

Before leaving the site, plot and identify the exact location of the control point and prepare a description. The type of description prepared depends on the type of control which will be provided by the point. For example, a point that is to provide both vertical and horizontal control must have two descriptions, one for each part of the data base. Vertical control descriptions are discussed in sec. 2.4.4. Horizontal control descriptions are discussed in the Federal Geodetic Control Committee publication, *Input Formats and Specifications of the National Geodetic Survey Data Base* (Pfeifer 1980: vol. I), the National Geodetic Survey Operations Manual (Greenawalt and Floyd 1980), and C&GS Special Publication 247, Manual of geodetic triangulation (Gossett 1959).

When recovering a relocated point ("reset"), confirm that the original monument has been destroyed. Search for both monuments if both descriptions are available. Since the two are likely to be close together, take care that they are not confused. Prepare a separate description for each monument.

If a control point apparently cannot be recovered, recheck the description, especially the starting location, distances, turns, landmarks, and supplemental notes. Compare distances to those given for points located immediately before and after it in the line. To find a likely site, compare elevations from map contours to the published elevation for the point. Look for remnants of flagging, signal cloth, wood, witness posts, or concrete. Pinpoint the likely location of the mark by measuring from the reference objects, then clear away any brush and dig. A metal detector may be helpful. Never report a monument destroyed unless enough of the metal surface is recovered for positive identification. If no identifiable part of a monument can be found, report in the station description that the monument was not recovered and note the amount of time spent searching.

Destroyed bench marks. If a destroyed or severely damaged NGS bench mark is recovered, NGS employees or other agents designated by the agency may remove what remains of the monument from the site. Report that it is destroyed in the description and return marks bearing a stamped designation to the project office for disposal. Report destroyed or damaged monuments of other agencies, but do not remove such monuments unless the agency specifically requests it. Occasionally a monument that is scheduled to be destroyed (e.g., during highway widening), or one that is unlikely to prove reliable in the future because of lowquality construction (quality D) or poor condition, is recovered. Set a new bench mark of better quality in the immediate vicinity and route a spur from the new mark to the old. After the spur has been leveled and checked, destroy the old monument if the responsible agency so requests, and prepare a new description accordingly. If an unreliable monument is not destroyed, be sure to note in the original recovery description that it is not recommended for future geodetic leveling.

The cap-and-bolt is an example of a type of monument likely to prove unreliable because of its construction. The monument includes two control points, a bolt set in concrete 1 to 2 m beneath the ground surface and a cap on a pipe extending to the surface from the bolt. In the past, both points were leveled from a temporary bench mark set nearby and three elevations were published (cap, bolt, and reset cap). This practice was very costly and has been discontinued. Because the monument is of quality D, replace it as described in the preceding paragraph. Since standard leveling rods are too wide to fit into the pipe, instruct the leveling unit to level only to the cap.

## 2.4.2 Bench Mark Setting

Only bench marks of the best possible quality (A or B) should be set to complete routing requirements for a line. The mark setting manual provides instructions for selecting sites and constructing such monuments. Information presented here is intended to supplement and update the manual.

Designations for bench marks. Only NGS employees and agents may set brass disks and aluminum flanges precast with the NGS logo. Such marks must be stamped with designations supplied by the agency in project instructions or via computer. Each designation includes a letter of the alphabet and a number, up to four digits. It should be unique among all the designations in the State where the control point is located.

Designations are normally assigned to an individual mark setter in a quantity sufficient for the line to be routed Several series are normally assigned; each includes one number and 24 letters of the alphabet. ("I" and "O" are omitted because they may be confused with "1" and "O".) For example: A 200-km leveling line, with the space between control points averaging 1.6 km and the recovery rate expected to average 50 percent, requires approximately 63 designations or three series, such as A 123 through Z 123, A 124 through Z 124, and A 125 through Z 125.

To avoid duplicate designations, keep a record of all designations that have been used (fig. 2-16). If a mark is set that bears a stamping duplicating that of another mark in the same State, stamp it with the letter "X" following the number if it has not been leveled. For example, "C 124" becomes "C 124 X." Be sure to correct the plots, the description, and the log. If the duplication is discovered after the mark has been leveled, do not stamp the mark but notify the project office. (See sec. 5.3.3, "Synoptic file comparison.")

Site selection. Set bench marks in bedrock whenever possible. In a region where expansive soil or local subsidence is prevalent, extend a spur as much as 4 km (2.5 mi) to set a high-quality bedrock mark.

# RECORD OF BENCH MARK DESIGNATIONS ASSIGNED BY MARK SETTER



Figure 2-16.—Example of bench mark designations assigned by mark setter.



When selecting sites for monuments, remember the length (3 m or 10 ft) and width (7.6 cm or 3 in) of the standard leveling rod. Avoid setting marks that are awkward to level, such as those sketched in figure 3-78 (chapter 3). When setting a disk in bedrock, chip away the surrounding rock and level the disk so the highest point corresponds to the center of the disk. Hold a carpenter's level vertically, as though it were a miniature leveling rod, to test this. If a disk must be set vertically in a structure, place it about 1.0 m (3 ft) above ground and make sure the long line at the center is horizontal.

Massive bridges. A massive bridge across a wide river or valley or an overpass of an interstate highway may provide a high-quality site for a disk. Such a bridge normally has five structural members. The deck is the surface on which vehicles are driven. It is made up of several sections, separated by expansion joints, and is designed to yield to various forces. The abutments provide support to the ends of the deck. The piers provide support between the ends of the deck. Abutments and piers are in turn supported by foundations. At the ends of a bridge may be wingwalls, which are retaining walls built to hold back earthen materials.

Three types of foundations are commonly used (fig. 2-17). A spread foundation is essentially a slab that provides support by transferring the load over a large area of the soil. A friction pile transfers the load to the soil through friction at the interface of the pile and the

soil. An end-bearing pile transfers the load to a resistant subsurface formation or bedrock. Piles are usually placed in groups and topped with a pile cap. Abutments and piers rest on the pile caps. A pile cap cannot be visually distinguished from a spread foundation. Check with the local highway engineer to learn the types and depths of the foundations of bridges you might use.

To be suitable for a monument of quality A, the foundation of the bridge should rest directly on bedrock or should be made up of end-bearing piles. Where soil is nonexpansive and maximum frost depth is less than 0.5 m (1.6 ft), the foundation may be made up of friction piles. The disk must be set on an abutment, pier, or pile cap.

Rod marks. Since the publication of the mark setting manual, various driving points (fig. 2-18) have been devised. When driving a string of stainless steel rods that have been screwed together, the wings of the driving point prevent the rods from unscrewing as a result of vibration and rotation induced by the hammering of the rock drill. Use these points when setting class A or B rod marks as described in the manual.

A rod string must be driven deeply enough to resist movement caused by hammering or pulling the rod at the surface. Initial experience in driving rods with a gasoline-powered rock drill indicates that, to ensure sufficient stability, a driving rate slower than 30 cm/min (1 ft/min) must be attained at or below the minimum



Figure 2-18.—Driving point for a rod mark.

depth required for the site. (See table 3 in the mark setting manual.) If driving through clay, do not drive more than 10 m (30 ft) when trying to attain this rate; after the rod has been in place for a few days, the clay should "set up" around it, anchoring the mark in place. If possible, return to such a mark to check that it has become anchored.

Two changes have been made to the instructions for a class B rod mark. First, instead of using a plastic cleanout fitting, protect and identify the control point with a casement exactly like the one used for a class A rod mark. (The 4-inch fitting used in the past is too narrow to permit a standard leveling rod to be placed with certainty on the highest point of every such mark.) Second, instead of crimping a brass disk to the top of the rod, round off the top end with a file or crimp a stainless steel cap onto the rod, as for a class A rod mark.

Before crimping on a stainless steel cap for either a class A or B rod mark, stamp the designation around the top of the cap, just below the edge of the rounded portion (fig. 2-19). This permits backup identification in the event that the flange of the casement is later mutilated or destroyed.

To crimp the cap, use a small hydraulic pump with a cutting attachment. After slipping the cap over the end of the rod, crimp it into the rod about halfway down from the top. Be sure that the cap cannot be raised, lowered, or twisted after crimping. Construct the casement so a leveling rod may be set and rotated on the control point without restriction.

## 2.4.3 Assigning Geographic Positions

The geographic positions of a small number of points in the national network of vertical control have been precisely determined by horizontal surveys. However, a position, even if approximate, should be assigned to every point for the following reasons: to permit automatic plotting and filing of vertical control data in the appropriate 30' quadrants, to permit interpolation of gravity values for correcting the leveling data, and to assist recovery without relying solely on a description of the route to the point.

Working plot. To obtain approximate positions for control points, maintain a working plot ("rough plot"). This is a 7.5 topographic map (or 15' map if the larger scale is not available) on which latitude and longitude lines have been penciled. On the map, plot and verify the location of each point as it is recovered or set. If precise positions are available for some of the points, such as triangulation stations, these should be plotted in advance to assist in recovery. Plot points found destroyed, but do not plot those not recovered unless project instructions request estimated positions for them.

The accuracy of the plot depends on the mark setter's good judgment and experience. Use all the information available on the map to estimate the location of each point in relation to the surrounding topography. Consider the following elements: distance along the route from the last point; distance from one or more landmarks along the route, such as overpasses, road or highway intersections, prominent buildings, towers, powerline crossings, railroad crossings, bridges, marked



Figure 2-19.—Stainless steel cap for a rod mark.

boundaries, gullies, and hills; and distance of the point from the centerline of a highway or a rail of a railroad. When working across country, use the contour intervals to aid in locating points. After leveling, elevation differences and leveling distances should be used to check such plots. (See sec 5.3.3, "Positions.")

Use only the most recent editions of topographic maps. If the map does not show a new highway, obtain local maps that show both the highway and section divisions (e.g., maps of the U S. Forest Service, U S. Bureau of Land Management, and route plans of State highway departments). Then, transfer the highway route to the topographic map by comparing section divisions. Use "bluelines" if they are available from the USGS (see sec. 2.3.1, "Maps for position plots..."), since they often show control points labeled with designations as well as new highways.

With waterproof black ink, indicate each control point with a dot, neatly surrounded by a 3-mm (1/8-in) circle Label it with the designation. Indicate points that are not to be leveled with red ink. Indicate points that are destroyed with the note "DEST" and those that are not recovered with the note "NR." Well-maintained working plots may be used subsequently as logs for the line. (See sec. 2.3.5, "Records.")

Scaling positions. The position of each control point without a previously published position must be scaled from the working plot and recorded in the description. To ensure accuracy and consistency, scale positions only from 7.5 or 15' USGS topographic maps. The position of a single point may be scaled while preparing the description, or the positions for all the points may be scaled after work on the area is completed. In the latter case, ensure that the positions are matched correctly with the descriptions.

The scales illustrated in figure 2-20 are used to obtain both latitude and longitude. The following instructions apply to a 7.5 map. The same procedure should be used for a 15' map; however, use the single scale marked in increments of 2" from 0'00" to 5'00".

1. Place the map on a flat surface. If latitude and longitude lines have not been previously drawn, draw them on the map using a straight edge and a sharp pencil. Draw latitude lines across the map at intervals of 2'30''. Draw longitude lines at the same interval. The intervals are marked on the edges and, with small crosslines, on the face of the map. (On a 15' map draw the lines at intervals of 5'.)

2. From the edge of the map, read the degrees, minutes, and seconds of the nearest latitude line below the point. If the seconds read "00," use the scale marked from 0'00" to 2'30" If the seconds read "30," use the scale marked from 0'30" to 3'00".

3. Place the scale, oriented vertically, near the control point. Adjust the lower end (0'00'' or 0'30'') until it rests over the nearest latitude line below the point. Adjust the upper end (2'30'' or 3'00'') until it rests over the nearest latitude line above the point. 4. The scale is now slightly tilted. Maintaining the same angle of tilt, move it sideways until it intersects the point, as shown in figure 2-21. Ensure that the lower and upper ends rest over the lower and upper latitude lines, as before.

5. At the control point, read from the scale the minutes and seconds to the nearest second. Add this to the reading of the lower latitude line to obtain the latitude of the point.

6. Proceed similarly to obtain the longitude. From the bottom of the map, read the degrees, minutes, and seconds of the nearest longitude line to the right of the point. If the seconds read "00," use the scale marked from 0'00" to 2'30". If the seconds read "30," use the scale marked from 0'30" to 3'00".

7. Place the scale, oriented horizontally, near the control point. Adjust the right end (0'00'' or 0'30'') until it rests over the nearest longitude line to the right of the point. Adjust the left end (2'30'' or 3') until it rests over the nearest longitude line to the left of the point.

8. Again, the scale is slightly tilted. Maintaining the same angle of tilt, move it up or down until it intersects the point, as shown in figure 2-22. Ensure that the right and left ends rest over the right and left longitude lines, as before.

9 At the control point, read from the scale the minutes and seconds to the nearest second. Add this to the reading of the right longitude line to obtain the longitude of the point.

When scaling positions, pay particular attention to the direction in which the scale is read. In the northern hemisphere, latitude always increases as you read up (north) from the bottom edge of the map. In the western hemisphere, longitude always increases as you move left (west) from the right edge of the map. Also, watch for consistent errors of 30" caused by reading the wrong scale, errors caused by aligning one end of the scale with a latitude line and the other with a longitude line, and errors caused by reading the wrong latitude or longitude line.

To prevent such blunders, positions should be scaled by two persons independently before they are submitted to the project office. A recommended routine is for one member of the mark setting unit to plot, scale, and record the position while at the site of the control point. Then, the other member rescales and checks all the positions at the time survey-point serial numbers are assigned and logs are completed.

Smooth plot. The smooth plot is a clean version of the working plot, prepared on a copy of the same map (fig. 2-23). It is the permanent record illustrating the precise locations of the control points recovered. Label each point to be leveled with the survey-point serial number. Label the lower, right-hand corner with the line number and a sequence number corresponding to the location of the map in the line.

$\begin{array}{cccccccccccccccccccccccccccccccccccc$
+3, 24 48 42 36 30 24 18 12 6 +2, 24 48 42 36 30 24 18 12 6 +1, 24 28 42 36 30 20" to $3^{1}00^{1}$ to $3^{1}0^{1}$
להלה להלה להלה להלה להלה להלה להלה להל
o'00" دە 2' 30" av o' 30" av

Figure 2-20.—Tools for scaling geographic positions.

2-30

(

(

(

- <u>-</u> \_



Figure 2-21 --- Scaling latitude: D 59 USLS is 42°01'33"N.



Figure 2-22.-Scaling longitude: D 59 USLS is 83°13'32''W.





## 2.4.4 Descriptions

To be useful, control points must not only be permanent and stable, but recoverable. Descriptions of points in the National Geodetic Vertical Network are the index for referring geodetic leveling data to actual points on the Earth's surface. Each description must independently provide, in a concise and consistent manner, all the information necessary to locate, positively identify, and level to a control point.

The degree of care with which a mark setter prepares descriptions is immediately evident to the other surveyors who use them. Do not repeat approximations, assumptions, and mistakes which may be present in previous descriptions. A description should contain only clear and accurate statements of facts that have been verified by the mark setter.

The National Geodetic Survey uses a form for coding much of this information, thus reducing the repetitive writing required when preparing numerous descriptions of similar points. It serves as a reminder, prompting the mark setter for detailed information about the location and the construction of the monument. It is designed to satisfy the requirements for vertical descriptive data presented in *Input Formats and Specifications* of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: volume II, ch. 7), hereinafter referred to as the input formats and specifications manual

Prepare a description on NOAA Form 76-186, "Bench Mark Description," for each point searched for or set. Complete it at the site. Never write a description from notes or from memory; a vital piece of information may be missed or be remembered incorrectly. Only two items may be completed after leaving the site: the survey-point serial number and the position. These may be respectively assigned and scaled while compiling logs for a completely routed segment of line.

Descriptions are original records that are referred to repeatedly during the course of a project and are filed for further use afterwards. Protect them while in the field by attaching them to a clipboard with a waterproof cover. Write neatly, printing if necessary, in waterproof black ink. Do not use abbreviations or codes except those specifically permitted in the input formats and specifications manual.

Specific items of information required for the complete description of a vertical control point are explained here in the order in which they appear on the description form. If the point is already included in the national network and only slight changes must be made to the previous description, a recovery note may be prepared on the same form. It must provide information that verifies the identity and construction of the monument; instructions to reach the point are not necessary unless the previous description is inadequate or incorrect. If the point is not recovered, do not enter information that cannot be verified. Table 2-5 summarizes the standard entries for different types of descriptions.

To begin, in the upper right-hand corner write the project line number and the name of the topographic map on which the point is plotted. (See examples in figs. 2-24 and 2-25.)

Survey-point serial number. The survey-point serial number (SPSN) is an identifier used to match leveling data with the correct control points. Assign a unique

Action	Type of description	Survey Point Serial number? (*10*)	Description recovery code <sup>1</sup>	Archival cross-reference number?	Other condition (*22*)	Condition <sup>1</sup> (*25*)
Set new bench mark	Complete	Yes	D	No	Appropriate code	not required
Recovered, no change or slight change in description	Recovery <sup>2</sup>	Yes	R	Yes	Not required	G or P
Recovered, major change in description	Complete	Yes	R	Yes	Appropriate code	G or P
Recovered, first connection to the National Geodetic Vertical Control Network	Complete	Yes	R	No	Appropriate code plus X	G
Staff, gage mark, or other temporary control point	Temporary'	Yes	None	No	Not required	Not required
Recovered, not leveled by this survey	Recovery <sup>2</sup> or complete	No	R	Yes	Appropriate code	G or P
Recovered destroyed	Recovery <sup>2</sup>	No	R	Yes	Not required	х
Not recovered	Recovery <sup>2</sup>	No	R	Yes	Not required	N

Table 2-5—Standard entries for descriptions

\* Refer to Input Formats and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: vol 11)

\* A recovery note requires the following entries: \*10\*, \*13\*, \*15\* (if any), \*23\*, \*24\*, \*25\*, \*26\*, \*27\*, \*28\*, and \*30\*

'A temporary description requires the following entries: \*10\*, \*13\*

EVAN	DIE	LINE 3 STANU PAINT
NOAA FORM 76-186 13-601 BENCH	I MARK DESCRIPTIC	U.S. DEPARTMENT OF COMMERCE U.S. DEPARTMENT OF COMMERCE NATIONAL OCEANIC AND ATMOSPHENIC ADMINISTRATION NATIONAL OCEAN SURVEY
******	**********	*** XRN
*10*SPSN-76039, DRC CODE-7D, APPROX LAT-7415813N, APP	ROX LON- 0831720W *11*QU	JAD
*13*DESIGNATION-7 E 234	, <u>*14</u> *STATE CODE/C	COUNTY-7 <b>M</b> //. <b>M</b> ONBOE; PARISH in Louisiana — State CENSUS DIVISION in Alaska)
*16*AREA-7	— <sup>1</sup>	DISTANCE AND
*17*NEAREST CITY OR TOWN-TO LDPORT		*18*NEAREST CITY OR TOWN-72.8KM(1.8M1)S,
*20*CODE/MONUMENT BY AGENCY-71,/, NG-S	*21*YEAR-71986	2, CHIEF OF PARTY-TLHW *22*OTHER CONTROL-T
*23*CODE/RECOVERY BY AGENCY-7_1/1	, *24*YEAR-7	_; CHIEF OF PARTY-T, *25*CONDITION OF MARK-T_,
*26*CODE/SETTING CLASSIFICATION- 164/1		$\star 27^{\star}$ MONUMENTATION- $_{\tau} f$ , DISK TYPE- $_{\tau}$
*28*STAMPING-E234 1980		NGS QUALITY OVERRIDE-T_
**************************************	RECOVERY DESCRIP	<u>TIVE TEXT</u> ************************************
1301.2. 8KM(1.75MI) SOUTH ALDNG DIX 1301. (HURCH IN OLDPORT, AT A PRI) 1301. ATOMIC POWER PLANT	LE HIGHWAY FROM	THE ST. CHARLES CATHOLIC
*30*		~ <i></i>
*30*.42.7 METERS (142 FT) FACT AFT	HE CENTERLINE OF	DIVIE HIGHWAY
*30*, 47/1 METERS (139 FT) EAST- SAUTA	LEAST OF THE TUN	r TIAN
*30*, 11. 6 METERS (38 FT) SOUTH OF	THE CENTERLINE OF	FTHE PRIVATE POAD.
*30*,6.10 METERS (20.0 FT) SOUTH 0	F THE NORTHEAST C	OPNER DE A CORRAL
*30*, AND 0.61 METER (20 FT) EAST 1	OF A WOODEN FEMCI	E.
*30*,		
*30*,		
*30*,		·
*30*,		
*30*,		
*40*(Units,,E-English/M-metric) A DISK SET INTO THE TOP OF	A CONCRETE POST,_, F-FLUSH WITH T	THE GROUND/P-PROJECTING/R-RECESSED,, IN/CM
*41*(Units M, Setting Code, 64; same as in *26*) ROD/PIPE DRI (leave preceding entry blank if no sleeve), EN	VEN TO THE DEPTH OF, 6.0, #/M KASED IN A PIPE, FF-FLUSH WITH T	1. IN A SLEEVE EXTENDING TO THE DEPTH OF, 4. 4. M. THE GROUND/P-PROJECTING/R-RECESSED,, IN/CM
*42*(Units,), FT/M, A-ABOVE/B-BELOW/L-ABOUT LEVEL	WITH (Units and FT/M blank if 'L	_');i
*43* (Units, [1]),, 61, FT/M,, (Point of Compass) FROM A	WITNESS POST — else specify obje	set,
, of Bench Mark,		(start entry with article (a,an,the) if applicable)

SUPERSEDES NOAA FORM 76-186 (8-78) THICH MAY BE USED.

U.S. GOV. PRINTING OFFICE: 1950-668-690

Figure 2-24.—Complete description.

	EXAMPL	E Line 3 GIBRALTAR
ноаа голи 76-186 13-802 ВЕNCН МАРК	DESCRIPTION	U.S. DEPARTMENT OF COMMERCE NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION NATIONAL OCEAN SURVEY
*10*SPSN-, 6051, DRC CODE-, R APPROX LAT-, 420455N, APPROX LON-, 08	3 1 2 3 9 W *11*QUAD-	TXRN TSN-TNEO SO6*12*LINE-T
*13*DESIGNATION-TL199	*14*STATE CODE/COUN	ΠΥ
*15*ALIAS-1'	(PARI	SH in Louisiana - State CENSUS DIVISION in Alaska
*16*AREA-7	T	DISTANCE AND DIRECTION FROM
*17*NEAREST CITY OR TOWN-T	* *1	8*NEAREST CITY OR TOWN-
*20*CODE/MONUMENT BY AGENCY-T_1/	*21*YEAR-7 C	HIEF OF PARTY-7 *22*OTHER CONTROL-7
*23*CODE/RECOVERY BY AGENCY-T.//NGS	*24*YEAR-1980, C	HIEF OF PARTY-TLHW, *25*CONDITION OF MARK-G
*28*STAMPING-T L/99 1968	·	UIALITY OVERSIDE
	·	
*30*, ONIT PRIOR REFERENCE TO WITNESS PO	ST.	VEIEXI **********************************
*30*,		
*30*,		
*30*		
*30*.	ne	
*30*,		
*30*,		
*30*,		<b></b>
*30*,		
*30*	······································	
*40*(Units, ,E-English/M-metric) A DISK SET INTO THE TOP OF A CONCRETE POS	ST, F-FLUSH WITH THE	GROUND/P-PROJECTING/R-RECESSED, , IN/CM
*41*(Units, E, Setting Code, 2_, same as in *26*) ROD/PIPE DRIVEN TO THE DEM (leave preceding entry blank if no sleeve) ENCASED IN A PU	PTH OF, $\underline{I}\underline{4}$ , FTA4, II PE $\mathbf{P}_{\text{E-FLUSH}}$ with the (	N A SLEEVE EXTENDING TO THE DEPTH OF,, FT/M GROUND/P-PROJECTING/R-RECESSED, $\mathcal{U}$ IN/ <del>O/</del>
*42* (Units, ,), , FT/M, ,A-ABOVE/B-BELOW/L-ABOUT LEVEL WITH (Units a)	nd FT/M blank if 'L').	
*43* (Units E.),, FTA4E, (Point of Compass) FROM A WITNESS POST -	- else specify object,	
',Mi/Km,,of Bench Mark,	' (!	start entry with article (a,an,the) if applicable
SUPERSEDES NOAA FORM 76-18	IS 19-78; WHICH MAY BE USED.	VILS. GOV. PRINTING OFFICE: 1980-668-690

Figure 2-25a.—Recovery note.

 $\sim$ 

NOAA Manual NOS NGS 3, Geodetic Leveling

D-self-standing original description R-self-standing recovery description

\*14\*STATE CODE: Use standard two-letter POSTAL CODE in the United States, its territories and possessions. Consult the User's Guide to the Input Formats and Specifications of the National Geodetic Survey Data Base for points elsewhere.

#### \*20/\*23\*AGENCY CODE:

#### 0-unknovn

I-NGS or CGS (USC6GS)
2-U.S. Geological Survey (USGS)
3-U.S. Department of Defense (DOD)
4-other federal or interstate agency
5-state agency
6-county, city, or regional agency
7-commercial organization or private firm
8-National Ocean Survey (NOS)
9-foreign government agency

#### \*22\*OTHER CONTROL CODE:

A-astronomic station F-fault monitoring aite G-gravity station H-horizontal control point M-magnetic station N-no vertical control (not presently or anticipated to be connected to the National Vertical Control Network) O-other (see descriptive text) T-tidal bench mark X-no vertical control (not connected to the National Vertical Control Network at time of recovery, but expected to be after this survey)

#### \*25\*CONDITION CODE:

G-GOOD or fair N-NOT RECOVERED or not found, lost O-other (SEE DESCRIPTIVE TEXT) P-POOR or disturbed, mutilated X-DESTROYED

#### \*26\*SETTING CLASSIFICATION CODE:

00-UNSPECTATED Shallow Settings (less than 10ft): 10-UNSPECTATED SHALLOW 11-METAL ROD WITH BASE PLATE 12-CONCRETE POST 13-SHALLOW-SET PIPE 14-SHALLOW-SET METAL ROD (without base plate) Unsleeved Deep Settings: 20-UNSPECIFIED DEEP 21-COPPER-CLAD STEEL ROD 22~GALVANIZED STEEL PIPE 23-GALVANIZED STEEL ROD 24-STAINLESS STEEL ROD 25-ALUMINUM ALLOY ROD Rocks and Boulders: 30-UNSPECIFIED ROCK 31-ROCK OUTCROP, rock ledge, rock cut. or bedrock 32-BOULDER Structures: 40-UNSPECIFIED LIGHT STRUCTURE 41-PAVEMENTS, such as streets, sidewalks, aprons, curbs, etc. 42-RETAINING WALLS AND BULKHEADS, including headwalls and wingwalls of culverts and small bridges 43-PILES AND POLES (e.g., spike in utility pole) 44-FOOTINGS AND FOUNDATION WALLS OF SMALL OR MEDIUM STRUCTURES 45-MAT FOUNDATIONS (bearing surface same size as structure), such as landings, platforms, steps, floors, tower foundations, bases of semaphores. etc. **50-UNSPECIFIED MASSIVE STRUCTURES** 51-MASSIVE RETAINING WALLS, including headwalls and retaining walls of very large bridges 52-ABUTMENTS AND PIERS OF VERY LARGE BRIDGES. including overpasses and underpasses 53-TUNNELS 54-MASSIVE CONCRETE, MASONRY, OR STEEL STRUCTURES WITH DEEP FOUNDATIONS 55-LARGE CONCRETE, MASONRY, OR STEEL STRUCTURES WITH FOUNDATIONS ON BEDROCK Sleeved Deep Settings: 60-UNSPECIFIED ROD/PIPE SLEEVE 61-COPPER-CLAD STEEL ROD IN SLEEVE 62-GALVANIZED STEEL PIPE IN SLEEVE 63-GALVANIZED STEEL ROD IN SLEEVE 64-STAINLESS STEEL ROD IN SLEEVE 65-ALUMINUM ALLOY ROD IN SLEEVE

Figure 2-25b.—Recovery note.

#### \*27\*MONUMENTATION CODE:

B-BOLT C-CAP-AND-BOLT D-SHRVEY DISK (any type) F-FLANGE-ENCASED ROD H-DRTLL HOLE T-METAL ROD N-NATI. O-CHISELED CIRCLE P-PTPE CAP **O-CHISELED SOUARE** R-RIVET S-SPIKE T-CHISELED TRIANGLE V-STONE MONUMENT X-CHISELED CROSS Z-SEE DESCRIPTIVE TEXT

#### \*27\*DISK TYPE (blank if not survey disk):

00-unspecified SURVEY DISK 01-BENCH MARK DISK 02-TIDAL BENCH MARK DISK 03-TRIANGULATION STATION DISK 04-TRAVERSE STATION DISK 05-TOPOGRAPHIC STATION DISK 06-SURVEY DISK (not listed) 07-REFERENCE MARK DISK 08-AZIMUTH MARK DISK 09-GRAVITY STATION DISK 10-GRAVITY REFERENCE MARK DISK 11-MAGNETIC STATION DISK

\*27\*QUALITY OVERRIDE (for NGS use only)

\*40\*/\*41\*/\*42\*/\*43\*INITS CODE:

E-English M-metric units

\*43\*POINTS OF COMPASS: N, ME, NE, ENE, E, ESE, SE, SSE, S, SSV, SV, WSW, W, WNW, NW, NW, serial number to every point that is to be leveled in the line. A series of numbers for each line is normally assigned in the project instructions. Assign the number "0000" to a point that is not to be leveled, such as one that is destroyed, unsuitable, or not recovered.

To prevent duplications, serial numbers may be assigned to all the points on a completed map at the same time that positions are scaled. They may be assigned in a sequence corresponding to the order in which the points should be leveled. However, they need not always appear in sequence. If points are added to the line after most serial numbers have been assigned, do not change the original numbers merely to correct the sequence.

Junction code. Write "JUNCTION" at the top of the description form whenever the control point is one of the following:

- 1. One of the three points at network junctions (including international boundaries).
- 2. A point at the base of a spur connecting a primary tide station.

A junction point leveled repeatedly within the same project must be described for each line in which it is leveled. If it is newly set, prepare a complete description for the first line leveled; prepare a recovery note for each subsequent line. For convenience, assign the same survey-point serial number to the point whenever it is described.

Description/recovery code (DRC). Enter a single letter on line  $*10^*$  to indicate whether the control point was searched for, "R," or is newly set, "D." The code "R" indicates a recovery note or a complete description, written for a point that existed prior to the arrival of the mark setter. The code "D" indicates a complete description, written at the time the mark is set.

Position Scale the latitude and longitude of the plotted position of the control point from the topographic map. Record each to the nearest second, latitude followed by the direction, "N" or "S", from the equator and longitude followed by the direction, "W" or "E", from the Greenwich meridian. The position is used by NGS to compute automatically the 30' quadrant and the number of the quadrant line in which data for the point are filed.

If a point is not recovered, enter on line \*11\* the number of the quadrant and on line \*12\* the number of the quadrant line in which the previous description was published. Do not enter a position.

Archival cross-reference number. The archival crossreference number (XRN or ACRN) is a unique identifier that is assigned to each control point in the national network. It permits the comparison of data collected during different epochs for the same point. It consists of two parts, a two-letter area code, indicating the Geodetic Control Diagram in which the point is positioned, and a four-digit number that is unique within the area. An index of two-letter area codes appears in appendix B.

Obtain the cross-reference number from either the published description or from a special listing of the

synoptic file provided by NGS. Copy the number on line \*11\* of the description form. If a point has not previously been connected to the network, leave the space blank; a cross-reference number is automatically assigned after the project is completed.

Designation To identify the control point, a designation is normally stamped on the mark by the agency which sets it The designation is a unique name by which the point can be conveniently recognized and referenced. In general, the designation given in the description should be identical to that stamped on the mark. It should not, however, include redundant or nonspecific information that may appear in the stamping.

When the point is recovered, assign the designation given in the published description (or listing of the synoptic file) If necessary, correct mistakes such as misspellings or transposition of numbers, and record the previous, incorrect designation as an alias. Use the following guidelines to standardize the designations. (Examples appear in table 2-6.)

Do not include information that is normally precast on a mark, such as warnings or type of control.

Do not include a stamped elevation unless it is the only means of distinguishing the point.

Do not include a stamped date, unless the point has been relocated or the date is the only means of distinguishing between otherwise duplicate designations.

Limit the designation to 25 characters or less, including imbedded blanks. Abbreviate the designation if necessary to conform to this limit. Separate groups of alphabetic and numeric characters with one blank. Do not retain any punctuation that is stamped, except the following: A plus (+); a minus (-), only if it indicates a negative elevation; a period (.), only if it is embedded in a group of numerals; an equal sign (=); or a slash (/). Do not separate such symbols from the adjacent characters with blanks. Note: If a minus (-) is included, it can only be the first character of the designation.

If the mark is stamped with two designations, totaling 24 characters or less, concatenate the designations with an equal sign. If the characters total 25 or more, enter one of the designations as an alias.

If the mark belongs to an agency other than the NGS (or its predecessor, the Coast and Geodetic Survey), include at the end of the designation the acronym for the agency's name that is precast on the mark. If no name is precast, include the same acronym that is entered for the agency that set the mark. (See "Setting information," this section.)

If the mark provides control for a tide or water-level station, but was set primarily to provide control for the national network, include the station number as an alias. If the mark was set solely to provide control at the station, include the seven-digit number in the designation. Enter the first three digits, a space, and the last four digits. Add the word "TIDAL" if the station is a tide station. Then, enter the designation stamped on the mark, observing the guidelines given in this section. Newer

## Table 2-6.—Examples of bench mark designations

Stamping as it appears on mark	Symbol for agency name as it appears on mark	Designation to appear on description
General:		C 124
C 124 1980	NUS	
2903 11 226 2046 220 406 EE		2903 0303 H 225
H 525 1945 250 695 FT	MORC	140 B MORC
T T17B 1065	USGS	TT 17 B USGS
TT. 17B 1965	USGS	TT 17 B USGS
595 ±00 1957	AHD	595+00 AHD
MI 14.7	None	M1 14 2
4419	USGS	4419 USGS
5301 29 FT	USGS	5301 29 USGS
CH1174, 297+00A	USGS	CH 1174=297+00 A USGS
STA NO. 3. MI 182.5	None	STA NO 3=MI 182 5
No stamping	None	Previous designation
		-
Tide station 872 8912:	10000	DTO POLO TIDAL 1
TIDAL 1 1937	USCGS	872 8912 TIDAL I
BASIC 1937	USCGS	872 8912 HDAL BASIC
H 14, 11DAL 3 1963	USCUS	H 14=0/2 0912 11DAL 3
8912 A 1977	NUS	672 6912 A HDAL
A 307 1974		972 8012 TIDAL Previous designation
No stamping Staff, rod hald at 6 (t stop	03003	TRM 872 8912 STAFF 6 FT
Start, foo neio at o it stop		IDH OF STE STEEL OF
Water-level station 906 3000:		
2 1934	USCGS	906 3000 2
BM 3 1934	USCGS	906 3000 BM 3
POOL 1934	USCGS	906 3000 POOL
M 104 1973	NGS	M 104 Alias 906-3000
3000 B 1977	NOS	906 3000 B
Nostamping	USCGS	506 3000 Previous designation
Electric tape-gage reading mark		IBM 900 3000 ETG KEAD MK
Other control:		
BOULDER 1935	USCGS	BOULDER
BOULDER NO 2 1935	USCGS (Reference disk)	BOULDER RM 2
BOULDER 1935	USCGS (Azimuth disk)	BOULDER AZ MK
BOULDER NO 2 1935	USCGS (Azimuth disk)	BOULDER AZ MK 2
CHARLOTTE	USGS	CHARLOTTE USGS
CHARLOTTE NO 1	USGS (Reference disk)	CHARLOTTE RM I USGS
CHARLOTTE	USGS (Azimuth disk)	CHARLOTTE AZ MK USGS
PALMER N E BASE 1940	USCGS	PALMER NE BASE
N WASH AZI	USGS	N WASH AZ MK USGS
Relocated points:		
Z 201 RESET 1980	NGS	Z 201 RESET 1980
3000 USGS RESET 1976	NGS	3000 USGS RESET 1976
CHICO 1948 1971	NGS	CHICO 1948 1971
CHICO 1948 NO. 3 1971	NGS (Reference disk)	CHICO RM 3 1948 1971
CHICO 1948 1971	NGS (Azimuth disk)	CHICO AZ MK 1948 1971
CHICO 1948 NO. 2 1971	NGS (Azimuth disk)	CHICO AZ MK 2 1948 1971
LIGHT 1950 RESET 1955	USCGS	LIGHT RESET 1955
WINSLOW 2 1962	USCGS	WINSLOW 2
LAKE WASHINGTON 1950 1970	USCGS	LAKE WASHINGTON 1970

tide and water-level marks are stamped with the last four digits of the station number and a letter; in this case, enter the three-digit state code followed by a space, the stamping, and the word "TIDAL."

If a point at a tide or water-level station has no stamping, append the designation given in the previous description to the station number. Because a staff or the reading mark of an electric tape gage is not a permanent monument, enter the letters "TBM" at the beginning of the designation, to indicate a temporary bench mark. Designate a staff by the station number, the word "STAFF," and the height of the stop on which the leveling rod is to be held. Designate the reading mark of an electric tape gage by the station number and the words "ETG READ MK." If the mark was set to provide control for a horizontal network or other types of networks, assign the designation as it appears in the published description, again observing the guidelines given here. Designate a reference mark with the horizontal station name, the letters "RM," and the number stamped on the mark Designate an azimuth mark with the horizontal station name, the letters "AZ MK," and the number, if any, stamped on the mark.

The mark for a vertical control point that has been relocated is normally stamped with the original designation, the word "RESET," and the year of relocation. Include all of this information in the designation. Other types of relocated control points may be stamped differently. For example, to reestablish a horizontal control point, a new surface mark may have been reset directly over an underground mark. The reset mark is usually stamped with two dates, the year the point was established and the year it was reestablished. The geographic position has not changed, but the elevation has. Indicate this in the designation by including both dates. If the designation exceeds 25 characters in length, include only the most recent dates.

A horizontal control point that has been entirely relocated and designated with the number "2" should not be confused with a reestablished point. Do not designate such a point with a date unless it, too, has been reestablished.

Alias. If the control point is known by more than one designation, enter the designation that is stamped on the mark on line \*13\* and enter all other designations as aliases. If the point has been designated incorrectly in a previously published description (or in the listing of the synoptic file), the incorrect designation should be shown as an alias in the corrected description.

General location. For control points in the United States enter a two-letter postal code indicating the State, commonwealth, province, or territory on line \*14\*. United States postal codes are given in table 2-2. Then enter the county, parish, census division, or independent city in which the point is located. Indicate a city with the phrase "C OF." For example, "C OF ST LOUIS" for St. Louis, Missouri.

For control points in other countries, enter the appropriate two-letter code and then the primary political division, such as a State, province, or district. The codes are published in annex A of the input formats and specifications manual.

If the point is part of a local network, such as a network for monitoring subsidence, an area title may be entered on line \*16\* to indicate this. The entry is optional and can be automatically added to all appropriate points after completion of the survey.

On line \*17\* enter the nearest city or town. It must have a post office, railroad station, or well-defined highway crossroads from which to begin distance measurements and it must appear on the official State highway map. If it is in a different State from the point, append the postal code for the state whenever the city or town is named. Also, append the code when necessary to avoid confusion. For example, "KANSAS CITY KS" as opposed to "KANSAS CITY MO."

The distance from the nearest city or town, which is to be entered on line \*18\*, is not a straight-line measurement. Instead, it is the total of the distances given in the instructions to reach the point, by following the most direct highway routes. Enter it to the nearest tenth of a kilometer, followed by the mileage in parentheses and the direction traveled Abbreviate kilometer, "KM," and mile, "MI." Use the eight-point compass, spelling out the four cardinal directions and abbreviating the four intercardinal points. For example, "3.4 KM (2.1 MI) EAST," "10.2 KM (6.3 MI) NW," and "1.0 KM (0.6 MI) SE."

If the control point is located within city or town corporate limits, enter the word "IN" before the name of the city or town and do not give a distance and direction.

Setting information. Whenever the control point requires a complete description, enter information about the agency which set the monument and its purpose. This includes a code indicating the agency classification, the agency's name, the date (year) the mark was set, and the initials of the project director of the party which set it. Agency codes are listed on the back of the description form and in table 2-7. Acronyms for agency names, as standardized for the use of the National Geodetic Survey, are given in annex C of the input formats and specifications manual. If the date the mark was set and the initials are not available from a previous description, enter the date stamped on the mark and leave the initials blank. If the date is not known, enter "UNK" in its place.

When the point serves another geodetic or geophysical purpose enter the appropriate code on line \*22\*. The purpose should be evident from the project instructions for newly established points and from the previous description or type of disk for recovered points. Spe-

 
 Table 2-7.—Agency classification codes designated by NGS for preparing NOAA Form 76-186

Agency	Code <sup>1</sup>
National Geodetic Survey	1/NGS
National Ocean Survey	8/NOS
U.S. Coast and Geodetic Survey (prior to 1970)	1/CGS
U.S. Geological Survey	2/USGS
US Department of Defense	3/DOD
Other Federal or interstate agency	4
State agency	5
County, city, or public regional agency	6
Commercial organization or private firm	7
Foreign government agency	9
Unknown	0

<sup>1</sup> Refer to Input Formats and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: vol 11). cific horizontal, astronomic, or gravity data must be available in the NGS data base if the point is to be coded as providing such control. Control classifications appear on the back of the form and in table 2-8.

If a recovered point has not been previously connected to the national network, but it is connected by the current survey, enter "X" in addition to any other codes. If a point is described that has not been connected to the network in the past and is not connected by the current survey, enter "N" (A description is only required for such a point if it is marked by a disk or flange belonging to NGS).

Recovery information. In each recovery description, enter information about the agency which searched for the point. This includes a code indicating the type of agency, a standard abbreviation for the agency's name, the year the point is searched for, and the initials of the project director of the party conducting the search. Use the same codes and abbreviations that appear in tables 2-6 and 2-7 to denote the setting agency.

Enter a code for the condition of the monument on line \*25\*. If the monument appears to be undisturbed and is suitable for leveling, in good or fair condition, enter "G." If it has been disturbed, or if it cannot be positively identified because of mutilation, enter "P". If the monument is found destroyed enter "X". (See sec. 2.4.1, "Destroyed bench marks.") If it is not recovered, enter "N", even if it is presumed to have been destroyed. Explain any condition other than good in the descriptive text.

Construction details. Describe the construction of the monument by entering the two-digit code that denotes the setting classification. Table 2-9 summarizes setting classifications. If the monument is a mark installed in a preexisting structure (codes 40 through 55), describe the setting specifically with a short phrase on line \*26\*. For example, "51/ VERTICAL IN OVER-PASS HEADWALL," "42/CULVERT," and "44/ VERTICAL IN BUILDING FOUNDATION."

### Table 2-8.—Control classification codes designated by NGS for preparing NOAA Form 76-186

Other control	Code
Fault monitoring	F
Tide or water-level	Т
Horizontal	н
Astronomic	А
Gravity	G
Magnetic	M
Other (explain in text)	0
Not previously connected to the National Geodetic Vertical Network:	
Connected by the current survey	x
Not connected by the current survey	N

<sup>1</sup> Refer to Input Formats and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: vol 11)

### Table 2-9.—Monument classification codes designated by NGS for preparing NOAA Form 76-186

Code	Setting classification Qu	ality'
00	Unspecified	D
Shallr	w settings (less than 3 m or 10 ft);	
10	Inspecified shallow	D
11	Metal rod with base plate	ř
17	Concrete post	č
12	Concrete post in the reason of the task of the task of the reason of the task of task	n n
14	Shallow metal rod without base plate	D
Unsle	eved deep settings:	
20	Unspecified deep	С
21	Copper-clad steel rod	В
22	Galvanized steel pipe	в
23	Galvanized steel rod	R
74	Stainless steel rod	Ř
ትግ ግና	Aluminum allow rod	D D
23	Additional anoy for the second s	D
Rocks	and boulders:	
30	Unspecified rock	В
31	Rock outcrops, such as a rock ledge,	
	rock cut, or bedrock	Α
32	Boulder	С
Light	structures:	
40	Unspecified light structure	D
41	Pavements, such as streets, sidewalks, aprons, curbs,	_
	and so forth	D
42	Retaining walls and bulkheads, including headwalls and	
	wingwalls of culverts and small bridges	C
43	Piles and poles, such as a spike in utility pole	ñ
40 A A	Footings foundation wells and abutments of small	U
44	rootings, roundation wans, and additions of sman	c
	or medium size structures	C
45	Mal loundations, such as landings, platforms, steps.	
	floors, tower foundations, semaphore bases, and other	
	bearing surfaces of the same size as the structure	С
Massi	ve structures:	
50	Unspecified massive structure	В
51	Massive retaining walls, including headwalls and	
	wingwalls of very large bridges	в
52	Abutments and piers of very large bridges	B
57	Tuppels	ĥ
57	Marriva concrete maconey or steal structurer	Б
2.4	with deep foundations	Δ
55	Large concrete macoury or steel structurer	A
, C	with foundations on hedrock	A
Sleeve	ed deep settings:	_
60	Unspecified rod or pipe in sleeve	В
61	Copper-clad steel rod in sleeve	В
62	Galvanized steel pipe in sleeve	Α
63	Galvanized steel rod in sleeve	Α
64	Stainless steel rod in sleeve	Α
65	Aluminum alloy rod in sleeve	Α

<sup>1</sup> Refer to Input Formats and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980; vol 11)

If the monument is a concrete post (code 12), describe the height of the post with respect to the ground surface in metric units on line  $*40^*$ . Similarly, if the monument is a rod or pipe which has been driven into the ground (codes 11, 13, 14, 20-25, or 60-65), give the depth, the sleeve depth (if any), and the height of the casement with respect to the ground surface in metric units on line \*41\*.

Describe the type of device marking the control point by entering the appropriate one-letter code on line \*27\*. If the mark is a survey disk, enter a two-digit code to indicate the type. A list of marks and corresponding codes is given on the back of the form and in table 2-10. Figures 2-26 and 2-27 show the bench mark disks used by the former Coast and Geodetic Survey and the National Geodetic Survey.

Quality classification. In addition to the variety of monuments set by NGS, monuments of even greater diversity, set by other Federal, State, and local agencies, have been and continue to be included in the national network. See appendix D of the mark setting manual. To assist the user of vertical control data in selecting the most reliable control points for the purpose of the intended survey, monuments have been classified according to their expected ability to remain stable in relation to the local topography. One of four quality codes is automatically assigned to each control point on the basis of the construction details provided in the description. The codes are defined as follows:

1. Quality A describes a monument of the most reliable construction, likely to be affected only by movement of the geological feature in which it is installed.

Table 2-10.—Bench mark classification codes designated by NGS for preparing NOAA Form 76-186

Type of mark	Codei
Survey disk:	
Unspecified	D 00
Bench mark (vertical control)	D 01
Tide or water-level bench mark	D 02
Triangulation station (horizontal control)	D 03
Traverse station	D 04
Topographic station	D 05
Reference mark	D 07
Azimuth mark	D 08
Gravity station	D 09
Gravity reference mark	D 10
Magnetic station	D 11
Other type (explain in text)	D 06
Flange-encased rod	F
Metal rod	I
Pipe cap	Р
Cap-and-bolt	С
Bolt	В
Rivet	R
Nail and the second	N
Spike	S
Chiseled cross	Х
Chiseled circle	Ö
Chiseled square	Q
Chiseled triangle	Ť
Drill hole	н
Stone monument	v
Other (explain in text)	Z

<sup>1</sup> Refer to Input Formats and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: vol 11).



Figure 2-26.—Survey disks of the former U.S. Coast and Geodetic Survey.

- 2. Quality B describes a monument of construction likely to prove reliable.
- 3. Quality C describes a monument that may be affected by surface movement, such as frost heave or the shrinking and swelling of certain clays.
- 4. Quality D describes a monument not likely to prove reliable, or a monument of unknown construction.

Of course, generalizations have been necessary to implement this system. However, the experienced mark setter may greatly improve the value of the quality estimate by assigning the code in the field. A monument of low expected quality in one location might require a higher classification in another. For example, an unsleeved stainless steel rod (quality B) is subject to movement due to soil shrinkage or expansion in regions where the soil readily absorbs and retains water. In regions with low humidity, rare frost, and nonexpansive soil, the same monument may be upgraded to quality A

Similarly, a monument of high expected quality might, when construction details are examined, require a lower classification. For example, a disk set in the wingwall of a very large bridge (quality B) is less likely to prove reliable if the wall is cracked and undercut by erosion (quality C).

More details on placement and construction of monuments are presented in section 2.4.2 and in the mark setting manual. Using table 2-9 as a guideline, apply your judgment in specific cases to assign a more realistic estimate of quality when it is appropriate. Enter the quality code after the setting classification.

Stamping A control point marked with a disk flangeencased rod, pipe cap, or cap-and-bolt is normally stamped with the designation and year it was set. Sometimes an elevation may be stamped as well. A precast inscription is not considered part of the stamping.

The stamping serves as final verification of the identity of the control point. Enter it exactly as it appears on the mark. If no stamping appears, make note of it in the descriptive text.

From the precast inscription (if any), determine the originating agency for the mark. In the space following the stamping, enter the acronym for the agency as it appears on the mark.

*Text.* The text ("body") of the description must provide thorough and accurate instructions for reaching the control point. It should also describe features affecting the utility of the point for vertical control, clarifying information coded elsewhere in the description when necessary.

The fact that a control point is recovered by following instructions from the previous description does not, in itself, imply that the instructions are correct in every detail or that they will apply in the future. Verify all items in the previous instructions and assess their future usefulness. If no changes are necessary, the recovery note need not include a text entry. If brief corrections or additions are necessary, enter them. Here are two examples: "The 1935 description is adequate except



Figure 2-27.—Survey disks of the National Ocean Survey and National Geodetic Survey.

that the sign 'JONES FARM' no longer exists"; and "The 1952 description is adequate. The point is 0.5 meter (1.6 ft) south of a witness post."

When a point is found destroyed or cannot be recovered, describe the situation. Here are three examples: "The concrete post was found broken off at the base and the disk was removed at this time." "After a 30-minute search by two persons, the point was not recovered. Reference objects given in the 1948 description were not found." "The bridge mentioned in the 1952 description has been rebuilt and the mark appears to have been destroyed."

If a point is particularly difficult to recover, the previous instructions are probably not adequate even though they may be correct. In such cases, and whenever major corrections are necessary, prepare a complete new description, including in the text a new set of instructions to reach the point.

Present instructions in a consistent, clear, and concise format. Start from a major highway intersection or a prominent, permanent landmark in the nearest city or town. Mention landmarks and reference objects in the order in which they will be used. Lead the surveyor first to the general vicinity of the monument and then to the exact point. For convenience, treat these two portions of the description separately, as shown in figure 2-24.

The instructions to reach the general vicinity should follow the most direct driving route (not necessarily the leveling route). Identify highways by Federal, State, or county number. To ensure that the measured distances between landmarks and turns are accurate, calibrate your vehicle's odometer. Enter distances to the nearest tenth of a kilometer, followed in parentheses by miles to the nearest half-tenth. Enter compass directions to the nearest cardinal or intercardinal point: north, northeast, east, southeast, and so forth. When describing a turn or the location of a landmark, enter a compass direction, followed by "right" or "left" in parentheses.

Since vehicle odometers differ, if driving distances are long, give distances to intermediate landmarks that can serve as check points. In addition to noting turns, mention a prominent landmark visible from the highway as near to the control point as possible, at least within the final kilometer. Within city or town corporate limits, include the address (if any) of the property on which the point is situated.

For a control point located about halfway along the line between two cities or towns, include driving instructions from each town, prefacing the second set of instructions with the word "also."

Once in the general vicinity, direct the surveyor to the point by listing measured distances and directions from nearby reference objects in an order convenient to the recovery of the mark. Begin with the most conspicuous object or the longest distance. A good practice is to enter first the distance and direction from the centerline of the highway, so the surveyor can search on the appropriate side.

Select and describe reference objects carefully. Consider the available objects when selecting the site for a new mark. (See page 6 he mark setting manual.) Keep in mind that a more sent may have become obscured by brush or may be buried under dirt and gravel. The lines of position for at least two of the objects should intersect at nearly a right angle Select objects that are well defined, permitting such precise measurements that extensive digging or clearing of brush is not required to recover the point. Avoid objects to which the distance must be estimated. A total of three objects, plus a witness post, are sufficient for most control points. When describing reference objects, indicate whether sizes are standard, approximate, or measured by observing the following conventions: A standard size is given in standard units, with a hyphen between the numeral and the unit; for example, "1-inch pipe" or "2- by 4-inch timber." An approximate size is given in metric units, prefaced by the word "about"; for example, "a walnut tree about 15 cm in diameter" or "a timber about 15 by 25 cm." A measured size is given in metric units; for example, "a post 10 cm in diameter."

Measure distances horizontally (not along the slope) to the nearest hundredth of a meter. After the metric distance, enter the distance to the nearest tenth of a foot in parentheses. If an object is more than 30.5 m (100 ft) from the point or is not definitive (such as the centerline of a highway), measure to the nearest tenth of a meter (integral foot). Precede estimated distances with the word "about."

Measure from the center of objects such as trees and telephone poles and perpendiculary to objects such as fences and walls. Otherwise state the point from which the measurement was made. For example: "6.31 m (20.7 ft) north of the northeast corner of a semaphore base."

Be sure to describe the height of the mark relative to the ground or highway surface. A standard entry for this is available on line \*42\* of the description form. A similar standard entry for the measurement from a witness post is available on line \*43\*.

Enter additional notes as necessary to describe the monument or to explain unusual leveling procedures that may be required. Do not confuse culverts with bridges, piers with abutments, and retaining walls with foundations. Clarify information that may be somewhat misleading as presented in codes elsewhere in the description. For example, note the following: a mark which might be expected to bear a stamping but does not; a mark stamped with a designation duplicating that of another; a mark mounted vertically; the precise location of the control point on a large structure or an unusual monument; the presence of ferrous material in or adjacent to a monument; and the name of the owner on whose property the monument is set. Include your initials at the end of the text.

Throughout the text the following standard abbreviations may be entered for units of measurement: "m" for meter, "mm" for millimeter, "cm" for centimeter, "km" for kilometer, "ft" for foot, and "mi" for mile. Unless undue confusion would result, avoid entering quarter-quadrant compass directions, such as eastsoutheast or north-northwest. Use the accepted terminology found in a dictionary or engineering textbook; avoid the use of jargon.

## 2.5 Relocating Vertical Control Points

Each vertical control point represents a large investment of resources. Since it is intended to provide a continuous record of elevation change, as well as control for many local surveyors, its preservation is vital to the maintenance of the national network. Although monuments are constructed in locations where they are unlikely to be disturbed, many are destroyed or damaged by highway widening and maintenance, bridge rebuilding, railroad maintenance, and building demolition and construction. Such deterioration of the network must be minimized.

Although regional mark maintenance engineers monitor the condition of the network, the areas they cover include many States. To preserve specific points, the cooperation of local engineers and surveyors is important. When recovering a mark, prepare a brief report of its condition (fig. 2-28). If a monument is about to be destroyed, notify NGS at once. Before it is destroyed, a vertical control point should be relocated as described here.

First, a new monument is constructed in the vicinity. Then the elevation difference is measured by leveling from the old to the new control point. Finally, the old monument is destroyed. Data are normally recorded on NOAA Form 76-60, "Report on relocation of bench mark." (See fig. 2-29.)

Setting a new monument. After NGS is given the designation and location of the monument about to be destroyed, the agency will send to the reporter a brass disk, which is prestamped with the designation and year of relocation. For example, the disk for M 346 is stamped "M 346 RESET 1980."

Select a site for the new control point, if possible within one leveling setup from the monument to be replaced. Construct a monument of the best possible quality in which the disk may be set. (See the instructions in 2.4.2 and in the mark setting manual.) Recommended settings for this purpose are a class B rod mark (with the disk crimped to the top), bedrock, or a large and stable structure. If a witness post is near the old monument, relocate it to the site of the new one.

Prepare a complete description of the new point, either on the report form or on a standard description form (NOAA Form 76-186). On the report form, be sure to include instructions to reach the point from the nearest city or town, construction details, and a position. If the position cannot be plotted and scaled as explained in section 2.4.3, note the 30' quadrant of Vertical Control Data and the number of the quadrant line in which the original point was described.

\* U.S. GOVERNMENT PRINTING OFFICE: 1974-665-074/1047 Region 6

RE	PORT ON CONDITION OF SURVEY MARK	Form Approved sudget Bureau No. 41-R1923						
U 150	NF	1941						
Name or Designation:	Year Established:							
State: PA Cour	ny: <u>Huntingdon</u> Organization Establis	hed by: <u>C&amp;GS</u>						
Distance and direction from nearest town:	bout 4.6 miles north along PA Iuntingdon Quad 400781	RR from station at						
Mark searched for or recovered by: Nam	e Roger C. Poe							
Orge	nization - Michael Baker, Jr., 1	Inc.						
Date of report June 17, 1975 Address - P. O. Box 280, Beaver, PA 15009								
Condition of marks: List letters and number	rs found stamped in (not cast in) each mark.							
Mark stamped:	Condition:							
U 150 1941	Good							
Marks accessible? X Yes Please report on the thoroughness of the se repairing or moving the mark, or other pertir	No Property owner contacted? arch in case a mark was not recovered, suggested cha ent facts:	Yes 🕅 No nges in description, need for						
	Witness P	ost? Yes No X						

If additional forms are	needed, indicate number required.	Witness Post set feet of mark
NQAA FORM 76-91	U.S. DEPARTMENT OF COMMERCE - NATIONAL OCEANIC AND ATM	IOSPHERIC ADMINISTRATION - NATIONAL OCEAN SURVEY

## Figure 2-28.—Report on the condition of a survey point.

Leveling to the new control point. Leveling should satisfy standards for second-order, class II accuracy or better. If the distance between the old and new points is greater than 140 m (460 ft), or if more than one setup is necessary, comply with the specifications for geodetic leveling given in chapter 3 of this manual. In most cases, however, leveling may be conducted as follows, with an engineer's level and a single rod. Be sure to record the type of instrument and the units of the rod scale (meters or feet). (See the example in fig. 2-29.)

1. Set up the instrument halfway between the old and new points, but no more than 70 m (230 ft) away from either point.

2. Plumb the leveling rod on the highest point of the old bench mark. Record the designation of the point. On the same line enter the published elevation.

3. Observe the intercept of the middle reticle line against the rod scale. Record the reading to the third decimal place. This completes the backsight, BS.

4. Compute the height of instrument, HI, which is the sum of the backsight and the elevation of the point. Record it on the same line.

5. Do not move the instrument. Plumb the rod on the highest point of the new bench mark. Record the designation of the point on the next line.

Witness Post set \_\_\_\_ feet \_\_

of

mark.

mark.

6. Again, observe the rod intercept. Record the reading on the same line as for step 5. This completes the foresight, FS.

7. Compute the elevation of the new point, which is the height of the instrument from the previous line minus the foresight. This completes the forward run.

8. Reset and relevel the instrument. Level backward, from the new point to the old, in the same manner as in steps 2 to 7. The elevation computed for the old point as the result of the backward leveling may differ from the published elevation by no more than  $\pm 0.003$  m  $(\pm 0.010 \text{ ft}).$ 

9 To compute the elevation difference from the old point to the new, subtract the mean of the two elevations for the old point from the elevation for the new point

Destroying the old control point Do not destroy the old point until the leveling data have been checked. Remove the old disk and any other parts of the monu-

í

Ć

ł

NOAA FORM 76-60 U.S. DEPARTMENT OF COMMERCE DESIGNATION OF MARK								
REPORT ON RELOCATION OF BENCH MARK								
(See instructions on reverse side)					NEW MEXICO	LUNA		
A DESCRI FROM NOR (38 WES WOOL PROS	TION OF ORIGINATION OF ORIGINATION OF ON INTERPORT	INAL MARK UTHERN PA G STATE H ST OF THE C ROW OF PO VESS POST. O.30 METER	ACIFIC CON 16HWAY I RENTERLIN LES, AND A STANZ 2 (1.0 FO	1PANY RAILROA   FOR 4.6 KM  E OF THE HIGH  O 34 METER  ARD DISK SET  OT) ABOVE GR	D STATION 1 (2.9 MI), 1 WAY, 19.5 M (1.1 FOOT) EA IN TOP OF 1 OUND DEST	N COLUMBUS, 1. 6 METERS ETERS (64. FEET) ST OF A WHITE A CONCRETE POST ROYED THIS DATE.		
PUBLISHED	ON	1082 LINE	101	STAMPING_K	133 1934			
B. WAS ORIGINAL BENCH MARK DESTROYED? VES NO HORIZONTAL CONTROL POINT? VES NO REMARKS INSTRUMENT: NI 2 50132 ROD: PHILADELPHIA, FOOT UNITS (Feet or Heter)*								
POINT	8.5.	H.t.	F.S.	ELEVATION	F	REMARKS		
K 133	3.791	4182.759	<u></u>	4178.968	CL	AR, CALM, WARM		
K 133 RESET 1977	5.177	4182.745	5.191	4177.568				
			3.789	4178.956 -	012 7 .009	REVECTED		
K 133	3.801	4182.769		4178 968				
K 133 RESET 1977	5.250	4182.820	5.199	4177.570	<u> </u>	4177.570		
			3.858	4178.962 -	006 MEA	V= 4178.965		
					ELEVATION DIFFERENCE	= - 1.395 <sub>FF</sub>		
			<u></u>					
C. DESCRIP	TION OF NEW N	ARK		L	L	33 RESET 1977		
FROM NOR7 (99) (40.0 UTIL RIG POST	THE SOU H ALON EEET) EA FEET) JTY POL HT-OF-WA F PROJE	UTHERN PAG IG STATE I ST OF THE SOWTH OF ES, AND O AY FENCE CTING 0.3	CIFIC CON HIGHWAY CENTERL MILEPO 0.30 METER O METER	STAMPING MPANY RAIL.ROA II FOR 4.6 ( INE OF THE H ST 6.13, IN 2 TER (1.0 FOOT, NDARD DISK R (1.0 FOOT) A	ON NEW MARK A. / D STATION A (M (2.9 MI)) IGAWAY, 12. LINE WITH A ) WEST OF T, SET IN TOF- BOVE GROUND	4T COLUMBUS, 30.1 METERS 19 METERS 4 ROW OF 4E HIGHWAY 0F A CONGRETE		
*Strike out unit NOT used. AGENCY NAME AND ADDRESS								

Figure 2-29.—Report of a relocated bench mark.

ment that bear a stamping (such as a logo flange or stainless steel cap on a rod) and return them to NGS.

Prepare a recovery note on the report form or a standard description form (NOAA Form 76-186), identifying and explaining the disposition of the old monument.

Submitting relocation records. To include relocation data in the national network, the following records must be forwarded to NGS:

- 1. Complete description of the new control point
- 2. Recovery note for the old control point.

3. Field records for at least two runs of leveling, one forward and one backward, including the computed elevation difference from the old to the new point.

4 Name, address, and telephone number of the individual who performed the relocation.

## 2.6 References

Dracup, J. F., 1976, rev. 1979: National Geodetic Survey data: availability, explanation, and application. NOAA Technical Memorandum NOS NGS 5, National Geodetic Information Center, NOAA/NOS, Rockville, Md. 20852, 45 pp.

- Floyd, R. P., 1978: Geodetic bench marks. NOAA Manual NOS NGS 1, National Geodetic Information Center, NOAA/NOS, Rockville, Md. 20852, 55 pp.
- Gossett, F. R., 1959: Manual of geodetic triangulation, Coast and Geodetic Survey Special Publication 247. National Geodetic Information Center, NOAA/ NOS, Rockville, Md. 20852, 352 pp.
- Greenawalt, C. B. and Floyd, R. P, 1980: National Geodetic Survey Operations Manual (unpublished manuscript). NGS Operations Division, NOAA/NOS, Rockville, Md. 20852.
- National Ocean Survey, published annually: NOAA Tide Tables: High and Low Water Levels. NOAA, National Ocean Survey, Rockville, Md. 20852.
- Pfeifer, L. and Morrison, N. L., 1980: Input Formats and Specifications of the National Geodetic Survey Data Base, vol. II: Vertical control data. Federal Geodetic Control Committee. National Geodetic Information Center, NOAA/NOS, Rockville, Md. 20852, 136 pp.

ć,

(

ĺ

## **3.1 Introduction**

In a widespread network of vertical control, geodetic leveling is the technique that provides the most reliable elevation differences between control points. It is a form of precise leveling where the observing team limits the magnitude of error by using calibrated instruments in combination with a rigorous, symmetrical observing procedure. In the following pages leveling procedures are described, the principal sources of error are discussed, and error tolerances are presented to provide a foundation for the instructions that comprise the rest of the chapter.

## **3.1.1 General Procedures**

Along each line of a vertical control network, leveling is conducted in increments called sections. Each section is an unbroken series of setups, made between two permanent control points. A setup consists of a point supporting a backsight rod, a point supporting a foresight rod, and a leveling instrument positioned between them (fig. 3-1). Two heights are measured by sighting through the instrument toward a scale on each rod and recording the values intercepted by a line on the reticle. The height difference, backsight minus foresight, corresponds to the elevation difference between the two points. The foresight point of one setup becomes the backsight point of the next; thus, the sum of the elevation differences of the series of setups is the elevation difference for the section (fig. 3-2).

Three conditions must be satisfied for this technique to provide reliable elevation differences between points. First, the lines of sight from the instrument to the rods must be level; in other words, the lines of sight must be



Figure 3-1.—Setup of leveling,  $\Delta h = B - F$  and  $s = s_B + s_F$ 

parallel at all times to the reference surface. Second, the values observed on the scales must accurately indicate heights above the points on which the rods rest, and third, the points in turn must be stable with respect to the topography.

These conditions cannot be perfectly satisfied in the "real world"; however, they may be approximated by limiting the known sources of error. Leveling is classified by the degree with which error magnitudes are limited. (See sec. 3.1.3, "Tolerances for geodetic leveling.")

## 3.1.2 Sources of Error

Error may be characterized as random or systematic. Random error in leveling results represent the effect of unpredictable variations in the instruments, the environment, and the procedure of leveling. Random error cannot be completely eliminated, although it can be kept small. Therefore, it represents the "noise level," the limit on the precision with which leveling may measure elevation differences.

Systematic error represents the effect of consistent inaccuracies in the instruments or in the leveling procedure. It also results from consistent, though not always predictable, environmental effects. Although systematic error may be small in a single measurement, it accumulates when measurements made under similar circumstances are totaled. Thus, it can result in a significant discrepancy in the elevation differences measured between two control points by different leveling systems and/or routes. For leveling to provide accurate elevation differences, systematic error must be eliminated, either by procedure or by applying corrections to the data.

The sources of error in leveling can be classified into three groups: those affecting the line of sight, those affecting the heights observed, and blunders. The line of sight cannot be exactly level because of the effects of imperfections in the instrument and variations in the human eye, refraction, curvature, and tidal accelerations. The heights observed are not exact because of imperfections in the rod scales and the turning points; furthermore, a perfectly stable relationship cannot be maintained between the equipment and the topography because of environmental effects on the equipment. Blunders may occur while attempting to limit any of these errors.

Leveling instrument. The instrument used for geodetic leveling should consistently provide a horizontal



Figure 3-2.—Section of leveling,  $\Delta H = \Delta h_1 + \Delta h_2 + \dots + \Delta h_n$  and  $S = s_1 + s_2 + \dots + s_n$ .

line of sight. The extent to which it achieves this determines its suitability for various orders and classes of leveling.

To be horizontal, the line of sight should be perpendicular to the direction of gravity, at the vertical axis of the instrument. If the line of sight is not horizontal, the angle by which it deviates from being horizontal causes an error in every observation (fig. 3-3). This angle is referred to as collimation error.

Collimation error can be limited by using a welldesigned, properly maintained instrument. The angle should be measured and adjusted to specifications. The effect of collimation error on each observation can be reduced by limiting the sighting distance. Furthermore, if the sighting distances in each setup are balanced, the errors resulting from collimation error become equal. They cancel when the foresight is subtracted from the backsight to compute the elevation difference (fig. 3-4).

Although it is impractical to balance every setup exactly, the total contribution of collimation error can be limited very effectively while leveling by keeping the imbalance small and random in sign. Any systematic contribution that accumulates with distance may be



Figure 3-3.—Effect of collimation error,  $\alpha$ .

eliminated by later applying corrections computed from the imbalance and a precisely determined value for the collimation error.

Collimation error should not change as the instrument is refocused or rotated about its vertical axis. A consistent difference between the collimation errors of the backsight and foresight causes a systematic accumulation of error (fig. 3-5). The precautions described here will limit this effect, but, because it is unpredictable, it cannot be eliminated. Additional precautions should be taken to ensure that a consistent difference does not exist.



Figure 3-4.—Consistent collimation error cancels in a balanced setup since  $s_B = s_F$ .

In a compensator-type leveling instrument such a difference can occur if the compensator is not suspended properly in each direction. To prevent gross error the spherical level on the instrument should be properly adjusted, and compensation checks should be performed routinely. To eliminate smaller systematic effects, the compensator should be repositioned during each setup.

In a spirit-level type of leveling instrument, the error caused by imprecision in centering can become systematic if the bubble is consistently affected by a heat source in one direction. Shading the instrument should reduce this effect.



Figure 3-5.—Inconsistent collimation error does not cancel in a balanced setup since  $\alpha_B \neq \alpha_F$ , even if  $s_B = s_F$ .

*Pointing.* Another effect on the line of sight results from the human inability to repeat a pointing exactly. Both imperfections in the instrument and atmospheric refraction may contribute to this effect, combining with the imperfection of the human eye, to create pointing error.

The magnitude of pointing error is reduced by the use of a precise instrument with a micrometer and wedge reticle that has been adjusted to remove parallax. Limiting the sighting distance and instrument movement can also reduce the magnitude of the error.

*Refraction* Variations in atmospheric density cause the line of sight to refract or bend in the direction of increasing air density. These variations seem to be primarily a function of air temperature.

Refraction is most noticeable when the line of sight passes through air of fluctuating density, as when "heat waves" are observed. The graduations on the scales appear to move up and down rapidly. This phenomenon, called shimmer (fig. 3-6), makes it difficult to intercept the scales precisely, thus increasing the magnitude of pointing error. It can be reduced by shortening the sighting distances or, in some cases, raising the height of the line of sight.

Whether or not shimmer is observed, the line of sight may be refracted. Since the error caused by refraction increases proportionately with the square of the sighting distance, refraction error may be reduced by limiting the sighting distance. As long as atmospheric conditions are similar along both the foresight and backsight, the error may be nearly eliminated by balancing setups.



Figure 3-6.—Shimmer.

However, conditions are often not the same along both lines of sight. Air close to the ground changes in density more rapidly than air situated 1 m or more above ground. This can be visualized by imagining air layers, of equal density, conforming to the topography. On a slope, even if setups are exactly balanced, the conditions along the foresight differ from those along the backsight (fig. 3-7). Because the sight uphill passes through a greater change in air density, it is refracted more Refraction error, then, accumulates with change in elevation.



Figure 3-7.—Refraction error, r, does not cancel on sloping terrain since  $r_B \neq r_F$ , even if  $s_B = s_F$ .

Leveling results may be corrected at least partially for refraction if atmospheric conditions are determined and recorded with the observations. Of the many mathematical models that attempt to predict refraction error, the most successful require that air-temperature differences be precisely measured during every setup

Two types of refraction error cannot be corrected. Therefore, the situations which cause them should be avoided when leveling. If the line of sight passes very close to the ground or to an intervening object, changes in air density cause the line of sight to be refracted in an unpredictable way. Similarly, when the air near the ground is cooler than the air above it, the relatively stable air layers may move slowly across the line of sight, causing long-period shimmer. The graduations on the scales appear to move up and down, but so slowly that an entire setup may be observed and checked before the movement is noticed. The observations may be significantly and unpredictably affected by the unnoticed shimmer. This effect usually occurs at night when the air is calm.

*Curvature.* Both the leveling instrument and the rods are oriented to the direction of gravity, to measure elevation differences with respect to the same reference surface. When the instrument is level and is rotated so

that the line of sight intercepts each scale, the line of sight should sweep out a horizontal plane. It would be parallel to the equipotential surface if the gravity field, at each setup, also defined a plane. This is not the case, however, since the gravity field defines a curved surface. Thus, a small amount of curvature error is introduced into each observation (fig. 3-8).

Curvature error is proportional to the square of the sighting distance. Assuming the equipotential surface is evenly curved, curvature error can be reduced by making the backsight and foresight distances in each setup nearly equal. If setups are thus balanced, within a tolerance, correction for curvature need not normally be made while leveling.



Figure 3-8.—Curvature error, c, where the line of sight is not parallel to an equipotential surface, cancels if  $s_B = s_F$ .

Because the surfaces defined by the gravity field are not evenly curved, minute differences in curvature error in every setup accumulate systematically in leveling conducted over large changes of elevation, particularly in a northerly or southerly direction. Even so, the magnitude of this error is so small that it may be neglected if the backsight and foresight distances are nearly equal.

Tidal accelerations. Because leveling instruments and rods are oriented to the direction of gravity, after curvature has been taken into account, the elevation difference of each section is computed along a route that approximately parallels an equipotential surface. However, the Sun and Moon create tidal accelerations that periodically distort this surface, generally more toward the equator than the poles (fig. 3-9).

The distortion is termed a deflection and is described by two component vectors. The vertical component affects only the magnitude of gravity along the route, resulting in a negligible effect on the elevation difference. The horizontal component, however, acts at 90° to the equipotential surface, resulting in a small error, especially if the section is oriented in a line with the Sun, Moon, and the north or south pole. The error accumulates significantly in leveling lines oriented north-south, particularly in the middle latitudes. To remove it, a correction must be applied.



Figure 3-9.—Effect, a, of tidal deflection,  $\epsilon$ , on a section of length and direction,  $\vec{S}$ .

Leveling rods. To observe accurate heights above the points on which the leveling rods rest, precise relationships must be maintained between the rods and the equipotential surface, and between the rods and the scales mounted on or within them.

The first relationship is ensured by plumbing, or aligning the rods with the direction of gravity, a task which is analogous to leveling the instrument. If they are not so aligned, an error is introduced into each observation. Although the error may be small, it accumulates systematically with change of elevation, especially on steep slopes where observations are made alternately low and high on the scale. (See figs. 3-10, 3-11, 3-12.) It can be limited only by plumbing the scales properly.



Figure 3-10.—Effect of rod plumbing error,  $\rho$ , on a height observation, *B*.

The second relationship depends upon the accuracy with which the scales are manufactured and mounted in or on the rods. If the graduations are not accurately marked above the scale zero or if the scale changes in length during leveling, error may accumulate in the observations. To limit the error, leveling rods should be well-designed, routinely calibrated, and properly handled and maintained.

The index error, which is the difference in height from the scale zero to the base plate of the rod, represents a constant portion of the error in the scale values. Index error can be eliminated by making an even number of setups for every section, thus using the same rod on



Figure 3-11.—Systematic effect of plumbing error (and scale errors) is small on flat terrain, since  $B_1 \simeq F_2$  and  $F_1 \simeq B_2$ .

every bench mark during the leveling. (It can also be eliminated if only one rod is used if accurate calibration corrections are applied.)

In much the same way as an error in plumbing, error in the scale values causes systematic error to accumulate with change of elevation. For the most accurate results, observed scale values should be corrected to standardize them with respect to the National Standard of Length. To do this the rods must be accurately calibrated. Since the scales are subject to thermal expansion, the coefficient of thermal expansion must also be measured, and scale temperatures should be recorded during the leveling.

Turning points. The leveling rods must be set on clearly defined, stable points. Rather than using natural objects, which may or may not prove to be adequate, portable, standardized turning points should be used. They should provide definite points on which to rest the rods. Both the base plate of a rod and the top of a turning point should be cleaned before placing the rod on the point.

To prevent error when leveling to and from control points, they too should provide definite points. Since this is not always possible, special procedures may be necessary for leveling to awkward points.

Stability. During each setup, the leveling instrument and the rods may change elevation because of settlement or rebound caused by the type of ground cover. To minimize error from such movement, the backsight and foresight should be observed nearly simultaneously. This requires two rods and an efficient instrument and leveling unit.

Instrument movement can be reduced by proper use and maintenance of the tripod. Systematic error resulting from consistent movement of the instrument can be eliminated by using one of two observing procedures. In one procedure, known as micrometer leveling, two elevation differences are measured in two directions during each setup. If the instrument settles significantly during either or both measurements, the observations will not lie within a limit imposed on the difference between the results. Averaging the results practically eliminates any remaining systematic error.

In the other procedure, three-wire leveling, the first reading of the setup is made on the backsight of oddnumbered setups, and on the foresight of even-numbered setups. Consistent settlement of the leveling instrument will cause the results of odd setups to be too large and the results of even setups to be too small by a similar amount. Thus, over a section with an even number of setups, the errors nearly cancel.

Rod movement is minimized by placing the turning points so they provide sufficient stability. Systematic error caused by consistent movement during each setup can be limited somewhat by the procedures described previously. If the rods are allowed to rest on the points for 20 seconds before making an observation, any remaining movement should be negligible.

When proceeding from one setup to the next, the forward turning point must not move. This can only be assured by the rodman's diligence and by comparing repeated levelings of the section. Averaging the results of two runnings, made in opposite directions, nearly eliminates any systematic accumulation of error resulting from consistent movement.

Blunders. Errors that result from failure to follow the specified procedures are termed blunders. In leveling, no unit is so experienced that the work performed by the group can be automatically considered free of blunders. Only conscientious attention to detail, combined with scrupulous checks and cross-checks, can ensure accuracy.



Figure 3-12.—Effect of plumbing error (and scale errors) accumulates on sloping terrain, since  $B_1 \neq F_2$  and  $F_1 \neq B_2$ .

Random blunders are usually large and occur at single setups. They include (1) mistakes in reading or recording scale values, (2) observing the rods in the wrong order, (3) improper placement or leveling of the instrument, (4) gross movement of the instrument or turning points during the setups, (5) improper placement or plumbing of the rods, and (6) gross movement of the forward point between setups. The first four of these blunders may be detected and eliminated with confidence by using double-scale rods, computer-recording equipment, and the micrometer-leveling procedure given in section 3.7.2. In this system, observations are made in such a way that checks for internal consistency can be computed. The last two blunders, if unreported by the responsible rodman, may be detected only by comparing repeated levelings of the section.

Systematic blunders are more difficult to detect. They are primarily caused by failure to follow the procedures necessary for precise leveling, particularly maintenance of the instrument and the rods. If the collimation check and the compensation check are not performed correctly, the use of an improperly adjusted instrument may introduce error, undetectable in each setup, but accumulating over the line. Similarly, if the spherical levels on the rods are not properly adjusted, error may accumulate. If a rod is dropped, or otherwise damaged, previous calibration values may no longer apply and the data cannot be corrected with confidence.

### **3.1.3 Tolerances for Geodetic Leveling**

To produce reliable elevations, the results of geodetic leveling must satisfy the appropriate standard of accuracy for the order and class of a survey. This standard is attained in three ways while leveling: first, by operating a well-organized and well-trained leveling unit; second, by selecting sufficiently precise equipment and calibrating it properly; and third, by adopting observing and recording routines that limit accumulation of error. Specific instructions for satisfying these requirements are presented in the remaining sections of this chapter. Tolerances, suitable for attaining each standard of accuracy, are presented in table 3-1.

Two types of tolerances must be applied. The first type is a set of practical limits on the dimensional factors within the leveling system that are known to increase the magnitude of error in the results. The second type is a set of limits on the precision of the results obtained from two or more levelings between the same two points. The tolerances should reflect the probable magnitude of random error that cannot be corrected or eliminated. Then, blunders and most systematic error can be detected and removed from the leveling data

The second set of tolerances is defined with the expectation that the tolerances will be exceeded 5 percent of the time (95 percent test level). This permits a check

on the validity of the limits imposed. If leveling results exceed the limits more than 5 percent of the time, either the first set of tolerances is not being satisfied or the second set has been made too restrictive (forcing observations that may not be accurate to appear precise). If leveling results exceed the limits less than 5 percent of the time, the second set of tolerances may not be restrictive enough.

Tolerances, revised periodically by the Federal Geodetic Control Committee, are published in *Classification, Standards of Accuracy, and General Specifications of Geodetic Control Surveys* (Federal Geodetic Control Committee 1974) and Specifications to Support Classification, Standards of Accuracy, and General Specifications of Geodetic Control Surveys (Federal Geodetic Control Committee 1980).

## 3.2 Organizing a Leveling Unit

During a leveling project one or more units operate under the supervision of a project director who supports and monitors the activities. Like mark setting units, each leveling unit must be capable of working selfsufficiently several hundred kilometers from the project office. Guidelines are presented here for organizing and maintaining such units.

### 3.2.1 Personnel

A leveling unit normally requires five persons: the observer, the recorder, two rodmen, and a pacer. Certain equipment or conditions may require that additional personnel be assigned. An experienced unit, equipped with a compensator-type leveling instrument and special vehicles, can operate efficiently with four persons.

The leveling unit is exactly what the name implies, a unit, and each person in the group, whatever the position, must be constantly on the alert to ensure that all duties are performed promptly and carefully. Complete and continuous attention to the details of the work is required of all members of a unit if it is to operate smoothly and efficiently. Any member of the unit who is slow, careless, or indifferent can ruin the work of the whole unit.

Unit chief. The unit chief, who usually serves as the observer, is responsible for managing the unit in the field. The quality and quantity of leveling produced depends on the experience and good judgment of the chief. Practical proficiency in each of the leveling duties is a prerequisite for the job. This individual should set an example which ensures correct, safe, and efficient leveling.

The chief should ensure that the specifications and guidelines contained in this chapter are strictly followed by all unit members. New employees, in particular, should be thoroughly instructed in their duties. The chief should ensure that project instructions are read
	First order class I	First order class II	Second order class I	Second order class II	Third order
Lines of sight Maximum sighting distance Maximum imbalance.	500 m	60 0 m	600 m	700 m	900 m
per setup per section	±20 m ±40 m	±50 m ±100 m	±50 m ±100 m	±100 m ±100 m	±100 m ±100 m
Leveling instrument Maximum collimation error. single line of sight		$- \pm 10''0$ (C $\leq \pm 0.05$ mm/m)		>	
Maximum collimation error. mean of two lines of sight Maximum angular difference		$\pm 4''0 (C \le \pm 0.02 \text{ mm/m})$		>	
in two lines of sight Setting precision Minimum reading	±0"25 ±0.1 mm Micrometer required	$- \pm 40''0 (Q \le \pm 0.20 \text{ mm/m}) - \pm 0.25 \\ \pm 0.1 \text{ mm} \\ \text{Micrometer} \\ \text{required}$	±0"50 ±0 5 mm	±0"50 ±05 mm	
Leveling rod Plumbing accuracy Maximum scale unit Calibration accuracy	±10'0 10 cm ±0 05 mm	± 10 '0 1 0 cm ±0 05 mm	±10′0 10cm ±005mm	±10′0 1.0 cm ±0.05 mm	
Agreement of observed elevation diff	ferences before co	rrection, observed backward and forward	l during		
One-setup section Two runnings of a section	±0 40 mm	±1 00 mm			·····
Two runnings of a section of one- way length K: $T \times \sqrt{K}$ mm, $T =$	±0 95 mm ±3 00	± 1.26 mm ± 4.00	$\pm 1 90 \text{ mm}$ $\pm 6 00$	±2.53 mm ±8.00	±3 /9 mm ±12 00
Three or more runnings of a section t given below For a section les	on: Each accepted s than 0 10 km in	I running must differ from the mean of length. let $K=0.10$	f all accepted runnir	ngs by no more th	an $t \times \sqrt{K}$ mm.
Number of levelings:					
3	2.10	2 81	4 21	5 63	8 44
+ 5	2 33	3 31	4.00	6 64	9 95
6	2.59	3 46	5.19	6 94	10 4
7	2 68	3 58	5.36	7 18	10 7
8	2 75	3 67	5 51	7 37	11 0
One leveling of a loop of one-way length $K$ , beginning and ending at the same point:					
$T \times \sqrt{K}$ mm. $T =$	±4 00	±5 00	±6 00	$\pm 8 00$	±12 00

and understood by unit personnel, resolving any technical questions before work begins. When substantial confusion or doubt about a field procedure exists, the chief should seek immediate guidance from the project director.

In addition to technical concerns, the chief is responsible for the safety of personnel, and for the security and maintenance of all equipment assigned to the unit Any deficiencies in these areas must be reported immediately to the project office The chief submits weekly and monthly status reports, and is also charged with the safekeeping of the leveling data until they are properly transmitted to the project office. The chief can improve the unit's efficiency by planning each day's activity in advance. By reviewing the logs in advance of the surveying date, sections which require unusual survey procedures can be pinpointed and the project director notified if any additional personnel or equipment are needed. It is helpful each morning to explain the day's goals to the unit. Whenever possible, duties should be rotated to provide opportunities for unit members to develop and upgrade their skills. When weather conditions or other factors prevent leveling, the chief should assign unit members other duties such as maintaining equipment or augmenting their skills and knowledge. Observer The unit chief or another individual who serves as the observer should have a thorough understanding of the leveling instrument, as well as the experience to operate it correctly and efficiently. The observer is responsible for maintaining the instrument and the tripod, and should follow the precautions outlined in subchapters 3.3, "Leveling instruments," and 3.6, "Atmospheric conditions" He or she should be constantly alert, to ensure that the route for the line is followed correctly, to plan ahead to avoid problematical setups, and to monitor the activities of the rodmen. The observer should check the recorder's work periodically and check and initial all records at the conclusion of each day's work.

Recorder. The recorder enters observation data on recording forms or into a portable computer. He or she should write legibly and be able to follow exactly the recording procedures outlined in subchapters 3.7, "Observing routine," and 3.8, "Field records." A clear understanding of the mathematical computations and checks is important, whether recording on a form or a computer, and the ability to compute these checks rapidly is essential when using recording forms. When using a computer-recording system, the recorder should be familiar with the various recording programs and skilled in operating and maintaining the equipment.

The recorder should be an alert and conscientious driver, since the truck that transports the rodmen and observer from setup to setup can be used as a recording station. In addition, the recorder should support the observer by checking constantly for blunders or oversights, which would lower the quality of the leveling,, and by serving as acting observer in the absence of the unit chief

Rodman. Although the position is often filled by an individual with little surveying experience, the rodman's duties are critical to precise leveling. With a computerrecording system, observing and recording blunders may be discovered and corrected before leaving each setup; however, no check can be made as to whether or not a turning point or leveling rod has moved between setups unless the entire section is releveled. The turning point and the leveling rod are the responsibility of the rodman.

He or she should be unquestionably reliable and conscientious, able to pace precisely, adept at handling the rod, and meticulous in setting turning points, following the precautions described in subchapters 3.4, "Leveling rods," and 3.5, "Turning points." The rodman is charged with the responsibility of maintaining the equipment properly and reporting deficiences promptly. When a mistake is made (e.g., pulling a turning pin too soon or accidently bumping a turning plate), the rodman should report it immediately. For motorized leveling, the rodman must be able to operate skillfully a specially equipped motorcycle or small vehicle, and perform minor maintenance. Pacer. The pacer, an optional position which is usually rotated with the position of the rodman, marks out precisely balanced setups and sets the turning points ahead of the unit. He or she should be able to select a satisfactory leveling route, based on the information provided in the logs. The pacer also sets up warning signs and returns for the leveling truck after sections have been walked. In motorized leveling, this position is not necessary if the vehicles are equipped with accurate odometers

Other personnel. A spirit-level instrument must be shaded during leveling, thus creating the need for a person to hold an umbrella. In three-wire leveling, this person assists the recorder by making the back check. (See sec 3.7.4). He or she may also be needed to shield the instrument in extremely windy conditions.

If a traffic lane must be closed, two flagmen are required to divert traffic. These are usually the pacer and a sixth person who is temporarily assigned to the unit.

#### 3.2.2 Equipment

Survey equipment, itemized in appendix B, should be readily available and routinely maintained by the personnel in the unit. An inventory should be conducted when the equipment is issued to the unit chief and at least once a year thereafter. All equipment should be stored securely, in an organized fashion, in the leveling truck or other assigned vehicles.

Items of survey equipment have been classified and coded according to type. The codes must be recorded together with the serial numbers of equipment used to collect data that are to be included in the national network. The codes are listed in *Input Formats and Specification of the National Geodetic Survey Data Base* (Pfeifer and Morrison 1980: vol II, annex F).

In general, the individual who uses a piece of equipment should be responsible for its maintenance. Any defective or damaged equipment should be reported immediately to the project director. When the unit is working at a substantial distance from the project office, a spare leveling instrument should be available, especially if the instrument's collimation error cannot be adjusted in the field.

When equipment requires maintenance that cannot be performed by the leveling unit, the equipment should be returned to the project office. If the equipment cannot be delivered in person, it should be packaged properly to prevent damage in shipment. Instruments and computer-recording equipment are particularly vulnerable to such damage.

Motorized leveling. The term "motorized leveling" refers to using vehicles to transport all members of the leveling unit from one setup to the next. When personnel proceed between setups, they must usually wait for the backsight rodman, who has twice the distance to travel. Traditionally, the leveling truck, driven by the recorder, transports the rodmen half this distance to reduce the cycling time of each setup. When the rodmen are equipped with vehicles, the cycling time can be further reduced, especially if each vehicle is provided with a mechanism for setting the turning point and raising and lowering the rod without leaving the vehicle.

A specialized observing vehicle driven by the recorder may allow the observer to set up and level the instrument more efficiently, perhaps at a height sufficient to reduce the effect of refraction. A built-in intercom can prevent misunderstandings between the observer and the recorder. In addition, both the instrument and personnel are protected from the weather, permitting leveling to continue under a wider variety of conditions.

In the traditional, partially motorized routine, the backsight rodman proceeds to the leveling truck and rides to the instrument position of the next setup. From there the rodman proceeds to the foresight position, at which the pacer should already have placed a turning point. When only four persons are available, the rodmen must pace to and set their own points.

Vehicles, equipped with precise odometers, may improve the rodmen's efficiency. In the absence of a pacer, however, the rodmen must still set their own turning points.

## 3.2.3 Safety

Because lines of leveling typically follow highway and railroad right-of-ways, personnel must be alert at all times to the hazards of working close to traffic.

When operating along a road, each member of the unit must wear a highly visible, orange vest. Work should be conducted as far from the road itself as possible, avoiding setups which unnecessarily expose personnel to the traffic. For added protection, warning signs should be placed within 1.6 km of the unit, in each direction. Any other safety requirements should be identified at the time liaison with the State highway district office is established.

Any vehicles used to transport the rodmen and the observer during a section of leveling should be driven on the right shoulder of the road, oriented with the flow of traffic. Vehicles should never be backed against traffic. A warning light should be flashing and the vehicle's hazard lights should be turned on. If the road shoulder is too narrow or if the highway is too heavily traveled to accommodate the leveling truck or other vehicles, the section should be walked. If there is insufficient room for the unit to pass safely on foot, the procedures cited under "Controlling traffic" (next subheading) should be used.

When working along a railroad, do not wear orange vests or use flashing lights, since these may cause a train to stop unnecessarily. Each person should be alert to the presence of approaching trains and be ready to warn other unit members. Each person should also have an escape route planned, especially when working on terrain where a train is not visible until the last instant.

All personnel in the unit should be trained in basic first aid and cardiopulmonary resuscitation. At each work area they must be aware of the location and telephone number of the nearest emergency medicaltreatment center, so prompt assistance can be obtained if an accident occurs. Any personal injury must be reported to the project director immediately. More detailed guidelines for safety are given in the National Geodetic Survey Operations Manual (Greenawalt and Floyd 1980).

Controlling traffic. When a traffic lane must be closed, as when leveling across a bridge with no sidewalks or leveling through a tunnel, permission to control the traffic must be obtained from local authorities A suitable configuration for controlling traffic is shown in figure 3-13. Six personnel are necessary: two flagmen and a fourperson leveling unit. The flagmen must be equipped with portable radios to coordinate their efforts. The recorder should drive the truck as for routine leveling. When added protection is necessary, two extra vehicles, each driven by an additional person, should precede and follow the unit.

After setting up warning signs, the unit proceeds down the right lane, as shown in figure 3-13. The forward flagman moves about one setup ahead of the unit, stopping traffic in the left lane (example 1). The rear flagman stops traffic behind the unit. After notification by radio that traffic in the left lane is stopped, the rear flagman begins directing traffic around the unit. When the setup is complete, the unit waits until the rear flagman has stopped traffic in the right lane (example 2). After notification of this, the forward flagman allows traffic to proceed in the left lane, and advances one setup ahead of the next setup. The unit advances at the same time to establish the next setup (example 3).

### 3.2.4 Reports

The project director should visit each unit at least once each month to inspect technical and safety procedures, and to provide personnel with updated information on project accomplishments. At all other times the unit chief should maintain regular communication with the project director. This can be accomplished in person when the unit is operating near the project office and by telephone when the unit is at a distant location. In addition, the reports described in the following paragraphs are required of the level unit.

Weekly leveling report. The weekly status report is submitted in person or mailed to the field office at the end of work each week. Any certified or registered-mail receipts for leveling data sent to the project office during the week should accompany the report. The report

STOP Flagman Observer SLOW <u>کۆ</u> <u>-)%</u> Flaoman 0 O A O Rodman Rodman Recorder With one vehicle: 1 (STOP) σ SLOW  $\sim$ 2 ō STOP lsion Flagman moves റ 3 SLOV Flagman moves  $\circ$ 

Figure 3-13.—Controlling traffic.

includes notes on progress, weather, activities, equipment condition and maintenance, suggestions, and requests for supplies. Figure 3-14 shows a properly executed NOAA Form 76-159.

When completing the form, enter distance in kilometers to one decimal place. Of primary interest is progress—the amount of the assigned line which has been completed. Report progress in two categories: double run and single run. Double-run progress is the total one-way distance of all sections that have been leveled in both the forward and backward directions. Singlerun progress is the total one-way distance of all sections that have been leveled in only one direction.

When a section is releveled in the same direction as an earlier leveling, total the distance as rerun. For example, if a double-run section fails to close, count the distance for the third leveling of the section as the first rerun. If still more levelings are required to obtain closure, count the distances in the column for additional reruns.

The total distance represents the amount of work performed by the unit each day. It is the total of all the leveling distances, including both directions of doublerun sections and all reruns. Check the statistic totals each day by this formula:

$$(2 \times \text{double-run progress}) + \text{single-run progress} + \text{rerun} = \text{total}.$$

The formula may be used to calculate double-run progress each day when an entire line is double run. Total the distance for the day, subtract all reruns, and divide what remains by two.

Report the number of setups of leveling completed each day. Do not count setups from the collimation check and setups from incomplete sections. Also report the number of section runnings completed.

The time at which the first setup began and the time when the last setup ended should correspond with the beginning and ending times recorded in the leveling data. Any delays of 15 minutes or more during the workday should be noted and explained in the remarks.

Note the number of collimation and compensation checks made. Also note the number of persons working with the unit for the day. Columns for weather conditions and remarks should be completed. Note such items as: cloud cover, wind speed, temperature, unusual climatic conditions, change of lines, change of observer, change of unit personnel, plumbing of rods, other adjustment and maintenance activities, equipment breakdowns, accidents, training activities, and any incidents which have a bearing on the quality of the leveling or condition of personnel and equipment.

Monthly leveling report. The monthly leveling report is submitted, either by mail or in person, to the project

With two vehicles:

3 - 10

.

office at the end of work on the last day of each month. The report contains the same information as that given in the weekly report. Unlike the weekly report, however, the monthly report is forwarded to the headquarters of the National Geodetic Survey, where it serves as a source of statistical information and suggestions concerning the progress of leveling in the national network.

Figure 3-15 shows a properly executed monthly releveling report. At the end of each work day, the unit chief should fill out the line on the report corresponding to that day's activities. Then it is a simple matter to complete the totals at the end of the month.

Vehicle report. Careful records should be kept of all expenses incurred for the repair and maintenance of the leveling truck and other vehicles. A report summarizing these expenses should be submitted, together with all pertinent receipts, to the project office at the end of each month.

*Miscellaneous reports*. The unit chief should be familiar with current organizational procedures for submitting per diem vouchers, accident reports, personnel reports, and expense receipts.

# **3.3 Leveling Instruments**

The instrument used for geodetic leveling must be precisely manufactured to ensure both sensitivity and ruggedness. This subchapter describes the mechanisms that are important in an optical instrument that is suitable for geodetic leveling. Specifications are included for proper maintenance and use of the instrument.

# 3.3.1 Classification

Optical leveling instruments are classified in two ways. The first is by the mechanism with which they provide a horizontal line of sight. Spirit-level instruments depend on a sensitive tubular level attached to the telescope. A fine-pitch tilting screw is turned manually to center the bubble in the vial of the spirit level. Compensator instruments depend on a pendulous reflecting component, the compensator, within the optical system of the telescope. Responding to the attraction of gravity automatically renders the line of sight horizontal.

Since the line of sight provided by either type of instrument cannot be exactly horizontal, the resulting collimation error must be regularly measured and ad-

N (3-64	FORM	76-159						······				NATIONAL OCEANIC ,	U.S. DEPARTMENT OF COMMERCE
OBSE	ERVER C.	Sm	itt					WEEI	(LY RI	EPORT RECORD	<u>– LEV</u> DER	ELING UNIT M. Dreath	MONTH/YEAR July 19 82
RODI J.	Spa	CER r <b>ks</b>	m.	cy	ウ!	n. C	)ren	th			PARTY	<i>๎</i> ๛. <i></i> ӡ <i>๛</i>	
LOC/ C	Ie F	) lum	,w	a	; wa	ina	tche	e,l	Na.		PROJE	83624100	
LINE	:151	27	9								SURVE	First	CLASS II
DAY	DBL RUN PROG	DISTAN SGL RUN PROG	ICE (k IST RE- RUN	M) ADL RE- RUN	TOT ALL RUN	NUM SET UPS	BER SEC- TION	TIN IST SET UP	AE LAST SET UP	COLL AND COMP CKS	ST AF F DAY	WEATHER CONDITIONS	REMARKS
Sun 4													
Mon													Holiday (4+h + July)
Tue		8.4			8.4	125	٩	0821	1620		Ч	Clear, Mild	Narrow winding mountain Road, Heavy treffic
₩ed 7		8.2			8.2	107	6	<i>•</i> 839	1520	vv	4	Cloudy - Am Ptcloudy-Rain - PM	Good effort by Rodmen, Heavy trathic ; Ruin last sating
Thu B	1.95	2.9			6.8	82	5	0845	1615		4	clear, Warm	Rytremaly Heavy traffic, Line Tic Spur Move on line
Fri <b>9</b>		6.4			6.4	9]	7	0840	1528		4	clear, Warm	Extremely Heavy traffic, Hlanc Hwy; Heat Waves
Sat 10													
Total	1.95	25.9			29.B	405	27				16	km/Staff-Days = 1.86	

TU S GOVERNMENT PRINTING OFFICE 1982-564-008-1149

NOA (3-8)	A FORM	76-104											NATIONAL OCEANIC	U.S. DEPARTMENT OF COMMERCE AND ATMOSPHERIC ADMINISTRATION
OBS	ERVER		<u> </u>				M	ОНТН	LY RE	PORT		VELIN		MONTH/YEAR
		<u> </u>	SW	T	-++					m.c	YR,	M. 1	DRENTH	JULY 19 <u>82</u>
RODA	MEN, PA	J.S	PAR	ĸs	m	CY.	R. /	m.D	RENT	гH	PARTY	G	36	
LOC	ATIONIS	) 2. E			···· •						PROJE	ст <u>о</u> .	2124100	****
LINE	(5)	<u></u>	EL	m,	, WH.	jw	ENA	CHEE	,wA.		SURVE		<u>362 7 100</u>	C1.455
		27	9									FIRS	ST	<u> </u>
		DISTAN	VCE (	km)	1	NUM	BER	тн	ME	COLL		Down		
DAY	RUN	RUN	IST RE-	ADL RE-	ALL	SET UPS	SEC-	IST SET	SET	AND COMP	DAY	TIME	CONDITIONS	REMARKS
	PHUG	41	KUN	HUN	16	141	7	09.50	1410	CKS_	4			STEEP GRADE, WINDWL
· 。		44			44	01	-	00 14	18/0		4		CLEAR-AM	MOUNTAIN PASS RD. STEEP GRADE, EXTREMELY
-		7.7		-		00	3	-100 IT	1328				CLOUPY-PM CALM	HEAVY HOLIDAY TRAFFIC
					<u>a.</u>	K			N	0				1. HAILE 085. BOTH 1= 520
			W	2	E	ĸ	-	E	N	<u> </u>		~		4 ZH OF JULY HOLIDAY
		01	н	0	4	<u> </u>	0	4	y .			10		NARROW WINDING MOUNT
<u> </u>		0.7			8.4	125	4	0821	1620		4		CLEAK, MILD	ROAD HEAVY TRAFFIC
7		0.2			8.4	107	6	0839	1520		4		PT. CLOUDY - PM RAW	TRAFER RAIN LAST SECTIONS
8	1.95	2.9	<u> </u>	<u> </u>	6.8	82	5	0845	1615		4	1	CLEAR, WARM	LIVE TIE SPUR ; MOVE ON LINES
		6.4		ļ	6.4	91	7	0840	1528		4	l	CLEAR, WARM	4 LANE HWY HEAT WAVES
10			w	E	E	к		Ε	N	D				
11			W	E	ε	ĸ	-	E	N	D				
12			м	0	V	E		D	A	Y	4	8	CLEAR, WARM	SWITCH EQUIPMENT TO NEW TRUCK-AM: MOVE FROM X
13		7.1			7.1	74	5	0828	1409	11	4	2	CLOUDY, WARM, RAIL	HEAVY TRAFFIC, 4 LANE HWY. DOWN TIME DUE TO RAIN
14		6.6			6.6	82	5	0807	15:43		4	2	CLOUDY - RAIN-AM PT. CLOUDY MILD - PM	DOWN TIME TO RAIN; VISITED I
15	3.25	2.5			9.0	114	4	0814	1627		4		CLOUDY, MILD	EXTREMELY NEAVY TRAFFIC; D BUN SPUR INTO WENATCHEE, WA
16	4.1				8.2	134	3	0815	1648		4		CLEAR, WARM	PBL. RUN SPUR THRU WENATCHEE TIME SPENT TO SET BACK TO TRUCK A
17			W	E	E	ĸ	_	E	N	D				
18			W	ε	E	ĸ	-	Ε	N	D	1			
19	6.75	8.1			9.6	110	6	0752	1637	11	4		CLEAR, WARM	20 MIN, MOVE DN LINE; HOT AL
20		9.2			9.2	112	8	0822	1544		4		CLEAR WARM	HIT BM ON ROCKY REACH DAM; HU
21		8.7			87	90	7	0809	1502		4	1	CLEAR, WARM	DOWN TIME TO STRONG GUSTY
22		8.9	┢───	<b> </b>	8.9	103	7	DALO	15.52		4		CLEAR, WARM	HEAVY TRAFFIC ; WORKING THRU
23		8.4		<u> </u>	8.4	100	6	0932	1609		4		CLEAR WARM	HEAVY TRAFFIC ALL MORNING
24		<u> </u>	141	-	-	2	-		10-7	~	<u> </u>			BAD HEAT REFRACTION - PM.
25						<u></u>				2	<u> </u>	<b> </b>	<u> </u>	
24			w	<b>*</b>		<u>n</u>	<u> </u>	=	N	<u></u>		1	ALEAR HET	75"- 101" F ; 1 HR. 45 MIN. LOST 7
27		1.2			1.2	87	5	0803	1692	11			CLEAR, HOT	GOING BACK FOR TRUCK WORK # 720-1020F : INE LOST TO MOVE
2/	ļ	7.6	<u> </u>	<b> </b>	76	96	3	0832	1635	<u> </u>	4	<u> </u>	CLEAK, HOT	DN LINE ; WORN AWN'S RE TRACKS
<u><u></u></u>	ļ	5.1		<b> </b>	5.1	124	6	0840	1558		+		CLEAR, HOT	NARROW GRADE ; NO USE OF TRUCK
		7.5	<b> </b>	[	7.5	124	7_	0902	1612		4		CLEAR, HOT	GRADE GOOD EFFORT BY CREE
30	L	<b> </b>	M	0	1	E		D	A	<u> </u>	4	8	]	PATEROS, WA.
31			W	E	E	ĸ	-	E	N	D	4	34	<u> </u>	
Total	10.05	121.8	}		141.9	1587	113	[			188	km/Sto	11-Days = 1.69	A SEE REVERSE

NOAA FORM 76-104 (3-82)

カリ S GOVERNMENT PRINTING OFFICE 1982-564-008-1148

Ć

(

(

justed within tolerance. Once the instrument is properly adjusted and the collimation error is known, the precision of the observations made depends in part on the consistency with which the spirit level or the compensator maintains the given line of sight. In a spirit-level instrument, this consistency is primarily determined by the degree of precision with which the bubble may be centered in the level vial. A coincidence prism or similar device enhances the precision of centering. However, since the bubble tends to "run" toward a point source of heat, the instrument must be shaded to ensure centering precision while leveling.

Compensator instruments maintain the consistency of the line of sight in various ways, some of which are described in this subchapter. In general, the automatic compensator mechanisms provide a setting precision superior to that attainable by centering a bubble. Thus, compensator instruments permit both greater productivity and improved leveling precision.

The second way in which leveling instruments are classified is according to whether or not they are equipped with micrometers. Without a micrometer, observation values may be estimated to, at best, 0.5 mm. The micrometer permits observations to be made with better precision. It permits values to be read directly to the equivalent of 0.05 mm or better.

For productive geodetic leveling, optical instruments should have the following features:

1 A telescope equipped with a large objective lens (50 to 70 mm diameter of the free aperture) to ensure brightness and good image definition.

2. A magnification power of 30 to 50 diameters.

3. A reticle etched between glass plates to ensure a consistent, dust-free image. It should include one vertical line, three horizontal lines (a middle line and two stadia lines) and, for use with scales with line graduations, a wedge. (See sec. 3.3.5, "Making an observation.")

4. Resistance to the effects of temperature change and vibration.

5. Units of the micrometer compatible with the graduations on the leveling rods. (See sec. 3.3.4, "Optical micrometer.")

Only instruments that are equipped with micrometers and that meet the appropriate standard of setting precision with respect to collimation error (table 3-1) can provide results considered to be of first-order accuracy. Although instruments with micrometers are preferred, instruments without micrometers can be used in combination with the three-wire leveling procedure to provide results considered to be of second-order accuracy.

Several different kinds of leveling instruments have been satisfactorily used for geodetic leveling in the United States. The Fischer level (fig. 3-16), designed by the Coast and Geodetic Survey, was the sole instrument used by that organization from 1900 to the early 1960's. Since then, commercial instruments that have been



Figure 3-16.—Fischer level.

extensively used by the National Geodetic Survey include: Breithaupt NABON, Jenoptik Ni004, Zeiss Ni2, Zeiss Ni1, MOM NiA31, and Jenoptik NI 002 (figs. 3-17 through 3-21). Some features of these instruments are given in table 3-2.



Figure 3-17.—Jenoptik Ni004.

#### 3.3.2 Telescopic System

In a simple telescope, such as that used in older spiritlevel instruments, the straight line between the optical center of the objective lens and the center of the reticle is called the line of collimation. A ray of light, passing through the optical center of a simple lens, continues to travel in its original direction as it proceeds beyond the lens. Thus, when the telescope is horizontal, the image of the center of the reticle, projected along the line of collimation to the center of the objective, defines a horizontal line of sight.

The ocular, or eyepiece, is a magnifying lens which permits the observer's eye to focus on the reticle lines and to see simultaneously on the reticle a magnified image of the scale of the leveling rod. The ocular and the observer's eye function jointly as a single lens. The ocular must be set at a fixed distance from the reticle, to enable the observer to see, without eyestrain or parallax, a sharper view of the reticle lines and the scale image. This setting must be adjusted to the eye of the individual observer. (See sec. 3.3.5, "Parallax adjustments.")



Figure 3-18.—Zeiss Ni2, with micrometer attachment.

To focus an instrument, the distance from the objective to the reticle,  $D_i$ , in figure 3-22, must be adjusted relative to the sighting distance from the objective to the rod scale,  $D_o$ . In older, external-focusing telescopes, the reticle and ocular are moved together in a combined mounting; the ocular is adjusted separately to focus distinctly on the reticle lines. In more recent versions, the reticle is in a fixed mount and only the objective lens is moved, usually by means of an external sleeve, to set the image distance as required.



Figure 3-19.—Zeiss Ni1.

In internal-focusing telescopes, both the objective and the reticle are mounted in fixed positions (fig. 3-23). This system requires the introduction of a third, focusing lens between the objective and the reticle. The spacing between the objective and the focusing lens determines the focal length of the system. Focusing is accomplished by moving the focusing lens back and forth until the focal length is that required to make a sharp image at the reticle for the particular sighting distance. An inverted image of the rod scale is formed. The internal-focusing system is much less likely to admit dust and moisture and is widely used in leveling instruments of high precision. In some instruments, the inverted image is made erect by a complex ocular system.

The presence of a focusing lens in the telescope requires a more general definition for the line of collimation. Because of imperfections in manufacturing, the center of the focusing lens cannot be set exactly on the line between the center of the reticle and the center of the objective Neither can the sliding motion of the focusing lens be adjusted to coincide exactly with the line of collimation. Thus, the line of collimation deflects slightly as it passes through the focusing lens. The line of collimation can be redefined as the path followed by a ray passing through the center of the objective and deflected by intervening optical elements, finally intersecting the center of the reticle.

The line of sight provided by the instrument is the forward extension of the line of collimation, through the center of the objective to the sighted object. In the case of a deflected line of collimation, the line of sight is the extension of the segment immediately behind the objective.



Figure 3-20.—MOM NIA31.

### 3.3.3 Compensator System

The presence of a compensator in an instrument eliminates the time-consuming operation of carefully centering the bubble in a level vial immediately before each observation. Instead, the instrument need only be roughly leveled with the tribrach screws, referring to a circular level rigidly attached to the instrument body.

Compensators can be grouped in two general classes: nonreversible and reversible. This manual explains two examples of nonreversible compensators: the Ni2 and Ni1 instruments. The NI 002 instrument is also described to provide an example of a reversible compensator.



Figure 3-21.—Jenoptik NI 002.

General description. The compensator consists of one or more reflecting elements, usually prisms, set in the line of collimation of the telescope. One of the reflectors is suspended within the instrument by fine flexible links (usually Invar bands). The orientation of this movable element is changed within the telescope system by the action of gravity, whenever the alignment of the instrument body is changed with respect to the local gravity vector. The lengths and geometrical arrangement of the suspension links, the position of the center of gravity of the suspension links, and the position of the center of gravity of the movable element are carefully designed to tilt the reflecting surface. The tilt assures that the path followed by a ray of light, entering at the center of the objective, is deflected to intersect the center of the reticle.

Since the housing of the instrument is made only approximately at each horizontal setup (within a range depending on the sensitivity of the circular level), the movable element within the compensator must assume an attitude such that the segment of the line of collimation immediately behind the objective is consistently

horizontal. The angle by which it deviates from the horizontal is the collimation error. The consistency with which the movable element repeats a given collimation error when the instrument is releveled, without otherwise being disturbed (e.g., in a laboratory), is termed the setting precision. It can be limited to less than  $\pm 0.25$  in a precise instrument. It compares with the centering error of about  $\pm 0.25$  for the tubular level of a precise spirit-level instrument.

Zeiss Ni2. The first commercial compensator instrument was the model Ni2, produced in 1950 by Carl Zeiss in the Federal Republic of Germany. It has a relatively simple compensator system which contains three elements. Two are prisms rigidly attached to the telescope body. The third is an inverted pendulum, suspended by two pairs of fine Invar tapes on which the element can swing into a stable position with respect to the gravity vector. A reflecting prism is attached at the lowest part of the element. The telescope system, into which the compensator is inserted, is of the internal-focusing type. A micrometer attachment is available for the Ni2.

The basic configuration of the compensator system, with the telescope axis horizontal, is shown in figure 3-24. The fixed reflector prism and the movable prism serve, in effect, as plane mirrors. By simple reflection, the prisms cause large deflections in the line of collimation along its path from the objective to the reticle. Because of the "floating" action of the movable prism, the deflection varies in magnitude as the vertical axis of the instrument changes slightly from setup to setup. The fixed roof prism picks up the reflected line of collimation from the movable prism and redirects it to the reticle, where the image of the rod scale is formed. The roof prism not only deflects the line of collimation, but, by double lateral reflection, also reverses the image of the rod, left to right and vice versa.

The focusing system in front of the compensator generates an inverted image of the rod scale, but the prism

Instrument	Micrometer	Image	Stadia factor	Stadia constant	Setting precision
Compensator instruments	i		******		
Jenoptik NI 002					
(reversible)	Integral	Erect	100	+0.4 m	± 0.''05
MOM NIA31	Integral	Erect	100	+0.37 m	±0''1
Zeiss Nil	Integral	Erect	100	0 0	±0!'1
Zeiss Ni2	Optional	Erect	100/333	0 0	±0."2
Wild NA2	Optional	Erect	100/333	0 0	±0!"3
Spirit-level instruments					
Wild N3	Integral	Inverted	100/333	-0.2 m	±0."2
Jenoptik Ni004	Integral	Inverted	100	00	±0!"23
Breithaupt NABON	Optional	Inverted	100	0.0	±0"2
USC&GS Fischer	None	Inverted	300/333	+0.6 m	±0"5

Table 3-2.—Features of some geodetic leveling instruments



 $\frac{1}{D_i} + \frac{1}{D_o} = \frac{1}{f}$ 

Figure 3-22.—External-focusing telescope.

system erects the image from top to bottom and reverses it left to right. Thus, the observer sees an erect, direct image.

With the telescope axis horizontal, the line of collimation is horizontal before reflection by the fixed reflector prism. The line which connects pivot points Aand B in the body of the instrument and the line which connects pivot points C and D by which the compensator is suspended are also horizontal. Points C and D are placed symmetrically with respect to A and B, by virtue of their suspension on either side of the movable prism. This requires four suspension ribbons in all, of nearly equal length. The line of collimation is deflected 45° by reflection from the first fixed prism, by 90° in the opposite direction from the horizontal surface of the movable prism, and again by 45° to coincide with the telescope axis. Thus, the image directed by the objective along the line of collimation is viewed at the ocular as if no deflection had occurred.

When the instrument is roughly leveled (fig. 3-25), the telescope axis is tilted within 4' from a horizontal position and the line of the pivots, AB, is tilted by the same amount,  $\alpha$ . Because of the flexible suspension ribbons, the supports C and D shift to a new position. The lengths, AC, BD, and CD, remain unchanged and the suspension ribbons bend slightly at A, B, C, and D, as if the supports were frictionless pivots. (Actually, the ribbons assume complex curved alignments, but if the axis is tilted only slightly, this causes no significant error.)



#### Figure 3-23.—Internal-focusing telescope.



Figure 3-24.—Ni2 compensator system, telescope axis horizontal.

Because of the shift of the support, CD, the lower reflecting surface of the movable prism is tilted in the same direction as the line of pivots, AB, but through a larger angle,  $\beta$ , by the relationship:

# $\beta = n \times \alpha$ .

The factor, n, is a function of the dimensions of ABand CD, the perpendicular distance from AB to CDwhen AB is untilted, and the height of the center of gravity of the suspended system above the axis CD. The magnitude of the factor in a particular instrument depends on the location of the compensator in the telescope.

In essence, the compensator must deflect a ray of light, entering a telescope whose axis is tilted by a small angle from horizontal, through the angle necessary to redirect the ray to the center of the reticle, as shown schematically in figure 3-26. In the Ni2 compensator the ray is deflected three times, but the net effect is that of a single deflection of the correct magnitude.

The position assumed by the movable element for any tilt within its functioning range depends on the dimensions of the suspension system and the position of the center of gravity. Figure 3-27 shows the condition that must be met for stability of the movable element during a setup. A, B, C, and D are the pivot points shown with the same notation as in figure 3-24. G is the center of gravity at a height, h, above the line of pivots, CD. Stability (no oscillation) occurs automatically when the gravity vector, acting downward from G, passes through the point of intersection, I, of the extensions of the suspension links AC and BD.

When the axis is tilted at a small angle, motion occurs at all four pivots. Since they are essentially the equivalent of frictionless pins, the system swings past the stable position, swings back, and then continues to oscillate about the stable position until the motion is damped. To damp it quickly, a loose-fitting, air piston-andcylinder damper is attached between the dynamic element and the instrument frame. Without mechanical contact between the piston and the cylinder, the device utilizes the resistance of the air pressure generated by the compression stroke to damp the oscillation in less than a second. Such a damper is vital to the productive use of a compensator instrument.

When stability is attained, the line of collimation is horizontal. Any small residual tilt resulting from imperfections in the instrument is adjusted by rotating a wedge-shaped optical element at the front of the objective. (See sec. 4.3.1, "Rotary-wedge attachment.")

This discussion of the basic configuration of the compensator system has been developed around the Ni2 instrument because the details of application are relatively straightforward. Application to other compensator instruments would follow the same general theory, but might vary considerably in detail.

Zeiss Nil. The Nil instrument, like the Ni2, is produced by Carl Zeiss in the Federal Republic of Ger-



Figure 3-25.—Ni2 compensator system, telescope axis tilted.

many. Designed some years after the Ni2, it is larger, more precise, and intended specifically to produce leveling results of better accuracy. It is equipped with a high-powered telescope and a built-in, plane-parallel plate micrometer. The compensator (fig. 3-28) of the Ni1 is generally similar to that of the Ni2. The movable element is an inverted pendulum suspended on four Invar ribbons. The line of collimation is deflected three times by reflection from prisms. Figure 3-29 shows the schematic arrangement of the compensator system when the telescope axis is tilted.

The compensator varies from that of the Ni2 in the following particulars. First, the Invar suspension ribbons cross at a point above the base of the movable element. This "X-suspension" responds to tilting and to the gravity vector in a manner similar to that of the Ni2 (a "V-suspension"). However, the axis of the movable element tilts in the direction opposite to the tilt of the telescope axis.

Second, the three deflections of the line of collimation are accomplished by only two prisms. The roof prism reflects once; the movable prism reflects twice, internally.

Third, the movable element is not symmetrical about the support system. The roof prism is mounted fairly high above the base pivots and to the rear of the support system, near the ocular. The resulting eccentricity is balanced by the placement of a large damping system at a position on the objective side of the sup-



Figure 3-26.- Deflection of the line of collimation by the compensator.



Figure 3-27.—Condition for dynamic stability of the Ni2.



Figure 3-28.—Ni1 compensator system, telescope axis horizontal.



Figure 3-29.—Ni1 compensitor system, telescope axis tilted.

ports. This does not affect the response of the movable element to gravity. The condition for stability still requires that the gravity vector from the center of gravity of the movable element pass through the point of intersection of the supporting ribbons, as projected on a vertical plane. A major attribute of the Nil is that the mass of the element is greater than that in the Ni2, which reduces its natural vibration frequency. This makes the system more stable in the presence of external vibrations.

Fourth, the Nil has a larger objective aperture and a higher-powered telescope than the Ni2. These features provide a sharper and more distinct view of the rod scale, a factor that improves the precision possible when intercepting the graduations.

Jenoptik NI 002. A major innovation in compensator instruments is the Jenoptik NI 002, manufactured in Jena, German Democratic Republic. Although its optical system is complex, the compensator operates on a simple principle by adapting a vertically suspended, plane mirror.

Figure 3-30 shows that the optical system defining the line of collimation has been reduced to only a few components. The objective lens, 2, transmits rays of light to the optically flat, compensator mirror, 3, which is supported in an essentially vertical position by links, 15, in an "X-suspension" array, similar to that of the Zeiss Nil. The instrument is focused by moving the mirror backward or forward. The reticle, 4, is engraved at the front of the objective lens. Therefore, the focusing distance is the total distance from the objective to the mirror and back to the objective.

The line of collimation, then, consists of two superimposed segments, both of which are horizontal if the compensator mirror is aligned with the direction of gravity. The important task of providing a horizontal line of sight is accomplished by the simple arrangement of directing rays from the objective to a suspended mirror, then reflecting them back to the reticle on the front surface of the objective. The only adjustment required is to distribute small counterweights so the compensator mirror hangs precisely in line with the direction of gravity.

The compensator mirror is manufactured to be a precise optical plane; both "front" and "back" surfaces are precisely flat and parallel within the fine tolerances of optical technology. The use of the double mirror as a reversible compensator is made possible by the provision of the rotary motion, h, with which, by the simple shift of a knob, the mirror and its suspension system can be rotated 180°. After rotation, the line of collimation is reflected from the reverse side of the mirror.

Since the line of collimation is tilted from horizontal by an angle,  $\alpha_1$ , in one position, it is tilted by a nearly equal and opposite angle,  $\alpha_2$ , after the mirror is rotated 180° (fig. 3-31). If the rod scale is observed twice, once



Figure 3-30.—Diagram of the NI 002.

- 1. Rotatable optical wedge for adjusting mean line of collimation.
- 2 Telescope objective.
- 3 Compensator mirror for telescope image.
- 4. Reticle engraved at front surface of objective lens.
- 5 Transparent plate with micrometer index line
- 6 Compensator mirror for micrometer index-line image.
- 7. Transparent plate with micrometer scale, attached to objective lens assembly.
- 8. Real image of combined images of micrometer index line, 5, and micrometer scale, 7, projected on face of prism c via optical train a
- 9. Circular level, for roughly leveling the instrument.
- 10. Real image of bubble in circular level, projected on face of prism c via optical train b.
- 11 Ocular assembly.
- 12. Compensator support.
- 13 Compensator support base
- 14. Compensator suspension block, attached to mirror, 3.
- 15. Flexible bands for compensator suspension block.
- 16. Air piston assembly for damping oscillation of compensator.
- a Optical train, transmitting image of micrometer scale, 7, and index line, 5; image formed on face of beam-splitting prism c, at 8.
- b. Optical train, transmitting image of circular level bubble, 9; image formed on face of beam-splitting prism  $\epsilon$ , at 10.
- c Beam-splitting prism, collecting images from the objective, at 4, and from optical trains a and b, at 8 and 10, to be transmitted to image plane e, via optical train d.
- d Optical train, transmitting combined images from 4.8. and 10. to image plane e
- e Image plane for optical train d, with images of rod and reticle lines from telescope combined with images of micrometer scale and index line (from optical train a) and image of circular level (from optical train b)
- f Pendulous motion of compensator system.
- g Sliding motion of compensator system, to focus telescope image on reticle at 4.
- h Reversing motion of compensator system; provides 180' rotation about vertical axis, interchanging mirror surfaces "" and ":" on 3; also reverses micrometer mirror 6
- i Vertical sliding motion of objective assembly, carrying micrometer scale 7, to set line of sight on integral rod graduation, thus serving as a mechanical/optical micrometer.
- j Location for plug to prevent vertical sliding of objective assembly during shipment

in position one () and once in position two (:), the mean of the readings is essentially free from the effect of the collimation errors. Any small residual tilt in the mean line of collimation ("quasi-absolute horizon") may be adjusted within 1."O of horizontal by rotation of a wedgeshaped optical element (item 1 in fig. 3-30).

Note that the difference between the readings is proportional to the angular difference,  $\alpha_1 - \alpha_2$ , between the collimation errors. This angle is normally kept within 20"0 by adjusting the counterweights on the compensator assembly.

During a setup of micrometer leveling with an imbalance,  $\Delta s$  (backsight distance minus foresight distance), the angular difference causes a difference between the elevation difference observed in compensator position one,  $\Delta h_L$ , and that observed in compensator position two,  $\Delta h_H$ :

$$\Delta h_I - \Delta h_H = Q \times \Delta s$$

This must be considered when checking observations for blunders. Since the angular difference fluctuates somewhat as the air temperature changes, its maximum magnitude should be used to compute the tolerance for the reading check. (See sec. 3.7.2, "Instructions: reading check.")

The precision of the NI 002 depends primarily on the relationship between the objective, the compensator mirror, and the reticle. The remainder of the optical system, which merely serves to transmit the scale image from the compensator to the ocular, is essentially described in the explanatory key to figure 3-30.

#### 3.3.4 Optical Micrometer

Around 1910, Heinrich Wild made possible a significant improvement in the precision of leveling observations when he invented the plane-parallel plate micrometer. This optical device eliminates the need to estimate the intercept of the middle line of the reticle as it appears against the image of a rod scale. The micrometer permits the image of the scale to be shifted vertically until an integral graduation is precisely intercepted by the middle line. The amount of vertical shift may then be measured.

The function of the parallel-plate micrometer depends on Snell's law. A ray of light, incident on a plane airglass surface at an angle, l, is refracted toward the perpendicular to the plane surface (fig. 3-32). After traversing a finite thickness of glass, t, and on passing through a glass-air surface parallel to the first, it is refracted through an equal and opposite angle. It thus emerges parallel to the original ray, but offset from it by a small distance, d. The distance may be computed from the index of refraction, n, for glass relative to air, as follows:

$$d=t[(n-1)/n]\tan l$$

On a leveling instrument, the plane-parallel glass plate is normally mounted at the front of the objective. The plate may be partially rotated about an axis perpendicular to the vertical axis of the instrument, by means of a knob. With the plate unrotated, the line of sight usually intercepts the rod scale at some point between two graduations, as shown in figure 3-33. Rotating the plate at an angle to the incident ray produces an apparent vertical shift of the field of view.

To measure the interval between the intercept of the line of sight and the first graduation below it, the plate is rotated until a graduation is shifted into coincidence with the middle line of the reticle (fig. 3-34). Since the interval depends on the tangent of the angle of rotation



Figure 3-31.—Lines of sight provided by the NI002.



Figure 3-32.—Snell's law applied to a plane-parallel plate micrometer.

of the plate, uniform rotation of the knob generates a tangent function of the angle of rotation of the plate. A drum scale, attached to the knob, permits the values of the function to be read, in units that measure the vertical shift of the line of sight.

The rotating range of the micrometer plate is 10 micrometer units (fig. 3-35). To avoid exceeding the limits imposed by the approximations involved in the equation, the range provides equal displacement on each side of the unrotated position of the plate. Rotating the upper edge of the plate toward the eyepiece causes an apparent lowering of the field of view, or raising of the line of sight. Rotating the upper edge of the plate away from the eyepiece causes an apparent raising of the field of view, or lowering of the line of sight. For convenience, the micrometer units are usually numbered from 0, with the line of sight at its maximum upward displacement, to 10, at its maximum downward displacement. Thus, when the plate is unrotated, the reading is 5 units.

As a result of this system of numbering, a micrometer reading is positive and must be added to the value of the intercepted graduation. Since the readings are referred to its maximum upper displacement (not to the undisplaced line of sight) the readings are 5 micrometer units too large. However, since they are too large by a constant amount, the difference of elevation obtained from the difference of the foresight and the backsight is mathematically correct.

The maximum vertical range of shift of the line of sight is set equal to the interval between two adjacent graduations on the rod scale, i.e., one rod unit. Because of this, only one graduation may be intercepted for any fixed position of the instrument and the rod.

The foregoing details apply to the most commonly used micrometer system Some instruments (e.g., the Jenoptik NI 002) achieve the same effect by shifting the objective lens itself vertically. The amount of shift is measured directly against an attached linear scale. Optical micrometers are available for use with rods graduated in any of a variety of units. One micrometer unit is normally equivalent to one-tenth of a rod unit. Readings are made to the nearest one-hundredth of a rod unit (equivalent to 0.05 mm with half-centimeter rods), as indicated by subdivisions, or can be readily estimated. Always use a micrometer compatible with the units of the rod scale; otherwise confusion and blunders will result.

## 3.3.5 Use and Maintenance

During field use the leveling instrument is subject to continual physical stress, including shock, vibration, windblown dust and dirt, moisture, and temperature change. Even the most rugged instrument cannot withstand this assault indefinitely The observer should take every possible precaution to protect it from these conditions.

Tripod. To ensure a stable platform for the instrument, the tripod on which it is mounted must be strong and rigid. It should be constructed of a material that is affected very little by temperature change. To provide stability in various types of soil, the feet of the tripod should have tapered metal shoes with projections that permit them to be pushed into the ground.

The legs should be of sufficient length to allow the observer to stand erect. For a given observer this means that the length necessary depends upon the location of the instrument eyepiece, which can vary considerably from instrument to instrument. The Jenoptik NI 002(fig. 3-21) requires a shorter tripod than the MOM NiA31 (fig. 3-20) or the Zeiss Ni1 (fig. 3-19). For routine leveling, tripods with adjustable legs are not recommended.



Figure 3-33.—Plane-parallel plate unrotated.

The observer should check the condition of the tripod head each day. A loose head makes it very difficult to maintain the instrument in a level position throughout the backsight and foresight of each setup. The hinge assembly should cover the legs snugly. Hinge tension is properly adjusted when each leg, raised to a horizontal



Figure 3-34.—Plane-parallel plate rotated to intercept a graduation.

position, slowly drops of its own accord.

Bolts, located between each pair of legs, should be tightened as necessary to reduce any play between the platform and the legs. Many modern tripods are constructed with plastic bushings that may crack or shatter if the bolts are overtightened, so care should be exercised when making this adjustment. The bushings may be replaced by taking the hinge assembly apart. The tripod hinge assembly should not be stressed by lifting a leg too high or by forcibly jamming the legs simultaneously into the ground. Shock from the latter activity may severely damage the leveling instrument.

The tripod should be placed firmly on the ground and each leg should be pushed in individually, to ensure a stable setup. Setups on bituminous pavement (such as asphalt) must be avoided, since significant tripod settlement may occur. The same precaution applies to frozen ground, where, if possible, the frozen layer should be broken through or shoveled away before setting up the tripod.

Set the tripod to afford as high a line of sight as possible. When leveling along a road shoulder that slopes down away from the road, if the instrument is used at the side (as is the NI 002), point one tripod leg away from the road and stand between the other two legs.

Tribrach The tribrach, which supports the instrument, is another factor in maintaining instrument stability. The tribrach should be mounted on the tripod head so the foot screws are aligned with the tripod legs. Then, the bell-shaped bolt below the tripod head is screwed up through the two plates forming the bottom of the tribrach. It should be just tight enough to prevent the tribrach from shifting when transporting the instrumenttripod assembly. If overtightened, the plate may eventually loosen enough to wobble. When this occurs, tighten the screws that hold the upper plate against the footscrew pedestals. These screws should never be so tight that the downward pressure causes wear on the pedestals.

On many leveling instruments the tribrach is an integral part of the instrument, not removable in the field. Some instruments (e.g., the NI 002) may be detached from the tribrach by loosening a setscrew on the tribrach. The function of this screw should not be confused with that of the tangent screws immediately above it. Check periodically to ensure that the setscrew is tight.

Foot screws. Each of the three foot screws used for leveling the instrument should be adjusted to the tension that the observer finds is practical. First, turn each foot screw until the small hole on the inside is aligned with a similar hole on the outside. Then, insert an adjusting pin and turn the entire screw assembly to loosen or tighten the tension (the direction depends on the particular instrument). One-quarter to one-half turn in either direction is usually sufficient. The tension should not be so great that it is difficult to feel when a stop has been reached. Do not force the foot screws past their upper or lower stops because this damages the threads.



Figure 3-35.—Plane-parallel plate micrometer scale.

Carrying the instrument. After the instrument is mounted on the tripod, carry the entire assembly vertically, resting it against a shoulder and supporting it by gripping the tripod rungs with one or both hands (fig. 3-36). The tripod should not be balanced across the shoulder, since this puts the instrument in a more vulnerable position and subjects it to greater stress from shock or vibration. Use your body to cushion shocks to the instrument. When it is carried on the side step of a truck, rest the tripod feet not on the metal step, but on the toe of a boot or a thick elastic cushion (such as a section of tire) mounted expressly for this purpose.



Figure 3-36.—Carrying the Instrument-tripod assembly.

Leveling the instrument. At every setup the instrument must be roughly leveled. (See sec. 3.3.3, "Compensator system.") A circular level (known also as a spot or bull's eye) is attached to the body of the instrument for this purpose. The bubble in the level is centered to ensure that the compensator (or the bubble in the tubular level of a spirit-level instrument) is freely suspended. To rough-level the instrument, first, turn it so the circular level and one foot screw form a line perpendicular to the line formed by the other two foot screws (fig. 3-37). Turn the two foot screws simultaneously, either toward or away from each other, to effect sideto-side movement of the bubble. Then, turn the first foot screw to move the bubble forward and backward, until it is centered.



Figure 3-37.—Centering the bubble in a circular level.

Many instruments are equipped with mirrors or other optical mechanisms for observing the circular level. These mechanisms may invert or rotate the bubble image. For example, the bubble on the NI 002 may be observed either in a mirror or through the eyepiece. In the mirror image, side-to-side movement is the same as, and forward-backward movement is opposite to, what the observer would expect when looking directly down on the bubble. The relationship of the eyepiece image to the direct view of the bubble is more complex, depending on the orientation of the eyepiece. If the eyepiece is turned to the side of the instrument, side-to-side movement appears forward-backward, and vice versa. With practice, the observer can learn to level the instrument efficiently while observing the bubble through the eyepiece.

Circular-level adjustment. At the start of work each day, and after any severe shock to the instrument, the observer should check that the circular level is properly adjusted. A circle is inscribed on the vial glass to provide a reference for assessing the bubble's movement. When the instrument is sufficiently level, the bubble should remain precisely centered as the instrument is slowly turned through a full circle about the vertical axis. If this is not the case, the level should be adjusted as follows:

1. Turn the instrument until the bubble displacement from center is at a maximum.

2. Using the tribrach foot screws, bring the bubble halfway back to center.

3. Using the three or four small screws supporting the circular level, adjust the bubble until it is centered. Each screw is adjusted by slipping an adjusting pin through the screw hole and turning it so as to maintain downward pressure on the base washer. Do not adjust by depressing one screw and retracting another. If the limit of downward motion is reached on the screw being turned, retract all the screws and start adjusting them again. Do not force the level vial against its base.

4. Rotate the instrument slowly through a full circle. If the center of the bubble does not stay within 0.2 mm (0.01 in) of the center of the level vial, repeat the entire adjustment procedure.

Parallax adjustment. Before observing with the instrument, the observer should check for parallax. Parallax results when the focused image of an object does not lie on the same focal plane as the focused image of the reticle lines. To test for this, point the instrument at the sky or at a distant light-colored surface. Focus the ocular so the lines appear sharp and black, without straining the eye. Then, focus the objective on the scale of a leveling rod about 40 m (130 ft) away, and move the eye up and down, slowly, across the ocular. If the lines appear to move over the rod graduations, parallax is present. To eliminate the parallax, refocus the objective until no parallax is evident. If the image of the graduations is not distinct, focus the ocular by the small amount necessary to make it appear distinct.

Making an observation After setting up and leveling the instrument, the observer should point it quickly at the appropriate leveling rod. With most instruments this is done by sighting along the top of the telescope. Since the observer stands at the side of the NI 002, in this case the system of prisms on the front of the instrument is used to line it up with the rod. Once the instrument is roughly pointed, the rod is observed through the telescope and the tangent screw is turned to align the image of the rod scale with the reticle lines.

The reticle normally includes a single vertical line with three horizontal lines crossing it. The top and bottom stadia lines are used to measure sighting distance. The longer, middle line defines the horizontal line of sight. When leveling with scales having block graduations, a straight line is sufficient. However, with scales having line graduations, the reticle should include a pair of divergent lines, the wedge, marked to one side in lieu of the middle line. The wedge (fig. 3-38) is a more sensitive tool for precisely intercepting a graduation. There are two ways to use it.



Figure 3-38.—Wedge reticle, graduation not centered.

In one method, set the wedge so the intersections of the upper and lower wedge-lines with the upper and lower edges of the graduation appear in vertical alignment (fig. 3-39a). In the second method, set the wedge so the interior edges of the wedge lines appear to just touch the corners of the end of the graduation (fig. 3-39b). Always make the final setting by turning the micrometer knob in the same direction.

In wind or on a vibrating surface, such as a bridge, it may be difficult to intercept a graduation precisely. If the vibration of the instrument is short in period, set the wedge so it appears to vibrate an even amount above and below the graduation. If the vibration is long in period or includes more than one graduation interval, mean the readings from a series of pointings to make each observation. If satisfactory results cannot be obtained, return to the section under better observing conditions.



Figures 3-39a. & 3-39b.—Wedge reticle, graduation centered.

Instrument maintenance. Before putting the instrument away each day gently wipe it with a clean soft cloth to remove dust and moisture. If the instrument has been exposed to rain or mist, it should also be allowed to stand at room temperature overnight. If the instrument is dirty or greasy, or if it has been exposed to salt water, clean all nonglass surfaces with a cloth dampened in denatured alcohol. Dust the objective and ocular lightly with a lens brush. Then, if necessary, wipe the lenses with lens paper that has been moistened with a small amount of lens cleaning solution. Care must be taken not to scratch the glass surfaces.

The instrument, with the lens cap attached, should be stored securely in a specially designed padded case. To protect the instrument from jolts and vibrations while the truck is in transit, the instrument should either be (1) carried in a foam-lined box bolted to the truck bed, (2) stored in its case and held on the lap of a passenger in the truck, or (3) strapped onto an empty seat in the truck. If the instrument is to be shipped to another location, it should be packed in its case and placed inside a well-padded carton, with an appropriate warning label affixed on the outside of the package.

At least once every 18 months, the instrument should be cleaned and adjusted on a collimator by a qualified technician.

Shipment of the NI 002. The National Geodetic Survey has found the objective assembly of the NI 002 to be particularly vulnerable to vibration during shipment Damage to the stop at the base of the assembly can be recognized by the inability of the user to read a full scale on the micrometer when the instrument is leveled. If the stop has been damaged, the instrument should not be used until it is repaired.

To prevent damage to the stop, screw a small, hardrubber plug into the instrument to limit movement of the assembly during shipment (fig. 3-21). The plug should be screwed in whenever the NI 002 is transported for long distances in a leveling truck or shipped via commercial carrier.

# **3.3.6 Sighting Distance**

The sighting distance between the instrument and a leveling rod is normally computed by the stadia method. To use the method, a full or half stadia interval must be computed from stadia readings made while observing. To balance the sighting distances of awkward setups quickly, before any observations are recorded, the stadia interval may also be determined by counting the number of rod units observed between the stadia lines (full interval) or between the middle line and one stadia line (half interval).

Derivation of the stadia method. By the stadia method, short distances may be measured with a precision of  $\pm 0.2$  m ( $\pm 0.7$  ft). The stadia method has been widely used for topographic surveys and is routinely used in geodetic leveling to measure sighting distances and control the imbalance between backsights and foresights.

Figure 3-40 illustrates the relationships which exist between the leveling rod, the objective, and the reticle when a stadia measurement is made with a leveled instrument. For the focused instrument, the distance,  $D_i$ , from the objective to the reticle, and the distance,  $D_o$ , from the objective to the rod, are governed by the Law of Conjugate Foci, where the focal length, f, of the objective, f, is a constant:

$$(1/D_o) + (1/D_i) = 1/f_o$$

The sighting distance, s, is equal to  $D_o$  plus the fixed distance, c, between the objective and the vertical axis of the instrument. By solving for  $D_o$  and adding c:

$$s=f\times(D_o/D_i)+f+c$$



Figure 3-40.—Relationship of stadia interval to sighting distance.

To a sufficient approximation, the line connecting points a and a' is a straight line through point o; therefore, triangles *abo* and a'b'o are similar. Thus, all corresponding dimensions are in the ration  $D_o:D_i$ , including the full stadia interval, I, and the distance, i, between the upper and lower stadia lines in the reticle:

$$D_o/D_i = I/i$$
.

By substituting the known ratio for the unknown:

 $s = (f/i) \times I + f + c$ 

The ration, f:i, is unique for the instrument and is called the stadia factor ("multiplication constant"). The constant, f+c, is also unique for the instrument and is called the stadia constant ("addition constant"). In most instruments with internal-focusing telescopes, the optical system is so designed that the stadia constant is essentially zero for the sighting distances used in leveling. Table 3-2 shows stadia factors and constants for some typical instruments.

Computing sight distance. From the full stadai interval, I, the sighting distance, s, is computed by the formula just derived:

 $s = (I \times \text{stadia factor}) + \text{stadia constant},$ 

For practical use in the field, the stadia constant is usually neglected. However, it should not be neglected when determining the stadia factor. During these operations sighting distance must be determined as accurately as possible. However, during routine leveling the error resulting from neglecting the stadia constant is insignificant when compared to the overall error of the stadia method.

To obtain the sighting distance in meters, a conversion factor must be included in the formula. Since the stadia factor is a unitless quantity and the stadia interval is measured in rod units (ru), the conversion factor required is the number of meters per rod unit. Thus:

$$s_m = I_{ru} \times \text{conversion factor}_{m/ru} \times \text{stadia factor}$$
.

In the past, the product of the stadia factor and the rod-unit conversion factor has been referred to as the "stadia constant." Do not confuse such usage of the term with the definition given here.

This formula applies in three-wire leveling and whenever the full interval between the stadia lines is computed. For example, if the instrument has a stadia factor of 100 and the rod units are centimeters,

$$s_m = I_{cm} \times 0.01_{m/cm} \times 100 = I_c$$

Thus a convenient relationship is derived: the number of rod units in the full stadia interval equals the sighting distance in meters.

ţ

However, if the rod units are in half-centimeters (hcm), the formula becomes

$$s_m \times I_{hcm} \times 0.005_{m/hcm} \times 100 = I/2$$

In micrometer leveling, this result is employed to reduce the number of readings necessary. Instead of reading the intercepts of all three lines, only those of the wedge and one stadia line (usually the lower) are read. The difference of the readings is half of the stadia interval. As long as the rod units are half-centimeters, the difference of the readings is equal to the sighting distance. When using the micrometer-leveling procedure with centimeter rods, the half stadia interval must be multiplied by two to obtain the correct sighting distance.

Determining the stadia factor. The stadia factor is set by the instrument manufacturer for converting the stadia-interval to sighting distance. In modern instruments used for geodetic leveling the stadia factor is typically 100. The reticle is etched in glass to ensure that the factor does not change. The stadia factor need not be determined in the field when using such an instrument.

Some older instruments may have a reticle consisting of fine strands of spider web, sandwiched between glass plates. The stadia factor for this type of instrument should be determined in the field whenever it is put to use and whenever the reticle is changed. To determine the stadia factor:

1. Lay out a course in a straight line along a level track, roadway, or sidewalk, marking the distances at 0, 25, 35, 45, 55, 65, and 75 m.

2. Position the instrument to allow for its stadia constant. For example, the Wild N3 has a stadia constant of -0.2 m; therefore, it should be set forward of the zero stake by 0.2 m. An instrument with a positive stadia constant should be set back from the zero stake by the corresponding amount.

3. Level the instrument. Read the low scale of a leveling rod at each of the six distances. Read all three reticle lines, top to bottom, at each distance.

4. Record the readings and identifying information as shown in figures 3-41 and 3-42, using the recording form appropriate to the type of leveling to be done. Compute each half-stadia interval.

5. Total the half-stadia intervals (in rod units). Total the measured distances (in meters) and divide by the total of the half-stadia intervals. The value obtained is the stadia factor, expressed in meters per rod unit, for the instrument-rod combination used

6. The result of step 5 should be checked by computing a similar value for each distance. Divide each measured distance by the sum of the two half-stadia intervals at that distance. The mean of these six values should agree with the value computed in step 5.

7. To obtain the stadia factor of the instrument, independent of rod units, the value expressed in meters per rod unit must be multiplied by a conversion factor. This factor is the number of rod units per meter.

8. Submit data for the stadia factor measurement to the project office, along with other leveling data for the day.

## 3.3.7 Collimation Check

Under field conditions the collimation error of a leveling instrument (sec. 3.1.2, "Leveling instrument") is measured by obtaining a set of observations called the collimation check ("C-shot" or "peg test"). Collimation error is limited by adjustment of the instrument. Because the adjustment can change easily under field conditions, thus changing the collimation error, make the collimation check at least once a day with most instruments. In addition, make the check any time that an instrument sustains a severe shock or seems to function abnormally.

The collimation check has two purposes: to prove that the instrument is properly adjusted within the standard of accuracy required for the survey and to provide a collimation factor with which to correct data from unbalanced setups.

Collimation factor. During the collimation check the angular value of collimation error is not measured directly Instead, the tangent of the angle, called the collimation factor (C-factor or C) is computed. A derivation of the formula follows.

Two elevation differences are observed in the same direction, but with different setups, between the same two turning points. Each setup has an imbalance of sighting distances. Since the elevation differences measured between the same two points should be equal, any difference between them is the result of the effects of collimation error, pointing error, refraction, and curvature.

The collimation error,  $\alpha$ , causes an error that is proportional to the imbalance,  $\Delta s$ , in each setup:

$$\tan \alpha \times \Delta s = C \times \Delta s.$$

The pointing error is limited by a reading check The sum, e, of refraction and curvature error is mathematically predicted for the appropriate sighting distances in each setup. Hence, if each observed elevation difference,  $\Delta h$ , is corrected for these errors,

 $\Delta h_2 - (C \times \Delta s_2) - e_2 = \Delta h_1 - (C \times \Delta s_1) - e_1.$ 

Solving for the collimation factor,

$$C = [(\Delta h_1 - e_1) - (\Delta h_2 - e_2)] / (\Delta s_1 - \Delta s_2).$$

If the first setup is balanced and the second unbalanced (as for "Kukkamaki's method," given later in this subchapter,  $\Delta s_1 = 0$  and  $e_1 = 0$ ,

$$C = \left[\Delta h_1 - (\Delta h_2 - e_2)\right] / (-\Delta s_2).$$

If both setups are unbalanced by the same amount, and they are leveled in opposite directions (as for the "10-40 method," given later in this subchapter), the corrected elevation differences are opposite in sign. Thus,

$$C = [(\Delta h_1 - e_1) + (\Delta h_2 - e_2)] / (\Delta s_1 - \Delta s_2)]$$

Since  $\Delta s_1 = \Delta s_2$  and  $e_1 \simeq e_2$ , the formula is simplified to the following:

$$C = \left[ \left( \Delta h_1 + \Delta h_2 \right) - 2e \right] / 2\Delta s.$$

The elevation differences, the refraction and curvature errors, and the imbalances are measured in the same units; the resulting collimation factor is dimensionless. However, for convenience, it is usually expressed in millimeters per meter:

$$C_{\rm mm/m} = \tan \alpha \times 1000_{\rm mm/m}$$

For an instrument with a reversible compensator, a separate collimation factor may be computed for each position of the compensator. The collimation factor for the instrument is the mean,

$$C = (C_1 + C_2)/2$$

If the collimation factor is not within the tolerance for the survey, the instrument must be adjusted. When the adjustment can be made in the field, the amount of

(

(

(

HDAA FORM 76-171 (8-77)	,AT IOH	U.S. DEPARTMEN L DEEANIC AND ATMOSPHERIC	ADMINISTRATION
	GEODETIC LEVELING MICROMETER OBSERVATIONS (4h)		//
I A D I YE /HO DAY CODE HUSTOURENT SECOND		STADIA E	EACTOR 11 -
· · · · 800728217 60813	34316 277890	EXAMPL	E
FRON AN DESIGNATION		TINE 764	ELATURE
	Roi	0108300 27	1 27502 645
SET STADIA Se STADIA SF UPP STADIA I STADIA	LOW SCALE AL. HIGH SCALE BACK/FORE E BACK/FORE	Даны Дац. Б. Дан	řemarki
254010 2503760 2501	B		25 = 0.500
3760 3510	# <u>      ! ] ] ] ] ]        </u> # ] <u> </u> # ] <u> </u>		50.0
354/11/11 135/13760 13501	a thill the transformed as a second sec		35 - 0 400
3760 6013410 6001			70.1
		<u> </u>	de
3750 10513301 1049		┨ <del>┝┝┝┇┥┥┫</del> ┥╋	$\frac{75}{899} = 0.501$
			<b>~</b>
554310 5503760 5500	B		55 = 0.500
3760 760 73210 7599			110.0
654400 6503750 6501	a		65 - 0 500
3750 22513100 2249+	P		130.0
			75
3760 30013010 2999	╩ <mark>┍╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷╷</mark>	╋┥┽┽┊╞┝┠┊┼╸	150.0 = 0.500
300 m \$600.0 h	hcm 300/600.0 = 0.500	****	MEAN 0.500
┝┵╉╌╡╌╡╴╎╴╎╴╎╴╎╴╎╴┥╌┥╸┥╸┥╴┥╸	<u>●</u> <u>↓</u> ↓ ↓ ↓ ↓ ↓ ↓ ↓ ↓ ↓ ↓ ↓ ↓ ↓		
┢┉┉┫┙ <u>┥╶╡╴╡╶┨╶┨╶┇╶┫</u> ┈┫╺┟╸┊╴┨╶┨╶┨╶╿╶┨ <sup>╒</sup>		HECK	a complex waster 1
	8 1 2 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4		U. JOOM ACAL ACONCAYA
			= 100
<u>╞╶╴┫┤┼┆╫╗┝╎┊</u> ┨┼┤┆┨ <mark>╞</mark>	<u>¯</u> <u></u>		
┠╍┶╼╊╍╊╍╊╍╊╍╋╍╋╦╋╋╌╋╼╋╼╋╝	╊ <mark>╆┝┥┥┧┊╎╎╏╎╏</mark> ┥┥┥┙╵	╅┼╍┼╌┼╍┼╌╢╍	
<u> </u>	┿╋╍╉╍╉┉╊┉╧ <u>╸┠╍┠┉╋┉┨╸┨╼╂┉╧╸┨╶</u> ┨ <u>╴╂┸┨<u>╴┨</u>╴┨<u>╴┨</u></u>	┺╍┶╍┶╍┶╍┶╍┶╍┶╍┶╍┶	

Figure 3-41.—Stadia factor determination, micrometer procedure, half-centimeter rods.

HOA# 14-77	k. # 3 mił Zśw 187 73							NATIONAL OCE	U. S. DEPAR	THENT OF CONNERCE	PADE
				THR	GEODETIC L EE-WIRE OB	EVELII SERVA	NG TIONS				OF
* 4	C + YA. 140. 10	AY CODE INSTAUN	ENT SEPIAL NO.	CODE	BOD SERIAL	нο.	CODE	NOD SERIAL HO.	STADIA	FACTOR	2 TINE
• 4	1 . 80072	82/7	30150	3/2		305			EXA	MPLE	
		(D)	T i		HDENCHATION			T + L	7146 6	TEMPERATURE	1. 00-
			IT.	ĔΤĹ				R084	5091140	280 2850	2 CWS
	*****	BACKSIG	нт					FQI	ESIGHT		COMMENTS
	UPPER		[	ſ	[	T	UPPER		T. T		1
SET HP	MIDDLE	NEAN	BACK GENTER	\$ <sub>U</sub>		,	4100LE	MEAN	BACK CENTER	5 <sub>U</sub>	
	LOWER			\$L	(U+L)		LOWER			51. {U+L}	1
	Σ	Σ	<u> </u>	ļ,,	Σ		Σ	Σ	- <b> </b>	Σ	
						ЦЦ			╻╻╷╷╷╷╷╷		
	//35/7						444			╏┯┯┯┫	
212	/ 30.0			3:7					┨ <sub>┛</sub> ╧┙	┠┼┊┟┬╌┌┈┲┉┰┉	4
	/262		1	3:8	75	<b></b>		┨╌┎╌┲╌┲╼┲	25 . 3 33	┝┙╧┠╌┼╌┼╴┿╴	-
			<b></b>			-+-+	┽┼┼┼	╶┟╌╏┈┟┉┟┈ᢥ┈╵	17.5	┟───┝─┟─└─┴	-
372	(35.7		<u></u>	177			╶┼╌┼╌┼╴┼	┨┰┰┱┯┯			
20	130.7	┟┉┹┉┹┉┹╌╹╌┹╌	f_L_L_L_i	5:2	TIME			╉┸╌┸╌╩╼┹	╶┧╤╧╵┥┙┈╸	┨┈┨┈╅┈┠┈┱╌┰╴┰╴	-
					10.5		┼┼┼┾	+	- 3.33	╞─┶╘╍┟╍┼╍┼╶┼	-
	1977	┟╴╽╶╷╷╶╸╸			-1-1/10.0		+++	┨╴┨╴┇╼╘╍╘╸╅		┋┉┈┉┠┉┖┈┶┉╅┉	1
45	1/3/0			20			┥┧┼┼	+			
	1242			68	। । त्रवास		++++	┉╁╍┶╾┶	45	╊ <del>┝┇┣┲┲┲┍</del>	1
			1		375		<u>+++</u>		13.5 = 3 33		1
	/32.5					T					1
55	1243			8.2							]
	1160			813	165	11			55		
					480				16.5 3.33		]
	/26.2										
65	1/16.5			9.7						<b>││┊│</b> <sub>┲─┲─┲─</sub>	1
	106.8	·····		917	19.4			<u> </u>	1 = 3 35	┠┖┊╂┤╡┞┊	1
					674				19.4	ļ	334 %
	/2/8						444	$\frac{1}{1}$		mequ= 3.34	× 100 04
715	1/06	┟┈┟┈┟┈┟┈┋┈╏┈		//:2		<b>-</b>		<u>↓↓↓↓↓</u> ↓			1
1200	099.4		300 = 3 34	1/12			┿┥┿	-1	774 = 3 35	<u>⊢}×f+++</u> +	334
5001	al and the second s	└┈ <del>╎╓╻<u>┟</u>╸<u>┟╻</u>┟╍╬╍╽╸</del>	l	L		cr71			A HEAR	ونصاصا صلحا ساحم	

Figure 3-42.—Stadia factor determination, three-wire procedure, centimeter rods.

error to be removed is computed from the following formula. It is referred to the rod at the farthest sighting distance,  $s_{F}$ , and computed in rod units:

# $\operatorname{error}_F = s_F \times C_{mm/m} \times \operatorname{conversion factor}_{ru/mm}$

The error is subtracted from the last reading made on the far rod. Then the instrument is adjusted in such a way that the corrected reading is observed. After performing the adjustment, the entire procedure is repeated to compute and check the new collimation factor.

Note that the collimation-correction factor, "C", commonly used to correct data from three-wire leveling, is not equal to the collimation factor defined here. In the past, "C" has been defined as the product of the stadia factor and the tangent of the collimation error. The imbalance—expressed in rod units as a stadia interval, not as a distance—could be multiplied directly by "C" to obtain the correction for collimation error in rod units. Modern instruments and rods make this practice unnecessary, since they usually provide a one-toone correspondence between stadia interval and distance.

General instructions. Two methods, sufficiently precise to satisfy the purposes of the collimation check, are presented here. The accuracy of the result with either of these methods depends on the assumption that the error observed is entirely a function of collimation error. Other effects that alter the line of sight, such as refraction, must be controlled. To achieve this, perform the following:

1. Make the collimation check on uniformly flat ground. The slope should be no more than 2 percent.

2. Allow the instrument and leveling rods to acclimatize for 5 minutes or more, if they have just been removed from their cases.

3. Make sure that the circular levels on the instrument and the rods are properly adjusted. (See secs. 3.3.5, "Circular-level adjustment" and 3.4.3, "Checking the circular level.")

4. Correct for refraction and curvature at each sighting distance. To use the values for refraction and curvature given in table 3-3, the collimation check should be made, if possible, when the air-temperature differential is less than  $1^{\circ}C$  ( $2^{\circ}F$ ) in magnitude and negative in sign (sec. 3.6.1). However, make a check immediately following an accident to determine if the instrument has been damaged, regardless of the atmospheric conditions. Then, make another check as soon as the conditions are suitable.

Kukkamaki's method. T. J. Kukkamaki of the Finnish Geodetic Institute developed this method for making a collimation check. The following instructions apply to any combination of instrument and rods. If the instrument does not have a reversible compensator, ignore references to compensator's position. Record the data either on a standard recording form for leveling or into a computer.

Table 3-3.—	Refra	action	and	curvature	errors
	in a	singl	e sig	ht	

	s	ightin	g distance,	s		Err	or in a	rod reading, e
	(m)			(ft)			(mm	) (ft)
0	to	28	0	to	92		0.0	0.000
28		48	92		157		0.1	0.000
48		61	157		200		0 2	0 001
61		73	200		240		03	0 001
73		82	240		269		0.4	0.001
82		91	269		299		0.5	0 002
91		99	299		325		06	0 002
99		106	325		348		07	0 002
106		113	348		371		08	0.003
113		119	371		390		0.9	0 003
119		125	390		410		10	0 003
125		131	410		430		11	0.004
131		137	430		449		12	0 004
137		142	449		466		13	0.004
142		147	466		482		14	0.005
147		150	482		492		1.5	0 005
	160	ŧ		525			1.8	0 006
	170	)		558			21	0 007
	180	)		591			23	0 008
	190	1		623			26	0.009
	200			656			2.8	0.009
	210	1		689			3.0	0.010
	220	1		722			3.3	0 01 1
	230	)		755			3.7	0 012
	240	)		787			40	0.013
	250	)		820			43	0 014
	260	)		853			47	0.015
	270	)		886			5.0	0 016
	280	)		919			54	0 018
	290	)		951			58	0.019
	300	)		984			62	0 020

1. On the flattest possible ground, lay out a setup with precisely 20 m between the turning points. Position the instrument in line between them, precisely 10 m from each turning point. (Use a tape to measure these distances.) (See fig. 3-43.)

2 As indicated on the recording form (line \*40\*), enter the date, instrument code and serial number, rod codes and serial numbers, time and time zone, type of temperature units, beginning air temperature, wind and sun codes, and the initials of the observer, recorder, and rodmen. Check all serial numbers against the equipment actually used. Label the recording form "Collimation Check." (See figs. 3-44 through 3-46 for examples.)

3. FIRST SETUP: Level the instrument. Check that the circular levels on the instrument and the rods are properly adjusted. Observe and record a set of readings by either the micrometer or three-wire procedure. Use rod 1 as the backsight, and be sure to change the compensator's positions between the low- and high-scale readings. Check that the imbalance is no more than  $\pm 0.4$  m. The reading check for the micrometer procedure should be  $\pm 0.25$  mm ( $\pm 0.05$  hcm for half-centimeter rods).







# Second setup

Figure 3-43.—Collimation check, Kukkamaki's method.

4. SECOND SETUP: Position the instrument in line with the turning points, 20 m from rod 1. Put the compensator in position one. With rod 1 as the backsight, observe and record another set of readings; however, leave the compensator in position one throughout the setup. Check that the magnitude of the imbalance is between 19.6 and 20.4 m. Enter the ending temperature. Remain in position until the collimation factor has been checked. 5. Convert the elevation differences,  $\Delta h_2$  and  $\Delta h_1$ , from rod units to millimeters. Compute or obtain from table 3-3 the values, *e*, for refraction and curvature error at the sighting distances of the second setup, 20 and 40 m. Compute *C*, in millimeters per meter, by the formula:

$$C = [(\Delta h_1 - \Delta h_2 + (e_{20} - e_{40})] / - \Delta s_2$$

If the instrument has a reversible compensator, label the value " $C_i$ " and skip to step 11.

6. Compare C with the tolerance (table 3-1). If the tolerance is exceeded, then adjust the instrument as follows.

7. ADJUSTMENT: Compute the error, in rod units, resulting from collimation error in the reading made at 40 m:

 $\operatorname{error}_{ru} = 40 \times C_{mm/mm} \times \operatorname{conversion factor}_{ru/mm}$ 

For half-centimeter rods, the error is  $8 \times C$ . For centimeter rods, it is  $4 \times C$ . Subtract this from the foresight reading, obtained on rod 2 (high scale) during the second setup. The result is the correct reading after adjusting the instrument.

8. Refer to the instrument manual for the mechanics of adjusting the instrument. Many compensator instruments have a setscrew on the front of the telescope that is loosened, after which a wedge-shaped lens is rotated to correct the angle of the line of sight.

9. Still in position for the second setup, point toward rod 2 (high scale) and adjust the instrument until the line of sight intercepts the corrected reading.

10. Return to the first setup (step 3) and repeat the entire collimation check to compute and check the new collimation factor.

11. THIRD SETUP (reversible compensator only): Put the compensator in position two and follow the same procedure as for the second setup (step 4). Compute  $C_2$ by the same formula as  $C_1$ .

13 Compute the mean collimation factor and record it:

$$C = (C_1 + C_2)/2$$

Check this against the tolerance for instruments with reversible compensators (table 3-1).

14. Compute the *Q*-factor and record it:

 $Q = C_1 - C_2$ .

Check this against the tolerance (table 3-1).

15. If the tolerances in step 13 is exceeded, the collimation check should be repeated to verify the original observations. If the tolerances are still exceeded, return the instrument to the repair shop or factory for adjustment.

10-40 method. Another procedure for the collimation check is called the 10-40 method because each setup is unbalanced, with one rod positioned 10 m away from the instrument and the other 40 m away. The following instructions apply to any combination of instrument and rods. Record the data either on a standard recording form for leveling, or into a computer.

N044 U 77	FOR	¥ 74-	191																_	-						•	EATI		. DC1	ZA HI	C AN	U.S. 0 AT	DE	PARTI	ENT D	FEDNE	RCE	PAGE	2
													M	ICR	) MO	ET I	DE ER	TI( OB		EV V/	ELI ATIO	NG DNS	{ <b>4</b> h	)														1	OF
Fi	٥].	1 70.	/wc	. þ.	17	C00	ε		UNK	нт	sen:		a] #	6	DE	T	Rot	) <b>62</b>	RIA	. +c	).	c	ODE	T	7.00	sen	IAL	нo.	Т	C	04	IM	A	TIOI	TO	IECK	Ť	1	TINC
• •	<u>, I</u> .	80	11	oli	5	23	3	1	[4]	5 4	5	11	Н	3	16			14	15	6	82	3	16		П	4	56	8	4			E	X/	1 M T	LE			T	
FR		<b>*</b>	22.830	NAT	OH							Τ	1	0	ŀ	<b>H</b> DI	10	NA7	10N	_					··		z		T) AH	HE 1		2	Æ	- 11 - 11-	HPER,	TURC		5	SERVER
μυ		K	KK/	₹MA	KI I	HET	HO.	D				1	1		L												R	0	110	1	03	0	C,	12	01	10.	10	2]	CWS
SET UPP	317/ 11/	LDIA ICK		i: I			TAD	iA E		1	-			LOW	1CA	LE ORE				۵	μ. Σ				HIGH	SCA	L E RE			۵	ћны С	•		∆ы∟. Данн			RENAR	×s	
1/	28	7	Π		9!4	2	6	7!	Π	Ι	9	61		Z	8 7	8	2			[]	8.1	3	в		8	0	43		-		29	0		+0	3 <i>Δ/</i>	, <i>≠</i>		T,	= <i>I</i> O.1
	217	7.4	Ш			2	5	7!4	Ц	1		1		z	69	6	9		Ι.				ŀ	_	82	3	33	5	4	Ļ	Ļ				_++/	ġ. 115		72	= 10.4
101	215	9	П	17	41-	72	4	0	гf	15,	# 0 10	41	^	17	59	77	9	Ť		7	812	5	1.	T	2		219		<u>a =</u>	12	1	74	T	-10	101	-		7	= 10.2
	2 5	913		ľ	ĹŢ,	2	0	2.4	Ħ	*	Ť	7		2	10	19	4	-	-	Η	Ť	Ť	F		8	4	44			Π	T			Ť	-+1	8.24	5	7	10 6
		<del></del>	<del>1 1</del>			<u>+</u>	58	5	-4	19.		9.9	, wi	, T T		<u> </u>				<b></b>	·	7-1	1	<b>T</b>	<u> </u>	- <u>-</u> r		т ¬	4	-2	4	1-3		- · -	- "	7.130	hen	7	
┝─┴┨	+	┼┼	+		H	╋		÷	╂╍╂	+	+	╡		$\square$	╇	╣	-	+-	+	$\mathbb{H}$	÷	╉┥	8	-	╞┼┥	H	+	+		+	┝┿		-		×	1.50		hem A	Lank
┝───┖	<b>-</b>		1_1		1*		1.1	<u>.</u>	11	l	- <u>1</u>	<u>I'</u>		1 6	-	• . :			-1	L	<b>`</b> -	<u>ل</u> ــــــــــــــــــــــــــــــــــــ	Ľ.	4	<b>I</b> I.			1			•	السل			ŢΓč	.200	,	<u> </u>	1 11 Z
	Т	Π	Π					Ţ	Π		Π	B			Ι	!	Ι	Τ	Π		Ţ	Π	в		Ш	TI	Τ	TΠ				Π	_	Ū.	-0	850	mm	- e	æ-⊄#0
	1							Į			]	1	1		1	!							ŀ								1			<u>.</u> F	÷	19.9	7 m	- 7	∆s <sub>2</sub>
r		<b>-</b>						· -	T				<b>1</b>	r	$\mathbf{T}$	7 1				<b>T</b>	-r	1.1	-	<b>T</b>	гт			<b>T</b> 1	<u>-</u>	T1		11	-	<u>- C</u>	티드	0.043	mm/	<u></u>	
	+		┢┥		l-i-	+		÷	┡╌┥		┥┥		-		╀	$\frac{1}{1}$	-		-		<del> </del>	-		╀╌	┝╌┼╍	+	-			+		++							
I	£	11.	11		<u> </u>	-11	<u>+</u> -		¥1	l.,	- <b>J</b> š	_ <u>_</u> :	.£	<u>, 1</u>	-	<u>•</u> -				I	<b>-</b> -		I	<u>.</u>	<u> </u>	1 .		<u> </u>	Ē		L	.i1	Ł	<u>.</u>	-				
3	25	9	П	1	9 7	12	4	1	П		19!	61		2	59	4	3	Т	Τ	7	7!9	1	8	Γ	8	17	93		T	T	3!0	8	T	-!0	T AA	a <b>"</b>		7	# 10.1
••	23	9.3	Π			2	01	1.4	Π			۶		2	4 I	5	2	L	L		<u> </u>		F	E	8	5	01	1	Γ		Ţ	П	_[	1	- +/	7.915	•	_ 7	2 10.6
		<b>7</b>	1 <sup></sup> T		<del></del>	F.	Sq.		-4	1	<u>* /'</u>	<u>, 9</u>	~~			· · · ·	-r					·····		·····	r				+	2	4				_++ {	0.200	hem		
<b>-</b> -		┢╋	╉┥┥		H	+	-+	- <del>i</del>	┡╌┥	╉	+	-	-	┝╍┼	+-	÷	┿	+	+	H	-i-				┝━╋╸	++	╞	Ļ		┢	⊢ <del>i</del> −				×	5	mm/h	œn į	
<b> </b> I	<b>i</b>	I	1		<u> </u>	1		•	<b>E</b>		1:		1.	ŀ ŀ	_L_	<u>.</u>	L	1		<u> </u>			1.		i.,		l	1	1	ł	;			اا	1.	2000	/hm	/۵ م	η -Δn <sub>3</sub>
П	1	ТТ	TΤ	Т	T	Л	T	T	ГΤ	T	П	1		ГТ	T		Т	Т	TT	Т	Ţ	TΠ	1.	Π	T-		Т	ТТ	-T-	TT	T	ТТ	T	T	Ŧ	800	AT A1	•3	1040
	1	TT-	Ħ	T	Ţ		1	ļ	M	T	11	P			T			T	Π		Ţ		F	Í			T				Ţ		Ť		-	19.9	m	-4	153
		· · · ·									-,,		· · · ·			<i>,</i> ,						·····			-									C	- +0	,040	mm/n	<u>.</u>	-
-	+	1	$\square$	_	┝┼	╇		1	╟╢	-	╇	- 10	$\square$	H	╇	$\left  \right $			+	-		⊢	8	+	┣-╂-	+	+	$\left  \right $	-	$\mathbb{H}$		┉	-		-1	-0.04	+0.04	ō	0000
<b>I</b>	1	L !	Ш	<u> </u>	. :			<u> </u>	L			11	1	i	l	: 1		I.	L			11		<b></b>	ш.	1 '	L	1		1_1				:	۲	4	-		1000 C
Π	Т	П	П	1	L İ.	Π	T	Τ	ΓT		TI		Π	Π	Т	Π	Т	T	Π	П	1	Π	в	Π	T	TI	T	TT	Т	Π	Ţ	TT	Т	<u> </u>	−lφ=	0.043	-0.04	<u>e</u> -	0.017
	T	<u>L</u>	П		ļ	Т		1			11	F	L		Τ		T	Τ	Ι		ļ	П	F			1		Π	1	$\square$	Ţ	П	1	Ţ	<u> </u>	5			hc=/m

Figure 3-44.—Collimation check, Kukkamaki's method, micrometer procedure, half-centimeter rods, reversible compensator.



Figure 3-45.—Collimation check, Kukkamaki's method, micrometer procedure, centimeter rods, nonreversible compensator.

1. On the flattest possible ground, lay out a setup with precisely 50 m between the turning points. Position the instrument in line between them, precisely 10 m, as measured with a tape, from the turning point supporting rod 1. (See fig. 3-47.)

2. As indicated on the recording form (line \*40\*) enter the date, instrument code and serial number, rod codes and serial numbers, time and time zone, type of temperature units, beginning temperature, wind and sun codes, and the initials of the observer, recorder, and rodmen. Check all serial numbers against the equipment actually used. Label the form "Collimation Check." (See figs. 3-48 through 3-50 for examples.)

3. FIRST SETUP: Level the instrument. Check that the circular levels on the instrument and the rods are properly adjusted. Make a set of readings by the micrometer or three-wire procedure. Use rod 1 as the backsight. Check that the magnitude of the imbalance is between 29.6 and 30.4 m. The reading check for the micrometer procedure should be  $\pm 0.25$  mm ( $\pm 0.05$  hcm for half-centimeter rods). Do not make the reading check if using a reversible compensator.

4. SECOND SETUP: Position the instrument in line between the turning points, precisely 10 m from rod 2. Make a set of readings as in step 3, using rod 2 as the backsight. Check the imbalance as in step 3. If using an instrument with a reversible compensator, check that

$$(\Delta h_L - \Delta h_H)_1 + (\Delta h_L - \Delta h_H)_2 \le \pm 0.35 \text{ mm}.$$

The tolerance corresponds to  $\pm 0.07$  hcm for halfcentimeter rods. (If this reading check is not satisfied, begin again at step 1.) Remain in position until the collimation factor has been checked.

5. Convert the elevation differences,  $\Delta h_2$  and  $\Delta h_1$ , from rod units to millimeters. Compute or obtain from table 3-3 the values, e, for refraction and curvature at 10 and 40 m. Compute the collimation factor, in millimeters per meter, by the following formula. Note that the imbalances are both negative.

$$C = \left[ (\Delta h_1 + \Delta h_2) - 2(e_{10} - e_{40}) \right] / (\Delta s_1 + \Delta s_2)$$

6. Check the collimation factor against the tolerance (table 3-1). If the tolerance is exceeded, then adjust the instrument as described in steps 7 through 10 of Kukkamaki's method.

7. If the tolerances in step 6 are exceeded, the collimation check should be repeated to verify the original observations. If the tolerances are still exceeded, return the instrument to the repair shop or factory for adjustment.

#### 3.3.8 Compensation Check

When a compensator instrument is roughly leveled, the compensator should be freely suspended, unaffected by its suspension and damping mechanisms. The range

NOAA FONM 74-187 - 18-375	U. S. DEPARTMENT OF COMMERC National deeanic and atmospheric administratio	PAGE
		1 DF 2
	REE-WIKE OBSERVATIONS	1.1. v. 4.
• 4 1 • 8 0 / 0 / 6 2 3 2 30 9 / 0 3 / 2	2 1063332312 1006334 EXAMPLE	
TO TO KUKK AMAKI METHOD	BIN DESIGNATION         Image: state of the state o	1/ Cws
BACKSIGHT	FORESIGNT	COMMENTS
SET MIDDLE MEAN BACK CENTER SU	UPPER UACK CENTER SU	
<u>ιανεη</u> <u>Σ</u> Σ	(υ+L)         LOWER         5L         //υ+L)           Σ         Σ         Σ         Σ         Σ	
1/ 1/546 //5460 / 54		0 A5 =
		0.0m
Z / / 40.0 / / 40.03 / / / 0.0	0 1/44:0 1/44:00 1/14:00	
$\frac{1}{2} \frac{1}{30!1} \frac{1}{14!57} B_1 - B_2$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	0 - 0 - 52= 0 20.1 m
X ( 0. mm/cm 1 1   -1/1310 mm Δh, -Δh.		
$+ -0.20$ mm $e_{20} - e_{40}$		
$ = \frac{1}{2} \frac$	┝┰╤┲╤╧┥┢┼┼┾┼┼┽┼╌╌╌╌╌┥╴╴┶┶╧┼┼┼┼	
· C - · · · · · · · · · · · · · · · · ·	<mark>╱<mark>╡┙┙╧┥</mark>╽<del>┝┥╞╎┊╎┊╽╝╧┶╵╧╵┨</del> ┑</mark>	
		4
		r
	U, 3. DEPARTMENT OF COMMERC	PAGE
NOAA 70nu 74-189 (4-71)	U. S. DEPARTMENT OF COMMERC NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATIC	PAGE
ND&A FDHW 76-189 14-771	U. S. DEPARTMENT OF COMMERC NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATIO GEODETIC LEVELING REE-WIRE GBSERVATIONS	PAGE 2 OF 2
НОАА ГОЛЫ 74-189 (1-71) • a 0. • YR. (40. (БАХ) СОВЕ МІХЛИЧЕНТ БЕЛІАТ НО СОВЕ	GEODETIC LEVELING REE-WIRE GBSERVATIONS RODSEMIAL ND. CODE RODSEMIAL ND. COLLINA ATION CHECK	PAGE <u>2 OF 2</u> <u>1</u> TINC
HOAA FORM 74-187 HI-TTI • A O • YA. AO. /DAY CODE HISTROUGHT SEMIAL NO. CODE • A I • 8 80//0/16 23 2 3091/0 31/2 FROM RM DESIGNATION TO 1	GEODETIC LEVELING REE-WIRE GBSERVATIONS ROD SERIAL NO. CODE ROD SERIAL NO. COLLI M ATION CHECK 2 1/063323/23/21066334 ROD SERIAL NO. CONTINUED	2 OF 2
HOAA FORM 74-187     THE       * a 0 * YR. /HO. /DAY CODE HISTRUMENT SERIAL NO.     CODE       * a 1 * 8 0/10/16 23 2     309110       * FROM     BW DESIGNATION       * A DESIGNATION     TO       * A After adjust ment:     TO	U. S. DEPARTMENT OF CONVERCINATION           INATIONAL OCEANIC AND ATMOSPHENIC ADMINISTRATIO           GEODETIC LEVELING           REE - WIRE GBSERVATIONS           ROD SEMIAL NO.         COLLI M ATION CHECK           IND SEMIAL NO.         COLLI M ATION           IND SEMIAL NO.         COLLI M ATION           IND SEMIAL NO.         COLLI M ATION           IND SEMIAL NO. <th< td=""><td>PAGE <u>2 OF 2</u> <u>1 TINC</u> <u>1 C WIS</u></td></th<>	PAGE <u>2 OF 2</u> <u>1 TINC</u> <u>1 C WIS</u>
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c} \textbf{W} \textbf{S} = DPARTMENT OF COMMERCY AND ATTACK OCCANE AND ATTACKA AND ATTACKA AND ATTACKA AND A$	Раде <u>2</u> OF <u>2</u> <u>1</u> тичс <u>1</u> Сента <u>1</u> Соммента Соммента
HOAA FORW 74-187 HOAA FORW 74-187 HI-TTI HA D O VR. (40. (5AV CODE HISTRUVENT SERIAL NO. CODE 1 1 0 8 0/10/16 232 3091/0 31/2 FROM 84 DESIGNATION HAFter adjust ment: BACKSIGHT DACK CENTER SU 1000LE MEAN BACK CENTER SU SI	GEODETIC LEVELING REE - WIRE GBSERVATIONS	Рлас <u>2 ог 2</u> <u>1 тис</u> <u>1 тис</u>
HOAA FORW 78-189 HOAA FORW 78-189 $\bullet   a   0   \bullet   YR. /NO. /DAY CODE [HIST RUMENT VERIAL NO. CODE \bullet   a   : \bullet   B   0 /   0 /   6   2   3   2   3   0   1 /   0   3   1   2   3   0   1 /   0   3   1   2   1   1   1   1   1   1   1   1$	$\begin{array}{c c} \textbf{GEODETIC LEVELING}\\ \hline \textbf{GEODETIC LEVELING}\\ \hline \textbf{GEODETIC LEVELING}\\ \hline \textbf{REE-WIRE GBSERVATIONS}\\ \hline \textbf{REE-WIRE GBSERVATIONS}\\ \hline \textbf{ROD SERIAL ND.} \hline \textbf{CODE} \hline \textbf{ROD SERIAL ND.} \hline \textbf{COLLIMATION CHECK}\\ \hline \textbf{ROD SERIAL ND.} \hline \textbf{CODE} \hline \textbf{ROD SERIAL ND.} \hline \textbf{COLLIMATION CHECK}\\ \hline \textbf{ROD SERIAL ND.} \hline \textbf{CODE} \hline \textbf{ROD SERIAL ND.} \hline \textbf{COLLIMATION CHECK}\\ \hline \textbf{ROD SERIAL ND.} \hline \textbf{CODE} \hline \textbf{ROD SERIAL ND.} \hline \textbf{COLLIMATION CHECK}\\ \hline \textbf{ROD SERIAL ND.} \hline \textbf{CODE} \hline \textbf{ROD SERIAL ND.} \hline \textbf{COLLIMATION CHECK}\\ \hline \textbf{SI / 0   6   3   3   2   1 / 0   6   3   4   E   1 / 0   5   C   2   0   2   2   0   1 \\ \hline \textbf{FORESIGHT} \hline \hline \hline \textbf{FORESIGHT} \hline \hline \textbf{UPPER} \hline \textbf{HDDLE} \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \textbf{LOMER} \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \hline \textbf{LOMER} \hline \hline \hline \hline \textbf{LOMER} \hline \hline \hline \hline \textbf{LOMER} \hline \hline \hline \hline \hline \textbf{LOMER} \hline \hline \hline \hline \hline \textbf{LOMER} \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline$	Рлас 2 оF 2 2 тис 2 тис 2 тис 1 тис 1 тис 1 тис Сомментs
HOLA FORM 74-189 HOLA FORM 74-189 12-711 THE a = 0 = 7n. (AO. (DAY CODE INSTRUMENT SERIAL NO. CODE a = 1 = 8[0/7]0/16[2]3[2] = 3[0]9/7]00 (CODE FROM 8M DESIGNATION After adjostment: TO 1 BACKSIGHT Statement Statement LOWER	$\begin{array}{c} \textbf{U. S. DEPARTMENT OF COMMERC HATIONAL OCEANIC AND ATMOSPHENIC ADMINISTRATIO GEODETIC LEVELING REE - WIRE GBSERVATIONS       nod semal ND     code     nod semal ND     COLLI M ATION     CHECK       2     //0/06/3/3/2/3/1/2     //0/06/3/3/4     EXAMPLE     CONTINUED       BM DESIGNATION     2     \frac{1}{100}6/3/3/4 \frac{1}{2} \frac{1}{2} \frac{1}{0}/0/6/3/2 \frac{1}{2} \frac{1}{2} \frac{1}{0}/0/6/3/2       BM DESIGNATION     2     \frac{1}{2} \frac{1}{0} $	Рлас <u>2 ог 2</u> <u>1 тис</u> <u>1 тис</u>
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Рлас <u>2 оF 2</u> <u>1 тис</u> <u>1 се<sup>21</sup>ти</u> <u>1 се<sup>21</sup>ти</u> <u>1 се<sup>21</sup>ти</u> <u>1 се<sup>21</sup>ти</u> <u>1 се<sup>21</sup>ти</u>
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c} \text{GEODETIC LEVELING} \\ \hline \text{GEODETIC LEVELING} \\ \hline \text{RED_WIRE GBSERVATIONS} \\ \hline \hline \text{RED_WIRE GBSERVATIONS} \\ \hline \hline \text{RED_SERVATIONS} \\ \hline \hline \hline \text{RED_SERVATIONS} \\ \hline \hline \text{RED_SERVATIONS} \\ \hline \hline \text{RED_SERVATIONS} \\ \hline \hline \text{RED_SERVATIONS} \\ \hline \hline \text{RED_SERVATIONS} \\ \hline \hline \hline \text{RED_SERVATIONS} \\ \hline \hline \ \text{RED_SERVATIONS} \\ \hline \hline \ \ \text{RED_SERVATIONS} \\ \hline \hline \hline \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \$	PAGE PAGE 2 OF 2 Truc 1 Truc 1 Truc 1 Truc 1 Councents Comments $0$ $\Delta s_{1} = 0.0$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	PAGE 2   OF 2 $2   Truc$ $2   Truc$ $3   constant of the second seco$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2 OF 2 2 TWC 2 TWC 1 TWC 1 TWC 1 CWMCHTS COMMENTS 0 Δs, = 0.0 9 -Δs, = 1/9 9
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	PAGE 2 OF 2 $2 Truc$ $2 Truc$ $3 Constant of the second secon$
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	PAGE 2 OF 2 2 THE 2 THE 2 THE 2 THE 2 THE 2 THE 2 OF 2 2 THE 2 OF 2 2 THE 2 OF 2 2 OF 2
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c} \textbf{GEODETIC LEVELING} \\ \hline \textbf{GEODETIC LEVELING} \\ \hline \textbf{REE-WIRE GBSERVATIONS} \\ \hline Rec.wire and structure and structure administration of the second structure administ$	2 оF 2 2 оF 2 1 тик 1 тик 1 селіта 1 селіта 1 селіта 2 сомменть 0 Δ5, = 0 0 со 9 - Δ5, = 9 19 9
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	U. S. DEPARTMENT OF COMMERC HATIONAL DECAME AND ATMOSPHENIC ADMINISTRATIO         GEODETIC LEVELING REE-WIRE GBSERVATIONS         ROD SEMAL NO.         COLLI M ATION CHECK EXAMPLE CONTINUED         INTEL AND ATMOSPHENIC ADMINISTRATION         THE AND ATMOSPHENIC ADMINISTRATION         INTEL CONTINUED         INTEL AND ATMOSPHENIC ADMINISTRATION         THE AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL AND ATMOSPHENIC ADMINISTRATION         INTEL ADMINISTRATION         INTEL ADMINISTRATION         INTEL ADMINISTRATION         INTEL ADMINISTRATION         INTEL ADMINISTRATION         INTEL ADMINISTRATION         INTEL ADMINISTRATION     <	$\frac{2 \text{ OF } 2}{2 \text{ OF } 2}$ $\frac{2 \text{ OF } 2}{2 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$ $\frac{1}{1 \text{ OF } 2}$
THE         ITHE	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\frac{2 \text{ OF } 2}{2 \text{ OF } 2}$ $\frac{2 \text{ OF } 2}{2 \text{ OF } 2}$ $\frac{1}{2} \text{ TWC}$ $\frac{1}{10000000000000000000000000000000000$
HOLA FORM 74-187         THE         ITHE         IT	U. 3. DEPARTMENT OF COMMERCE ADMINISTRATION         GEODETIC LEVELING         REE_WIRE OBSERVATIONS         Indo semial wo.       COLLI M ATION CHECK         Indo semial work work work work work work work work	Рлас 2 ог 2 1 тис 1 тис 1 сельта $1 сельта 7 Соммента 0 \Delta s_{r} = 0.00 \Delta s_{r} = 0.09 .\Delta s_{r} = 0.0$
THE         THE         INDEA FORM TA - 187         THE         THE         THE         THE         INDEA FORM CODE [HIST RUMENT SERIAL NO.         CODE [HIST RUMENT SERIAL NO.         CODE [HIST RUMENT SERIAL NO.         INT RUMENT SERIAL NO.	U. S. DEPARTAENT OF COMMERC HATIONAL DECAME AND ATMOSPHENIC ADMINISTRATIO           GEODETIC LEVELING REE-WIRE OBSERVATIONS           non-structure and structure and structure administration           CODE ROD SERVATIONS           Intel colspan="2">COLLI M ATION CHECK EXAMPLE CONTINUED           Intel colspan="2">CONTINUED           Intel colspan="2">COLLI M ATION CHECK EXAMPLE CONTINUED           Intel colspan="2">CONTINUED           Intel colspan="2">Continues           Intel colspan="2">Continues           Intel colspan           C	PAGE 2 OF 2 $2 \downarrow THEC$ $1 \downarrow COLOR 1 \downarrow C \downarrow OSCOMMENTS0 \Delta s, = 0.09 - \Delta s_4 = 919 9$
HOLA FORM TA-189         THE         I I I I I I I I I I I I I I I I I I I	U. 3. DEPARTMENT OF COMMERCIZATION         GEODETIC LEVELING REE-WIRE GBSERVATIONS         rado semial No.       CODE       ROD SEMIAL NO.       COLLI M ATION CHECK         2       1/10/6/3/3/2/3/1/2       3/1/2       1/10/6/3/3/4       COLLI M ATION CHECK         BM DESIGNATION       1       1       1       CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUED         THE CONTINUE         THE CONTINUE         THE CONTINUE         THE CONTINUE         THE CONTINUE         TOTO         TOTO       State	2 оF 2 2 оF 2 1 тик 1 тик 1 сельти 1 сельти 1 сельти 0 Δ5, = 0.0 9 - Δ5, = 19 9 19 9

Figure 3-46.—Collimation check, Kukkamaki's method, three-wire procedure, half-centimeter rods.



First setup



# Second setup

Figure 3-47.—Collimation check, 10-40 method.

of arc in which the compensator is expected to suspend freely should be somewhat greater than the arc described by a 2-mm movement of the bubble in the circular level (Table 3-3 contains the sensitivities of the circular levels mounted on various instruments.) When the circular

level is properly adjusted the compensator should provide a line of sight having a consistent collimation error, within 1''0 (0.00485 mm/m) of that measured during the collimation check, no matter in which direction the instrument is pointed.

ſ

HOA	FORM 74	-191					~~~~			······											 N	AT10	***	00.57	HIC	U.	5. DE	PAR	THE	NT OF	CON	MERC	E "^	at
											міс	ROM	GEO Ete	DE1 R (	TIC I DBSE	E' RV	VELI	NG ONS	(ձհ)															/ 07/
•••	0 • Y	R. /	40. p	Y C	COL	+ <b>*</b> 5T	RUKE	нт \$I	ERIAL	. но	¥ (	obt	Ţ	RCD	SERI		10.	_ co	DE	RO		AL H	o.	Γ	ÇC	JLL.	MA	177	ON	Ch	IEC.	ĸ	I	TIME
• •	1 1 8	00	(02	6 2	33		4	5]6	51	111	HJЗ	116	<u>.</u>	2	2 3 4	1 5	6	7]3	16		2 3	4 5	68	1	*		E/	XAA	ЛР	LE_			Ц	
		//	0 - 4	0 /	ЧE	THO	٥			Н	Ť	Ц	IN DE	310 1	(AT (0								/	30	/2	0			z 7	1	2	2.0	00	CWS
SET UPP	STADU BÁCX		5# 1		57	101A 396		ŝ,			LO BÅ	₩ SCA CK/F	OF E				۵ <u>۲</u>			HIG	1 3 CA1	Ē			Δh Σ	H 74		Å	L- H			ħ.	MARKS	
Π	301	8	17	02	30	6.	$\overline{+}$	4	00		3	01	3	Z 4	F	Ŧ	4.0	7	B p	8	73! /9!	75 06	F-	-	25	; э/	-	-	36	2	0.7 59	10	hсм M	001,-004, 15.+15
			······································		,							4			-fl			ii-				- 1		<u>d=</u>	21				2=	+	0.0	12	hem/	—,
2	302		4	00	29	9!	┽┼	3	98	8	3	02	17	7	뷤		44	43		17		2/	ŀĽ	4-1	25		-	4	35					
	<u>, en 718</u> ,	<u>* 1</u>	.]_[#		<u> </u>	+Δ	52 *	-5	9.6	<u> </u>	14	17.15	·· r		1 1			171	لمنار	101	<u>, i, :</u>			<u> </u>	21	Δ.	h,+	۵4	- -		0.2	35	hom	
Ц.							++	+		8		$\left  + \right $	+	_	$\square$	╇	H	4-4	B		┽┽	_		+	_	╬╌╋	+		<b>X</b>	Ļ		5	*/hc	• ** • * *
	L	<u> </u>			L	<u>l</u> .	<u> </u>		<u></u>	1.1	<u> </u>	LL,	<u>.                                    </u>		1,1	Ц.	1 :	<u></u>	<u> </u>	e	= e,		240	<u> </u>	0.2	2				Ξ,	1. 1 0.4	75 . 00 /	nyurs Mars	<i>∆h,+∆h</i> 2× €
		Т	T	1	П	ΠŢ	П		П	8	1	П	Π		П	<b>_</b>	T !		P	11	11			П		-	Т			- 1	0.7	75	et/14	
				<u> </u>		[]	11			[ <b>7</b> ]		I. I.			1				1*		1:			.[]		! ]		!		=	<u>59</u> .	6 /	7	D5,+Q5,
T		Τ		ļ	Π	ΓŢ	П	Т		n	Τ	П	ŢŢ	Т	П		ļ	T	8					Π		!		Ţ	Τ	É				*
				ļ		<u>l i</u>				•			11		Ш		] ]		<b>r</b>					Ш		! [		!	Γ					
H		Т	1		П	T	TT	Π	L İ		Т	П	T	F	TT	1	11	ТΤ	T <sub>B</sub> T	TT	TI	Т		ТТ	Т	İΤ	Π	T	Т					
				I		ļ				ŀ			Ţ		11	Τ	11		F		ļ			Π		!		ļ						
	r m	Т	111	T	<b>—</b> —		тт		TT-	тт		m	TT		ГГ	<b>T</b>	TT	ТΤ	<b>T</b> .	1	11	1	П	ТΤ	Т	<u> </u>			Т					
		T		Ť.		<u>t</u>	Ħ		<u>-</u> †-	Ħ	-	$\square$	; †		$\Box$		ŢŢ	$\mathbf{t}$	Ī,		<u>†</u> :					<u>i t</u>	$\square$	-i-	$\Box$					
			111			<del></del>	TT	77	- m	<b>T</b> . T		r T	TT		тт		τr	TT	1.1			Т		тт		<del>r r</del>			-					
				Ţ		† †	++	+	H	i,	+		††			┢	t	$\mathbf{H}$		+	ti	+		+	+	††	+	Ť	H					
<b></b>						1-1-			<b></b>	т т			<del></del>		т т		<del></del>		<b>T</b> T	1 1	<u> </u>			11	-	T1			Ţ					
<u> </u>		+				H	++	╉┥	H				ήł	+	┢╌┝╴		Η÷	+	-	┼┼	Ηİ			╉	+	<del>i  </del>	+	÷	╋					
<b></b>					r	1-1-						 E T	17	-		- -								 				- 1	,					
		+		÷		┝┼	┼┼	+	$\vdash$	╞	+	$\left  \cdot \right $	ił	+	╟	┥	H	+	╞	++	+	+	+	╉╉	_	$\frac{1}{1}$	+	+	-					

Figure 3-48.—Collimation check, 10-40 method, micrometer procedure, half-centimeter rods, reversible compensator.

HOAA FORM 74-191 (8-77)	NA	U.S. DEPARTME	NT OF COMMERCE PAGE					
м.	GEODETIC LEVELING ICROMETER OBSERVATIONS (۵ħ)		//					
+ 4 0 + YR. MO. DAY CODE INSTRUMENT SERIAL HO M	CODE ADD SERIAL NO. CODE ROD SERIA	L NO. COLLIMATION	CHECK Z TINE					
· · · · 801026231 50695H	316 123456316 123	468 EXAMPLE						
FROM BH DESIGNATION	D BH DESIGNATION		ERATURE US SERVER					
S///4///2/15/ //8/8/0// ews								
SET STADIA BE STADIA BE LUPP BACK I PORE I	LOW SCALE AL HIGH SCALE	Е Дъни Дъц. Е Σ дън	REMARKS					
1/150 49/52 1980	150.67 - 2.28 = 447.4	2 - 1228 +00	1 . 0 000					
	/ 5 2:7 5	+ 10	ΔH <sub>1</sub> = 0.000					
2/57! 51/56 2000	15102 + 227 = 4756	5 + 1228 -01						
$\frac{1}{53!7!} \frac{1}{50!} $	/ 516:715     0.0	$- / O \Delta h_{1} + \Delta h_{2}$	Ди,==000s = −0.005 hcm					
		<u> </u>	5 m/hcm					
	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2.2 mm -	- 0.400 MM IXC					
			+ 0.375 mm					
┟──┠╶┠╶╏╶╏╶╏╶╏╶╏╶╏╶╏╶			- 59.6 m 135,7 - 2					
B								

Figure 3-49.-Collimation check, 10-40 method, micrometer procedure, centimeter rods, nonreversible compensator.

Systematic change in the collimation error, resulting from consistent variation in the response of the compensator to gravity, is limited by strictly following the micrometer leveling procedure. The instrument is leveled while pointing in one direction, and then releveled while pointing in the opposite direction. The releveling is omitted if the compensator can be mechanically repositioned, as is the case with an instrument having a reversible compensator.

However, a compensator may "hang" or "stick" in such a way that releveling or repositioning may not remove all of the compensation error that is introduced (fig. 3-5). With a nonreversible compensator, the observer should lightly tap the side of the instrument or tripod leg during each pointing to make sure that the image oscillates. This indicates that the compensator is freely suspended. In addition, a compensation check should be performed weekly.

With a reversible compensator, whenever the compensator is repositioned the observer should check that the image oscillates. Compensation error in such an instrument results in a lack of symmetry in the two lines of sight. The error is revealed if the mean collimation factor exceeds the tolerance during the collimation check; therefore, a separate compensation check is not necessary.

Instructions. The form commonly used for this test is NOAA Form 77-81, "Geodetic Leveling Compensation Check." The form is designed for use with the Nil leveling instrument and half-centimeter rods. An example is given in figure 3-51. The following instructions apply to any compensator instrument with a micrometer, used with double-scale rods. If a computer program is not used, prepare a standard recording form as illustrated in figures 3-52 and 3-53.

1. Make sure that the circular levels on the instrument and the rods are properly adjusted before beginning this test. (See sec. 3.3.5, "Circular-level adjustment" and 3.4.3, "Checking the circular level.")

2. Set up a leveling rod 40 m from the instrument. For convenience, use rod 2 from the last setup of the collimation check.

3. Point the instrument at the rod. Bring the bubble in the circular level to position "a," where the left

NOAA FORH 76-189 U. S. DEPARTMENT OF COMMERCE 14-77) NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION									PAG	E															
GEODETIC LEVELING THREE-WIRE OBSERVATIONS									1	0F															
* 4 * 4	0 *	YR. BO	/ua. 10	/6. 2 1		2086	<u>o.</u> I			∧∟ ⊋[:	но. Э (	dı	со: Э I	e J	R	0D 5	ERIAL NO.	0	COLL E	/ M A T	ION ( PLE	CHE	CK	z	TIME
F RI	эм	. <b>ВН О</b>	esigni 10	4TIC )	»» 40 МЕТНО	D	T		N DESIGNATI								Z BEGIN S 0900		92	- [8 0 C	TEMPER 0COIN 11,3	ATU 7	RE ND 131	s /	C WS
					BACKSIC	HT											FOR	ESIC	нт					CON	MAENTS
		UPPE	R	_					~	I	Τ		UPP	ER										]	
τιρ'	<u> </u>	MIDDL		+	MEAN	BACK CENT	ER	s <sub>υ</sub>	38	-	┢		MIDO	CE			MEAN	1-		NIER	<u>ະ</u>		9 E		
		Σ		╉	5			aL	το τ ωτ	┥	+		5			<u> </u>	Σ	-			<u> </u>		Σ		ŀ
		ΠĨΊ		┉┝	TTTTT					1	$\dagger$		Ť	П	Ţ		TĪŢŢ	╈	TTT	1			$\overline{1}$		1
		1	<u>र हैं</u>	đ	مستحص المستقصيل ومتعمر				,,,,,,,,,,,,,,,,	1	F		1	7	0.5		,	+-	<u></u>					1	
11		1	5 3.	6	153.63		Ţ	5.0			Γ	$\top$	1	5	0.5	Т	1505	3			20:0				1
		1	4 <del>0</del> .	7				4!9	9	2			1	3	0.6						19.9		39.9		1
		<u></u>	<u> </u>	1						_	F			Ľ.						······	ļ			1	
<u> </u>		14	62	4				لمقر		l	F		_1/	8	<u>0'8</u>	Į	THE LEAST	-	<del></del>	<del>- (* · z · ·</del>					
- d		- 1	57	5	157.47	landa i d		4.4			ŀ	-	-1	6	0.8	ļ⊥	160.8				200		I dala	ł	
		1		5	31110			15:0	<u> </u>	갦	┢		1	4	0:8	┟╌	3///3	-			20:0		799	1	
		++	- <b>[-</b> ī	÷	- 311.73	}			- 79	쥙	ŀ			+	Ť	┟┈╹╸		4-						1	1
						Con TT	<u> </u>		- 60.	71	"ŀ		$\neg$	1	Ţ			-		ŢŢ				1	
			$\neg$ i		X /0	мм/см		Hi		-	T	-1-1			İ	┟┈┷	or daarahaan daarahaan da		1, 1, 1	<b>.</b>		T		1	1
					-2.30	mm △h,+	∆h,				Ľ				Ţ									1	
				ŀ	0.40	mm 2x	e			٦	L													]	
				4	-1.90		!		<u></u>	_	L							_					·····		
	$\square$			╞	<u> 60.1</u>	$\square \Delta S_i + I$	723			_	╞	1		$\downarrow$	_ <b>.</b>	<b> </b>							<u></u>		i
		+		-	C =   + 0:0 3	MM/M			┍╾┻┻┻┻┻┻	4	┝	_		┼╍┤		┞┈┴					h	<u> </u>		•	
	$\vdash$			+				-			ŀ		-	┝┤	-				$\overline{\mathbf{T}}$		$\left  \right $				l
		┼╌┼╌│		╈		┟┉└┈└─╵└				┥	ł	+		┼╌┼	-i-	┟─┴		+						1	
	$\vdash$	++	1	T		1				-1	F			$\dagger$	Ţ			1			┝╍╧┥	+	<u>++</u> !-	1	
		┼┼┤	T	1		1				4	F				Ţ	† †	<u></u>	╈		- ****				1	l
											Γ			Π	ļ	Π									1
						]				]				П		ļ,.		_					44-		Î
						L					1				5										

Figure 3-50.—Collimation check, 10-40 method, three-wire procedure, centimeter rods.



Figure 3-51.—Compensation check for the Ni1 and half-centimeter rods.

edge of the bubble is just under the black line of the left edge of the circle.

4. Read the low scale of the rod. Estimate to onethousandth of a rod unit (the third decimal place) on this and all subsequent scale readings during the test. Record the reading in column C. Record the integer portion in column A.

5. Read a stadia intercept. Record it in column A and compute the half stadia interval, HSI, in column B.

6. Read the high scale of the rod twice (two pointings). Record the readings in column E. Compute the mean in column F.

7. Read the low scale again, record the reading in column C, and compute the mean of the two low scale readings in column D.

8. Bring the bubble in the spherical level to position "c," where the right edge of the bubble is just under the black line of the right edge of the circle. Repeat the rod readings in the same order (steps 4, 6, and 7), omitting stadia readings.

9. Convert rod units to millimeters. Compute the sighting distance, s, in meters. Compute tan  $\alpha$  in millimeters per meter by the formula

$$\tan \alpha = \left[ (a-c)_L + (a-c)_H \right] / 2s$$

Tan  $\alpha$  is the maximum change that may be introduced into the collimation error when the compensator is suspended within the maximum left-right range of arc permitted by the circular level.

10. Tan  $\beta$  is observed and computed in the same way. Substitute bubble position "b" for "a" and bubble position "d" for "c," and follow steps 3 through 9. Tan  $\beta$  is an expression for the maximum change that may be introduced into the collimation error when the compensator is suspended within the maximum forward-backward range of arc permitted by the circular level.

11. Both tan  $\alpha$  and tan  $\beta$  must be less than  $\pm 0.0097$  mm/m ( $\pm 0.0000097$ ). If this is not the case, the instrument should be adjusted at the repair shop or factory.

12. Submit data from the compensation check to the project office along with the leveling data acquired that day.

# 3.4 Leveling Rods

Leveling rods must be carefully designed, precisely manufactured, regularly calibrated, and properly used if they are to provide accurate heights above turning points and control points. In the following pages leveling rods are described, and the requirements for their use, calibration, and maintenance are presented.

#### 3.4.1 Description

Leveling rods have often been made of seasoned hardwood, with a scale printed or stamped directly on the wood. Typically, such a rod is constructed of two or

ł

(



Figure 3-52.—Compensation check, micrometer leveling, half-centimeter rods.

HOAA FORH 74-191				HATIOHAL	U.S. DEPARTMEN OCEANIC AND ATMOSPHERIC	AT OF COMMERCE PAGE
		GEODET MICROMETER C	FIC LEVELING DBSERVATIONS (4h)			1 or 1
· 4 0 · YA. / NO. DAY C	ODE HASTRUMENT SER	ALHO N CODE ROD	SERIAL NO. CODE	ROD SERIAL NO.	COMPENSATION	CHECK Z TIME
- 1411 - 20171701512	31/ 1 208				EXAMPL	
- FROM				3 10	00/0300 /5	5 / 5 4 / 0 CMS
SET STADIA Sa UPP BACK I		LOW SCALE	AL MEAN	HIGH SCALE	- MEAN AN	AZUANKS
a / 4 4 2/!7 / 223 x 2	<u>en</u>	a 144361 r 350	/44.356	440606	440412	
5 = 43.4	/m					
		384	-0024	440:652	-0034	-0029 cm 4-C
			·····		×	10 mm/cm
	┟┼┟┊┠┼┼┼	<b>B</b> <b>7</b>	B F			-0.290 mm 43.4 m \$
				111111	-±an ∝ =	-0.0067 mm/m
	┝┼┼┲╋╋					
6/44 21.6		8 /44468	1444640	440248	440233	(A) I
/22!4x 2!		P 461		1 2/8		ROD
		B 144248	1442423	440004	440002	
		1 205	+0222	000	+0231	+0.226
		8	В			+ 2.260
┠┈╫╹┚┊╫╫╢╢╝┚						43.2
		a				
			B			
<u>                                     </u>					<u></u>	Instrument must be
		B	a			Adjusted
						L

Figure 3-53.—Compensation check, micrometer leveling, centimeter rods.

more telescoping sections. The overall length of the scale on a wooden rod changes significantly with normal variations in atmospheric humidity. Furthermore, when assembling two or more telescoping sections, mismatches at the junctions introduce further errors in the scale. Therefore, for geodetic leveling, a much more precise rod is required. The elements essential to such a rod are described next.

Scale. The scale should be a continuous (not collapsible or folding) metal strip with a small coefficient of thermal expansion. Steel-nickel alloy (Invar) is a suitable, commonly used metal for this purpose. A scale length of 3 m (10 ft) is efficient when the height of the line of sight is expected to average from 1.5 to 2.0 m (4.9-6.6 ft). A 3.5 m (11.5 ft) rod may be preferable if the line of sight averages 2.0 to 2.5 m (6.6-8.2 ft), as in the case when observations are made from a vehicle in motorized leveling. The length of the scale sets an upper limit on the amount of scale-related error that may be accumulated in a setup.

Graduations must be accurately marked on the scale during its manufacture, and the scale must be calibrated regularly thereafter. The intervals indicated by the graduations, termed rod units, should be no larger than 1 cm (0.03 ft). Half-centimeter or centimeter units are preferred for geodetic leveling, although "centifeet" (0.01 ft) and "centiyards" (0.01 yd) have been used. The units must be compatible with the units of the micrometer in the leveling instrument. (See sec. 3.3.4, "Optical micrometer.")

The scale or its housing is usually labeled with numbers corresponding to tens of rod units. To obtain a reading, the observer must count the rod units between the point where the line of sight intercepts the scale and the first labeled number below that point. The reading is the sum of the labeled number and the number of rod units counted. To prevent blunders as a result of miscounting, every graduation could be labeled. This is possible with centimeter rods, but not with half-centimeter rods since the numbers are too small to be read at the usual sighting distances. The orientation of the labeled numbers must be compatible with the instrument, depending on whether the image provided by the instrument is upright or inverted. An inverted scale, for use with an inverting instrument, is pictured in fig. 3-54).

The type of scale must be compatible with the leveling procedure. For micrometer leveling, two parallel scales with line graduations are necessary on each rod. Line graduations are lines of finite height (0.8 to 1.6 mm), the centerlines of which define their actual positions on the scale. When intercepted with the wedge reticle (sec. 3.3.5, "Circular level adjustment"), precise readings can be made without estimation. The parallel scales are marked on the same metal strip, one offset slightly from the other, so a constant relationship exists between them even though the overall length of the scale may change. The graduations are labeled so the base of the low scale is zero, and the corresponding point on the the high scale is about 300 cm (600 hcm). The resulting offset, in rod units, is termed the rod constant. (See fig. 3-55.)

The pair of double-scale rods must be unmatched; each rod must have a different rod constant. This permits a check to detect observations that are transposed during a setup. The rod with the smaller constant is always labeled rod 1. It should be marked with a piece of flagging or reflective tape so it can be easily recognized.



Figure 3-54.—Inverted scale for a leveling rod.

Because the scales are independent of the rod housing on which the graduation values are labeled, the rod constant may be changed if necessary. If this is done, a record must be included in the observation data specifying the date of change and the new constant. For example, if both rods of a pair have identical rod constants of 592.50, one constant can be changed to 602.50 by affixing a metal strip over the high-scale graduation values marked on the housing. The strip should be engraved with values increased by one ("60" becomes "61," etc.), to obtain the desired 10 unit change in the constant.

For three-wire leveling, rods must have single scales and block graduations. In addition, each rod should have a check scale in different units on the back. Block graduations are the borders between rectangular spaces, which usually alternate black and white in color. Each block is of a height equal to one rod unit. This type of



Figure 3-55.—Double scale with half-centimeter line graduations.

graduation permits the intercept of the line of sight to be estimated to one-tenth of a rod unit. Black blocks must not be thought of as black graduations on a white background.

Figure 3-56a shows a type of block graduation (known as the Philadelphia graduation) commonly used in engineering surveys, with a rod unit of 0.01 ft. At short sighting distances, readings may be estimated to 0.001 ft. Figure 3-56c shows a similar type of graduation, in which the unit is 0.01 yd.

Figure 3-56b shows the block graduation devised and used by the U. S. Coast and Geodetic Survey from 1916 to 1962. The unit is 1 cm, but the rod appears to have two adjacent, parallel scales. In each centimeter, there are a black block and a white block side by side, causing a checkerboard appearance. The purpose of this scale is to make a white block available, in every centimeter, against which the black reticle line shows distinctly, thus facilitating estimation of tenths of units. This scale must not be confused with the offset parallel scales used for the micrometer-leveling procedure.

Housing. The housing of a precise leveling rod is made of durable metal or seasoned hardwood. It provides a frame in which the scale may be suspended under constant tension, independent of expansion or contraction of the housing. Padding between the edges of the scale and the housing prevents wind-induced vibration and the resulting wear on the scale and decrease in reading precision.

A base plate ("footpiece") made of noncorrosive metal should be an integral part of the housing. It should be designed so that the rod may be placed easily on the various types of control points. The bottom surface of the base plate should correspond to the zero of the scale. Typically, the scale is attached rigidly at the foot of the rod and attached flexibly at the top, by a springloaded clamp.

Circular level. A circular level should be mounted on the back of the housing, where it can be readily observed by the rodman. It should have a sensitivity of 10'0 or less per 2 mm displacement of the bubble. Thus, the rod may be plumbed within 10'0 (equivalent to 1.0 cm at the top of a 3-m rod) of the direction of gravity. A mirror or reflecting prism placed above the level facilitates centering.

Brace poles. The rod should be equipped with a pair of adjustable length poles, attached by swivel joints to the center of the top of the rod (fig. 3-57) The pole assembly should be lightweight, but it must be sufficiently strong to withstand the stresses of bracing the rod against the wind. Mounting the assembly at the top center permits the rod to be pivoted on the turning point between setups, without moving the poles. (If the circular level is not properly adjusted, the fact should become evident while pivoting the rod.) The use of brace poles minimizes swaying and unsteadiness of the rod, allowing it to be plumbed more precisely.



Figure 3-56.—Leveling rods with block graduations.


Figure 3-57.—Brace poles for a leveling rod.

#### 3.4.2 Calibration

The rods used for geodetic leveling must be calibrated routinely by a reputable laboratory against a standard that has been compared to the National Standard of Length. A complete set of calibration data should be obtained for each rod before the rod is first used, every 5 years during its use, and just before the rod or scale is retired. The data should include the index error, length excess, and coefficient of thermal expansion. In addition, during each year of use, at least one calibration should be made to determine the index error and length excess. If a rod is dropped, or otherwise handled in a way that may have damaged it, it should not be used again until it has been recalibrated.

Each calibration consists of the precise measurement of at least four overlapping intervals on the scale (usually between the base plate and the bisected graduations at 0.2, 1.0, 2.0, and 3.0 m). The measurements, with respect to the National Standard of Length, should be accurate to  $\pm 0.05$  mm.

From one calibration a value for the index error and a factor for the length excess can be computed. The differences between the calibrated and assigned lengths of the graduations are plotted versus the assigned lengths (fig. 3-58). The intercept of the plotted line with the axis of the differences is the index correction which is equal and opposite in sign to the index error. The slope of the line is the length excess.

To obtain a complete set of calibration data, at least four calibrations must be made, each at a different temperature between 15° and 35°C. The variations in graduation values due to changes in temperature are described by the coefficient of thermal expansion, computed from these data. To verify the amount of scale offset, at least one additional calibration of the high scale of a doublescale rod is required.

Calibration with a laser interferometer. A very accurate method for calibrating the rod scale is employed by the National Bureau of Standards, using a laser interferometer. All leveling rods used by the National Geodetic Survey should be calibrated by this method at least once during each year of use. Instead of values for four intervals, a calibrated value is obtained for every graduation on the scale. With this information, observed heights may be converted directly to "calibrated" heights.

Submitting calibration data. The complete set of calibration data must accompany any leveling data submitted to the National Geodetic Survey for inclusion in the national network. Formats for the NGS "Leveling Instrument and Rod File" are presented in *Input For*mats and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: vol. II, ch. 6).



Figure 3-58.—Rod calibration data.

Calibration data should include the actual measurements and the temperatures at which they were made for each rod used. All data should be identified by the serial number of the rod. If a rod housing is reused with a new scale, the rod should be assigned a new serial number and recalibrated The same applies if a scale is reused in a new rod housing.

#### 3.4.3 Use and Maintenance

The leveling rod, like the leveling instrument, must be properly maintained if good quality data are to be collected. The rodman should guard the rod against physical damage and protect it from the effects of exposure to the outdoor environment. The guidelines and adjustment procedures given in this section should be strictly followed.

General handling. Because of its length, the leveling rod is so easily struck against objects that great care must be exercised when handling it. Paint does not stick easily to the special materials used for rod scales. As a result, striking the rod against any hard object (such as poles, trees, signs, rocks, vehicles, or the ground) may chip paint off the scale, making it difficult to continue observing precisely and efficiently. Such shocks may also damage the inner spring assembly or the housing, causing unpredictable change in the length of the scale.

Carry the rod by holding it with one hand at the middle and balancing it face up across your shoulder (fig. 3-59). Do not touch the painted scale. Use your arm as a spring to absorb shocks. When riding on the rear step of a truck, hold the rod parallel to the side of the truck, taking care that it does not hit the truck or any obstacles along the leveling route.

During a halt in the leveling operation, hold the rod vertically on the toe of your boot. Otherwise, place the rod in a protected location where it can be easily seen. Do not place it near the path of a vehicle Lay it gently on the ground, with the scale facing up and the circular level protected from stress. See that no dirt or rocks are kicked on the rod while it is on the ground.



Figure 3-59.—Carrying the leveling rod.

At the end of work each day, gently wipe the rod to remove moisture and dirt. Store it, supported firmly by padding, in a box large enough to accommodate a pair of rods with brace poles and turning point guides attached Protect the circular levels and any attached mirrors with padding or the rod handles (which may fold over the attachments).

Setting up the rod Since the base plate corresponds to the zero point of the scale, it must be kept perfectly clean and free from corrosion if precise heights are to be observed. Before setting the rod on a point, carefully wipe away any dust or dirt on the base plate and the point. Do not allow the base plate to rest directly on the ground, since scratches may result. A rod with a defective or damaged base plate should be repaired and calibrated.

Set the rod gently onto the point; never drop it. Dropping a rod on a turning point or control point is comparable to hitting the point with a 7-kg sledgehammer. Even the best quality bench mark may suffer from such treatment.

Place the rod so that the exact center of the base plate rests on the highest point of the turning point or control marker. (See fig. 3-76 for the locations of control points on various types of monuments.) The National Geodetic Survey uses a turning-point guide ("flipper"), attached to the bottom of the rod, to assist with this centering (fig. 3-60). The guide is designed to fit around the standard turning point. If the rod is set against a flatter, broad surface, such as a bench mark disk, the guide flips back out of the way, exposing the entire base plate.



Figure 3-60.—Turning-point guide.

Be alert to the possibility that the rod can be set down in such a way that some part of the turning-point guide, instead of the base plate, is the contact point. Prevent this by turning and jiggling the rod after placing it on the turning point. A visual examination of the point of contact may be necessary. The guide can also "hang up" inside the protective 4- or 5-inch pipe which encircles many bench marks. When setting the rod on a mark of this type, remove the turning-point guide by loosening the two finger screws.

Always place the brace poles downwind from the rod, and arrange them so they do not interfere with the observer's line of sight. Face the rod toward the instrument and plumb it by adjusting the lengths of the poles until the bubble in the circular level is centered precisely. After completing one setup, without moving the poles, rotate the rod to face the instrument at the next setup. Do not remove the rod from the turning point at any time during this cycle of two setups. If necessary, replumb it by adjusting the poles again.

Lightly hold both poles throughout each setup. Stand where the circular level can be seen to ensure that the rod is plumb, and always be ready to prevent the rod from falling (fig. 3-61). Do not press down on the poles or lean against them, since this can cause movement of the turning point.



Figure 3-61.—Setup of leveling rod.

The top assembly of the brace poles must withstand a great deal of stress during leveling. Examine it regularly for loose, bent, or broken parts. The large screw, which secures the assembly to the top of the rod, may loosen and its shaft may bend if it is subjected to strain. Prevent this by checking at least once a day to see that the screw is tight. Do not twist the poles, since this may bend the fittings which connect the poles to the top assembly. The grips lock the poles at the desired length by means of friction. They may become smooth with use and should occasionally be taken apart and roughened with sandpaper.

Checking rod plumb. During a setup each leveling rod must be plumb, in other words, vertically aligned with the direction of gravity. Achieve this by centering the bubble in the vial of the circular level attached to the rod housing. To center the bubble precisely, use a mirror or reflecting prism mounted on the rod to observe the level from as nearly overhead as possible, not from the side.

Each day, while leveling, the observer should check the plumbing of each rod by comparing the alignment of each rod to the vertical line in the reticle of the leveling instrument. Check the rod while it faces the instrument; then, after the rodman turns the rod 90° (at a right angle to the instrument), check it again. This procedure quickly reveals gross errors in the plumbing of the rod.

If the circular level is not properly adjusted, centering the bubble does not ensure that the rod is plumb. Check the level by one of the following methods at least once a week and immediately after any shock to the rod. Include a note in each weekly report, stating the time and date that the check was performed, and explaining the adjustments required.

If the level cannot be adjusted within tolerance, the rod housing may be warped or the level may be defective. In this case, check the housing by stretching a string from one end of the rod to the other and looking for bends or twists. If the rod has such defects, replace it immediately. If not, replace the circular level.

Checking the circular level. The simplest method for checking the adjustment of the circular level is the bracepole method. To use this method, the rod must be equipped with a turning-point guide and the brace poles must be in good condition. In particular, the shaft of the large screw which centers and secures the poles to the top of the rod must be straight. If this is not the case, three guy wires, attached to the top center of the rod, may be used in lieu of the brace poles.

Set the rod on a turning point and precisely center the bubble by adjusting the length of each pole or guy wire. Lock the poles and slowly rotate the rod  $360^{\circ}$ . The bubble should not move more than 2 mm (0.08 in) from the center, a distance which represents an angular deviation from vertical of 10'.0 or less. If the bubble moves more than this distance, adjust it in the same way as the one on the leveling instrument. Follow the instructions in section 3.3.5, "Circular-level adjustment," and adjust the poles or guy wires as though they were the tribrach foot screws of the instrument.

Another method for checking the circular level is the suspension method. Begin this method by suspending the rod from an eave or other similar structure with a short piece of strong line fixed at the top center of the rod. Anchor a piece of paper under the rod. (The back of the weekly report form is suitable.) The check and adjustment follow:

1. Plumb the rod, with the base plate resting against the paper. Trace around the front and sides of the base plate.

2. Without moving the paper, rotate the rod to face the opposite direction. Plumb it again and retrace the front and sides of the base plate. The two tracings should be parallel. If not, the rod has not been rotated completely.

3. Measure the distance between the corners of the tracings (fig. 3-62). For a 3-meter rod, if the distance is more than 2 cm (0.8 in), the circular level should be adjusted.



Figure 3-62.—Tracings for checking the adjustment of the circular level.

4. Before adjusting the level, move the base plate of the rod halfway back toward the first tracing and hold it there.

5. Center the bubble in the level by adjusting the small screws supporting the vial with an adjusting pin. Always turn the screws downward, to maintain pressure on the base washer. Do not retract one screw and depress another. If more adjustment play is needed, retract all the screws and begin adjusting them downward again. Do not force the vial tight against its base.

6. Turn the rod back to the original direction and perform the test again.

7. Submit the tracings and measurements to the project office with the day's leveling data or with the weekly report.

## **3.5 Turning Points**

A turning point is the temporary support on which a leveling rod is placed during a setup. The foresight point for one setup becomes the backsight point for the next, "holding" the elevation while the leveling instrument is moved between setups. Therefore, turning points must be stable, well defined, and properly spaced to measure accurately elevation differences from one setup to the next. The degree of care taken by the pacer and rodmen in setting and using turning points greatly affects the quality of the leveling. This is especially true in singlerun leveling, since movement of the turning points between setups is not easily detected

#### 3.5.1 Selection and Use

A turning point should not settle or rebound. It should have a definite high point on which to hold the rod and should be easy to set and remove. Do not use objects such as stones, fire hydrants, railroad spikes and ties, or marks on the pavement. Instead, use a pair of standardized points, suitable for the terrain likely to be encountered. Any of the three types described in this section is acceptable. The most satisfactory of these for routine use, under a variety of conditions, is the turning pin. On concrete sidewalks and gravel, the turning plate is suitable. If the route crosses sandy or marshy ground (and the line cannot be relocated to avoid this), a wooden stake with a double-headed nail provides the most reliable turning point.

General instructions. When setting a turning point of any type, avoid loose soil, sand, or unpacked gravel. If necessary, dig away the loose layer with a shovel. Avoid frozen ground, but if it cannot be avoided, dig through the shallow frozen layer to unfrozen ground in which a turning point may be placed.

Once set, any turning point can settle, rebound, or be displaced surprisingly easily. Often these changes are not detectable. Be alert to the possibility of such movement by observing the following precautions: Do not drop the leveling rod onto the turning point. Do not remove the rod once it has been placed, since removing and replacing the rod can cause a significant change in the elevation of the point. Maintain a constant weight on the point by allowing the rod to stand free from downward pressure as much as possible. Do not walk around the point unnecessarily.

*Turning pin.* A turning pin and hammer for driving it are illustrated in figures 3-63 and 3-64.

To set a turning pin, place the driving cap over the high point and use a 2-kg (4 lb) hammer to drive the pin into the ground. The faces of the hammer should be made of replaceable soft plastic, to prevent damage to the driving cap. Do not hold the hammer with its head crossways when hammering, since this may damage the metal part of the head and make replacement of the faces difficult. Drive the pin straight, not at an angle, until the driving shoulder is within 5 cm (2 in) of the ground. If the pin is set at an angle, or not driven deeply enough, it is more likely to rebound or settle.

After setting the pin, remove the driving cap and carefully place the rod on the high point. After the setup has been checked, gently strike the side of the pin from



Figure 3-63.—Turning pin (a) and hammer (b).

two directions. This should loosen the pin in the ground so it can be pulled "by hand." Avoid pulling the pin by tugging on its cap or chain.

Turning plate. When a turning pin cannot be driven efficiently, such as in the hard-packed gravel of a road shoulder, in very firm clay, or on a concrete sidewalk, use a turning plate (turtle). A turning plate should have (1) a definite high point, (2) three flat feet, and (3) weigh at least 7 kg (15 lb). Figure 3-65 shows a recommended design.

Exercise caution when setting the turning plate. Simply dropping it onto the ground will not provide enough stability. Stamp the turning plate firmly in place, even on concrete. Do not set it over small clumps of grass. Instead, use a shovel to clear a space, then tamp the plate down. When rotating the rod, maintain a constant weight on the plate, so it is not disturbed. Do not place your feet near the handle of the plate.

Wooden stake. In sandy or marshy ground or the loosely packed soil sometimes encountered on highway and railroad embankments, use a wooden stake. The stake is made from a piece of 2- by 2-inch wood cut 60-90 cm (2-3 ft) long. To use the stake as a turning point, hammer it into the ground to a firm depth, leaving about 5 cm (2 in) exposed. Then, drive a doubleheaded nail into the top, until the bottom head rests against the wood. The top of the nail serves as the turning point.



Figure 3-64.—Turning pin dimensions.

#### 3.5.2 Pacing Balanced Setups

Many errors in leveling are reduced in magnitude by balancing setups. The distance from the leveling instrument to the foresight point must equal the distance from the instrument to the backsight point within the tolerance for the survey. When a pacer is a member of the leveling unit, he or she should lay out a balanced setup and set the foresight point before the rodman arrives from the previous setup. This permits leveling to proceed more efficiently than when rodmen set the turning points.

To lay out a balanced setup, count paces from the backsight point to the next position for the instrument and make a mark. Then, take an equal number of paces to the spot where the foresight point is to be set.

Paces need not be of any specific length; rather, they must be consistent. A beginning pacer (or rodman) should mark a 50-meter course on level ground and pace it several times, striving to develop consistency in the number of paces taken. Most individuals take paces about a meter in length; therefore, to make a change of less than 5 m in the length of a setup, take the number of paces equivalent to the change.

Some individuals find that they routinely pace "long" (foresight distance longer than backsight distance) or "short" (foresight distance shorter than backsight distance). Learn to compensate for such tendencies, to prevent accumulating a large total of imbalances during a section of leveling. For this reason, do not move more than one setup ahead of the rest of the unit. If the observer or recorder alerts the pacer as soon as the total imbalance for the section exceeds the tolerance for one setup (see table 3-1), the pacer can reduce the total gradually instead of making one or two extremely unbalanced setups.

There are several ways to mark positions for the instrument and the turning points. If using lumber crayon ("keel"), draw a " $\wedge$ " to show where the instrument should be placed and draw a line, circle, or "X" for the turning point. Do not use the crayon on buildings or



Figure 3-65.—Turning plate.

private property without explicit permission from the owner. Flagging may be useful for marking setups in brushy areas. In dirt or gravel, use a shovel or the heel of your boot to scrape a line marking the instrument setup and to clear a spot for the turning point. Do not use spray paint for marking setups. In general, the leveling unit should leave little or no evidence of its passing.

In addition to marking setups and setting turning points, use the instructions provided by the mark setting unit in the log to route the unit to the next control point. Consider both the observer's line of sight and terrain irregularities, such as slope and brush, when laying out the route. Slopes may require shorter sighting lengths than usual; a hand level is helpful when determining just how short to make them. To level to some points, it may be necessary to clear brush or hold branches away from the observer's line of sight. When an even number of setups are required between control points, the pacer must plan the final setups of a section carefully.

Four-person unit. When a leveling unit operates with four persons instead of five, the rodmen assume the tasks of pacing and setting their turning points. First, the observer paces to a position for the leveling instrument. Then the backsight rodman (from a previous setup) counts the number of paces from the last foresight point to the instrument. He or she takes the same number of paces to the next foresight point.

Specialized vehicles for motorized leveling have precise odometers with which to balance setups. Used carefully, the odometers can prevent the troublesome task of correcting a poorly paced, unbalanced setup. When advancing to the next setup, set the odometer at zero upon passing the backsight point, to count up the backsight distance. Upon passing the instrument, switch the odometer to reverse, to count down the distance to the foresight position.

When on foot or using a single truck, count regularly spaced objects such as railroad rails or highway markers to establish balanced setups.

Hand level. The hand level (fig. 3-66) is a small telescope in which a spirit level is mounted in such a way that, by means of a prism, the level may be viewed at the same time that an object is viewed through the telescope. When the bubble is centered in the spirit level, a reticle line indicates the horizontal line of sight.

When leveling up or down a slope, use the hand level to select the positions for the foresight rod and the instrument at the next setup. The following instructions apply to standard, 3-meter leveling rods. (See fig. 3-67.)

1. Begin at the foresight rod of the previous setup. When leveling uphill, point and level the hand level toward the next foresight position. Select a reference object intersected by the horizontal reticle line. Proceed to the object, counting paces. (When leveling downhill, pace a conservative distance to an estimated position for the next foresight.) This procedure limits the elevation



Figure 3-66.—Hand level.

difference for the setup to the height of eye of the pacer (1.6-1.8 m). With experience, the pacer can take a few extra paces to increase the elevation difference to a maximum of 2.0 m.

2. Check the foresight position by pointing and leveling the hand level toward the previous foresight rod. Looking downhill, the line of sight should pass no more than 0.5m (1.6 ft) above the top of the rod. (Looking uphill, it should be just above the base of the rod.) Set the turning point.

3. Pace halfway back to a position for the instrument. Check it by pointing and leveling the hand level toward the previous foresight rod. Looking downhill, the line of sight should be below the top of the rod. (Looking uphill, it should be at least 0.5 m (1.6 ft) above the ground level at the rod.) Mark the position for the instrument. The pacer can omit this step and simply call out the necessary number of paces to the observer.



Figure 3-67.—Pacing a balanced setup on a slope.

## 3.6 Atmospheric Conditions

During geodetic leveling, atmospheric conditions must be determined for two purposes: to correct for the effects of atmospheric refraction on the line of sight and to correct for the effects of thermal expansion of the scales in the rods. Three measurements, recorded at the beginning and end of each section, can provide satisfactory corrections. These are air temperature, intensity of solar radiation, and wind speed. To correct more accurately for refraction, special equipment must be used to determine conditions at every setup. A method for measuring temperature differences, essential to many mathematical models of refraction, is presented here. Certain extreme atmospheric conditions may require special precautions or procedures when leveling. These are also discussed.

## 3.6.1 Air Temperature, Sun, and Wind

Measure the air temperature at the beginning and end of every section, as required for the observing procedure. To measure the air temperature, use a mercury thermometer with a range of at least  $-10^{\circ}$  to  $45^{\circ}$ C (14° to 113°F), accurate to  $\pm 0.1^{\circ}$ C ( $\pm 0.2^{\circ}$ F). Mount it rigidly in a shaded, protected place on the tripod (fig.3-68). If a pair of special sensors are available for measuring temperature differences, use the top sensor. Read and record the temperature to the nearest tenth of a degree.



Figure 3-68.—Thermometer, mounted on tripod.

Record the intensity of solar radiation, as a singledigit sun code, at the beginning and end of every section. The sun code is defined as follows:

1. "0", if less than 25 percent of the setups are performed in sunny conditions.

2. "1", if 25 percent to 75 percent of the setups are performed in sunny conditions.

3. "2", if more than 75 percent of the setups are performed in sunny conditions.

If the day is clear and sunny, but all the setups of a section are made in the shade (in other words, the entire line of sight is shaded), the code is "0" If the sky is half covered with clouds, but all the setups are made in the sun, the code is "2".

Record the wind speed, as a single-digit wind code, at the beginning and end of every section. The wind code is defined as follows:

1. "0", if wind speed averaged less than 10 km/hr (6 mi/hr).

2. "1", if wind speed averaged 10 to 25 km/hr (6 to 15 mi/hr).

3. "2", if wind speed averaged greater than 25 km/hr (15 mi/hr).

Measuring air-temperature differences. The best mathematical models for computing refraction corrections depend upon measuring at least one airtemperature difference during every setup. Temperatures are measured at two different heights above the ground. The difference is computed by subtracting the temperature at the bottom from that at the top. It is usually negative during daylight hours when the Sun heats the ground, warming the lower air layers. It is positive when the air near the ground is cooler than that above it.

Use aspirated thermometers, accurate to  $\pm 0.1$  °C ( $\pm 0.2$  °F), to measure the temperature difference. A typical measuring system includes two thermistors, each shaded by a metal tube and aspirated by a small, battery-powered fan (fig. 3-69). A digital display permits readings to be made from either thermistor.

Air movement across the thermistors should be equal at all times. Check for air movement by placing your hand in front of each thermistor. A small voltmeter should indicate when batteries must be replaced to power the fans adequately. In addition, the display should indicate when its batteries must be replaced. A display battery having low voltage can cause the observed temperatures to be in error by several degrees.

Once a week, and whenever a problem is suspected, check the thermistors for consistency. Place them next to each other at the same height and pointed in the same direction. After allowing them to stabilize for 3 minutes, the temperatures should agree within  $\pm 0.1^{\circ}$ C ( $\pm 0.2^{\circ}$ F). Note the result of this check on the recording form. If the thermistors fail to check and replacements are not available, continue leveling, but do not measure temperatures with them.

Locate the thermistors 1.3 m and 0.3 m above the ground unless specified otherwise in project instructions. These heights have been selected so the thermistors can be mounted on a tripod leg. If the tripod is used by different observers, relocate the thermistors as necessary for each observer. The thermistors must be at the correct heights above the ground when temperatures are measured.



Figure 3-69.—Aspirated thermistors for measuring air-temperature differences.

Keep the thermisters inside their shields and pointed into the wind and out of sunlight at all times. Allow them to stabilize for at least 1 minute during each setup, then read the display after the leveling observations are checked. Record the temperatures to the nearest tenth of a degree, top one first. Stand away from the thermistors while reading the temperature; body heat can falsify the results.

If the tripod must be positioned temporarily in a way that lowers or raises the thermistors from the required heights, reset the tripod after the observations are complete and allow the thermistors to stabilize for 1 minute. Then, read and record the temperatures.

#### **3.6.2 Leveling in Extreme Conditions**

Do not level if the weather becomes so severe that the health or safety of personnel is threatened. Adverse weather may also affect the correct operation of equipment and result in unreliable observations. Safe working conditions are discussed in the National Geodetic Survey Operations Manual (Greenawalt and Floyd 1980).

Leveling equipment is designed to operate correctly within a broad range of temperature. However, precipitation and wind may interfere with normal operations. If precipitation is light (less than 0.5 mm/hr) continue leveling by using an umbrella to protect the instrument. Afterwards, be sure to dry each piece of equipment thoroughly before storing it. Do not level in heavy precipitation. Wind often causes vibrations in the leveling instrument which makes observing difficult. At wind speeds in excess of 30 km/hr (19 mi/hr), leveling may be impossible. In windy conditions, use the leveling truck or an umbrella as a shield. Each rodman must be particularly attentive to maintain the rods plumb. If possible, find a less windy section of line on which to continue work until conditions improve.

Observations are not reliable if made when long-period shimmer is present. It occurs most frequently when the air is calm and the air near the ground is cooler than the air above it. It is unlikely to occur during daylight hours between 1½ hours after sunrise and ½ hour before sunset, when the air near the ground is warmer than the air above it. For this reason, restrict leveling to these daylight hours whenever possible.

Sometimes the unit must level when long-period shimmer may be present, such as in calm air over frozen ground or snow, over water, or at night. In these circumstances, double run every section

## 3.7 Observing Routine

This subchapter describes the routine to be followed when leveling a line of the National Geodetic Vertical Control Network. First, general instructions are given. Second, two procedures are described: micrometer leveling and three-wire leveling Third, precautions and special instructions are given for control points. Finally, instructions are provided for releveling and closing sections.

## **3.7.1 General Instructions**

The leveling unit is normally assigned a line or portion of a line within a vertical control project. The line should be leveled by sections, each section beginning and ending on described, permanent control points. Each section is an unbroken series of setups, observed by a procedure that meets specifications for the order and class of survey given in the project instructions. Micrometer leveling is preferred by the National Geodetic Survey for all first- and second-order surveys. Three-wire leveling produces results of lesser precision; therefore, it is included here only as a second-order technique. The following instructions apply to either procedure.

Leveling the line. Plan the leveling route by examining the logs (instructions) prepared by the mark setters for the assigned line. Chapter 2 contains detailed requirements for routing lines of control points.

Establish the direction of leveling by giving the line a title corresponding to the intended beginning and ending points. Normally, a line begins and ends at junctions with previous or concurrent leveling. Sections leveled away from the beginning point and toward the ending point are termed forward runnings. Sections leveled in the opposite direction are termed backward runnings. Check the project instructions and the National Geodetic Survey Operations Manual (Greenawalt and Floyd 1980) to determine the order and class of the survey, special requirements, and changes to standard procedures. Determine if the line is to be double run or single run. Review the tolerances given in table 3-1.

When a line is to be double run, level each section once forward and once backward, under different environmental conditions if possible A good practice is to level in one direction in the morning and return in the afternoon, leveling in the other direction Alternate the starting direction each day. The two runnings of each section must close as described in section 3.7.6, "Rejection procedure." This procedure minimizes blunders and the accumulation of most types of systematic error.

When a line is to be single run, level each section once. Alternate the direction of the runnings from one section to the next. An alternative procedure is to alternate the direction of leveling from one day to the next, maintaining an equal distance of forward and backward runnings along the line. This procedure prevents the accumulation of most types of systematic error for more than one section or day. Single-run leveling must form loops with other leveling, preferably during the same field season, to enable the detection of blunders.

Double run all spurs. A spur is a portion of line that branches off the main line of leveling, usually ending at one or more points not otherwise connected to the network. The second running provides the only check on the leveling. If a spur requires only one setup, change the height of the instrument by several centimeters between the forward and backward runnings. If the purpose of a spur is to tie a previous line of leveling, a single running is sufficient, provided it closes with the previous work.

Make single-run ties at each connection with a previous or concurrent line of leveling. Connections are normally noted in the logs. To tie to a first-order line, level to three points in the line (two sections). To tie to a second-order line, level to two points in the line (one section). To tie to an area survey, with several lines crossing the new work, level to three points from the survey that span its area.

At each connection, check the new leveling for closure with the previous, unadjusted leveling, using tolerances for whichever survey has the lower order and class Elevation differences for the previous survey may be obtained from an unadjusted abstract for the survey. (See sec. 5.2.3, "Preliminary data."). If the sections do not close after one running, relevel them until they close to verify the new work and contact the project office immediately to find out if additional points must be leveled to complete the tie.

If the leveling of a line is interrupted for 2 weeks or more, establish a tie, as just described, before resuming work on the line. If the leveling is resumed within the same field season, a two-mark tie at least 1 km (0.6 mi) in length is sufficient, regardless of the order and class of survey The tie ensures that no significant movement of the control points has occurred since the leveling was interrupted.

Leveling the section The entire leveling unit must exercise the greatest possible attention to detail while leveling each section of the line. Each day check and maintain the instruments and equipment as described in this chapter. A checklist appears in table 3-4.

Explain specifically any deviation from the leveling routine. Record the following: (1) date, (2) time at which the leveling of each running began and ended, (3) personnel, and (4) equipment involved Describe any symptoms of potential equipment failure at the time they occur.

Keep sighting distances within the prescribed tolerance for the order and class of the survey. If shimmer presents a challenge, shorten the sighting distance until acceptable readings can be made. In addition, do not allow the line of sight to pass closer than 0.5 m (1.6 ft) to the ground or to any intervening object. These precautions should reduce the error caused by refraction.

Balance the sighting distances as closely as possible on every setup, at least within the prescribed tolerance. This reduces the effects of collimation error, refraction, and curvature

Keep the total of the setup imbalances as small as possible, at least within the prescribed tolerance. If the accumulated imbalance becomes large during the levelings of the section, adjust the remaining setups to diminish it gradually. Do not try to correct for it with a few extremely unbalanced setups.

Complete the section as efficiently as possible, with no breaks during the series of setups. Only properly described, permanent monuments may serve as beginning and ending points. If the section cannot be completed or if a blunder occurs, reject the data collected so far. Enter on the recording form "999" for the next setup number and write a note explaining the rejection.

Except in the case of an incompleted section, do not reject any data in the field. If a blunder is found to have occurred after the fact, document it as clearly as possible on the recording form or by letter to the project office.

After completing one section, advance without a break to the first setup of the next section whenever possible The recorder may complete the ending and beginning running records from the new setup.

Equipment	Check	Section reference
Leveling instrument		
Tripod condition	Daily	3 3.5
Tribrach condition	Daily	3 3 5
Footscrew tension	Daily	3.3.5
Circular-level adjustment	Daily	3.3.5
Collimation check,		
Nonreversible compensator	Daily	3 3.7
Reversible compensator	Weekly	3 3 7
Compensation check	Weekly	3.3.8
Clean	Weekly	3.3.5
Stadia-factor determination	As necessary	3.3 6
Leveling rods		
Condition (base plate, brace poles, housing)	Daily	3 4 1
Circular-level adjustment		
Check with instrument	Daily	3.4.3
Check precisely	Weekly	3 4 3
Clean	Weekly	3.4.3
Calibrate	Annually	343
Condition of turning points	Daily	3.5.1
Temperature sensors		
Battery condition	Daily	3.6.2
Calibrate	Weekly	3.6.2
Computer-recording equipment		
Charge batteries	Daily	3.8.2
Precondition data tapes or disks	Daily	382
Clean recording head	Weekly	382
"Cycle" batteries	Monthly	3.8.2
Clean	Monthly	3.8.2
Check programs	At start of project	382

#### Table 3-4.—Checklist for leveling

## 3.7.2 Micrometer Leveling

Micrometer leveling is the most precise procedure currently available for geodetic leveling. Readings are made directly from the instrument's micrometer, which provides more precise results than those provided by estimation in the three-wire leveling procedure.

The primary feature of the procedure described here is the measurement of two elevation differences during every setup. The first difference is measured from backsight to foresight, the second from foresight to backsight. Since two runnings are completed in opposite directions during every setup, each section is leveled twice, simultaneously. When this procedure is used for a single leveling of a line, it is sometimes called doublesimultaneous leveling. It can be used for any order and class of survey by varying the tolerances specified in table 3-1.

To provide two different lines of sight, an instrument with a reversible compensator is preferred. An instrument without a reversible compensator can be used to provide two lines of sight if, between the two sets of observations, it is disleveled, adjusted to a slightly different height, and releveled.

To provide two different readings at each turning point, double-scale rods are used. Two rods are necessary, not only to improve efficiency, but to permit nearly simultaneous observation of the backsight and foresight. The constants of the two rods must differ, to permit a mathematical check ensuring that rods are observed in the correct order. The rod-constant difference, rod 2 minus rod 1, is labeled "d" on the recording form.

During each setup, six readings are made. First, wedge and stadia intercepts are read, backsight then foresight, from the low scale of each rod. From the two half stadia intervals the sighting distances,  $s_B$  and  $s_F$ , and the imbalance,  $s_B - s_F$ , are computed and checked. From the wedge intercepts the elevation difference of the low scales,  $\Delta h_I$ , is also computed.

Second, after changing the line of sight, wedge intercepts are read, foresight then backsight, from the high scales of each rod. From these two readings a value,  $\Delta h_H \pm d$ , is computed. To obtain the elevation difference of the high scales,  $\Delta h_H$ , the rod-constant difference, d, must be added or subtracted, depending on which rod is observed as the backsight. Then  $\Delta h_L$  and  $\Delta h_H$  are compared.

Balancing the sighting distances should reduce the effects of collimation error, moderate refraction, and curvature to immeasurable amounts during the setup. However, error in the pointings may still cause a slight difference between  $\Delta h_L$  and  $\Delta h_H$ . The difference,  $\Delta h_L - \Delta h_H$ , is compared to a tolerance to check for a value greater than might be expected as a result of random pointing error. This comparison is called the reading check:

$$\Delta h_L - \Delta h_H \leq$$
tolerance.

To determine a suitable tolerance for the reading check, the other tolerances for the leveling procedure must be considered. If the collimation error does not change between the low- and high-scale observations, the tolerance may be derived from the standard deviation of a single pointing, which, in turn, depends on the sighting distance and the optics of the instrument. With a sufficiently powerful computer-recording system, a formula might be used to compute the tolerance appropriate to the actual sighting distances of each setup. Usually, however, a single tolerance is derived, appropriate to the maximum sighting distance permitted for the order and class of the survey. The tolerances given in table 3-1 are of this type.

If the collimation error is different for the low- and high-scale observations, as is the case with a reversible compensator, the setup imbalance affects the tolerance for the reading check. For example, the difference in the collimation errors for position one and position two of the Jenoptik NI 002 affects  $\Delta h_L - \Delta h_H$  by an amount that depends on the setup imbalance and  $C_1 - C_2$ . Since the angular difference,  $C_1 - C_2$ , changes somewhat unpredictably both in magnitude and sign, the tolerance for the reading check must be derived from the maximum setup imbalance permitted for the order and class of the survey and the maximum magnitude of  $C_1 - C_2$  in addition to the standard deviation of each pointing.

After satisfying the reading check, temperatures are recorded so a refraction correction can be computed and applied to the data before adjustment. Then, the two elevation differences are averaged to obtain the elevation difference for the setup:

$$\Delta h = 0.5 \times (\Delta h_L + \Delta h_H).$$

As each setup is completed, the new difference is added to the sum of the previous differences. Thus, at the end of the section, the total elevation difference between the marks is obtained. For convenience when setup data are recorded on paper,  $\Delta h_L$  and  $\Delta h_H \pm d$  are summed separately. The elevation difference is then computed at the end of the section.

Though this leveling procedure is designed to prevent blunders and the accumulation of systematic error, both may still occur without warning. Note, for example, that there is no check for movement of the turning points between setups. As with any procedure, strict attention must be paid to the precautions and guidelines presented in the rest of this chapter.

Instructions. The following instructions apply to a properly adjusted leveling instrument with a micrometer, used with a pair of calibrated, double-scale rods. If the instrument does not have a reversible compensator, substitute the following procedure whenever instructed to change the position of the compensator: Dislevel the instrument using the footscrew closest to the foresight rod, then relevel it using the other two footscrews. If possible, record the observations in a computer and, if required, prepare a backup recording form (NOAA Form 77-82, "Geodetic Leveling") The instructions presented here include the procedure for using NOAA Form 76-191, "Geodetic Leveling Micrometer Observations." Use a separate recording form for each running. (See figs 3-70 through 3-72 for sample records.)

$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Lun 1 22	Cent 1		241				TAT	17					
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	DEAL NO A	VND DA		a Tri	7	APEN	0. /	<u>v m</u>	<u>الا ا</u>	_	0			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	PROJ. NO. /		1	<b>r</b>		AGE		1	0	F	đ			
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	51.0	51		2 2	51	\$ 								
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		YH. MO.		ZONE	INS	T. NO		110		. 1	22	ioe Tarla		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	• 4 0 •	8004	04	20		4	5	65	<u>//</u>	/	• Z	33		
NOD NG. 1       CODE       ROD NO. 2       CODE         I       I       I       I       I       I       I       CODE       ROD NO. 2       CODE         ST 4       ST 7       ST 8       ST 9       ROD NO. 2       I <td< td=""><td>514</td><td></td><td></td><td></td><td>51</td><td>5</td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	514				51	5								
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	ROD NO. 1	0101010		E /   4	80	D NO. 2	। नि	00	1	7.		TIL		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	6	1917 212	- 2	10		12	Щ	717	4	b	· 3	1/16		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	51 0	IHT TUT		ST B				ST 9	Ro	P		Ľ∕⊧-		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	1200	add	100			নন		<u></u>	1	~		++		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	1200	6006	30	-10	Ľ	00	<u> </u>	피우	Ļ#	L				
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	NO JIIMW - V	IAPE - 9				81001		10	<u>1</u>	_	07			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	STO	57 1	DESIGN	ATION										
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		PROM:		M	19	R I	93	15						
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	• 4 1 •													
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		51 2	DESIGN	ATION										
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		7 7 1 1 17	5	32	26	01	98	10						
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		ST 3	5T 4	ST	5	57 6	157	7			1	Баск		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	J	7145	TENP	<b></b>			١ <u>, </u>	76			E			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $			1	7		307	-	10	1	~		tah		
Image: State of the s	0	0 0 0 0	드슬		۲Ľ	╘╼╂ᡬ	P	90	7	2	410	181		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	U-WHITE ON	( TAPE - 7	7.		I		<u> </u>		1		L	~ ~		
$\frac{ 1 }{ 1 } - 5 \cdot 7 6 4 3 5 / \cdot 2 8 / 2  $		~-	· · ·								┍ー└┘	.5.5		
<u>III - DI · / Юн- J Э / · / D / / / Д В</u> ST 0 ST 1 ST 2 ST 3 ST 4 ВLOCК			1	26	7	. KCL	ia		10					
131 M 831 1 131 Z 131 3131 41 1 16400K		$\frac{-101 \cdot 17}{57}$	67 2	23	4	- 14	Ø	<u> 1</u>	14	<u>م</u>				
	DATA	TINE	TENP		ы 0 и	51194		as.	<b>1</b>	9 F C		12T-		
		alalac		1 1		200	h	ul-	1	6		12-1-		
					Ц	LUĽ		<u>11</u>	12	2	2010	60		
DEMARKS REMARKS	W - WHITE ON	DEMARKS		<u>ει /</u>	50									
COSAL and Cost and Cost	<u> </u>	C 0C	1			naminis								
Wrong Dron M 198 to checkpoint:	Wra	ong oron	- OK	7		M 198 to checkpoint:								
tape. "2212" 382.31	tap	e. "2212	4			382.31								
should be "2211." - 371.02	sho	uld be "	2211.	"		- 371.02								
+ 11,29									7	-	11.2	19		
800403: - + 11.41						800403: - + 11.41								
- ,12 hor						12 how								
× 5										x		5		
		7		1	11 <	0694	o Y LI		- 00		D.6	U 111		

GEODETIC LEVELING

SUPERSEDES NOAA FORM 77-42 18-79 WHICH SHOULD BE DESTROYED.

Figure 3-70.—Micrometer leveling, backup form for a computer-recording system.

1. EQUIPMENT RECORD: At the start of each day, or at any change of the observer, equipment, or collimation factor, prepare the survey-equipment record (line \*40\*). Include the date, the instrument code and serial number, the rod codes and serial numbers, the local time zone and time, the type of temperature units, the collimation factor at the last check, and the initials of the observer, recorder, and rodmen. CAUTION: Each work day check all serial numbers against the actual equipment used. Also, check that the record of the collimation factor is correct. 2. Establish a balanced setup with the backsight rod plumbed on the beginning control point of the section, the foresight rod plumbed on a turning point, and the instrument set in line between them. Check for parallax Using the instrument, check that the rods are plumb.

3 BEGINNING RUNNING RECORD: Immediately before leveling begins, prepare the beginning running record (line \*41\*). Include the following items of information: The survey-point serial number of the beginning control point, the stamping on the point (or designation if no stamping exists), the local time, the air temperature, the wind code, the sun code, and the number of the rod on the control point. CAUTION: Check that the rod identified is, in fact, on the point identified. The recorded stamping must correspond to that on the mark leveled. Check and initial the surveypoint serial number entered on the form.

4. BACKSIGHT, LOW SCALE: Level the instrument while pointing at the backsight rod. In compensator position one (odd-numbered setup) or two (evennumbered setup), read the wedge intercept on the low scale. Pause slightly between the three- or four-digit rod-unit reading and the two-digit micrometer reading, so the recorder can enter the decimal correctly. The recorder should call out any uncertain numbers for the observer to verify. Record the entire reading in column F, backsight. Record the rod-unit reading only in column B.

5. Read the intercept of the lower stadia line, estimating to tenths of rod units, on the low scale. Again, pause during the reading to indicate the decimal. Record it in column B. Compute the half-stadia interval by subtracting the stadia reading from the rod units of the first reading. With half-centimeter rods this value is the backsight distance,  $s_B$ . With centimeter rods, multiply the interval by 2 to compute the distance.

6. FORESIGHT, LOW SCALE: Point toward the foresight rod Make sure the rod has rested on the point for 20 seconds or more. Still in the same compensator position, read the low scale as before. Record the readings in column F, foresight, and column D. Compute the foresight distance,  $s_F$ . Check that the imbalance,  $s_B - s_F$ , is within tolerance for the survey. If it is not, relocate either the foresight rod or the instrument and begin again at step 4. Compute the low-scale elevation difference,  $\Delta h_L$ , by subtracting the foresight from the backsight.

7. FORESIGHT, HIGH SCALE: Still pointing toward the foresight rod, change the compensator position (or dislevel and relevel the instrument). Read the high scale. Record the reading in column H, foresight.

8. BACKSIGHT, HIGH SCALE: Point toward the backsight rod, in the same compensator position, and read the high scale. Record the reading in column H, backsight. Compute  $\Delta h_H \pm d$  by subtracting the foresight from the backsight

×.	<b>#</b>	For	¥ 74	191		******																,	ATIONA	LOCE	ANIC AN	U.S. DI O ATM	PARTME	NT OF CONNERCE	PA0	ž
						-						j	ICRO	GE Dhe	ODE		LEY	ELIN ATIO	IG NS (	∆h)	EX	AN	1PLE	5	First	Orde	er, Cia	ss.I		1 or 1
F	4		T ya	The	o. bi	4	ODE	ната	UMEH.	SER	IAL.	HO B	l co	DE I	Rot	-	AL N	0.	cor	DE	Rot	-	IAL NO.	- <u>T</u> -					12	TINE
F	4	+	80	17	17	42	3/		5	80	2	14	311	6	Π	14	89	72	3/	6	TŤ,	4	897	4			• <u>==</u> ==		1 T	TTT
F			182	0.5.5.10	INATI	OH.			·····			[		1 au	05510	NATIC	*		*****					Y	RE.			PERATURE	T.T	- 45
	2	36	Ŀ	TOP		0.2	1	148		_		12	32	L	м	300	5 1	979	, 			Ļ	R 74	20		र्डि		2 1780	7	CWS
	2	Ţ,	NCK	Γ	Ç.	)	<u>اب</u>		Ľ	Ę			LOL		Ľ		4	ę			HID		LE RE		A.		₩ A	new.	ARKS	
Π	7	315	7	T	5	1.4	15	5	T	53	0	в	35	7	72		10	2.Z	4	8	191-	- 9!	53	İΤ	812	7	-03	Reading che	ck=.	108 ha
<b>—</b>	Ť	30	516			Ţ	1	2:0				Ŧ	25	5	1					F	84	8	26	IT	111	П		Imbalance	±2.	0 m
		·				Ø	5	5	5 . =	- /	7.7													d.	+21	· · · · · · · · · · · · · · · · · · ·	1	Section imb	here	e * 40,
	2	37	2	1.	5	25	25	0.		51	0	в	37	2	80		1/2	1.8	8	1	22	16	97	14	428	3	+ 03	1		
-		31	19!	٩,J	10	13:9	1/9	<u>o e</u>	Ц/	04	0	F	25	' <i> 0</i> ''	7[2]	11	212	4!/	2	ľ	19	14:	114	Lļģ	214! /	0	+!02			
പ്പ	3	317	9	ÍΤ	5	25	25	4		< 1	Ы	8	32	9	9 <b>1</b> 61	ŦŦ	A2	48	9	8	191	77	79	TΤ	038	0	+09	b		
5	-	32	6	11	T		20	10			f	F .	25	4.	47	1-1	~	Ţ	H-	F	86	, 7!	39	t-t-				1K		
E										71	5						~							4	* <u>z</u> 7			]		
	3	37	9		5	2.5	25	4		51	0	B	37	9	35		/ Z	4.8	1	8	2	71	20		03.8	0	+01			
	_	3 2	6.5		15	6.4	20	110		55	2	F	2	4	54	Ц	314	89	3	Ir i	81	17	40	LĘ	<u>12 7:9</u>	0	[+!0]3			
Ь	-	7	10	1 1	Te				<b></b>	7		-	Tala	- a	नग	<b>1</b> 1	गंव	E C	ed 1	ыт	Tal	10	ma	ГT	5			{		
μ	7	<u></u>	ian,	+	77	9.2	14	20		24	2	-	1713	2	10	╉	10	31		,	12	P	60	┢┤┧	820	2	107	4		
		SIX.	<u>.</u>	11	<u> - 1</u>		1/1/	1	L	70	.3	L	1815		-1-1	<u> </u>	<u>[]</u>			F. I	152	<u></u>	<u></u>	<u>-</u>	17	<u></u>	0.017			
đ	5	34	6	П	5	45	26	1		51	17	8	34	6	74	Π		$\overline{\cdot}$		в	П			П						
Ē		29	1/5				20	99				<b>r</b>	26	1	30		Ι.			P	П			Ш						
		_	171-	•			1.21.2		·····	+ 3	4	~~	2.0	भ रिज्ञान	<del>a a</del> r	····	-	1.		<b>1</b>	1.00	101	20	<b>r</b> -r-		1-1		4		
μ	5	र्भूट	6	+	5	45	1216	4		<u>ş,</u>	0	5	14	6		+-+	Ť	4.8	3 -	8	나라	1	뒤뷔-	H	65.8	2	-02	3		
<b>-</b>	i	47	17:5	1_1	216	[ <u>5:2</u>	1210	6:0	ЦĘ	<u><u><u></u></u></u>	<u>.</u>		146	1/	17	4	210	6 7	<u> </u>	I	<u>1er</u>	1	744	ГĠ	27	10}	17. <u>0</u> 0	BH VONTO	1	
	71	ষাদ	0	ान	k	00	105	7		<u>5</u> 2	6	8	34	0	14	T	30	25	3	8	1914	74!	Z 8	T	234	18	+ 015	Used I-met	11 50	ale.
1	Ť	317	tole		31	512	10	9:0	3	16	0		lois	7	an	T,	86	94	6	P	6	, d	80	8	69!3	6	010		-	
E	K		*****	+	31	60		فيتتنب عيه		- 2	.0	£	13.	0	8 m		<u>-</u> .	میکنند. 							- I' M	Pan .	0H =	+ 869.410%	c##	
		T	ŢŢ	П	63	12	d.							[]		П				в	$\Box$							200 1	cm/x	<u>H.</u>
L	_1	L	Ц	<u>IX</u>	0	01	×41/	H				۴	LL	Ľ		Ш			<u> </u>	7	Ш							+ 4.34705	m	
-	-1	·	<u>_</u> 5	<b>1</b>	0.1	15	tm I	T T T				<b>.</b>	т"т"	TΤ		- <b>1</b> -7	<b></b>			тт	17	<b></b>		TT	<u></u>	1 1	<b>1</b> -1-1-1-	1		
μ	-		╀┼╴	╞┼		$\left  \cdot \right  =$	┝─┝─	$\left  \cdot \right $			$\vdash$	<u>-</u>	┿┿	+	╋	┥┥	-+-	-,		╊	╇	+-		┠╌┼╸	╋╋	┿	┠┼┝	1		
1			1				ii.	أستلك		1	·	L	_ل_م	1.		بليبية.	لىك		L.	سينا	سأسسك			L		.ii		<u></u>		

Figure 3-71.-Micrometer leveling, half-centimeter rods.

AGAA FORM 74-191		U.S. D	EPARTMENT OF CONMERCE PAGE
1-11	GEODETIC LEVELING	HATIONAL OCEANIC AND ATM	OSPHERIC ADMINISTRATION
	MICROMETER OBSERVATIONS (A	h) EXAMPLE First	Order, Class II / or /
+ 4 0 + YR. HO. DAY CODE INSTRUMENT SER	ALNO N CODE NOD SERIAL NO. CODE	ROD SERIAL HO.	¥ 7/¥Ľ
· · · · 801114231 580	210316 10832231	6 108324	
PROU DESIGNATION	TO DE DESIGNATION		TEUPENATURE WAS BEAVEN
40/3 B 301 1967	6012 M 31 1930	5/302/3300	153 16002 CWS
			Δ
UPP BACK I PORE I	BACK/FORE E	BACK/FORE E	Akie REWARKS
1/155 24.0/64 24	10 15520 - 896	a <u>95191 - 1894</u>	- OZ Reading check=+ . 10 cm
		* [4]7[0:8[5]	Timbalance == 3.0 m
21/48 225/46 23	40 1/4856 21/3	45484 1210	403
1255 4651226 47	5 1 4 6 4 3 - 6 8 3	+ 4 4 Z 7 4 - 6 8 4	701
-0	9 x 2 = -1.8	-10	
3/88 240/73 23		<b>a</b> 48455 416	
		- FIBU: 3[7] 1-1 [2:6]8]	14:02
6 14 1 6 4 1 1 / 18 6 / 6 / 1 1 / 18	0 1/64/211 11111		
1454 1450	16109		
+2.	xt= 5.2>5.0	-10	
4164 171162 177	0 16465 264	<u>471/15 1265</u>	-94
1/14/6:71 18/7:61/14/5:01 18/7	$ S  ^{1}   / G 2!O /      Y!Y 8   $	- 10	<u>H:01/1</u>
5 5 147 225 157 22	5 1/47/67 - 987	1444.08 - 1973	I-I/A P
1245 1345	1 5754	- 46381	
0.		+/0	
	5 = 14766 - 989	B 44398 - 1988	
	.0	+10	
6/66 234/43 23	01. 1/6625 2262	47275 3263	-101
1426 13351 200 133	0 143.63 22.7/	+ + + 0! 1 2 2272	-101
+0	5 X Z = 1.0	- / 0	Palas True culled
	┉╏╝╎╶┥╌┥╌┥╴┥╴┥╴┥╸┥╸┥╸┥	╸┼╶┥╶┥╴┥╴┥╴┥╴┥╴┥╴┥	Abaman Jond punch
┠╍╍╇┋╪╹┊╌╋╧╋┥╧╎╋╝╋╌┿╧╋╧┥╧┿╧			the force igni print

Figure 3-72.-Micrometer leveling, centimeter rods.

9. READING CHECK: Compute  $\Delta h_L - \Delta h_H$ . To do this, if rod 1 is the backsight rod, add d to  $\Delta h_H \pm d$ . If rod 2 is the backsight rod, subtract d from  $\Delta h_H \pm d$ . Then subtract  $\Delta h_H$  from  $\Delta H_L$ . Compare the result to the tolerance for the survey. CAUTION: Do not move the backsight turning point until the reading check is complete. If the point is moved, even only slightly, the data collected so far must be rejected and the running of the section must be begun again at step 2.

10. If the readings fail to check, begin the setup again at step 4. If they repeatedly fail to check, try one or more of the following procedures:

(1) Check for parallax and remove it if present

(2) Check the plumb of the rods and the leveling of the instrument, including the adjustment of the circular level.

(3) Check that the reversible compensator is correctly positioned

(4) Reposition the instrument to balance the setup exactly.

(5) Shorten the sighting distances by repositioning the instrument and the foresight rod.

(6) Check that the rods are being observed in the correct order (backsight firs).

(7) Check that the rod-constant difference, d, is entered with the correct magnitude and sign.

(8) Check that the starting rod for the section was identified correctly and that the current setup number is correct.

(9) Make a collimation check (sec. 3.3.7).

(10) Make a compensation check (sec. 3.3.8).

(11) Consider the possible presence of unusual environmental conditions (sec. 3.6.3), such as long-period shimmer. Return to the section when conditions improve.

11. After obtaining a reading check, record the air temperatures for the refraction correction if the appropriate equipment is available (sec. 3.6.2). Similarly, record temperatures of the rod scales if the appropriate equipment is available. With a computer-recording system, check that the temperatures are reasonable.

12. Check the accumulated imbalance for the section. If it is more than the tolerance for one setup, alert the pacer or rodmen to adjust the next setup or setups accordingly. When recording on paper, sum the backsight and foresight distances separately; then compare the overall imbalance. Also, sum the low-scale and high-scale elevation differences separately; at the end of each evennumbered setup the differences should agree closely (though not necessarily within the tolerance for the reading check).

13. NEXT SETUP: Leaving the compensator in the same position, advance to the next setup. Leave the foresight rod (and turning point) in place to become the next backsight rod. Move the former backsight rod and turning point to the next foresight position. Set the instrument on a line halfway between the two rods. CAU-TION: Do not remove the former foresight rod from its turning point. Keeping it centered on the point, pivot it to face the instrument. 14. Repeat steps 4 through 13 at the next setup Remember that the backsight, low scale is observed first on every setup, but the rods, the compensator positions, and the sign of d alternate on every setup. (See table 3-5 for a summary of the procedure.)

15. As the unit approaches the end of the section, if required, adjust the sighting distances to make an even number of setups for the section. The final setup should have rod 1 on the control point and rod 2 on the last turning point. An even number of setups is not required if all graduations of both rods have been calibrated by laser interferometry (See sec. 3.4.2, "Calibration with a laser interferometer.")

16. ENDING RUNNING RECORD: After completing the last setup of the section, prepare the ending running record (line \*41\*). Include the following items of information: the survey-point serial number of the ending control point, the stamping on the point (or designation if no stamping exists), the local time, the air temperature, the wind code, and the sun code. Compute and record in meters the elevation difference of the section. Compute and record in kilometers the length of the section. Record the number of setups. If the leveling is the second running of a section, check for closure (sec. 3.7.7). CAUTION: Make sure the stamping entered for the ending point corresponds exactly to that on the mark on which the rod is resting.

# Table 3-5.—Summary of the micrometer leveling procedure

#### During each setup

I Balance setup.

- 2 Point instrument at backsight.
- 3 Level instrument and plumb rods
- 4 Read backsight, low scale: wedge and stadia
- 5 Point to and read foresight, low scale: wedge and stadia
- 6 Check sighting distances and imbalance against tolerances
- 7. Change compensator position or dislevel-relevel
- 8 Read foresight, high scale: wedge
- 9 Point to and read backsight, high scale: wedge.
- 10 Check low- and high-scale elevation differences against tolerance
- Read temperatures.
- 12 Check accumulated imbalance against tolerance
- 13 Move to next setup

#### During the section

Number setup	Scale observed	Compensator position
1	low	•
	high	**
2	low	**
	high	•
3	low	•
	high	**
•	•	
•	•	
•	•	

(

## 3.7.3 Three-Wire Leveling

Until the advent of leveling instruments equipped with micrometers, three-wire leveling was the most precise method for measuring elevation differences. However, without a micrometer, scale readings must be estimated and only one elevation difference can be measured efficiently during each setup. The method described here can provide differences with a precision sufficient for second- or lower-order surveys.

The instrument need only provide a single, consistent line of sight during each setup. Two nearly identical rods are used to permit nearly simultaneous observation of the backsight and foresight. Only one calibrated scale is necessary on each rod; however, a second uncalibrated scale on the back of the rod should be read as a check for blunders. To estimate readings with the greatest precision, the calibrated scale is graduated with blocks, preferably in a checkerboard pattern. (See sec. 3.4.1, "Scale.").

During each setup, eight readings are made. First, the intercepts of the upper, middle, and lower reticle lines ("wires") are read from the calibrated scale of rod 1. Then, the intercept of the middle line with the back scale is read. The four readings are estimated to the nearest tenth of a rod unit.

Two checks are made. To check the internal consistency of the three precise readings, the half stadia intervals,  $s_U$  and  $s_L$ , are computed and their difference is compared to a tolerance. To detect gross blunders of decimeters or meters, the reading of the middle reticle line is compared to the back-scale reading. Since the precise readings are made nearly simultaneously, a consistent error in the first or second digit might otherwise go undetected. This is particularly likely when leveling moves from flat to sloping terrain and the readings increase abruptly by a full meter.

Second, four more intercepts are read and checked from rod 2. The sighting distances are computed by summing the half stadia intervals for each rod. Then, the imbalance is computed and checked against the tolerance.

Rod 1 is observed first during each setup. Thus, the backsights and foresights are observed in alternate order on alternate setups, to reduce systematic error. This is especially important when using a compensator instrument, to prevent the accumulation of systematic error due to consistent variation in the collimation error. (See sec. 3.3.8, "Compensation check.")

The elevation difference for each setup is the difference between the mean of the three backsight readings and the mean of the three foresight readings. It is not usually computed until the entire section is completed.

With three-wire leveling certain blunders cannot be detected. For example, a mathematical check to detect transposition of the foresight and backsight observations is not possible. Neither is a mathematical check to detect disturbance of the instrument between the backsight and foresight observations. Instructions The following instructions apply to a properly adjusted instrument without a micrometer (or with the micrometer locked in position), used with a pair of calibrated rods, each having one calibrated scale and a check scale. A spirit-level instrument must be shaded with an umbrella

Record observations on NOAA Form 76-189, "Geodetic Leveling Three-Wire Observations." (See figs. 3-73 and 3-74 for sample records.) If observations are recorded in a computer, maintain a backup recording form as required by project instructions.

1. EQUIPMENT RECORD: At the start of each day or at any change of the observer, equipment, or collimation factor, prepare the survey-equipment record (line \*40\*) Include the date, the instrument code and serial number, the rod codes and serial numbers, the local time zone and time, whether temperature is measured in Fahrenheit or centigrade units, the collimation factor at the last check, and the initials of the observer, recorder, and rodmen. CAUTION: Each work day check all serial numbers against the actual equipment used.

2 Establish a balanced setup with the backsight rod plumbed on the beginning control point of the section, the foresight rod plumbed on a turning point, and the instrument set in line between them. Check for parallax Using the instrument, check that the rods are plumb.

3. BEGINNING RUNNING RECORD: Immediately before the leveling begins, prepare the beginning running record (line \*41\*). Include the following items of information: The survey-point serial number of the beginning control point, the stamping on the point (or designation if no stamping exists), the local time, the air temperature, the wind code, the sun code, and the number of the rod on the control point. CAUTION: Check that the backsight rod is, in fact, on the point identified. The recorded stamping must correspond to that on the mark leveled. Check the survey-point serial number entered for the point.

4. ROD 1: Level the circular level on the instrument while pointing at rod 1. With a spirit-level instrument, use the tilting screw to center the bubble precisely in the tubular level. Read the intercepts of the three reticle lines with the precise scale, top to bottom. Estimate each reading to tenths of a rod unit.

5. When rod 1 is the backsight rod, record the readings in the backsight column, column B in the example. When rod 1 is the foresight rod, record the readings in the foresight column, column G. CAUTION: The recorder should watch to ensure that rod 1 is observed at the beginning of the setup. Be sure the readings are recorded in the correct column because there is no mathematical check for this type of error.

NDAA PONU 34-189 IL-773 U. S. DEPARTMENT OF COUNERCE NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION													
				C THRI	EODETIC L	.E 35	veling ervations EX	AMPLE	Second Order	, Clas	sI	_/0F/	
1	0 * YR. /40. /0	AY CODE INSTRUM	ENT SEMIAL HO.	CODE	ROD SERIAL	нı 17	0. COOE NO	00 SERIAL HO.				2 7148	
7 R	ON DU DESIGNAT	052477			DESIGNATION		A (AGG	2 <u><u><u></u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>	INE EHO	TENPE		SERVER	
R	0/1/BM 800	S ISOL SIAFF			33 736	<u>.</u>	<u> </u>			6 ZI		COMMENTS	
	UP <del>r</del> en	Careksie	<u>"' (b) -</u>	Ð	-(E)-	Π	Userea	- U)	J J	r 🕑	<u>г-Ө-</u>	Reading	
SET IP	MIDDLE	NEAN	BACK CENTER	\$U 5.	(11+1)		MIDDLE	HEAN	BACK CENTER	5.U	(1)+11	Imba lance	
	Σ	Σ			Σ		Σ	Σ			Σ	= = 5 m x 2004 m/	
				11	11[.					<u> </u>		1000 km	
	478:7	474.07		14			2874	276.51		4.8	63=-0.4	- 333	
	469.4			4.7	9.3		2717			4.9	9.7		
<u>}</u>	19222	47407			1.1.7.3		8297	2761517	/		9.7	1	
12	238.8	238.80		13.0			425.2	425.11		12:1	11.7	}	
	225.8			130	26.0		4/30			12.2	243		
<u> </u>	2/38:6	- 701.74			- 34.0		2/052	1 17 10 17 :714	·		- 35.3		
	- 433.4	= +////3	Acres 1	1!	= <u>/</u> 3						= 69.3		
					x 3 3 3.	2	╞╌┼╌┼╌┼╌┼╌┝	╏┈┍┈┯╌┰╌┰╌┲╴┲╴	4	<u> </u>	x 333	1	
<u> </u>	÷ 200	3 Ment 1			- 200	Â		╽			200000	hem/km	
	AH=+ 0.0	556 .		TT.	ED- 2.2 m					T.	S=0.12 km		
	┍┿┽┽				╶┼┉┼╍┼╺┿╍		┝╼┿╼┾╼┾╼┿╍		4	┟└└	┠╂╂╞╬┉	-	
	+++++	<u>}iii</u>		<u>├</u>			┝┼┼┼┼┼	┟╼┵╌┵╌╌┖┈╩┉┶		<u> </u>	╏┉┶╌┷╍┷╺┷┉	1	
											<u> </u>		
1	┝╼┨╾╂╼╋╍┨╼╄╼╬╼╵			┝┷┷┥	┽┽┽┼		┝┼┤┽┽┾┿		-	┝┷┷	<del>╞╶┝╶┝╶┝</del>	-	
Ĺ												1	
L	╺╌╆╌╄╌┦╌┠╴╪╌	┟┷┷┹					┝┼┼┼┼┾┊╸			;		ļ	
	<b>╶┼┼┼┼┼</b>				╋		┝╾╪╼┥╸╆╍┝╼┢╼┇╼		-				
		Fi	gure 3-73.	.—Th	ree-wir	e	leveling, ha	lf-centimet	er rods.				
HOAA (4-7)	, ≠ciste 76_189 1)					_		NATIONAL OCE	U. S. DEFAR		F CONNERCE	PAGE	
				-									
				G Thre	EUDETIC L	.œ' 351	ERVATIONS FX	AMPLE	Same A.L		• e e T	/0F /	
E	<u> </u>								Secona UYOU	r Ch			
	0 * YA. /m. /b.	4 7 / 7	10827	31/7	AINSE CON	4	2.83/2	D SERIAL NO.					
F-	00 0E10044	16-16-14-14-14-14-14-14-14-14-14-14-14-14-14-	<u>, 1, mai an internet services</u> 		DESIGNATION				IME E	TENPE	AATURE		
<u>D</u> D	01 X 3	350 1957	100	oz	X 35	0	RESET 1980	र रेडे हैं।	16100	/34	/ 3 3 /	1 RUB	
F_		PACKSIG	нт	,		_		FOR	ESIGHT		r	COMMENTS Readi-	
SET	MIDDLE	MEAN	BACK CENTER		[			MEAN	BACK CENTER	5 <sub>11</sub>		check= 3cm	
<del>ا</del>	LOWER			S.	10+1)	ł	LOWER			۔ ۲	(U+1)	Imbelance	
$\vdash$	Σ	Σ			- <u>γ</u> - <u>Σ</u>	4	Σ	Σ	+			1255 m x 100 cm/m	
	1638	┝─┶┶┷┷┷┥					1370		┨╍┖┶└╧		┟┷┶┶┷┷	500 CM	
1/	1549	154.90	50.8	89		ļ	1282	12820	420	P.P	A5++0.2	= 332 SF	

Ć

Ľ	1_1	/12	14.17	1 1 5 4: 9 0	1 20:0	7 7			L			12	8.2	I	128:20		4212	1 8.8	L	As ++0.2	2+10
		14	60			819		178				11	94					8.8	[ ]	17:6	
		46	4.7	15490				178			1 3	318	4.6	Г	12820				TT~	17.6	1
	tΠ	1/17	00					and since the second				12	50	1	يستعدم والمسترقين والمستري			1	1		1
17	++		1	TUUda		lole				-+-	++	17	210	┢	112000		T latai		1		
<u> </u>	┼┉┥	1/10	42	1 1/10/100	1 131310	1013						13	6.0	<u> </u>	11/12/0-717	┝╍└╴	1 7 7	7.0	ł		
	ff	- 1/15	3:0		4 1	18:5	++	17.0		_	<u>↓</u> /.	12	6.9	1_	T T T T T T T T T T T T T T T T T T T			19.1	4-4-	11811	1
ļ		94	22	31640				34.8	Ļ	_	تلب	19	2.5	L	26417					357	1
		116	5.7								1.12	5	4.9						ł		
3	Π	115	6.9	156.87	513	8.8			I	Τ	$\Box$	4	5.9	Г	14590	1	# 7.5	9.0		-0.3	
	П	1/4	80	[		8.9		177	ľ	Т		13	6.9	T				190	П	180	
	П	141	98	47327				51915		1	1/2	3	0.2	t	A10.07				T	537	1
	11	1/6	8:0	<u>]</u>	1				ľ	í	11	12	49	t				+	<u>†</u>		1
	H	15	911	1/59.07	521	189			h	+	t t	17	67	t	11161	1	382	182	1		
	† †	1/5	ht,			90		100		+-	Ηť	10	94	1	1 17.11 (27.07.17			1019	ht	Li/e	
	H	189	710	632 34		1/		7/1/4		-†-		10	517	1~	57674			1.000	- <u>†</u> -	707	1
	Li	1/5 8	517	- 52674				70 2	ł	+	$\uparrow$	<del>ľ</del>		┢	1 1312 01 113			1	┢╧┷	70 4	
	E	120	Fié			-1-1-		10.2	ł		╈			┢╸	TTTTT		TTTT	1	1	1407	
l	1-1	1+1211	10.7	-11/03.60	CM ] ]		-	0.2	ł		┾┾	+		╀		ļ		- <b>  i</b>	1	1-1-1	1
	크	╺╺┟╼┝╼┝╍	2				X	2 Z			<b>_</b>	-L-I		1_					지권	12	ļ
	=	1-10	5.6	00				664	4	1				L				= 4	616	79,2	C/H
	÷	110	0	Cra/m			÷	100		Cm	Im		L					i ÷	10	0000	ZM/KM
	1	H= +	1.0	560 m.			ΣΔs=	0.7 -						1		Τ			15=0	0.47 km	
	П		ΓŢ							1	TT	Г		Т					$\square$		]
	П		ļ						ľ	1	$\Box$	]	11	1					TΤ	111	]

Figure 3-74.— Three-wire leveling, centimeter rods.

6. READING CHECK: Compute and record in column E (or K) the half stadia intervals. The upper interval,  $s_{U^3}$  is the difference between the upper and middle readings, and the lower interval,  $s_L$ , is the difference between the middle and lower readings. Check that  $s_U - s_L$  is no more than 0.3 rod unit. If it exceeds this tolerance, reobserve rod 1, beginning at step 4.

7. If the tolerance is satisfied, compute the full stadia interval,  $s_U + s_L$ , and record it in column F (or L) Compute the sighting distance and check that it is within the tolerance for the survey. For convenience, if recording on paper, convert the tolerance to a permissible stadia interval in rod units, and use this value to make the check. For example, if the rod units are centimeters, the stadia factor is 333, and the tolerance for sighting distance is 60 m, then the tolerance for the stadia ininterval is 18.0 cm [(60 m×100 cm/m)  $\div 333=18.0$  cm]

8. Compute the mean of the three readings The mean may be computed quickly by examining the last digits of  $s_U$  and  $s_L$ . If  $s_U$  is three tenths larger than  $s_L$ , add 1.0 to the middle reading. If  $s_U$  is two tenths larger, add 0.7. If  $s_U$  is one tenth larger, add 0.3. If  $s_U$  equals  $s_L$ , the middle reading is the mean. Similarly, if  $s_U$  is three tenths smaller than  $s_L$ , subtract 1.0 from the middle reading. If  $s_U$  is two tenths smaller, subtract 0.7. If  $s_U$  is one tenth smaller, subtract 0.3. The mean always ends with the numeral "0", "3", or "7".

9. BACK CHECK: Read the intercept of the middle reticle line with the back scale. Record the reading in column D. The recorder or the person holding the umbrella must convert the reading to the units of the precise scale, then compare the result to the reading of the middle reticle line made in step 4. If no blunders have been made, sum the three readings together with the sum from the previous setup. Also, sum the means computed so far. CAU-TION: Do not move the backsight rod and turning point until the data for the entire setup have been checked.

10. ROD 2: Point the instrument toward rod 2. Do not relevel the circular level. With a spirit-level instrument, recenter the bubble in the tubular level Make three readings as before.

11. When rod 2 is the foresight rod, record the readings in the foresight column, column G. When rod 2 is the backsight rod, record the readings in the backsight column, column B. CAUTION: Again, check that the readings are recorded in the correct column.

12. Perform the reading check and back check as in steps 6 through 9.

13. Compute and check the imbalance (backsight distance minus foresight distance.) If it exceeds the tolerance for the survey, reobserve the setup, changing the position of the foresight rod or the instrument.

14. Check the accumulated imbalance for the section. If it is more than the tolerance for one setup, alert the pacer or rodmen so the next setup or setups can be adjusted accordingly.

15 NEXT SETUP: Advance to the next setup. Leave the former foresight rod (and turning point) in place to become the next backsight rod. Move the former backsight rod and turning point to the next foresight position Set the instrument on a line halfway between the two rods. CAUTION: Do not remove the former foresight rod from its turning point. Keeping it centered on the point, pivot it to face the instrument.

16 Repeat steps 4 through 15 Remember to read rod 1 first on every setup, regardless of whether it is in the backsight or the foresight position

17 As the unit approaches the end of the section, if required, adjust the sighting distances to make an even number of setups for the section. The final setup should have rod 1 on the control point and rod 2 on the last turning point. An even number of setups is not required if all graduations of both rods have been calibrated by laser interferometry (sec. 3.4.2), which permits sufficiently accurate correction for index error.

18. After the last setup, compute and record the elevation difference for the section. Total the backsight readings and the foresight readings separately. Subtract the foresight total from the backsight total, and divide by 3. Similarly, total and subtract the backsight and foresight means. The two elevation differences should agree within 1.0 mm (0.003 ft).

19. Compute and record the length of the section. Total the backsight and foresight stadia intervals, multiply the result by the stadia factor, and convert it into meters. A summary of the three-wire procedure is given in table 3-6.

20. ENDING RUNNING RECORD: After completing the last setup of the section, prepare the ending running record (line \*41\*). This includes the surveypoint serial number of the ending control point, the stamping on the point (or designation if no stamping exists), the local time, the air temperature, the wind code, and the sun code. Compute and record in meters the elevation difference of the section. Compute and record in kilometers the length of the section. Record the number of setups. If the leveling is the second running of a section, check for closure. (See sec. 3.7.7, "Releveling and closing sections."). CAUTION: Make sure the stamping entered for the ending point corresponds exactly to that on the mark on which the rod is resting.

## 3.7.4 Precautions to Take at Control Points

The beginning and ending control points of a section are critical in three ways. First, the marks must be positively identified. Second, the control points must be clearly defined and properly used. Third, the stability of a mark that represents the end of a previous day's or week's leveling must be proved.

Identify the mark. Because of the proliferation of survey marks of all types and the possibility that control points may have been relocated without the knowledge of the leveling unit, each mark must be identified carefully. At the start of a section, identify only the

#### Table 3-6.—Summary of three-wire leveling procedure

1	Balance setup
2	Point instrument at rod 1
3	Level instrument and plumb rods
4	Read three reticle lines, top to bottom
5.	Check half-stadia intervals and sighting distance
6	Read back scale: middle reticle line
7	Check mean of three readings against back-scale reading
8	Point instrument at rod 2
9	Read three reticle lines, top to bottom
10	Check half-stadia intervals and sighting distance
11	Read back scale: middle reticle line.
12	Check mean of three readings against back-scale reading
13	Check imbalance and accumulated imbalance
14	Move to next setup

beginning mark; identify the ending mark when the section is completed. When the rodman prepares the control point for the leveling rod, he or she should call out the stamping to the recorder. The recorder enters the stamping, as it is called out, and the corresponding survey-point serial number as it is given in the log

Locations where marks may be easily misidentified include triangulation stations and tide or water-level stations. At these locations, marks are set in clusters, often only a few meters apart. To further complicate matters, many marks may be stamped with similar designations, perhaps bearing only one or two inconspicuous symbols to distinguish them. Level through such clusters with care. Avoid unnecessary spurs. For example, at a triangulation station, do not level separate spurs to the reference marks. Instead, level from one reference mark to the station, to the next reference mark, and then to the next control point in the line (fig. 3-75).

If an apparent mistake or duplication is found in the log, investigate and correct it, adding a thorough explanation on the recording form for leveling. Some examples follow: If a control point is not found where it was plotted on a map, explain the situation and plot the point correctly. If a survey-point serial number has not been assigned in the log, use "0000" and note the fact. If a mark does not exactly match the description given in the log, check for a similar mark in the vicinity, and be sure to record the stamping that appears on the mark actually leveled.

Sometimes the unit may find a control point that is not indicated in the log. In general, do not level to such a point unless it is already included in the national network. All control points in the national network should have been accounted for by the reconnaissance team. If, in fact, the point represents an accidental omission by the mark setter or an amendment to the original leveling route, it should be leveled. Prepare a complete description in the standard format (sec. 2.4.4) and submit it with the leveling data.



Figure 3-75.—Leveling routes through a cluster of control points at a triangulation station (a) and a tide station (b).

Identify the control point. The leveling rod must rest on a clearly defined point. On a horizontal surface, the control point is the highest point of the mark. On a vertical surface, it is at the intersection of a pair of crossed lines (the exact center of a bench mark disk, fig. 3-76). The one exception to these rules occurs if some other point is specifically defined in the description of the control point. If any other point is used, describe it clearly in the leveling records.

Center the base plate of the rod precisely on the control point. If the point is in a location that does not permit this, use the center of the back or front edge before resorting to a corner of the base plate. If a mark is very flat, use a spacer placed exactly at the center. Sometimes the concrete around a disk may prevent centering or rotating the rod. Use a spacer in this situation as well.

A spacer ("plug") is a calibrated cylinder of solid metal, often magnetized. (If observations are recorded on a magnetic tape or disk, keep the spaces away from the recording media.) The spacer is used to elevate the rod above the control point in a restricted setting, thus permitting the rod to be centered and rotated. When a spacer is used, either the same spacer or another one identical in height must be used on the turning point of the same setup. Make a note on the recording form whenever a spacer is used.

To use a single spacer, place it under the backsight rod of the setup. After reading the backsight, low scale, place the spacer under the foresight rod. After the foresight readings, return it to the backsight rod for the final reading. The height of the spacer is included in

3-60



Figure 3-76.—Control points on horizontal surfaces (a) and vertical surfaces (b).

the observed heights of both the backsight and the foresight; therefore, the extra height is canceled when the elevation difference is computed (fig. 3-77).

A pair of spacers, manufactured and calibrated to agree in height within 0.05 mm (0.0015 ft), can be used simultaneously for a more efficient operation.

Set a check point. The elevation of the control point at the end of a line segment leveled during one day must remain unchanged until leveling resumes the next day. To ensure this, set a check point wherever a day's leveling is not connected to previous leveling. The check point should be a solid point, clearly defined, and not on the same structure as the control point. Locate it no more than one setup away. Examples of check points include a marked point on a building foundatior, a doubleheaded nail driven into a telephone pole, or a marked point on a rock outcrop.

From a balanced setup between the control point and the check point, read the low scales of each rod. Write the readings in the "remarks" column of the recording form and compute the elevation difference, control point to check point. Also, write the difference on the field abstract for future reference.

When resuming work, observe a new difference between the control point and the check point, recording the difference as before. It should not differ from the first difference by more than 1.0 mm (0.0030 ft). If this check indicates movement has occurred (or that the rod was incorrectly placed on the control point), the original section leveled to the questionable point should be releveled in the opposite direction and closed.

If leveling is single run, set check points as follows: When leveling forward on the line, set one check point at the end of work each day; when leveling backward, set one check point at the start of work and set another at the end of work only if the previous day's leveling is not connected.

If leveling is double run, a check point need only be set at the end of a completed segment of line. It is practical to level the last section of the day in only one direction, setting no check point. The next day, level the section in the opposite direction, thus checking the stabil-



Figure 3-77.—When using spacers the height, e, of each spacer cancels when the elevation difference,  $\Delta h$ , is computed:  $\Delta h = (B+e) - (F+e) = B - F$ .

ity of the end mark, as well as satisfying the doublerun requirement.

#### 3.7.5 Leveling to Awkward Control Points

In the National network many existing control points are located where they cannot be easily leveled. These awkward points fall into two categories: (1) those mounted or etched on vertical surfaces, and (2) those mounted or etched on horizontal surfaces in positions requiring unusual equipment or procedures.

Locations where a control point may be found on a vertical surface include foundations and footings of large buildings and headwalls, and abutments of highway overpasses. The best way to level to a point in such a location is to intercept it directly. Another way is to use a short scale in lieu of the standard leveling rod.

Awkward locations for points on horizontal surfaces include the following: the top of a post or pedestal (too high to plumb the leveling rod properly and not large enough to support a rodman); a point less than 3 m (the length of a leveling rod) below an overhang or ceiling; and the underside of an overhang or eave. The best way to level to a point in the first location is to set the line of sight tangent to the highest point of the mark. At the second location, use a short rod. The third location requires that a rod be read while placed upside down against the point. Figure 3-78 illustrates each of these cases.

The following instructions explain how to level to awkward points.

Level directly to the point Follow the micrometer leveling procedure. (See fig. 3-79 for an example.)

1. During the setup, which includes the control point, adjust the instrument height so that when the instrument is leveled the line of sight is within one full turn of the micrometer (one rod unit) of the control point.

2. To make the low-scale reading on the point, intercept it with the middle reticle line. Read the micrometer Record the observation as "000." plus the micrometer reading.

3. To make the stadia reading, hold a short scale at the point. Read a stadia intercept on the low scale.

4. To make the high-scale reading on the point, intercept it with the middle reticle line again. Read the micrometer. Record the reading in the remarks column, and add the appropriate rod constant to the reading. Enter the sum as the high-scale reading.



Figure 3-78.—Leveling to awkward control points.

HOAA #QRM 76-191 (8-77)	L+AJTICAJ	U.S. DEPARTMENT OF COMMERCE PAGE
	GEODETIC LEVELING NICROMETER OBSERVATIONS (AL) EXAMPLE	lor/
· 40 · YR. NO. DAY CODE INSTRUMENT SERIAL	HO M CODE ROD SERIAL HO. CODE ROD SERIAL NO.	
	1115/10 1/4300 03/10 1/4300	
FROM BY DESIGNATION	TO BA DESIGNATION	
	00/2/10/ 85/ 87/2 CIAIT 71. K/1/	30/1/502 1/01/ 1/030/1 2005
SET STADIA SH BTADIA SH UPP BACK I PORE I	LOW SCALE AN HIGH SCALE BACK/FORE E BACK/FORE	Анны Ань неманка Е Анн неманка
1000 10:3310:100	a 000.519 - 309.67 a 59301	-33065 -02
0103 3000	• 3/020 • 923.66	Vertical mark rdy . SI
+ 0.3		+ 2 / Rod / Constant + 592.50
2267 430321 450	<b>26792 - 64/6 880</b> 72	- 33/0 - 06
2240 6334760 650	<b>                                      </b>	<u>FI31609:7151 F:018</u>
4780 388316 398	N 28095 - 3585 N 874 00	- 5695 00
24/2 92/2762 948	1 51680 - 39968 1 93985	-42060 -05
-1.0.		+ 2/
4276 502 900 7500	B Z7652 27626 B 89004	297.24 +01
2258 1423 8 18 18 1448	0 0 0 2 7   -1/ 2 3 43      5 9 2 8 0	-1/12131316 -017 Staff inaccessible
		Lavelie To 7 00 +F
	╂╬┧┼╍┝┽╍┰╉╶┨╼╄┥┼╋╌╦┥┥┹╏╬┨┥╋┱╅╶┊┨╌╸	+ <u>5772.50</u>
8=0.29 km Id	<u>                                      </u>	Mean AH = - 123-395 592.50
	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	Z00
		- Q. 6/698 m
· · · · · · · · · · · · · · · · · · ·		
╺┹┱╘╼╏╶┟╸┫╸┥╴┨╴┇╴┨╸┨╸┧╴┥╴┥╸╕╸┥		╊╋╋╌╪╌╄╌┇╌
	· · · · · · · · · · · · · · · · · · ·	
<del>╺┺╋╋╋╪╋╪╋╋╪╋╋╞╋╋┿╋</del> ╧╧	<mark>╋<mark>╞╶┼╌╋╍┼╍┊╶┼╶┠╍┦┙<mark>┊╶┼</mark>╌┥┥┥<mark>┥</mark>╝┊╞╍┼╍┨╸┊╶<mark>┊</mark>╶╎</mark></mark>	<del>┨╶┧╸┠╶┇╶╏╴┨╶┊╺┨</del>
		<del>I I I I I I I I</del>

Figure 3-79.—Leveling directly to a control point.

5. Note in the remarks column that the point was intercepted directly.

Use a short scale. A short scale is shown in figure 3-80. For micrometer leveling use a double scale with a constant identical to that of rod 1.

1. During the setup, which includes the control point, adjust the instrument height so that when the instrument is leveled the line of sight is as close to the control point as possible.

2. Two people should hold the scale, one at the top and the other at the bottom. Center the index line of the scale on the point as illustrated in figure 3-81.

3. Plumb the scale by observing it through the instrument and aligning the scale with the vertical reticle line. When necessary, as in case 3 of figure 3-78, support and plumb the scale with a carpenter's level.

4. Observe the scale exactly as though it were rod 1 in a routine setup. Make a note that the short scale was used.

Note that the short scale must not be used with the weighted end placed on the control point. In such a position the scale is upside down. If it is used on both the backsight and foresight of the setup including the control point, the resulting elevation difference will have the wrong sign.

Use a short rod. A short rod should be identical to rod 1, except in length. The auxiliary scale included with Zeiss River Crossing Equipment is suitable. Use it as explained in section 4.3.1, "Auxiliary scale." Read down from the point. If the rod or short scale cannot be plumbed in an upright position, it may be observed upside down. However, a special computation is necessary. (See fig. 3-82 for an example.)

1. Align the index line of the scale or place the base plate of the rod on the control point. Use a hand level to plumb the rod. If an upside-down, circular level is already mounted on the rod, be certain it is properly adjusted.

2. Follow the usual leveling routine, but on the upsidedown scale count graduations down. If a micrometer is used, read it as usual. Record each upside-down reading with a negative sign.

3. If a micrometer is used, correct the upside-down readings. (With a computer this should be done automatically.) The micrometer effectively counts up, but the negative sign implies that it counted down; therefore, add twice the micrometer value to each reading. For example, if the reading is -313.61, add  $2 \times 0.61$  (=1.22) to the value. The corrected reading becomes -312.39.

4. Compute the stadia distances and checks as usual.

5. Compute the elevation differences and reading checks as usual, remembering that the upside-down readings are negative. With the micrometer leveling procedure, the difference between the high-scale readings must be converted to  $\Delta h_H \pm d$  as follows: If the upside-down scale is the backsight, add twice the rod constant. If the scale is the foresight, subtract twice the rod constant.

#### 3.7.6 Tide and Water-Level Stations

Leveling connections from the national network to tide and water-level gages are important in two ways: The connections make it possible to refer elevations to the surface of a local body of water, and they provide a means to monitor relative changes in the levels of rivers, lakes, and oceans. Leveling to connect the national network to a gage is distinguished from leveling to monitor movement of the gage relative to the control points at the station. The latter type of work, conducted by the National Ocean Survey, is discussed in *Publication* 30-1, Manual of tide observations (Coast and Geodetic Survey 1965) and User's Guide for the Ettablishment of Tidal Bench Marks and Leveling Requirements for Tide Stations (Bodnar 1977).

A description and sketch of each station to be leveled should be attached to the log, showing the exact location of the points to be leveled and a recommended route through them. (See sec. 2.3.4, "Tide and water level stations.") Since the points are numerous and the stampings sometimes duplicated, identify the points with extreme caution. Level to the staff or gage and make a water-level measurement according to the following instructions.

Before leveling to a tide or water-level station, make an appointment to meet the station observer at the station. The station observer should be present to calibrate the water-level recorder.

*Staff.* Level to a staff by using a well-defined point on the staff to support the leveling rod, or level to the staff directly. The first method is preferred, but the second may be used if the staff is inaccessible.

Mounted on most tide staffs is a rod stop, a galvanized metal angle with a round-head bolt through the top. The highest part of the bolt is the point on which to place the rod. If possible, confirm the location of this point with the station observer. Include the height of the point, above or below the zero of the staff, in the designation.



Figure 3-80.—Short scale.



Figure 3-81.—Centering the index line of a short scale on a control point.

If no rod stop is available, or the staff is inaccessible, level directly to the staff. Follow the procedure described in the previous subchapter, using a convenient graduation as the control point. Include the graduation value in the designation (not in the readings). To make a stadia reading, count the number of staff graduations in the half stadia interval, estimating to tenths. Record this in the remarks column and convert it to rod units, then enter it as the stadia reading. For example, if the staff is graduated in tenths of feet and the rod units are halfcentimeters, then 2.3 staff graduations are counted in the half stadia interval, corresponding to 0.23 ft. Multiply this by a conversion factor, 60.96 hcm/ft, to compute the stadia reading, 014.0 hcm.

Immediately after leveling to the staff, record on the recording form the water-level height in staff units. If possible, record the staff's identification date, which should be stamped on an attached brass tag.

*Electric-tape gage.* The electric-tape gage is usually located in a small house along with the water-level recorder. Level to the reading mark, directly if possible. It is usually a horizontal line, etched on the vertical edge of a metal plate supporting the electric-tape reel (fig. 3-76). Perform the accompaning water-level measurement as follows:

1. With one hand, hold the crank handle of the tape reel. With the other hand, release the ratchet pawl and hold it clear.

HOAA (4-77)	Fan	M 78	191								••••••															,	-	HAL		-	C AN	U.1. C	E PAI	RTHE	NT OF	COM	ERCE	PAG	۲.
														ы	CR	GI			IC I BSE	EV			5 (Δ)	<b>.</b> ,	ΞX	AN	IPL	F									]		larl
		,	<i>,</i>		;																	1		<u></u>				_	1.										
	•	YR.	/*	-/	DAY TA	2	300	1**	17A	uw¤ ∡L	HT B	ERI/	717	1	20	0K		1	CRI/	- 7	o. I a l'	7 2	1002	<u>.</u>  -	noc	SEA	CTZ	10. 10					: -		<u></u>	<u></u>	-	2	TIME
<u>Fitt</u>	; .	1810	1/	1	~ 14	.4	21	21		7		121	11	1771	214			1/1			1.1	213	1/1	<b>?</b>	1.1/	771	မျင	0	<u>97</u>										
FRO	ж	<b>1</b> B M L	SES I	3 H A	T+0+	•							ļ	T	<u> </u>		OES	GNA	17 101 -	*	<b>י</b> ש						ž 🗖		14	1	ENG	Ţ	4	1.0		KND.	- *	4	SERVER
/3	1 2	<b>I</b>	^	10	<u> </u>		. 2						1/	3	/]:	<u>9</u>	<u> </u>		<u> </u>	17	~ 5	·					<u> </u>	4	10	1	43	0		12	2	/2	131 /	4	Charles
SET UP#	\$T,	NDIA ICK	Γ	\$	Ē		57 P	AD). '011	î.		s, E				LOW	5C A L (/T O	Ξ.			4	54. E					5CA K/# 0	L Ľ Re			۵	ьны E		4	нц МЯ			REMA	9×5	
17	Т	11	Π	T	I	<u> </u>	T	Т	Ţ	Т	Т	ĒΤ	- 8		3	13!	61	П	T	Т	L i			F	96	्रान	01	Π	E.	- 50	H.	I	Π	Τ	Rod	1 00	side	des	Un.
	-	ΤŢ	Π		1	1		1	T			Γİ	1		+	17!	zz			Τ	Ţ		T,		T,	.0	OZ		-/ 8	27	ds	4	]]		1		_		
																¥.								_		¥.		4	11	8	5.0	00 *	2	* Cos	stan	d of	Re.d	10	592.50
	1	3!			/ S	3	3	72	!	4	4	4	8 .	-	3/	z	39	↓_∤	- 6	66	<b>≁</b> !,	<u>5 5</u>		╞╴	20		99		-4	28	<u>s;s</u>	14		01	1				
ī	<u>3   z</u>	18!3					31	-12	2			<u>!</u> !	<u>_</u>		35	12.	16		L-				<u> </u>	1	96	,5	55	1	L.,	Ļ	Ļ!				1				
6	-1-		<b>m</b>	<b>T</b>	17			-12	T		T	121	2	<b>T</b> -7	_	1-1	1	11	<u>-</u>			-1-	F 1.	1	नि	तस	नत		<u> </u>	-	1	1.	77		{				
	25				44	12	4	12	12	+	-17	6		+	2	i Ei		-		10		02		╀	8	1/1	83			1	1:0	4	÷		1				
┝ <sup>↓</sup>	40	17.0		<b>.</b>			214	<u>, 16</u>		1	10	-0.	z		219	.[	<u></u>	غغ	L			<u> </u>	<u>L.I.</u>	<u> </u>	1.45	15.	01/		<u> </u>	- 2	7	<u>.</u>	<u> </u>						
31	26	5	$\Pi$		45	0	3	2 9	T I	T	4	3	4 0	Π	26	5	21	П	-1	6	4	56	1	T	P	5 8!	Z 7			8	5!5	2	Ĩ	04	1				
	27	00		1	06	3	2	8 5	16		10	4	41		3	9!	77	T	- 5	73	71.	13	T,	T	91	(3	79		- 1	15	21	ø	1-1	07	1				
											1	r I.	6																	÷ 2	7				]				
4	27	7			3 5	.3	\$	01			3	6	0 8		Z	19	75			68	0	50		L	8	7 <u>2</u>	84	1	2	10	1:5	0,		00					
	24	12!2		1	<u>4</u> /	.6	41,	-12	0	1	/14	0	<u>4 </u> =		44	0!	75	اللبا	-1/	15	0!	63	LĿ	F	9	93	66	Ц	-14	<u>کا</u>	<u>¢: 6</u>	0	<u>]-i</u>	07					
	<b>T</b>	т-т-	1	4	40	4	<b>—</b> —	-			 -		7	1 1		-	- 1-	Å		1	<u>к</u>				1=5			h	-	-	2	1.1	╁╌᠇	-	0.1	1.	onid.	. da	4071
14		┼┼	+	2	X   Z	2		+-	-	-+	╇	$\vdash$	-#	F	£44	44		Ы			Η÷		12	╞	121-	Ŧ	317	H	12	0	<u>त</u>	2	Нi		100	, <i>•</i> ,			
	<b>!</b>	13	- S		27	2	k.				51		+/	1 (z		10	5 0	21	ł	-			1_15		117	· 10 ·	op/	1		يعد	100	In A	<i>w</i>		- /	50.	615		
	1	T	Ť	T	Ť	Ť	Ť	1	1	ľ	Ĩ	ĒΤ	Te	Ē	- 	TT	T	T	T	T	L.İ.	· [	В	Τ.	П	TI	Τ			T	Ţ	ΤŤ	Ť1	4	1		200		
	1	ΤŤ	П		1	Ť		1	!	T	1			Π		11					İ		-			1		Π					11	T	-0.	.75.	308	m	
Ш		Π				Ţ						Ц	Į.		_			Ц	_		1		B	1						Ц	!	Ц	1						
	_		Ш			!			!		1		1								[ !			L	Ш		_						1	1.	4				
		7 1				1.1	_	-				r - 1	-	1 1		<b></b> .		<b></b>		· · · · ·	r-r			·····	<b>r-</b> r			<b>1</b> 1		·	····T····	<b>.</b>			Į				
┠╨┨		┼┼	+			┼┨	+	+	┼┨	+		Η	-#	┢		Η÷		⊢			Η÷		H	╋	┢╋	+	_	$\vdash$	_	+	-i-	++	+		ł				
$\vdash$	1.	1	Ļ	Ļ	1	:			<u>.  </u>		L	I	1		. <b>I</b> .	1	1	1					<u>. 1</u> °	1	11	1.,	ĺ	<b>L</b>		.ii	L.	<u></u>	L	L	ł				
Ш	T	Π			Т	П	T	1	Π	T	1	П	A			11	Т			TΊ	[]	Т		Τ	П	П	T	Π	Т	Π	ŀ	ТТ	11	Т	1				
H++	1	t†	T		┢	† I		T	Τ	+	-	h	Ţ,		-	11				T	Ţ		T,	1		11		Ì			İ	Ì	Ti	1	1				
and the second second second second second second second second second second second second second second second															_		_									_													

Figure 3-82.—Leveling to an upside-down scale.

2. Slowly lower the weighted tape until the voltmeter moves. This indicates that contact has been made with the water surface.

3. Read the tape value at the reading mark. Record in the "remarks" column the value (usually in feet), the time, and the gage identification date (which should be stamped on an attached brass tag).

4. Reel the tape back up to about 0.5 ft below the reading mark, and secure it.

#### 3.7.7 Releveling and Closing Sections

Often in single-run leveling, and always in doublerun leveling, two or more runnings of a section are required. Just as two elevation differences observed on a single setup must be checked for agreement, the elevation differences obtained from multiple runnings of a section must be checked for agreement. The check is analagous to closing a loop (sec. 5.4.3, "Checking loops") by summing the differences of a series of setups beginning and ending on the same point and comparing the result to a tolerance. Thus, the check for agreement is called "closing the section."

If all leveling errors were eliminated, section runnings would agree exactly. However, since small amounts of random error are always present, tolerances have been set to ensure that runnings retained in the uncorrected leveling data disagree by no more than an amount consistent with the precision of the prescribed leveling procedure. When multiple runnings of a section disagree by more than the tolerance, one or more blunders may have occurred Blunders occur when prescribed procedures are not followed. The section must be releveled to obtain a sample of data, excluding blunders, that more accurately provides the elevation difference.

Requirements for releveling and procedures for closing sections are discussed next.

*Releveling.* If any of the following situations occurs during single-run leveling, relevel the section in the opposite direction and close the section. Releveling may be required by the project office for other reasons as well, after the data have been checked and analyzed.

1. A procedural mistake is reported or suspected

2. A tie check with previous leveling is not within tolerance.

3. One or more readings are missing or unclear, either in the computer or on a recording form.

4. The stamping recorded for a control point is missing or apparently incorrect, either in the computer or on a recording form.

5. One or more observations that do not meet specifications have been accepted because reading checks were computed improperly.

Closing the section. After a section is double run, check that the elevation differences from the two runnings agree within the tolerance for the order and class of the survey.

First, compute the difference, the misclosure, between the two elevation differences. Because they are opposite in sign, the misclosure is the sum of the backward running plus the forward running.

Then, compute the tolerance by the formula  $T \times \sqrt{K}$ , where T is the factor from table 3-1 for the order and class of the survey and K is the length of the section in kilometers. The length of the section is computed by totaling the distance leveled and dividing by the number of section runnings. For example, if the distance leveled in each running is 1.60 km and the survey is first order, class I, the tolerance is  $3.00 \times \sqrt{1.60} = \pm 3.79$  mm.

If the misclosure is within the tolerance, additional runnings are not required. If it exceeds the tolerance, relevel and recheck the section to satisfy two criteria: (1) All runnings likely to contain blunders (as determined by the rejection procedure) are rejected, and (2) at least one forward running and one backward running are accepted. A standard rejection procedure is used to reject statistically unreliable runnings.

When releveling to close the section, alternate the direction of leveling on each running, to maintain an equal number of forward and backward runnings. If systematic error persists in the leveling, the mean of the elevation differences may be biased in favor of the running direction that is in the majority. Equalizing the number of runnings in each direction prevents this bias.

Notify the project office if more than four runnings are required to close the section. To prevent excessive releveling, check that the leveling rods are being placed on the control points correctly and check that the points themselves are stable. Record any unusual features of the points in the leveling data. Check and adjust the rods and the instrument, especially the circular levels. On subsequent runnings, try shortening the sighting distances, balancing setups more precisely, varying the routes, and changing the duties of the unit members. Try to make subsequent runnings when the airtemperature difference is negative and less than  $1^{\circ}C$  $(2^{\circ}F)$ . (See sec. 3.6.1, "Measuring air-temperature differences.")

*Rejection procedure.* After three or more runnings of a section, check for agreement and reject blunders as follows. (See example in table 3-7.)

1. Compute the mean of all the runnings (disregard the signs), including those that have been rejected before.

2. Compute the differences between the mean and each running, the mean value minus each running value.

3. Compute the allowable difference from the mean. It is based on the order and class of the survey, the number of runnings that were meaned, and the length of the section. The formula is  $t \times \sqrt{K}$ , where t is the appropriate factor from table 3-1 for the order and class of the survey, and K is the section length in kilometers. For example, if the survey is first order, class I, four runnings were meaned, and the section length is 1.51 km, and the allowable difference is  $2.33 \times \sqrt{1.5} = \pm 2.85$  mm. 4. Compare the differences from step 2 to the allowable difference. If any runnings exceed it, reject only the running that exceeds the allowable difference by the largest amount.

5. Check the runnings to see if there are at least one forward and one backward running remaining. If this is the case, and no running was rejected in step 4, the section is closed. If a running was rejected in step 4, compute a new mean with the remaining runnings and return to step 2. If only runnings made in one direction remain, relevel the section and begin again at step 1.

Notice that certain runnings, rejected when the number of runnings is three or four, may be accepted when the number increases. This is acceptable because the accuracy of the mean improves with a larger sample of data; it becomes easier to recognize if a running is different enough to be considered a blunder.

## 3.8 Field Records

Original records of field observations are the primary source of information for all future analysis and adjustment of a survey. As such field records must accurately and completely present the results and conditions of leveling. The records may include both data recorded on forms and printouts of data recorded in computer memory. To be incorporated into the National Geodetic Vertical Control Network, they must provide data sufficient to meet the requirements of the Federal Geodetic Control Committee, which are published in *Input* Formats and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: vol. II, ch. 6). Whatever form is used, the records must be carefully preserved by the leveling unit until they are submitted to the project office. Their preparation and submittal are the subjects of this subchapter.

#### **3.8.1 Recording Observations**

Leveling observations can be collected by using one of two methods. Observations may be either (1) written on a recording form, or (2) keyed into computer memory and recorded on a magnetic tape or disk, supplemented with a backup form if required. The computer-recording method is preferred to reduce errors and to simplify checking and archiving the data. Nevertheless, both methods require that the recorder prepare clear, correct, and complete records.

When observations are written, the recording form serves both as a record and as a reminder of the checks and computations to be made. When observations are keyed, the backup recording form serves as a record to verify information that cannot be checked by the computer. Use standard recording forms, as illustrated in the figures in this manual.

Be sure to record correct identifying information. This cannot be overemphasized. Mistakes in any of the following entries will cause false elevation differences to be computed when leveling data are processed.

Date	Direction (km)	Distance (m)	Elevation difference (mm)	Difference from mean	Explanation
10/11 10/12	F B	0 82 0 83	-2 43152 2 43752		Two runnings of a section are made Their difference is $B+F=600 \text{ mm}$ The tolerance is $3 \times \sqrt{0.82} = \pm 2.72 \text{ mm}$ Since the difference exceeds the tolerance. the section is releveled
10/11 10/12 10/12	F B F	0 82 0 83 0 82	2 43152 2.43752 R 2 43052 Mean=	+1 67 -4 33 +2.67 =2 43319	The mean of the three runnings and the differences from the mean are computed. The allowable difference is $210 \times \sqrt{0.82} = \pm 1.90$ mm. The second and third runnings are both outside the allowable. Rejecting the worst leaves only two forward runnings, so the section is releveled.
10/11 10/12 10/12 10/13	F B F B B B	0 82 0 83 0 82 0 82 0 82 0 82 0 83 0 82	2 43152 2 43752 2 43052 R 2 43810 Mean = 2 43152 R <sup>2</sup> 2 43752 2 43810 Mean =	+2 90 $-3 10$ $+3 90$ $-3 68$ $= 2 43442$ $+4 19$ $-1 81$ $-2 39$ $= 2 43571$	The mean and differences are computed for all four runnings The allowable is $2.33 \times \sqrt{0.82} = \pm 2.11$ mm. All are outside the allowable Rejecting only the worst, the third running, a new mean is computed. The allowable for three runnings is, as before $\pm 1.90$ mm Rejecting the worst, (the first) leaves only two backward runnings, so the section is releveled.
10/11 10/12 10/12 10/13 10/13	F B F F	0 82 0 83 0 82 0 82 0 83	- 2.43152 2.43752 - 2.43052 R 2.43810 - 2.43812 Mean=	+ 3 64 - 2 36 + 4 64 - 2 94 - 2 96 = 2.43516	The mean and differences are computed for the five runnings. The allowable is $2.48 \times \sqrt{0.82} \pm 2.25$ mm Rejecting the worst, a new mean is computed.
	F B F B B F	0 82 0 83 0 82 0 83 0 83 0 82 0 83	- 2 43152 R 2 43752 2 43810 - 2 43812 Mean 2 43752 2 43810 - 2 43812 - 2 43812 Mean =	$ \begin{array}{r} +480 \\ -120 \\ -178 \\ -1.80 \\ =24632 \\ +0.39 \\ -0.19 \\ -0.21 \\ =243791 \end{array} $	The allowable for four runnings is, as before, $\pm 2.11$ mm The first running is rejected, and then a new mean computed The allowable for three runnings, is as before, $\pm 1.90$ mm. Since all differences are within the allowable, and at least one forward running and one backward running are accepted, the section is closed!

	Table 3-7.—Example	of rejection	procedure for	first-order, c	lass II leveling
--	--------------------	--------------	---------------	----------------	------------------

The recorded stamping, or designation of each control point, and its corresponding survey-point serial number determine the order in which the line is abstracted. This order must correspond to the way the line was actually leveled.

The recorded date, time zone, and time must be accurate because observations are stored and sorted chronologically. Corrections for collimation error, refraction, tidal accelerations, and scale error are applied accordingly. Time zones should be coded according to the alphabetical system of the U.S. Navy. (See table 3-8 for the codes applying to North America.)

Finally, recorded descriptions of the equipment and the corresponding serial numbers determine which collimation and calibration corrections are to be applied. The make and model of rods and instruments should be specified by code if possible. (See sec. 3.2.2, "Equipment.") Write observations and remarks neatly in ink. Never recopy original records. If a mistake is made when computing, draw a straight line through the error and write the correct value above it. (See fig. 3-71.) If a mistake is made when recording a rod reading, reobserve the

Table 3-8.—U.S. Navy time zone designations

Standa time	ırd	Daylig time	ght 2	Time meridian	Time Zone description	U.S. Navy designation
Atlantic	AST	Eastern	EDT	60W	+ 4	Q (Quebec)
Eastern	EST	Central	CDT	75W	+ 5	R (Romeo)
Central	CST	Mountain	MDT	90W	+ 6	S (Sierra)
Mountain	MST	Pacific	PDT	105W	+ 7	T (Tango)
Pacific	PST	Yukon	YDT	120W	+ 8	U (Uniform)
Yukon	YST	AK/HI	HDT	135W	+ 9	V (Victor)
AK/HI	HST	Bering	BDT	150W	+10	W(Whiskey)

setup and record the new readings on a new line. Crossedout or illegible readings in written data are like readings missing from a computer-recorded tape: the entire section must be rejected.

When computing, round results to the appropriate number of decimal places. Elevation differences should usually be expressed in meters to five decimal places (four decimal places in three-wire leveling). Distances should be expressed in meters to one decimal place or in kilometers to two decimal places. Some calculators do not automatically round the result when displaying fewer decimal places than were computed. If using such a calculator, look at one extra decimal place and then round the result. If the number to be rounded ends with the numeral "5", round to the nearest even number. For example, the collimation factor +0.0135 mm/m rounds to +0.014 mm/m, and the elevation difference -2.148145 m rounds to -2.14814 m.

Compute reading checks carefully. If they are computed incorrectly and one or more setups are accepted which do not meet the tolerance, the section must be releveled. This type of blunder can occur anytime that observations are written; it can also occur when preliminary information is stored incorrectly in a computer. The observer should check for such blunders as work progresses.

Use the "remarks" column (or computer comment lines) freely. Record any activity or event which may affect the quality of leveling, such as frequent failure of reading checks, unusual atmospheric conditions, or difficulties with the computer-recording equipment. Give a complete description of the situation: time at which the activity occurred, the setup number, what happened, and the way it was handled. Be specific. Include the names of personnel involved and the serial numbers of the affected equipment.

At the end of the day, the unit chief should check and initial the recorder's work. Be sure to check that the designations and survey-point serial numbers are correct, as well as the other identifying information. If mistakes are found that may require editing of a data tape or disk, do not attempt to correct them. Instead, note and explain the correction required on a comment record in the computer or on the appropriate backup form.

## 3.8.2 Computer-Recording Equipment

Portable computers have revolutionized recording procedures. Specially designed computer-recording equipment permits observations to be checked at the time they are made, nearly eliminating reading and recording blunders from accepted data. The equipment also permits all field observations to be stored in data packages that can be easily transferred to more powerful computers, simplifying the subsequent checks, correction, analysis, and adjustment of the data. To obtain the greatest benefit from a computer, the equipment selected should meet the general requirements given in this section. To collect data reliably, the computer and its power supply must be maintained properly. To collect data accurately, the computer must be programed and operated correctly.

General requirements. Computer-recording equipment for leveling should be manufactured for the field environment. The equipment must withstand frequent exposure to dust, water, and vibration. It should be portable, weighing no more than 10 kg. It should be capable of operating continuously from an internal power supply for at least 6 hours and from both 115 V AC and 12 V DC external power supplies. It should not lose stored data if the power supply is interrupted or exhausted.

Data may be recorded and stored on a tape or disk, or in the computer memory. If recorded in memory, data must be easily retrieved for rapid transfer onto a tape or disk or by telephone into a central computer. The output format should be compatible with the computer facilities of the project office.

The computer should be programmable and should have an alphanumeric keyboard with a numeric cluster. The keyboard permits the stampings on control points to be entered in addition to survey-point serial numbers. It also permits direct entry of remarks and simplifies the coding of other information. The display should be easy to read and should have at least 32 characters. At least 20,000 bytes of memory should be available for storing both programs and data.

Maintenance. One person in the leveling unit (usually the recorder) should be responsible for maintaining the computer-recording equipment. Carefully follow the instructions provided with the equipment. Protect it from exposure to dust, water, and vibration. The more self-contained and rugged the equipment is, the better; however, it must still be kept clean and be handled with care.

Sticky or dirty keys on a computer keyboard make recording difficult. Protect the keyboard by taping a thin sheet of plastic over it. When leveling data are recorded on magnetic tapes or disks, the data are highly vulnerable to alteration or erasure caused by dirty recording apparatus. Therefore, learn the proper method for cleaning the equipment, and clean it at least weekly.

Data may also be altered or erased if the equipment is exposed to shock or excessive vibration during the recording process. As a result, do not transport the equipment while data are being recorded. When data are stored in the computer memory, dropping the equipment or striking it against objects may erase or render inaccessible the results of an entire day's work. For the same reason, keep magnetic material away from the computer and data packages.

Battery care. For computer-recording equipment to operate reliably, the batteries must be properly charged and maintained. The nickel-cadmium batteries used in most portable computers should be recharged at regular intervals, either nightly or as instructed. They should be protected from overcharging by a voltage regulator. They should also be "cycled," or discharged completely and recharged, at least once a month and after a period of storage. They should be stored in a cool, dry place.

Built-in batteries, if properly charged, should provide enough power for a day's work. However, to prevent loss of time if they fail, keep two alternate power supplies available: a spare battery pack and a hookup to the battery of the leveling vehicle Maintain the spare pack in the same way as the built-in batteries.

Data packages. For submittal to the project office, leveling data are normally stored on a tape or disk. Use a separate tape or disk to record each day's data. If data are recorded directly onto a tape, precondition the tape before work begins. Keep a supply of clean tapes in the leveling vehicle, in case a tape is defective. If data are stored in computer memory, print out or record the data onto a tape or disk immediately after the day's leveling is completed. Instructions for submitting data packages appear in section 3.8.4.

*Programs.* When recording observations into a computer, use only the standard programs specified in the project instructions. The recorder should become thoroughly familiar with the programs by studying accompanying documentation and instructions. Although knowledge of programing techniques is not essential, it is helpful when troubleshooting in the field.

Recording programs should prompt the recorder for the necessary data at the appropriate time in the observing routine. They should incorporate as many checks for blunders as possible. They should record observations twice on tapes or disks, to increase the chance of retrieving at least one complete set of observations if some data are destroyed or made inaccessible by imperfections in the system. For leveling data to be included in the National Geodetic Vertical Control Network, the programs should format the data as described in *Input Formats and Specifications of the National Geodetic Survey Data Base* (Pfeifer and Morrison 1980: vol. II, ch. 6).

Depending on the capabilities of the equipment, a system of programs can control the observing routine at one of three levels: the setup, the section, or the day.

The most basic system prompts the recorder for the observations, checks them for blunders, and records them at each setup. Using a backup form as a guide, the recorder must enter the running records for each section and the survey-equipment records for each day. Separate programs are used for each order and class of survey, and to observe and compute the collimation and compensation checks. Although this system nearly eliminates blunders at each setup, it allows blunders in section and equipment information to remain undetected in the field. Therefore, a backup recording form, such as shown in figure 3-70, must be maintained to permit verification of the data. More blunders can be eliminated if the system includes prompts and checks for running records and surveyequipment records. In such a system the recorder must again select the appropriate program to use, but programed control of the observing routine is extended to include the entire section.

The most sophisticated system should prompt for and check all the data collected by the leveling unit each day. Prompts should include the following: date, line number, survey-equipment record, daily equipment checks (such as the collimation check), weekly checks (such as the compensation test), running records, and observations. Subroutines, such as the collimation check, should be accessed by the master program when required. Constants, such as the collimation factor, should be computed and retained in computer memory from day to day. Separate options should be available for each order and class of survey. The results of multiple runnings of a section should be checked automatically.

*Precautions*. When using computer programs, the recorder must observe the following precautions.

Learn to glance at each displayed rod reading immediately after keying it and before entering it. Gross mistakes (such as misplaced decimal points and doublekeyed numbers) may then be deleted and corrected before the entire setup is rejected.

Never write over or erase the data for a setup. Enter a remark if necessary. If a section must be terminated, enter a rejection record. Learn how to avoid erasing data accidentally. Learn how to reprogram the computer if the program or data are lost because of a battery change or other power failure.

Avoid using computer-recording equipment near a source of high-energy radio-frequency emissions Depending on their frequency, such sources as airnavigation, television, and radio transmitters may interfere with normal computer operation.

## 3.8.3 Field Abstract

The field abstract is a complete summary of the elevation differences measured between the control points in a line. On the abstract, points are listed in the order in which they are connected by leveling. For every pair of points, the elevation differences from one or more runnings are recorded. These are added to the elevation assigned to the beginning point, to obtain the field elevation for each point on the line.

If all observations are written on recording forms, a field abstract must be prepared and submitted to the project office. If observations are recorded with a computer, the abstract, although not required, should still be prepared. Since observations must be submitted daily, the abstract provides the leveling unit with the only record of which sections require additional leveling and which portions of the line have been completed. Multiple runnings, check points, and ties can be recorded and checked on the abstract. The abstract, by itself, is not an acceptable substitute for the actual observations, no matter how they are recorded. It must always be accompanied by the original recording forms or printouts of the original observations recorded on a computer.

Instructions. For leveling to be included in the national network, use NOAA Form 76-188, "Geodetic Leveling Line Identification" (fig. 3-83), and NOAA Form 76-187, "Geodetic Leveling Field Abstract." A single-run portion of line (fig. 3-84) is abstracted in figure 3-85. A double-run portion of line (fig. 3-86) is abstracted in figure 3-87. Prepare and compute the abstract as follows:

1. Prepare a cover sheet for the abstract. Only the portions of the form that are circled in the example need be filled out by the leveling unit. When observers, recorders, rods, or instruments change, add the initials or serial numbers and dates of change to the form. Submit the abstract form to the project office when the leveling line has been completed.

2. On the abstract, enter the accession number for the line, if known. If the accession number is not available, enter the line number, job code, and project title. Enter the line title, corresponding to the direction of leveling. Be sure to circle the appropriate units of distance and elevation difference. Include this information on every sheet of the abstract. 3. On the first sheet, skip the first line. Enter the survey-point serial number and designation of the beginning point on the second line. Enter the beginning distance, 0.00, in column D. In column H, enter the adjusted elevation from the most recently published description of the point. If the point is new to the national network, do not enter a beginning distance or elevation until a point with a published elevation is leveled. The beginning point and elevation are normally assigned by the project office, as described in section 5.4.2.

4. Enter the results for each section on subsequent records. The ending point of one section must correspond to the beginning point of the next. Write the designation of the beginning point for a forward running on line \*30\*. Write the survey-point serial number and designation of the ending point on the line labeled "TO." As shown in columns C, D, E, and F, write the date, direction, distance, and elevation difference for each running. Always write the results of the first forward running on line \*30\*. If more than two runnings are made, write the designation of the ending point on the next available line labeled "TO."

5. Check that sections with more than one running are closed. Indicate rejected runnings with the letter "R" in red ink.

NOAA FORM 74-148 (8-78)	U.S. DEPARTMENT OF COMMERCE HATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION						
GEODETIC LEVELING LINE IDENTIFICATION							
HOZ HUHBER LINE YEAR UD DAY YEAR UD DAY SANDAY	ER- UCA TOR GOU TOR GOU						
·10- L24133 5R1976051119761209mm 3	· 0///L/A BKMMGS						
	· · · · · · · · · · · · · · · · · · ·						
· III· LEVELING FOR CORPS OF ENGINEERS N	ew drileaws diistrijettiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii						
112. LOTTIE TO 9.8 KM (6.1 M1449) M OF	KROTZ SARINGS						
	CONNENTS						
<u>~1151~1 ] [ ] [ ] [ ] [ ] ] [ ] [ ] [ ] [ ] [ </u>							
R 362 5200	316-121178						
CHIEF OF PARTY BKM	316-127951						
ODSERVERS RLW							
RECORDERS	HITRUMENTS						
DBC	231-90825						

a V. G. Government Frinting Office: 1978—765-005/1082. Region to 8

**Chapter 3—Geodetic Leveling** 



Figure 3-84.—Sketch of single-run leveling.

	EXAMPLE			##110a.		06 P A A	THE TOP COMPANY
	GEODETIC LEVELING FIELD ABSTRACT						
"' HGZ	L24479 LINE 368 N	'GV	/ Ľ	>	· / •	3-663	;s
JON	JONESVILLE TO BARKERSVILLE MT SINGLE -RUN						
	$\sim$	4	D	ň	FIRST	~ of	DEK, CLASSI
<u>ка — </u>	<u>(B)</u> (	÷	T	P(E)	(E)	( <u>G</u> .)	(H)
DISTANCE UNITS (		Der	4	6/E	0	2/2	B1+.£.
1		$\vdash$	H				ļ
18 TRON			T		~		
Raal	A. A. 93		Ц	0.00			
	1. AD3	47	F	0.16	+ 5.11130		+ 5. 19230
9,0,00	3525 RESET 1906 11111111		Н	0.16			981.13250
	3525 RESET, 13.04	$\overline{\eta_i}$	F	0.10	+ 15.01 542		+ 15.01162
71							
Peas .	Millannennennen	ļ	H	0,16		L	996.21612
· 10· *807	Villa manager	71	Ē	1.40	+ 1,20312		+ 1.20312
	u , 4	$\vdash$	Н				and diard
nielie -o	<u> </u>	Η	-	- <u>4</u> .32		A.16	1111-11147
N 7304	X 1/13. 1 + 1 + + 1 + + 4 + + + + + + + + + + +					5000	+ D.15530
		11	B	1,31	- 0.15530		007 67464
	16 14 Pt1	E	H	7.01			417.37454
	AAA	$\eta_{c}$	F	1.19	+ 0 15501		+ 0.15738
		4	8	-like	- 0.14100	(1)	1-0,15111
۰۰ سنساسا	<u>[]]]]]]]]]]]]]]]]]]]</u>	₩.	-		+ 0.15612	1.01	4 0.1555¶
		12	۲		- 0.13301		
- Me. 6 #04							
3163	Juliz concernence of the second	<u> </u>	1	5.30			117,73013
	Y 1.1	7.	R.	1.57	- 0 (711)	+ 17	-061117
-38- 5804		H.	B	152	+ 0.67 104	1.11	
3.7.04	<u># 164</u>	F		6.12		4.25	991,05701
1	Y. 1.3	1	Z	0.41	-1 12510	2050	- 1 (7 1A1
30" "H (M)		1	8	0.11	+1.12256		1
37.05 .	<u>M.1.1.2</u>	Ē	Г	4,21		+ Ζ, 6J	996,60630
	A 1 14	F	F				1 1 19345
H+ 1904	9.6.97.	4	8	0.11	-1.12345	<u> </u>	1- 111-512
\$7,96 10	<u>Edefinition and a second seco</u>	Ľ	Ľ	4.65			998.69997
AUMPEREEDEE WASIE SOME IN-(AU 13-(A) MONTEM IN SEMALE SA END Aristing Statu Annologo an a setting statu setting varian material sa enda							

Figure 3-85.—Abstract of single-run leveling.



Figure 3-86.—Sketch of double-run leveling.

-741	CEOBE	EXAMPLE	F		*****	AL GERAMIC AND A	OF PAR	TRENT OF COMMERCE
HGZ	L 23 797					1-111 / C	1	' <i>12</i>
LONGVIEW TO 6.2 KM W OF HILLSIDE, LA FIRST ORDER, CLASS							- RUN DER , CLASS	
	@ <b>&gt;</b>	10.1. cant. @ 11	4. Dir	Π	A/E	ð	6/L	Brr.e.
- ter Percu	****	<u></u>	F	H				
Leas	N WASA AZ MK	والمستعد المستحد المراجع المراجع		Н	0.00			1.97151
an Phone	N MASH AZ MK	للاستناقية القدمانية	7.	E	<u>au</u>	+ 2.22450	-,79	+z.zz436
aaz .	<u>д 332</u>		11	Ц	0,12	<u>- 1.21911</u>		4,11587
· 50·· * RD4	<u> </u>		$\eta_i$	F	0,10	+ 3,14467	#/	4 3 14445
ees	สมปรีระบบเนื่อง		Ē	Ê	010 112	- 3.14923	73	7. 34032
aar Pincin	5 Ide		7	Ē	1.17	- 3.45322	H24	- 3,46275
694	H diff and a second	<u>+</u>	Ē	Ľ	2,59	+ 3. 73168	t., 81	3.11117
· jaj- #RQar	H.1.H.1.1.1.1.1.1		71	F	0.40	-0.93541	1.37	- 0,13472
ه. بيت	سسيسيس		Ŧ	8	0,10	+0,13465		
-	***		ļ	П				
1965 To	303 REJET, 1.967		F	Ц	1.11		łZ.18	2. 15315
M- 7807	203, RESET. 1.947.	┷╍┶╍┶┶┶╼┷╼┙╼	×4	F	· 1.02	+2.43176	= ¥	+ 2. 43966
	<u>Z.331</u>		-		4.01		+ 1.97	5.39281
10 THUS	51644	-	75	F	0.70	-1.94763	II	-1.94774
1.994	Z 331	<u>+111111111</u>	Ê	П	2,10		44	5.39258
· M··· * • Cas	4.331	-	7.	F	0.12	+3,22449	+.3 <b>1</b>	+ 3. 22.468
1007 -0	17.8.1. <u>13</u>		F	Π	2.92		-,56	8.61726
10 TRON	T.B.M., 63.		2/2	F	0.51	-0.17501 +0.27511	-,10	~D.27513
1091 10	ATIZITI		2	<u> </u>	7. 37	K. AHD		8.34213

Figure 3-87.—Abstract of double-run leveling.

6. Compute the mean of the accepted elevation differences for each section. Write the mean in column H, on the same line as the first forward running. Assign to the mean the sign of the forward running.

7. If there are two runnings, compute the difference in millimeters between the forward and backward runnings. Enter the difference in column G, with the sign of whichever number is smaller. If there are more than two runnings, compute the difference between the means of the forward and backward runnings.

8. Draw a box in red ink around every spur of the line, as shown in the example. For a spur-on-a-spur, draw an additional, nested box. If any sections connect to form a loop, show the longer route as a spur and the shorter route as part of the main line.

9. To the beginning distance, add the shortest distance for each of the sections leveled so far. Write the total in column E, on the dark line across from the designation of the point leveled to. When the end of a spur is reached, be sure to return to the point at the base of the spur to continue computing the distance along the main line.

10. Similarly, to the beginning elevation add the elevation differences of the sections leveled so far. Write the total in column H, on the same line as the distance. When the end of a spur is reached, return to the point at the base of the spur to continue computing the elevations of points on the main line. Total column G in the same manner.

11. Start subsequent sheets of the abstract on the first line, beginning with the point from the last line of the previous sheet.

#### 3.8.4 Submitting Records to the Project Office

The unit chief is responsible for field records until they have been submitted to the project office. Submit records of observations daily. Submit completed abstract sheets weekly, together with the weekly status reports.

Package each day's records separately. Thus, if a package is lost or damaged, only one day's work at most need be repeated. Label the package with the following information: accession number or line number, date, initials of the unit chief, and, if data are recorded on a tape or disk, the serial number of the recording equipment. If many units are operating from the same project office, each unit should use labels of a unique color, so the data packages can be easily distinguished.

When working near the project office, submit records in person at the end of work each day. When working at some distance from the office, transmit observations recorded in a computer directly over a telephone line. Send other types of records, including backup material (duplicates of transmitted data) by certified mail. Retain the receipts and send them with the weekly status report Arrange the work schedule to allow visits to the post office when the facility is open. If this is not practical, send records by first class mail. When records are needed immediately, as at the end of a project, make special arrangements to send them to the office by courier or express mail.

Magnetic tapes and disks. Data recorded on magnetic tapes or disks are particularly vulnerable to alteration or loss. This may occur especially if they are stored for long periods under field conditions or exposed to X-rays or strong magnetic fields while enroute to the project office.

If data packages must be collected and stored in the field for any length of time, keep them in a rigid, nonmagnetic container. When they are sent through the mail, enclose them in an appropriate mailer Magnetic tapes may be sent in aluminum boxes and placed with the accompanying backup forms inside preaddressed, padded envelopes. Magnetic disks should be similarly packaged

## 3.9 References

- Bodnar, Jr., A. N., 1977: User's Guide for the Establishment of Tidal Bench Marks and Leveling Requirements for Tide Stations. National Ocean Survey (C233), NOAA, Rockville, Md. 20852 (in press).
- Coast and Geodetic Survey, 1965: Manual of tide observations. *Publication* 30-1, National Ocean Survey (C233), NOAA, Rockville, Md 20852, 77 pp.
- Federal Geodetic Control Committee, 1974: Classification, Standards of Accuracy, and General Specifications of Geodetic Control Surveys, 14 pp. (Sold as a set with the following companion publication).
- Federal Geodetic Control Committee, 1975, revised 1980: Specifications to Support Classification, Standards of Accuracy, and General Specifications of Geodetic Control Surveys. Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402, 50 pp.
- Greenawalt, B. and Floyd R. P., 1980: National Geodetic Survey Operations Manual. NGS Operations Division, NOAA/NOS, Rockville, Md. 20852 (unpublished manuscript).
- Pfeifer, L. and Morrison, N. L., 1980: Input Formats and Specifications of the National Geodetic Survey Data Base, vol. II: Vertical control data, Federal Geodetic Control Committee. National Geodetic Information Center, NOAA/NOS, Rockville, Md. 20852, 136 pp.

## Chapter 4 RIVER OR VALLEY CROSSING

## 4.1 Introduction

In geodetic leveling, when the sighting distance and setup imbalance cannot be confined to the maximum specified for the order and class of the survey, a special procedure must be followed. Because it is most commonly required when a line of leveling crosses a river, the procedure is termed a river crossing. However, the procedure may be required over any type of terrain. In the following paragraphs, general procedures and requirements for a river crossing are explained Detailed instructions for two river-crossing routines are presented in the remainder of the chapter.

Leveling errors that are a function of sighting distance are greatly magnified when an unusually long sight is observed, as shown in figure 4-1. The graduations on a geodetic leveling rod either cannot be seen or do not provide a sufficiently definite image, especially when shimmer is present. As a result, pointing error is greatly increased. Since the sighting distances of such a setup usually cannot be balanced, the effects of collimation error and curvature cannot be limited in the usual manner. (See sec. 3.1.2, "Leveling instrument.") Furthermore, even if the setup can be balanced, refraction may be large and unpredictable because of variations in the atmospheric conditions along the lines of sight, particularly if the line of sight passes over water. Intercepting targets. To reduce pointing error when the sighting distance exceeds the limits for the order and class of the survey, a series of repeated scale readings should be made for each required observation Pointing error may be further reduced by observing the crossing under favorable environmental conditions. Even so, a scale with line graduations does not provide an image which can be intercepted with sufficient precision at sighting distances greater than about 80 m (260 ft). Similarly, a scale with 1-cm block graduations does not provide a sufficiently precise image at distances over about 160 m (530 ft).

A better image is provided by a target or directional light, attached to the rod or other vertical support. A target should provide an image similar in size to that of a scale graduation viewed at about 50 m (165 ft) through the instrument. It should have a sharply defined shape and be symmetrical about the horizontal axis. An isosceles triangle is practical because the slopes of the sides define the center of the target throughout a wide range of sighting distances.

A precise method for making scale readings employs two targets. One target is placed above and the other below the point where the horizontal line of sight is estimated to intercept the support. The line of sight is then tilted by the amounts necessary to intercept each target and to return to a horizontal position. The inter-



Figure 4-1.—Curvature, c, refraction, r, and the effect,  $C \times s$ , of collimation error,  $\alpha$ , are magnified by a long sighting distance.

cept of the horizontal line of sight is computed from the proportional relationship between the distance intercepted on the support and the vertical angle of tilt, as illustrated in figure 4-2.

Three tilt measurements are made: (1) when the line of sight intercepts the upper target, a; (2) when it is horizontal, r; and (3) when it intercepts the lower target, b. The distance, F-T, between the unknown foresight intercept and the lower target is related to the angle, b-r, from lower tilt to horizontal in approximately the same ratio as the distance, D, between the targets is related to the angle, b-a, from lower to upper tilt (provided both angles are measured in radians):

$$(F-T)/(b-r)=D/(b-a)$$

Solving for the foresight intercept,

$$F = T + [D \times (b-r)/(b-a)]$$

The sighting distance, s, is computed without stadia observations by the following equation:

$$s=D/[\tan(b-r)+\tan(r-a)]$$

Since the angles of tilt are very small,

 $\tan (b-r) + \tan (r-a) \simeq (b-r) + (r-a) \simeq (b-a)_{\text{radians}}$ 

Thus, to a sufficient approximation,

$$s = D/(b-a)_{\text{radians}}$$

This method for making scale readings is suitable for the longest sighting distances and is employed by the river crossing routines outlined in subchapters 4.3 and 4.4. For the most precise results, the angular difference b-a should be between 100" and 250", and D should be no more than 1.0 m (3.3 ft). This is possible at sighting distances of 2.0 km (1.2 mi) or less. Reciprocal setups. To limit the effects of curvature, two setups should be observed in opposite directions, one from each side of the crossing (fig. 4-3). Each setup should be unbalanced by the same amount,  $\Delta s = s_B - s_F$ . Such reciprocal setups are like those used in the 10-40 method for making a collimation check.

To compute the mean elevation difference,  $\Delta H$ , the observed difference from the backward setup,  $\Delta h_B$ , must be subtracted from the observed difference for the forward setup,  $\Delta h_A$ . The effects of curvature then cancel. If the adjustment of the instrument does not change between the setups, the effects of collimation error cancel as well, but atmospheric conditions may change significantly, particularly over water, during the time required to move the instrument from one setup to the other. Another technique must be employed.

Simultaneous observations. To reduce the effects of refraction, the two reciprocal setups should be observed simultaneously through the same atmosphere. This condition can be approximated by employing two instruments, one on each side of the crossing, in positions that permit the lines of sight to be equal in length, parallel, adjacent, and at the same height above the water or ground. In addition, to ensure that the lines of sight pass over similar terrain, the bench mark and the instrument on one side should be positioned at distances from the edge of the water that are equal to the corresponding distances on the opposite side.

Such a set of simultaneous reciprocal observations is not enough, however, to limit the effect of any difference in the respective collimation errors of the instruments ( $\alpha_1$  and  $\alpha_2$  in fig 4-4). The mean  $\Delta H$  becomes

$$\Delta H = [\Delta h_{A1} - \Delta h_{B2} - (C_1 - C_2) \times \Delta s]/2.$$



Figure 4-2.—Intercepting two targets to reduce pointing error.



∆h<sub>8</sub>=8-F

Figure 4-3.—The effects of curvture, c, cancel if a pair of reciprocal setups are observed and the results,  $\Delta h_A$  and  $\Delta h_B$ , are meaned.

Carefully determined values for collimation factors  $C_1$ and  $C_2$  might be used to correct the result, but at long sighting distances the magnitude of the correction is very large, as is the corresponding uncertainty.

Reciprocal collimation. To limit the effect of different collimation errors more effectively, four instruments with special attachments should be employed, two on each side of the crossing. During one set of simultaneous observations, each instrument pair should be repeatedly placed in reciprocal collimation; that is, the collimation error of one instrument should be adjusted to be equal and opposite to that of the other (The lines of sight thus obtained are analogous to those provided by the two compensator positions of the NI 002, section 3.3.3. In this way, the mean collimation error of the instruments is minimized, and, as shown in figure 4-5,

$$\Delta H = (\Delta h_A - \Delta h_B)/2$$

This principle is applied in the observing procedure used with Zeiss River Crossing Equipment. (See sec. 4.3.)

Another technique for reducing the effect of different collimation errors, requiring only two ordinary instruments, is to interchange the instruments between two sets of simultaneous observations, each set including the setups illustrated in figure 4-4. The mean of the results of the sets is the elevation difference for the section:

$$\Delta H = \left[ \left( \Delta h_{A1} - \Delta h_{B2} \right) + \left( \Delta h_{A2} - \Delta h_{B1} \right) \right] / 4$$

The collimation errors of the instruments can and do change somewhat when they are transported from one side to the other (as a result of temperature changes, vibration, and shocks during transport). increasing the uncertainty of the results. However, when less than firstorder precision is required, a procedure employing this technique may be used. (See 4.4.)

Multiple runnings. One complete running of a river crossing, then, must include the equivalent of four setups: two sets of simultaneous reciprocal observations, with the instruments for one set providing lines of sight reciprocally collimated to those provided by the instruments for the other set. To permit a check of the results, a second complete running must be made. Although both runnings are computed as though they were observed in the forward direction, they constitute a double-run section because forward and backward levelings (the two reciprocal setups) are incorporated within each running. The results of the two runnings should close within the tolerance for the order and class of the survey, as described in subchapter 3.7.7.

## 4.2 Preparations

In most vertical control projects, river crossings occur infrequently. As a result, when a river crossing is necessary it requires special preparations. To ensure that it is conducted correctly and efficiently, the project director should first review the procedures outlined in



Figure 4-4.—The effects of refraction, r, cancel when the results of a set of simultaneous, reciprocal observations are meaned; however, the collimation error of one instrument,  $\alpha_1$ , is only partly offset by the collimation error of the other instrument,  $\alpha_2$ .



a3=-a4 AhB=Bmean-Fmean

Figure 4-5.—Simultaneous reciprocal observations with four leveling instruments, the two instruments on each side having reciprocal collimation errors.

this chapter when planning the crossing. Then, the site must be selected and prepared, the necessary personnel and equipment must be assigned, and the work must be scheduled under suitable environmental conditions.

#### 4.2.1 Reconnaissance and Mark Setting

An unusually long line of sight may be necessary when a leveling route crosses a river, the mouth of a bay, a deep ravine or valley, or a passage between two islands. When possible, the route should bypass such obstacles, taking advantage of bridges, dams, breakwaters, or even land exposed at low tide. However, if a river crossing is unavoidable, select a location to satisfy the following conditions:

1. The crossing is readily accessible to the leveling lines to be connected. Any connecting spurs or lines should be as short as possible.

2. The lines of sight, from one side to the other, are as short as possible.

3. The lines of sight are as high as possible, clearing water by at least 3 m (10 ft) and clearing land or objects (such as trees, buildings, or bridges) by at least 0.5 m (1.5 ft). Consider the possibility of difficulties resulting from wave action, vessel traffic, or extreme refraction. For crossings of 1.0 km (0.6 mi) or more, the lines of sight should be at least 4.5 m (15 ft) above the highwater line to ensure sufficient clearance.

4. The lines of sight pass over similar terrain. For example, avoid a site with a steep bank on one side and a gentle beach on the other.

Additional requirements for ordinary instruments. In most cases, special instruments are necessary for river crossings; however, ordinary leveling instruments may be used if all of the following conditions are satisfied:

(

5. The instrument has a mechanism for tilting and measuring the tilt of the line of sight (Compensator instruments, such as the NI 002 and Nil, generally do not have such a mechanism) Two such instruments and four targets must be available for this type of crossing

6. Less than first-order precision is required for the survey.

7. The instruments can be quickly transported (preferably within an hour) from one side of the crossing to the other.

8. On each side, a bench mark can be set in such a position that a 3-meter leveling rod placed on the control point is entirely visible from the instrument station on the opposite side.

Setup specifications. On each side of the crossing, mark a location for the instrument station and set a permanent bench mark (sec. 2.4.2) to satisfy the following specifications. The setups for Zeiss River Valley Crossing Equipment are shown in section 4.3.2, "Setups," and the setups for ordinary equipment are shown in section 4.4.1, "Equipment and setups."

1. The lines of sight between each instrument station and the opposite bench mark should be equal in length, parallel, and at the same elevation.

2. The instrument stations should be at equal distances from the edge of the water or obstacle.

3. The bench mark should be 5 to 50 m (16-160 ft) from the instrument station. This backsight distance should be the same on each side of the crossing.

4. The bench mark should be no more than 1.0 m (3.3 ft) above or below the instrument station, to permit observation of a standard leveling rod. Furthermore, conventional geodetic leveling must be possible between the bench mark and the target station (which will be located next to the instrument station).

## 4.2.2 Personnel

Six individuals are normally required to conduct a river crossing: two observers, two recorders, and two rodmen. If an extra truck and equipment are available, a leveling unit assisted by one or two additional persons may perform the crossing. The unit chief coordinates the overall operation and serves as observer or recorder for one side. If two leveling units perform the crossing, one unit chief should be designated as the coordinator, and advantage should be taken of the opportunity to train the extra individuals present.

Observers. The observers must follow the appropriate crossing procedures, paying particular attention to the details of intercepting the targets. They must ensure that observations on each side are made simultaneously. Before leaving the site, the observers must verify that all required measurements have been accurately recorded. *Recorders.* Recorders should ensure that all steps in the crossing routine are completed in the correct sequence. They must be thoroughly familiar with procedures for recording and computing the data, either on paper or with a computer-recording system. Accuracy and efficiency are especially important, since many readings must be entered during a short period of time.

Rodmen. At each side of the crossing a rodman is responsible for one leveling rod. The rod is guyed in place over the bench mark for the duration of the crossing, the rodman ensures that it is plumb and secure at all times. If more than one setup is required between the target station and the control point, the rodman holds a second rod on a turning point. The rodman may be responsible for setting the target(s), and measuring and reporting the target heights, as required.

## 4.2.3 Equipment

Each side of a river crossing should have a leveling truck and a complete set of geodetic leveling equipment (appendix A). Equipment appropriate to the crossing routine must be available. (See subchapter 4.3 or 4.4.) Even if Zeiss River Crossing Equipment is used, an ordinary leveling instrument and rods should be used to level between the bench mark and the target station.

In addition, two-way radios should be available to permit communication between the two stations. Signal flags may be used if radios are not available. The observers raise the flags to indicate when foresight observations are being made.

When instruments must be interchanged between stations, as for the method employing ordinary instruments, a boat and operator may provide the most efficient service. Other means of transport, which may have to be obtained by contract, include a ferry or helicopter.

#### **4.2.4 Environmental Conditions**

Refraction effects are most unpredictable when a large vertical air-temperature difference exists, as is usually the case over water on a sunny, windless day. Shimmer can be extreme, making observations difficult. A crossing should be conducted when refraction effects are apt to be small and uniform—when the sky is overcast and the wind is moderate Air-temperature differences (sec. 3.6.1, "Measuring air-temperature differences") may be measured to assist in determining whether conditions are favorable.

Other environmental conditions may interfere with visibility, such as vessel traffic, large swells, wind waves, fog, or having to observe directly into the Sun. Whenever possible, plan the location and time of the crossing to avoid these difficulties.

The leveling unit(s) assigned to a crossing should check each morning to see if conditions are suitable; if not, the units should resume routine leveling. To permit this, the necessary equipment should be available in the leveling trucks when a crossing is assigned.

## 4.3 Instructions for Zeiss River Crossing Equipment

Because the compensator instruments employed by the National Geodetic Survey have no mechanism for precisely tilting and measuring the tilt of the line of sight, Zeiss River Crossing Equipment is normally used to perform a river crossing. The equipment includes instruments with rotary-wedge attachments that enable the observers to tilt the line of sight by a precisely measured amount.

The equipment has two other versatile features. First, so instruments need not be transported from one side of the crossing to the other, four instruments are employed. Each pair of instruments is placed in reciprocal collimation, enhancing accuracy by limiting the mean collimation error on each side to  $\pm 0.22$ .

Second, to permit more flexibility when selecting sites for bench marks, target stations are used. The stations are located close to the shoreline, where in methods using ordinary equipment the marks must be set. As a result, the marks may be set farther away from the shoreline, improving their quality and potential for survival.

#### 4.3.1 Description of Equipment

Two complete sets of Zeiss River Crossing Equipment are necessary, one for each side of the crossing. Each set, labeled "A" or "B," includes the following:

- 1. Two Zeiss Ni2 leveling instruments, marked "1" and "2."
- 2. Two rotary-wedge attachments for the Ni2 instruments.
- 3. One base plate for the Ni2 instruments.
- 4. Two targets.
- 5. One target column
- 6. One tribrach.
- 7. One tripod, with adjustable legs, for the instrument station.
- 8. One tripod, for the target station.
- 9. One auxiliary scale, 0.6 m in length, graduated in units compatible with the ordinary leveling equipment.

Ni2 instrument. The Ni2 is illustrated in figure 3-18 and described in section 3.3.3. Before the river crossing begins, the collimation error of each instrument should be checked. If one of the instruments is to be used for leveling from the nearby target station to the nearby bench mark, a micrometer attachment may be required. If so, the micrometer units should be compatible with the units of the leveling rod used. (See sec. 3.3.4, "Optical micrometer.")

Rotary-wedge attachment. Each rotary-wedge attachment slips over the objective end of an Ni2. The attachment includes a rotatable optical element, the wedge, to which is attached a graduated scale. To be sufficiently precise, the attachment must be of the best optical quality. The wedge is a slightly deviating prism, cut in the shape of a circle and mounted to permit rotation about an axis parallel to the line of collimation of the instrument. It functions in the same way as the prism that is used to adjust the collimation error of such instruments as the Zeiss Nil.

The wedge deflects the line of sight by a small angle (257") in a plane perpendicular to the line of intersection, or "edge," of the two main faces. (The "edge" does not physically exist because it is cut away when the prism is cut to a circular shape.) When the wedge is rotated so the "edge" is parallel to the vertical axis of the instrument (fig. 4-6a), the deflection is entirely in the horizontal plane of the instrument, causing no "tilt" in the line of sight.

However, when the wedge is rotated to some other position (fig. 4-6b), the line of sight is deflected vertically as well as horizontally. The vertical deflection,  $\delta$ , is observed as a tilt of the line of sight and is proportional to the sine of the rotation angle,  $\omega$ :

The attachment is designed to permit a maximum rotation angle of  $\pm 51^{\circ}6'$ . Thus, the line of sight can be smoothly tilted as much as 200" above or below the horizontal plane of the instrument.

Units representing equal increments of sine of the rotation angle are graduated on the attached scale (fig. 4-7), that rotates with the wedge. To avoid difficulty with algebraic signs, the unrotated position (no vertical tilt) is numbered "10", the position of maximum upward tilt is numbered "0", and the position of maximum downward tilt is numbered "20". Each wedge unit is graduated into ten subunits, and each subunit is large enough to permit estimation of tenths of a subunit. Therefore, the angle of tilt can be measured to one hundredth of a unit. Since each unit represents 20" (0.0001 radian), the smallest angle that can be measured is  $\pm 0.22$  (0.000001 radian).

The wedge is rotated by turning a knob that is provided with both coarse and fine modes of motion, similar to the focusing knob on the instrument itself. In the coarse mode, which is available for the entire range of rotation, the knob is slightly difficult to turn. Reversing the knob shifts the mode from coarse to fine. The knob remains in the fine mode, where it is much easier to turn, for only 0.5 to 1.0 wedge unit (depending on which part of the scale is in use), after which it reverts to the coarse mode. The line of sight can be directed most precisely with the knob in the fine mode. Therefore, make all final pointings in this manner.

*Base plate.* The base plate is an elongated metal casting designed to support the pair of instruments. It screws onto the head of an ordinary tripod. On the plate, positions



Line of sight not deflected vertically.





Figure 4-6.—Rotary wedge, unrotated (a) and rotated (b).


Figure 4-7.-Rotary wedge scale.

for each instrument are labeled and marked with red lines. The attached clamping screws, similar to those found on tripods, are used to secure the instruments.

*Tribrach*. The tribrach is an ordinary, separate, threescrew, leveling base intended for use with any surveying instrument provided with a matching stub. In this application, it provides the means for mounting and plumbing the target column on a tripod.

Target column. The target column (fig. 4-8) is a piece of channel section, with a stub suitable for mounting it in the tribrach. A circular level at the foot of the column provides a reference for precisely aligning the column with the direction of gravity. On the column are four pairs of precisely spaced studs on which the targets can be mounted.

The height stud is attached horizontally near the base of the column and terminates in a circular knob. The top of the knob is at precisely the same height as the target center defined by the lowest pair of studs. It serves as the control point from which leveling is conducted to the nearby bench mark.

Targets. Two metal targets, each 20 by 30 cm, are provided with each set of equipment. Each target is black, with a white isosceles triangle painted on it. Each target is equipped with a mounting bracket that fits over a pair of studs on the target column The horizontal centerline of each target can be mounted 0.00, 40.00, 80.00, or 120.00 half-centimeters (hcm) above the height stud (fig. 4-9).

Auxiliary scale The auxiliary scale is a short, wooden leveling rod, 0.6 m in length. It is used to level from the height stud on the target column to the nearby bench mark. When in use, it is held vertically against the target column, with the base resting on the top of the knob of the height stud (fig. 4-10).

The scale is graduated in half-centimeter units. The graduations are grouped and numbered in sets of ten blocks, which alternate black and white, forming a checkerboard pattern. The first ten sets of blocks are numbered from 0 through 9, to be read as tens of rod units (010., 020, 030., etc.). The second four sets of blocks are numbered from 0 through 3, with a large dot over each number to indicate that the readings should be preceded by "1" (100., 110., 120., and 130).

#### 4.3.2 Setups

On each side of the crossing, position an instrument station opposite a target station located on the other side. A correct pair of setups is illustrated in figure 4-11. The foresights, from each instrument to the opposite target station, must be as similar in length as possible, within 1 m in height and adjacent within 3 m.

Instrument station. Set up the instrument station as follows:

1. Set the adjustable-leg tripod firmly in the ground. It must not move during the entire running.

2. Orient the base plate on the tripod so the long axis is perpendicular to the line of sight to the opposite target station, with the indicated position for instrument 1 on the left.



Figure 4-8.—Target column.

3. Level the base plate roughly by centering the bubble in the circular level. There are no foot screws or similar devices on the base plate; therefore, level it by varying the lengths of the tripod legs or by changing the positions of the tripod feet in the ground. This ensures that the lines of sight of the two instruments will be at nearly the same elevation.

4. Secure the rotary-wedge attachments to the instruments.

5. Attach the instruments to the base plate in the indicated positions, aligning the red marks on the instrument tribrachs with the numbered, red marks on the base plate (fig 4-12). In this way, one foot screw of each instrument is positioned in line with the long axis of the base plate and the other two are aligned perpendicularly to it.

6. Set the three foot screws on each instrument to the middle position of their range of movement. The top edge of each screw should be even with the indexing ring marked on the dust skirt. This procedure limits the elevation difference between the two instruments Make all subsequent leveling adjustments by using only the screws aligned perpendicularly to the long axis of the base plate. Do not use the screws that are aligned with the long axis again during the entire running.



Figure 4-9.—Targets mounted on target column.

Target station. Set up the target station as follows:

1. Attach the target column to the tribrach and mount it on the tripod (fig. 4-9). It must not move during the entire running.

2. Secure two targets to the target column Position them at the same heights on each side of the crossing, placing the lower target at zero, if possible.



Figure 4-10.—Using the auxiliary scale.

3. One target must be above, and one below, the mean line of sight from the opposite instrument pair. The angle of tilt from the lower target to the upper target should be five wedge units or more. Consult with the observer on the opposite side to ensure that these conditions are satisfied. Adjust the height of the tripod or targets as necessary.

4. Plumb the target column by centering the bubble in the circular level, using the three leveling screws on the tribrach.

Leveling rod. Set a geodetic leveling rod on the nearby bench mark. To preserve the usual value for the rod constant difference when leveling from the nearby target station to the mark with only one setup, use rod 2 of the standard pair. If two setups are necessary, use rod 1. During the series of observations from the bench mark to the opposite target station, the rod constant is not important because only the low scale is observed.

Plumb and guy the rod in place with three wires attached at the top center. Rotate the rod to face the nearby instrument station.



Figure 4-11.—Setups for Zeiss River Crossing Equipment, top view (a) and side view (b).  $\Delta h = \frac{1}{2} [(\Delta h_{\text{Series A}} + \Delta h_{\text{Leveling B}}) - (\Delta h_{\text{Series B}} + \Delta h_{\text{Leveling A}})].$ 



Figure 4-12.—Instruments mounted on base plate.

#### 4.3.3 Observing Routine

One running of the observing routine includes, on each side, the measurement of two elevation differences. The first is obtained by conventional leveling from the nearby target station to the nearby bench mark. The second is obtained by a series of observations from the nearby bench mark to the opposite target station. The first is then measured again to ensure that the nearby target station has not moved. To compute the total elevation difference from the nearby bench mark to the opposite bench mark, the result of the series of observations from one side is added to the result of the conventional leveling from the opposite side. The total computed at one side is meaned with the total computed at the other side to obtain the elevation difference for the running. (See fig. 4-11.)

For a complete river crossing, two runnings are required that close as described in section 3.7.7. If both runnings cannot be completed in one day, the second running may be conducted another day. (See table 4-1 for a summary of the observing routine.)

If possible, record the observations on a computer, using backup forms if necessary. The instructions that follow include procedures for using the standard recording forms. Collimation checks. On each side of the crossing, check the collimation errors of the pair of Ni2 instruments, following a procedure described in section 3.3.7. Since the instruments are to be set in reciprocal collimation, adjustment is not necessary unless the difference in collimation errors exceeds 40" ( $C_1 - C_2 > 0.20 \text{ mm/m}$ ).

Perform a collimation check with the instrument to be used for the nearby leveling. One Ni2 may be removed from the base plate of the instrument station and used for this purpose. Adjust the instrument if it does not satisfy the tolerance for the order and class of the survey (table 3-1).

Leveling, nearby target station to nearby bench mark. On each side of the crossing, double run a section of leveling from the height stud of the nearby target station to the nearby bench mark. Follow the usual observing routine, satisfying specifications for the order and class of the survey and preparing a survey equipment record (i.e., line \*40\*) and beginning and ending running records (i.e., line \*41\*) (Pfeifer and Morrison 1980: vol. II.)

The auxiliary scale must be used on the height stud. Make sure the units of the scale are compatible with the leveling equipment; convert the scale readings to appropriate rod units, if necessary. If employing the micrometer leveling procedure, described in section 3.7.2, a special computation is necessary to obtain a high-scale reading Read the scale in the same manner as for the low-scale reading. Then, add the rod constant of rod 1 to the reading before recording it. (See fig. 4-13 for an example.)

Series, nearby bench mark to opposite target station. On each side of the crossing, make a series of observations to determine the elevation difference from the nearby bench mark to the height stud of the opposite target station. This series follows the pattern backsightforesight-backsight. The backsights each include one set of observations to "lower" and "upper" graduations on the leveling rod placed on the bench mark. The foresight includes four sets of observations to the lower and upper targets on the opposite side. Every set includes observations from both instruments at the station. Do not interrupt a series once it has started.

1. Prepare the beginning series record (line \*42\*). Include the following information: date, the instrument codes and serial numbers, the rod code and serial number, the survey-point serial number and stamping of the bench mark (or designation if no stamping), the survey-

#### Table 4-1.—Summary of crossing routine for Zeiss River Crossing Equipment

During each running, on both side A and side B:

Set up instrument station, target station, and leveling rod

Make COLLIMATION CHECKS with the leveling instruments.

Double run a section of LEVELING from the nearby target station to the nearby bench mark.

Observe a SERIES, simultaneously with the opposite side:

- 1 Point away from the opposite target station
- and level the instruments
- 2 Observe and record reciprocal collimation.3. Observe and record backsights to nearby leveling rod
- 4 Point toward the opposite target station
- and level the instruments.
- 5. Observe and record reciprocal collimation
- 6 Observe and record foresights to opposite target station
- 7. Point away reciprocal collimation foresight observations
- 8. Point toward reciprocal collimation foresight observations
- 9. Point away reciprocal collimation foresight observations
- 10 Point toward reciprocal collimation backsight observations

Double run a section of LEVELING from the nearby target station to the nearby bench mark.

During one day:	OR	During two days:	
COLLIMATION CHECKS LEVELING SERIES	RUNNING I	COLLIMATION CHECKS LEVELING SERIES LEVELING	RUNNING 1
LEVELING SERIES LEVELING	RUNNING 2	COLLIMATION CHECKS LEVELING SERIES LEVELING	RUNNING 2

point serial number and designation of the opposite height stud, the running number, the local time zone and time, the temperature units and temperature, the wind code, the sun code, and the initials of the observer, recorder, and rodman (Refer to the annotated recording form shown in fig. 4-14. An example is given in fig. 4-15.)

2. Record the positions of the nearby targets as indicated on the form. Communicate this information to the recorder on the opposite side.

3. Obtain the positions, T and t, of the targets on the opposite side. Enter the positions as indicated on the recording form, in units identical to those of the leveling rod. For example, with a half-centimeter rod, enter 0, 40, 80, or 120 hcm for each target. Compute the difference, D, between the target positions.

4. At the instrument station, turn both instruments to point away from the target station on the opposite side. This procedure is indicated on the recording form as "1". Carefully center the bubble in each circular level, using only the foot screws aligned perpendicularly to the long axis of the base plate.

5. RECIPROCAL COLLIMATION: Perform a reciprocal collimation immediately before every set of observations in the series, as follows:

a. Looking through instrument 1, focus precisely on a distant, sharply defined object. Then, turn the instruments toward each other. Looking through instrument 2, focus instrument 2 until the reticle lines of instrument 1 are sharply defined (fig. 4-16a). This procedure focuses both instruments at infinity.

b. Rotate the wedge knob on instrument 1 until the scale is set at 10.00. Make the final setting with the knob in fine mode. (See sec. 4.3.1, "Rotary-wedge attachment.")

c. Looking through instrument 2, rotate the wedge knob to bring the middle reticle line into coincidence with the image of the middle line of instrument 1 (fig. 4-16b). Make the final setting by turning the knob clockwise, in fine mode. Record the reading of the wedge scale.

d. With the knob still in fine mode, move the middle line of instrument 2 out of coincidence by a small amount and set it again. This time make the final setting by turning the knob counterclockwise. Record the second scale reading. Compute the mean, r, of the two readings

6. BACKSIGHT: Turn the instruments to point at the leveling rod placed on the nearby bench mark. First, with instrument 1 and then with instrument 2 obtain a set of observations by carefully following the instructions here. With each instrument, intercept the same lower and upper graduations on the low scale. Make all final settings with the wedge knob in fine mode.

a. Refocus the instrument to obtain a sharp image of the rod scale.

b. Set the wedge scale to 20.0 by turning the wedge knob clockwise (the scale moves counterclockwise). The setting need not be precise.

c. Observe the lower graduation: turn the knob counterterclockwise until the middle reticle line, moving upward on the low scale of the rod, appears at a position slightly above the first graduation encountered. The knob is probably in coarse mode. (If the line already intercepts graduation when the wedge scale is set at 20.0, go to the next graduation.)

d. Reverse the knob and, in fine mode, intercept the lower graduation. Record the graduation value, G, and the reading of the wedge scale.

e. Continue turning the knob (clockwise) until the middle line is slightly below the same graduation. The knob should remain in fine mode. Release your fingers and grasp the knob in another place. Then, reversing it, intercept the graduation. Record the scale reading. Compute the mean, l, of this reading and the reading obtained in the previous step, 6d.

f. Set the scale to 0.0, turning the knob counterclockwise. The setting need not be exact.

g. Observe the upper graduation: Turn the knob clockwise until the middle reticle line, moving downward on the low scale of the rod, appears at a position slightly below the first graduation encountered. The knob is probably in coarse mode. (As before, if the line intercepts a graduation when the wedge scale is set at 0.0, continue on to the next graduation.)

h. Reverse the knob and, in fine mode, intercept the graduation. Record the graduation value, g, and the reading of the wedge scale.

i. Continue turning the knob (counterclockwise) until the middle line is slightly above the same graduation. The knob should remain in fine mode. Release your fingers and grasp the knob in another place. Then, reversing it, intercept the graduation. Record the scale reading. Compute the mean, u, of this reading and the one obtained in the previous step, 6h.

7. This step may be performed after all crossing observations are completed. Compute the mean backsight interval, I, and the sighting distance,  $s_B$ , as follows (fig. 4-17).

a. Compute the distance, d, between the upper and lower graduations, in rod units:

d = g - G

b. Compute the angles of tilt from the settings on the lower graduation to the collimated lines of sight:  $l_1 - 10$  for instrument 1, and  $l_2 - r$  for instrument 2.

c. For each instrument, compute the angle of tilt from the lower graduation to the upper graduation:  $l_1 - u_1$ and  $l_2 - u_2$ .

d. For each instrument, compute the interval, *i*, from the lower graduation to the intercept of the collimated line of sight, in rod units:

$$i_1 = [d \times (l_1 - 10)] / (l_1 - u_1)$$
  
$$i_2 = [d \times (l_2 - r)] / (l_2 - u_2)$$

Ć

Ċ

ALL FORM 74-191 International octanic and preference of the second seco						
4 0 + YA. MO. DAY CODX HATRUMENT SKAIAL NO W CODE ROD SKAIAL NO. CODZ ROD SKAIAL NO. 2 TIME 4 1 + 8000708233 44565999 1 316 316 3						
PRON OF DESIGNATION 0002 HEIGHT STUD B	TO BE DESIGNATION Z	1400950C 1714 17512 R127				
SET STADIA Se STADIA SF UPP BACK <u>I</u> FORE <u>I</u>	LOW SCALE ALL HIGH SCALE BACK/FORE E BACK/FORE	Данза Дац. Икмаяка Е Дан Икмаяка				
V 069 205352 200 0485 3320	• 0692/ -28324 • 66/175 • 35245 • 96598	- 30 423 - 01 Auxiliary scale 69.25 Rod 1 constant + 592.50				
	B  I					
$(a) \underbrace{5 = 4 \ 0.5}_{1} \\ \underbrace{\Delta S = 0.5}_{1} \\ \Delta S$		$\frac{\Delta h - 283.235}{200}$				
12 3 4 9 1 9 8 6 6 207 3 2 9 2 1 4 5 3 1 6 6 1 2 0 7	a 34950 28322 a 96304 r 06628 1 r 65878	- 0 4 30426 - 04 Auxiliary scale 66.28 Rod 1 Constant + 592.50				
$ \begin{array}{c} -0.7 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 2 \\ 0 \\ 7 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1$	b	-27 630.78 28326 4				
		200 +1.41620 m B				
		Section check: -1.41618 F				
	<b>8 1 1 1 1 1 1 1 1 1 1</b>	Tolerance, K ≤ 0.1 km				
	<b>B I I I I I I I I I I</b>					

Figure 4-13.—Leveling from the nearby target station to the nearby bench mark, forward (a) and backward (b).

RIVER CROSSING SERIES, SIDE Aorb								
TAR. AND. DAY CODE INSTRUMENT SERIAL NO CODE INSTRUMENT SERIAL NO CODE NOD SERIAL NO. TARGET POSITIONS, THIS SIDE								
Ŀ								
FROM DESIGNATION 2 THE END ALLER ALUE AND WE DESIGNATION								
F	1	Т	I INSTRUMENT No. 1	INSTRUMENT No. 2	COMPUTATIONS 1			
E	T	Ħ	RECIPROCAL COLLIMATION		GRADUATIONS: 8 - G = d	G		
5	[•B	R	LOWER G T To Li	LOWED G T I L.	$l_i = 10$ $l_i = u_i$ $l_i$ $I_{eraff}$ +	Laneau		
ă	14	ŏ	UPPER g 1 2 U	UPPER g 1 1 1	$l_1 = r$ $l_2 = u_2$ $l_3$ $S_{START}$	Backsight		
L		1	RECIPROCAL COLLIMATION	Tarrana	TARSETS OFFOSTIS I _ T = D	T		
	5	Ī			$\frac{-10}{h_1 - 10}$ h	HMENT		
	ļĘ	] 6			$-\begin{bmatrix} \mathbf{b}_1 - \mathbf{a}_1 \end{bmatrix}$ $\mathbf{H}_1 \end{bmatrix} \mathbf{S}_1$	Foresight		
	1:	ļř			$-b_{a}-r$ $b_{a}$			
	<u> -</u>	Ļ			$= \frac{\mathbf{D}_1 - \mathbf{B}_1}{\mathbf{D}_1 - \mathbf{D}_2}$	Backsight		
	<b> </b>	<b>!</b> *	KOVER KOLAL COLLIMATION	L a st	-	Foresight		
	ĮŞ	I.	(10 WEDGE SCALE TO WEDGE SCALE)	(10 WEDGE SCALE TO WEDGE SCALE)	Δ	h frod units		
	1Ŧ	2	SETTINGS LOVER SETTINGS UPPER 10.00	SETTINGS LOWER SETTINGS UPPER	H <sub>2</sub> S <sub>2</sub> Co	nversion		
1±	] z	F	(TARGET INSTR 1 TARGET INSTR 1)	TARGET INSTR 2 TARGET INSTR. 2		∆h		
18	┢	h	RECIPROCAL COLLIMATION					
Į.	ł	ŀ:				80		
	ΙĔ	Ľ.						
	Ī	lë.				S		
	3	Ť.			•			
	[	Π	RECIPROCAL COLLIMATION		Mallal MATA Shirth States Street			
	s	7						
.	Ē	Â	10.00					
		Ę.						
H	4	Ľ						
FI.	_	LT.	RECIPROCAL COLLIMATION					
ŧ.	0	0	LOWEN	LOWER	<b>I</b>			
la:	Lu Lu	L.,	Correal M. i i		State			
*	<b>AA</b> K.	K3						
L								

Figure 4-14.—Annotated recording form for the series of observations from the nearby bench mark to the opposite target station.

e. Compute the mean backsight interval, I:

$$I = (i_1 + i_2)/2.$$

f. Compute the backsight distance,  $s_B$ , in meters. First compute the mean of  $l_1 - u_1$  and  $l_2 - u_2$ . Then,

$$s_B = [d_{rod\ units} \times \text{conversion}\ factor_{m/rod\ unit}]/$$
  
 $[(l-u)_{mean} \times 0.0001_{radian/wedge\ unit}].$ 

For d in half-centimeter units:

$$s_B = (50 \times d) / [(l-u)_{\text{mean}}]$$

For d in centimeter units:

$$s_B = (100 \times d) / [(l - u)_{\text{mean}}]$$

8. Turn both instruments toward (" $\uparrow$ ") or away from ("]") the opposite target station, as indicated on the recording form. On each instrument, displace the bubble in the circular level toward the station. Then, carefully recenter it.

9. Repeat the reciprocal collimation, step 5.

10. FORESIGHT: Before beginning each set of target observations, coordinate the activites with the observing team on the opposite side, using radios or visual signals. The two observing teams must begin each set simultaneously. For each set, carefully follow the instructions given here, first with instrument 1. Then, repeat the procedure with instrument 2. Make all final settings with the wedge knob in fine mode.

a. Point the instrument at the opposite target station and observe the lower target. Turn the wedge knob clockwise until the middle reticle line appears slightly below the lower target. If the middle line is already below the lower target, move it to a position substantially above the target before starting.

b. Reverse the knob and, in fine mode, intercept the target. Record the scale reading.

c. Continue turning the knob (counterclockwise) until the middle line appears slightly above the target. The knob should remain in fine mode. Release your fingers and grasp the knob in another place. Reverse it and, in fine mode, intercept the target. Record the scale reading.

d Repeat steps 10a through 10c until ten settings (five counterclockwise and five clockwise) have been made on the lower target. Compute the mean, b.

e. Observe the upper target Turn the wedge knob counterclockwise until the middle reticle line appears slightly above the upper target.

f. Reverse the knob and, in fine mode, intercept the upper target. Record the scale reading.

g. Continue turning the knob (clockwise) until the middle line appears slightly below the target Release your fingers and grasp the knob in another place. Reverse it and, in fine mode, intercept the target. Record the scale reading.

h. Repeat steps 10e through 10g until ten settings (five clockwise and five counterclockwise) have been made on the upper target. Compute the mean, a.

11. Repeat steps 8 through 10 to obtain a total of four sets of target observations. In step 8, prior to successive sets, point the instruments alternately toward ("]") and away ("]") from the opposite target station. The correct positions are indicated, as a reminder, on the recording form (fig. 4-14). Regardless of the direction in which the instruments are pointed, always displace the bubbles toward the opposite target station.

12. This step may be performed after all the crossing observations are completed. Compute the mean foresight interval, H, for each set as follows (fig. 4-18):

a. Compute the mean value for the reciprocal collimation of Instrument 2 from the values observed before and after each set:

$$r_{\text{mean}} = (r_{\text{before}} + r_{\text{after}})/2$$

b. Compute the angles of tilt from the settings on the lower target to the collimated lines of sight:  $b_1 - 10$ , for instrument 1; and  $b_2 - r$ , for instrument 2.

c. For each instrument, compute the angle of tilt from the lower target to the upper target:

$$b_1 - a_1$$
 and  $b_2 - a_2$ .

d. For each instrument, compute the distance, h, from the lower target to the intercept of the mean line of sight, in rod units. Recall that D is the difference in the position of the opposite targets, t - T:

$$h_1 = [D \times (b_1 - 10)] / (b_1 - a_1),$$
  
$$h_2 = [D \times (b_2 - r)] / (b_2 - a_2).$$

e. Compute the mean foresight interval, H:

 $H = (h_1 + h_2)/2$ .

PALE

			RIVE	R CROSSING SERIES SIDE A	PALE
				EXAMPLE	7.7
F		T			
	+	+	YA. MO. DAY CODE INSTRUMENT SERIAL NO	CODE INTERNALING CODE ROD SERIAL NO. TARGET POSITIONS, THIS SIDE	AUM NO
Ľ	4   2	<u>ــــــــــــــــــــــــــــــــــــ</u>	8007082321 1723385622	2/3/2   1/ 2/5/7/6/0/3/7/6   2/4/7/6/0/3 TOP  / 2/0 BOTTOM   0	
	180	4	DESIGNATION TO	DESIGNATION	TS SEAVER
0	[]]3	3	CAISSON USE 1000	02 HEIGHT STUD B Q //0/0/0//2/0/5/C 715 81/	/12   K W S
			INSTRUMENT No. 1	INSTRUMENT No. 2 COMPUTATIONS	
5		1	RECIPROCAL COLLIMATION	1048 1045: 10.46 GRADUATIONS: 102 - 97 = 5	. 97.
No.	•8	Ŕ	LONCH 097 1696 1700 16.98	LONER 097 1572 1574 15.73 6.98 12.92 2.70 2.32 +	2.33
6	[1]	0	UPPER 102 0405 0408 04.06	01111 102 0224 0225 02.24 5.27 13.49 1.95 18.9	99.33 hcm
		I	RECIPROCAL COLLIMATION	1060 1060 10.60 TARGETS OFFISTEL 120 - 0 = 120	0.
	ş	Ť	1710 1711 1712 1712 1715 17.12	16 00 1518 1514 15 17 1516 15.97 9.74 97.76	04 B3 (mail
	Ť	۶ ۲	1711 110 110 113 115 10.00	0705 0709 0709 0709 0709 0709 010198 10.58 6.14 85.06 678.7	14.85 ncm
	1	t	0840 0850 0833 0835 0847 08.38	0704 070 0101 0100 0498 07.03 8 94 72.35	
		1	RECIPROCAL COLLIMATION	055 1058 10.56	99,33
	~	÷	1719 1709 1705 1705 1702	1585 1589 1589 1591 1584 7.07	84.83
	n E		1710 1709 1702 1706 1702 17.07	1588 1510 1512 1586 1582 15.88 8.68 97.74	14.50 hcm
.	T	5	0139 0142 0142 0141 0936 10.00	0700 0702 0698 0698 0698 069	2.00 ncm/m
5	2	Ť	0840 0140 0140 0135 0812 08.51	0699 0701 0695 0699 0695	0.01200 -
823		1	RECIPROCAL COLLIMATION	1070 1071 10.70	
5.	5	ţ	1718 1715 1709 1708 1702 17.09	1535 1511 1537 1511 1510 15.88 7.09 97.91	11.0m
	Ę	ŝ	1719 1710 1708 1702 1703 10.00	1590 1519 1519 1511 1515 10.62 9.69 84.54 683.4	-614.0 m
	1	Ĩ	0840 08400840 0142 0838 08.40	0700 0619 0619 0102 0707 07.01 5.26 71.16	703.0 m
•	-	-	0842 08410843 0835 0838 0 + F m	0611 0613 0101 0103 8.11	
		ł	RECIPAULAL COLLIMATION	1514 1519 1510 1510 1510 1510 151	
	Ş	Ă	1710 1112 1112 1110 1107 17.11	1579 1577 1579 1578 1572 15.77 9.64 98.75	
	ĭ	B	0150 0149 0149 0145 0146 10.00	0105 0102 0701 0707 0701 10.52 5.25 95.46 691.2	
	4	Ť	0 845 0851 0841 0845 0845 0845	0704 0703 0708 0706 0702 07.04 8.73 72.16	
F.		t	RECIPROCAL COLLIMATION	1050 1050 10.50	
0153	D	R	LOWER 097 1700 1702 17.01	LOWER 097 1580 1510 15.80 7.01 12.92 2.71 2.54	
3	14	ò	UFFER 102 0408 0410 04.09	UPPER 102 0242 0243 02.42 5,30 13.38 1,98 19.0	
R	CA M	K\$			

Figure 4-15.—Series from the nearby bench mark to the opposite target.



Figure 4-16.—Reciprocal collimation, before (a) and after (b).



Figure 4-17.-Backsight, wedge settings on rod graduations.



Figure 4-18.—Foresight, wedge settings on targets.

f. Compute the foresight distance,  $s_{F_2}$  in meters. First compute the mean of  $b_1 - a_1$  and  $b_2 - a_2$ . Then,

$$s_F = (d_{rod units} \times conversion factor_{m/rod unit})/$$
  
[(b-a)<sub>mean</sub>×0.0001 radian/wedge unit].

For D in half-centimeter units:

$$s_F = (50 \times D) / [(b-a)_{\text{mean}}]$$

For D in centimeter units:

$$s_F = (100 \times D) / [(b-a)_{\text{mean}}]$$

13. Repeat step 4, with the instruments pointed toward (" $\uparrow$ ") the target station. Follow this with a reciprocal collimation, step 5.

14. BACKSIGHT: Conclude the observations of the series by repeating a set of observations to the leveling rod, steps 6 and 7. Intercept the same graduations used for the first set. If this is not possible, the instrument station has moved and the series must be started over.

15. Prepare the ending series record (line \*42\*). Include the time, temperature, wind code, and sun code.

16. This step can be performed after all the crossing observations are completed. Compute the elevation difference for the series, as follows:

a. Compute the backsight intercept, in rod units:

Backsight = 
$$G + (I_{start} + I_{rad})/2$$

b. Compute the foresight intercept, in rod units:

Foresight = 
$$T + (H_1 + H_2 + H_3 + H_4)/4$$

c. Compute the elevation difference,  $\Delta h$ , for the series, in meters. For half-centimeter rod units:

$$\Delta h = (\text{Backsight} - \text{Foresight})/200.$$

For centimeter rod units:

$$\Delta h = (Backsight - Foresight)/100$$

d. Compute the backsight distance,  $s_{R}$ , in meters:

$$s_B = (S_{\text{start}} + S_{\text{end}})/2$$
.

e. Compute the foresight distance,  $s_{F}$  in meters:

$$s_F = (S_1 + S_2 + S_3 + S_4)/4$$

f. Compute the total distance for the series:  $s_B + s_F$ .

Leveling, nearby target station to nearby bench mark. As before (sec. 4.4.3), double run the section from the height stud of the nearby target station to the nearby bench mark. Check for closure with the previous leveling.

#### 4.3.4 Final Computations

If time permits and if data are recorded on a computer, make the final computations for each running while at the crossing site. In any case, all observation records should be submitted to the project office at the end of the day's work.

Each completed crossing should include the records listed in table 4-2. Compute the elevation difference on a cover sheet for each running, as follows (fig 4-19):

1. From the leveling data for side B, enter the elevation differences,  $\Delta h$ , and distances, S, observed from the height stud of the nearby target station to the nearby bench mark. Use the data from the two sections leveled before and the two sections leveled after the series. Mean the elevation differences.

2. From the series data for side A, enter the elevation difference,  $\Delta h$ , and distance, S, observed from the nearby bench mark to the height stud on the opposite target station.

3. Add the leveling results from side B to the series results from side A. The values obtained are the observed elevation difference,  $\Delta h_A$ , and the distance,  $S_A$ , from bench mark A to bench mark B.

4. Repeat steps 1 through 3 with the leveling results from side A and the series results from side B. The values obtained are the observed elevation difference,  $\Delta h_B$ , and the distance,  $S_B$ , from bench mark B to bench mark A. Do not expect  $\Delta h_B$  to agree with  $\Delta h_A$ , since neither has been corrected for curvature or refraction.

_	Running record	Section	Form
1	River crossing record	BM A to BM B. forward	Cover sheet
	Side A		
	Survey equipment record (line *40*)	Collimation checks	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud A to BM A, forward	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud A to BM A, backward	NOAA 76-191 or 76-189
	Series records (line *42*)	BM A to height stud B	River crossing series
	Beginning and ending running records (line *41*)	Height stud A to BM A, forward	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud A to BM A, backward	NOAA 76-191 or 76-189
	Side B		
	Survey equipment record (line *40*)	Collimation checks	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud B to BM B, forward	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud B to BM B, backward	NOAA 76-191 or 76-189
	Series records (line *42*)	BM B to height stud A	River crossing series
	Beginning and ending running records (line *41*)	Height stud B to BM B, forward	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud B to BM B, backward	NOAA 76-191 or 76-189
2	River crossing record	BM A to BM B, forward	Cover sheet
	Side A		
	Series records (line *42*)	BM A to height stud B	River crossing series
	Beginning and ending running records (line *41*)	Height stud A to BM A. forward	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud A to BM A, backward	NOAA 76-191 or 76-189
	Side B		
	Series records (line *42*)	BM B to height stud A	River crossing series
	Beginning and ending running records (line *41*)	Height stud B to BM B, forward	NOAA 76-191 or 76-189
	Beginning and ending running records (line *41*)	Height stud B to BM B, backward	NOAA 76-191 or 76-189

Table 4-2.-Data set for a complete crossing with Zeiss River Crossing Equipment

If runs are conducted on separate days, the data set for each day should be as presented for running 1 above.

5. Compute the mean elevation difference for the running, in the forward direction along the line:

$$\Delta H = (\Delta h_A - \Delta h_B)/2.$$

As explained in subchapter 4.1, curvature and refraction error are largely eliminated from this result.

Also compute the mean sighting distance for the running:

$$S = (S_A + S_B)/2$$

Closing the crossing. Check the runnings for closure, as described in section 3.7.7. Make additional runnings until the section is closed within the tolerance for the order and class of the survey (table 3-1). All records should be filed with the other observation records for the line of leveling.

# 4.4 Instructions for Ordinary Leveling Instruments

In 1929 the Coast and Geodetic Survey published an observing routine suitable for performing river crossings with the Fischer level, the leveling instrument in use at the time. The routine employs the basic procedures necessary for a precise crossing: multiple observations of targets, simultaneous reciprocal observations from two instruments, and reciprocal collimation by interchanging the instruments from one side to the other. Since the instruments must be interchanged and since the targets are mounted on leveling rods directly over the bench marks, the routine requires that the site meet criteria (sec. 4.2.1) which are somewhat less flexible than those for the Zeiss River Crossing Equipment. The routine is presented here for use (with suitable leveling equipment) when less than first-order precision is required and the Zeiss equipment is not available.

#### 4.4.1 Equipment and Setups

At each side of the crossing, one leveling instrument, one leveling rod, and two targets should be available. Position each instrument across from the rod on the bench mark at the opposite side. A correct pair of setups is illustrated in figure 4-20. The foresights from each instrument to the opposite rod must be as equal in length as possible and within 1 m (3.3 ft) in height.

Instrument. After performing a collimation check (sec 3.3.7), set up the instrument as usual, placing the

#### RIVER CROSSING Zeiss Valley Crossing Equipment MO. DAY YR. 800708 \*42\* BESIN FROM 0/33 CAISSON USE TO 0900 WINDSOR 1 @1000 RUN 1205 (BM B) Δh Δh ន S LEVELING At Side B: LEVELING At Side A: --1.41618 12.4 40.5 +0.48137 Before F Before Before B +1.41620 40.5 -0.48165 12.0 Before 40.3 12.4 -1.41622 +0.48150 After After -0.48167 B +1.41620 40.3 12.8 After After В 40.4 12.4 Height Stud B to BM B -1.41620 (Mean) Height Stud A to BM A +0.48155 (Mean) SERIES At Side A: SERIES At Side B: +0.43665 693.7 BM A to Height Stud B +0.07250 703.0 BM B to Height Stud A s<sub>n</sub> = <u>706.1</u> -0.96675 s, = <u>743.4</u> BM B to BM A = $\Delta h_{\rm p}$ = +0.91820 BM A to BM B = $\Delta h_A$ = $\Delta H = \frac{1}{2} \times (\Delta h_A - \Delta h_B) = -0.94248 \text{ m}$ $S = \frac{1}{2} \times (S_A + S_B) = 0.72$ km

#### Figure 4-19.—Computation of river crossing with Zeiss River Crossing Equipment.

tripod firmly in the ground to ensure that it does not move throughout the crossing routine. The instrument must be equipped with a mechanism for precisely measuring the tilt of the line of sight. The mechanism of the Fischer level is an example of the type required.

The Fischer level (fig. 4-21) is equipped with a precise tilting screw, normally used to center the bubble in the level vial. The screw is graduated to serve as a micrometer, one unit representing one hundredth of a rotation The position of rotation can be estimated to one tenth of a unit, which in the 1934 model corresponds to a tilt in the line of sight of approximately 0."5. This precision is compatible with the precision with which the bubble may be centered. Thus, the tilting screw may be read to measure a tilt of the line of sight as small as  $\pm 0."5$ .

A compensator instrument does not usually possess such a mechanism. The compensator cannot be "tilted" by any means analogous to the tilting screw. An alternative mechanism is the rotary wedge (sec. 4.3.1) but this attachment is not available for most instruments.

*Rod.* Set a leveling rod on the nearby bench mark. Either rod 1 or rod 2 of the standard pair may be used, since only the low scale is required. Plumb and guy the rod in place with three wires attached at the top center. Rotate it to face the instrument on the opposite side. Targets. Use two targets that satisfy the requirements given in subchapter 4.1, "Intercepting targets." To permit precise measurement of the target's height against the rod scale, an index line should be clearly marked at the centerline of each target. ť

Ć

Mount the targets securely on the leveling rod Position one target above and one below the estimated intercept of the horizontal line of sight from the instrument on the opposite side. The distance between the targets should be no more than 1.0 m (3.3 ft) and should be similar to the corresponding distance on the opposite side of the crossing. The angle of tilt from horizontal to each target should be at least  $\pm 100''$  (20 units on the tilting screw of the Fischer level). Consult with the observer on the opposite side to ensure that these conditions are satisfied.

Enter the positions of the targets in rod units at the top top of the recording form for the first series. If possible, communicate these heights to the recorder on the opposite side.

#### 4.4.2 Observing Routine for Two Targets (Three-Wire Leveling)

One running of this crossing routine includes two sets of simultaneous reciprocal observations. The locations of the instruments for the first set are reversed for the second set. The results of each set are meaned to compute the elevation difference for the running.

Two runnings are required for a complete crossing. They can usually be completed in one day (See table 4-3 for a summary of the routine.) Conduct the following procedure simultaneously at each side of the crossing, recording the data on the standard forms for three-wire leveling (fig. 4-22)

l Prepare the beginning series record (line \*42\*). Include the following: The survey-point serial numbers and stampings of the bench marks (designations if no stampings), running number, set number, local time zone and time, temperature units and temperature, wind code, and sun code.

2. BACKSIGHT: Level the instrument and point it toward the nearby rod. Observe and record the intercepts of the three reticle lines with the scale, as for three-wire leveling. Compute the backsight intercept and distance.



SET 1



Figure 4-20.—Setups for simultaneous reciprocal observations with ordinary leveling instruments, top view (a) and side view (b).  $\Delta H = \frac{1}{2} (\Delta h_{\text{Set 1}} + \Delta h_{\text{Set 2}})$ .



Figure 4-21.—Fischer level, with micrometer tilting screw.

3. FORESIGHT: Check with the observer on the opposite side to begin the foresight observations simultaneously. Point and focus toward the rod on the other side of the crossing. Make 25 sets of three tilt measurements, as follows:

a. Tilt the line of sight to intercept the center of the upper target with the middle reticle line. Read the tilting screw (or equivalent scale) and record the value to the nearest tenth of a unit.

b. Center the bubble in the level vial, bringing the line of sight to horizontal. Read and record the tilt measurement as before.

c. Tilt the line of sight to intercept the center of the lower target with the middle reticle line. Read and record the tilt measurement as before.

4. Compute the foresight intercept and distance:

a Compute the mean tilts to the upper and lower targets and to horizontal (fig. 4-2): a, b, and r, respectively. b. Obtain the heights of the opposite targets and compute the difference between them, D.

c. Compute the distance, h, from the lower target to the intercept of the horizontal line of sight:

$$h=D\times(b-r)/(b-a)$$

d. To h, add the height of the lower target to obtain the foresight intercept in rod units.

e. Compute the foresight distance by the formula:

$$s_F = [(b-a)_{\text{tilt units}} \times \text{conversion factor}_{radian/tilt unit}]$$

For example, with the 1934 Fischer level (0.000025 radian/tilt unit) and centimeter rod units:

$$s_{F} = (400 \times D)/(b-a).$$

### Table 4-3.-Summary of the crossing routine for ordinary leveling instruments

#### During each running, on each side of the crossing

- 1 Collimation check
- 2 Set up instrument station and leveling rod
- 3 Position targets above and below level line of sight from opposite instrument
- 4 Observe backsight to nearby rod (1 set of readings)
- 5 Observe foresight series to opposite rod (25 sets of readings)
- 6 Move instrument to opposite side. Do not change focus
- 7. Observe foresight series to opposite rod (25 sets of readings)
- 8 Observe backsight to nearby rod (1 set of readings).

#### During one day

Side A	Side B		
COLLIMATION CHECK	COLLIMATION CHECK		
INSTRUMENT 1	INSTRUMENT 2	SET 1	RUN I
INSTRUMENT 2	INSTRUMENT 1	SET 2	
INSTRUMENT 2	INSTRUMENT 1	SET 1	RUN 2
INSTRUMENT 1	INSTRUMENT 2	SET 2	

HDA (47	А ГОПИ 76—189 71	2	TARGETS THI	s side	(A) AT 143	3.	1.0	, <i>CH</i>	AND 19	3.	O CAL. NATIONAL OCEAN	U. S. DEPAR NIC AND ATMOSPH	THENT O	F COMMERCE	PAGE
	RUN I SET I			THR	GEODETIC L EE-WIRE OB	ESS	EV SE		ig tions E	2	XAMPLE	- Second ord RIVER CR	ler, Cla 055/N	35 I (G	2.0F <u>5</u>
* 4 * 4	0 * YA. /40. /0 F * 80022	AY CODE INSTRUM 8 2 1 1	IENT SERIAL NO.	CODE 3/2	ROD SERIAL	3	но. Э	58	CODE 3 1 2	<u>но</u> р	SERIAL NO. A	Tischer Instr 200 UNIT = CE	umen ntime	t ter	2 8350     /
FR 00	OM BM DESIGNAT	ION	00	0   D   Z	B 133						Z BEGIN Q / 0 2 2	HE END E .	TEMPE DEGIN /80	HATURE W END / 850	S SERVER
		BACKSIC	нт								FORE	SIGHT			COMMENTS
	UPPER						Ļ	!	UPPER	_				1	
11p	HIDDLE	MEAN	BACK CENTER	\$υ -			-		AIDOLE	╀	MEAN	BACK CENTER	59 E.	111411	
	5	5	1	- 21	5		1-		S UPPER	╉	S ( EuF)	IAWER	- <sup>2</sup> L	5	TARGETS
<u> </u>					┟╌┎╌┠		$^{+}$	TT	22	╡	490	728		UPPER	- 127.00
[	844	┟┉┧┈┟──┼──┦┈╸╦╺╍┢┉ ╎			╏╌┙╼┶╸╧╺┨		t		22	ģ†	4.8.8	72.8		£ 9.1	
17	833	83.30	27.3	11	1		ſ	11	22	2	487	728		a= 22.36	
	822		1-1-1-1-1-1	17.7	2.2		T	11	22.	8	48.8	728	<u> </u>		
L	2499	83.30	1		22		ľ	11	Z 2.	7	486	72.5		LEVEL	7
	- 3				× 332		Γ	T	22	5	4 8.4	72.4		£ 6.8	
	83.3	0			= 730.4		Γ		22	3	48.6	72.5	111	r= 48.27	]
	- 1000	2			- 100.				Z 2.	8	48.7	72.6			]
<u> </u>	16.7	2		SB	- 73			$\square$	Z 2.	5	481	72.3		LOWER	= 75.00
I	H 100		I		+ 416.5				2 Z Z	<u></u>	48.0	72.1		2 7.6	D= 52.00
	$\Delta h = 0!!$	672 m.			= 423.8	m	٦Ľ	11	22	2	481	7 Z O		6= 72.30	
Ì	┟┊┊┊	] •					L		22	1	48.0	72.2	ļ L !		
<u> </u>	┟┼┥┥┥╧	▋┸┸┚┛┷			┟╌└─└──└──┤		F	╺┿┿	22	4	48.0	72.1		6-13	-
<b> </b>	<u>┨╺</u> ┫┥ <u>┫</u>	┇ ┱┉┰┉┎┉╌┰╌┎╶┰╴┲	<u> </u>		{		╞	-+-+-	22	4	48.1	72.0	L.,	24.03	x 52.00
	┠╌┞╌┦╼┠╍┠╍┝╍┢╼	┟┼┈┽╢╧┼	┝┛╹╤╹╤	i	┨─┬─┲╌┰┉┯┛		-		22.	71-	48.2	7215		49.94	
	┠┉╪╍╄╍┠╍╄╌┿╌	<b>┣</b> ──┬──┬──┬──┬──┬──┬──┬──	4	┝─┴─┟	<u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>		Ł		22	4	48.1	72./		6-6-	= 25.02
<b> </b>	<del>┨╶┨╌┇╌┥╶┥╺</del> ╸	┠┉┟┉┟┈┟╴└╴┊╌┧┉			╂╍└╍┶╍┶╍┤		ŀ	<del>-++</del>	- 22	2	48.0	16.1	<b> </b>	┢┝┟┖┊╸	+ 15.00
	╂╌╀╾╄╼┼╸┽╶┼╶╆╴				-		┢	++	22	2	48.0	72./		4	/00.02
╏╌┵╌	╊╍╋╍╄╼╄╌╊╌╋╍	┨╶╹╶┖╌┠┈╹╌┠┈	╂╌┦╌┦╌╿╌┊╌	┝┼┿╸	┠┲┲┲┲		$\mathbf{F}$	+++	22.	#	1 4 8:0		┝┼┿	+	528400
Į			1	┝┉└╴╩┈	<del>╏┼┼┼</del> ┨		┢	-+-+	27	;-	1 dela	771	┞┷	<del>┨                                    </del>	49 94
1	<del>╏┊╎╎╵╎</del>	<u>  · · · · · · · · · · · · · · · · · · ·</u>	<u> </u>	<u> </u>	+		F	++	+		481	721	<u> </u>		- 111 -
F	╊╋╋				1		H	-++	22	ź	1 480	1 701		۶ <i>۴</i>	-4/6.7 M
<b>_</b>	┠ <del>┝┝╞╎</del> ┼┼┊╴	<u> </u>		Hİ			F	-†-†	22	7	48.2	72.1	Hİ		1
I							T	- _ -	Z 2.	z.	48.Z	72.2			

5. Prepare the ending series record (line \*42\*), which includes time, temperature, wind code, and sun code. Compute the elevation difference and the total distance in the same way as for a setup of three-wire leveling.

6. Leaving the tripod and the leveling rod in place, transport the instrument to the opposite side of the crossing. Do not change the focus on the instrument during this operation.

7. Observe and record a second series, this time observing the foresight (steps 3 and 4) and then the backsight (step 2). The leveling instrument should not have to be refocused to observe the foresight. In conjunction with the series from the opposite side, this set completes the first running

8. Begin the first set of the second running from the same setup. After completing steps 1 through 5, transport the instrument back to the original side and observe a second set. Again, do not refocus the instrument between sets and begin the second set with the foresight. 9. Compute the results for each running on a cover sheet (fig. 4-23). First, mean the results from the two sides in each set. Then mean the results of the two sets to obtain the elevation difference for the running.

10. Check the runnings for closure, as explained in section 3.7.7. Obtain at least two runnings which close within the tolerance for the order and class of the survey.

#### REFERENCE

Pfeifer, L. and Morrison, N. L., 1980: Input Formats and Specifications of the National Geodetic Survey Data Base, vol. II: Vertical control data. Federal Geodetic Control Committee National Geodetic Information Center, NOAA/NOS, Rockville, Md. 20852, 136 pp.



Figure 4-23.—Cover sheet for a running with ordinary equipment.

# Chapter 5 PROJECT DATA PROCESSING

# 5.1 Introduction

Vertical control data should be compiled, checked, and edited as soon as possible after collection. To be included in the National Geodetic Vertical Network, the data should be formatted according to *Input Formats and Specifications of the National Geodetic Survey Data Base* (Pfeifer and Morrison, 1980: vol. II). These tasks, collectively termed "processing," must be performed so thoroughly that, after the project is complete and survey personnel have left the area, data may be verified and adjusted and elevations may be published without additional field work.

# 5.2 Organizing a Project Office

To correct mistakes, resolve discrepancies, and ensure compliance with project instructions, processing should take place in a locale that permits immediate communication with and monitoring of both mark setting and leveling units. A project office, centrally located within the project area, serves this purpose. It includes the personnel and equipment necessary to support and monitor mark setting and leveling units while processing their data. The mobile field offices of the National Geodetic Survey must perform many administrative and processing services in addition to those described here. [See National Geodetic Survey Operations Manual (Greenawalt and Floyd 1980) for further details ]

### 5.2.1 Personnel

The project office normally requires a project director, an assistant project director, and, if many units are operating, an assistant computer.

Project director. The project director is responsible for ensuring that project goals are achieved. The project director's duties include planning and coordinating the activities of the mark setting personnel and leveling units, establishing liaison with local officials, directing the processing effort, assigning reruns as necessary, reporting monthly accomplishments, and preparing a final project report. The director should visit each unit during a workday at least once a month, not only to check for compliance with procedural specifications, but to provide feedback concerning the quality and progress of the unit's work and to provide training as appropriate.

Assistant project director. The assistant project director ("computer") is primarily responsible for compiling, checking, and editing the data submitted by the mark setting and leveling units. This individual must be well-organized, attentive to detail, and adept at operating the necessary computers. A thorough understanding of mark setting and leveling practices is vital to successful and efficient troubleshooting when discrepancies are found in the data. A computerprocessing system cannot substitute for this type of understanding, which is gained best from experience in the field.

Assistant computer. In a large project requiring many leveling units for its timely completion, an assistant computer contributes the added staffing necessary to keep the processing up-to-date. This position is suitable for rotational training of senior mark setting and leveling personnel, permitting them better to understand the significance of their work and the purposes behind many field specifications. An individual who has served as an assistant computer is better prepared to supervise a unit conducting a special project at a location remote from the project office.

## 5.2.2 Equipment for Data Processing

In addition to standard items of office equipment, the project office should be equipped with the specialized desk calculators, computer system, and telephones necessary to process data efficiently. If leveling data are recorded with a computer in the field, at least one matching system should be available in the office for troubleshooting purposes.

If all data are handwritten on forms or in books and are not to be keyed into a computer data base, a programmable desk calculator is sufficient for processing. It should have a large keyboard and display, as well as an option for printing. Such a calculator is also useful for preliminary analysis of computer-recorded data

For processing computer-recorded data the following components are essential: A terminal with a keyboard and a CRT display, which is preferable to a printer for on-line editing; a data retrieval and storage unit, such as a reader for magnetic tape or disks; a high-speed printer capable of producing permanent copy; and the computer itself. The high-speed printer and the computer, if not physically located at the office, may be accessed by telephone. This requires either a "dataphone" or a standard telephone used in combination with an acoustic interface. Arrangements must be made for printouts to be forwarded to the project office when they are needed

Study the maintenance and operating instructions that accompany the computer system. Learn how to connect various components together and how to set the switches correctly for different modes of operation. In particular, learn the symbols or "error flags" generated when the hook-up or switch configuration is incorrect. Do not confuse these with flags generated as a result of errors in data. Much frustration and loss of time can be prevented if mistakes in maintenance and operating procedure are quickly recognized and corrected.

Computer equipment may operate erratically or fail when exposed to dust, extreme heat, power fluctuations, static electricity, and vibration. Keep the project office clean, the temperature moderate, and electrical cables and outlets in good condition. Clean keyboards and recording heads weekly. Ground all computer components. Check cables and outlets on office trailers frequently, since they may suffer from exposure to weather during moves. Have available a list of local servicing agents and their phone numbers so repairs may be obtained quickly.

The computer itself must be protected from power fluctuations, especially if a temporary loss or reduction of power results in loss of memory or circuit damage. For a computer installed in the project office, an uninterruptable power source with a backup battery bank may be necessary. Voltage regulators are sufficient for most other components.

If a telephone line is used to transmit data to a computer at another location, interference occasionally causes loss of data. At the project office, have available backup on paper, tapes, or other media so lost items of data may be restored after transmittal

Magnetic tapes and disks Before the project begins, prepare a sufficient supply of the magnetic tapes or disks on which data are to be recorded and stored. Label each with the job code plus a unique number. This identifier should be written on the backup form for any data recorded directly on a tape or disk by the leveling unit Similarly, the identifier for a tape or disk on which data are stored in the project office should be noted on the status sheet for each file of data. Such a crossreferencing system is especially helpful when processing data for numerous leveling units or lines.

Most tapes or disks may be reused. Indicate each use by placing a small check next to the identifier. After a day's set of observations has been printed, checked, and recomputed, the tape or disk may be erased. Tapes or disks containing backup files of data for the project should be not be erased and reused until notice is received that the data have been stored in the data base. Depending on the quality of the media, the chances of acquiring errors due to dirt or strain on a tape or disk increase each time they are reused. Do not use a tape more than five times.

#### 5.2.3 Organizing the Data

Organize preliminary data for the project as well as original records received from the field. A logical and well-maintained filing and labeling system permits processing to proceed systematically and efficiently, especially when the project office supports numerous units operating on many different lines.

Preliminary data. One copy of each of the items listed and discussed in section 2.3.1 should be available at the project office. In addition, obtain from the National Geodetic Survey abstracts of unadjusted, corrected observed elevations for all previous leveling surveys in the project area. These are called "phase one abstracts" or "summary sheets." They are identified chronologically by accession number ("L-number"). Consult the indexes of the appropriate 30' quadrants of Vertical Control Data to determine the accession numbers to order.

File quadrants of vertical control data, horizontal control data, and summary sheets separately in numerical order. Label each cover sheet with the numbers of the project lines or loops covered. File miscellaneous data, such as tide station reports and unpublished descriptions, by line.

Maintain a notebook or file in which project instructions, correspondence, processing status sheets, statistics, notes, and miscellaneous preliminary data may be stored for quick reference. This is very valuable when preparing the project report.

Index Maps of Control Leveling or Geodetic Control Diagrams are suitable maps to display for an overview of the project area. Highlight and label the project lines and note other useful cross-referencing information, such as accession numbers of past surveys.

Original records. Prior to leveling each line, obtain from the mark setting unit bound volume of descriptions for all control points to be leveled, a set of smooth plots, and a set of logs, prepared as described in subchapters 2.3 and 2.4. On a daily basis, obtain from the leveling unit observations and computed elevation differences for each section, recorded as described in subchapters 3.7 and 3.8. If the unit transmits data over a telephone line directly into the computer, obtain a backup tape, disk, or printout daily. The unit should also submit pages of the handwritten field abstract as they are completed.

Store original records in a fireproof safe or file as soon as they are received. The storage container should protect computer-recorded data from magnetic interference as well. Store daily data packages in chronological order. Store backup tapes or disks in numerical order and maintain an index of their contents. Store master program tapes or disks in the same protective location.

Original recording forms and printouts of original observation records are repeatedly referred to during processing in the project office and, later, during verification and adjustment. To protect the records and make them readily accessible, bind large quantities of forms and printouts chronologically into volumes of manageable size that correspond, if possible, to the files of data stored on the computer. File small quantities (such as records for a river crossing) in labeled envelopes. Editing original records. Make notes and corrections on original records with ink of a different color from that used in the field. Each individual who performs editing should adopt a color, such as red or blue, and use it throughout the project. Do not repeatedly mark changes when trying to resolve a discrepancy. Show the final correction and affix your initials.

Once records are stored in computer memory, they may be inadvertently altered Whenever a record is edited in the computer file, note the change on the original recording form or on a printout of the original computerrecorded data. The note must explain the change. For example, if a distance is remeasured to resolve a discrepancy in a description, state this fact. If a surveypoint serial number is changed, explain why the original number was wrong and why the new number was chosen. If there is no backup record or other means to validate a change, do not alter the record.

Identifying the data. Key items of identifying information must be given in the heading or on the cover sheet for each file of data. Heading requirements for data stored in computer files are explained in *Input Formats* and Specifications of the National Geodetic Survey Data Base (Pfeifer and Morrison 1980: vol. II).

For adjustment purposes, each project is assigned a job code in the project instructions. Use the code to identify the beginning and ending of each file of data stored on magnetic tape or other media.

Every survey included in the national vertical network is assigned a unique accession number. Normally, each line of leveling is assigned a separate accession number. However, if a survey covers a small area and includes multiple lines, the entire project may be assigned a single accession number, in which case each line is assigned a "part number."

An accession number includes the letter "L" and five numerals. A part number is indicated by appending the letter "L" and the line number to the accession number. For example, "L20815.L1" or "L1302-L5".

If the project instructions include accession numbers, use them to identify the data for each line. If they are not assigned, use line numbers until accession numbers become available.

A group of lines is normally assigned an area title in the project instructions. (See fig. 2-3.) The area title must be included in the heading of each file of data for the group. Likewise, the line title, also assigned in the project instructions, must be included in the heading of each file of data for a line. Change the line title as appropriate to indicate the actual direction and route of leveling, but do not change the area title.

Neatly label each binder of original or supporting records for a line with the job code and line number (or accession number if available), name of the corresponding computer file, project title including the state, beginning and ending dates of leveling on the line, accounting task number, agency acronym and field party name, and the initials of the project director. (See fig. 5-1.)

*C2*L23021.L2.HGF1
BASIC NET A, REGION 1, GEORGIA
8/2/78 - 10/5/78
NGS Field Party G-37 DCF
Task 8362241V

Figure 5-1.—Sample label for project records.

#### 5.2.4 Submitting Data to the National Geodetic Survey

Project data for inclusion in the National Geodetic Vertical Network should be submitted to the National Geodetic Survey in the formats specified. This permits efficient verification and adjustment.

Data should be recorded on the standard forms illustrated in the examples throughout this manual. Blank forms are included at the end of the manual. Such records may then be keyed onto magnetic tape or other media for entry into the data base. Data recorded in a computer may be formatted either during leveling or during processing. If computer-recorded data are submitted, include all supporting documents, such as backup forms and printouts, showing the changes made during processing.

*Records for submission*. Submit the following original records for each line:

1. Descriptions, bound in line order. For points not leveled, submit descriptions bound by areas corresponding to the Geodetic Control Diagrams.

2. Rough plots or map-type logs, numbered in line order.

3. Smooth plots, numbered in line order.

4. Leveling observations, bound chronologically, including a cover sheet for the line, collimation and compensation checks for the instruments, and calibration records for the rods. If observations are recorded with a computer, submit bound sets of both the original printouts of the data and any backup forms that have been prepared. If calibrations are performed at the National Bureau of Standards, the records should be forwarded directly to the National Geodetic Survey.

5. River crossing observations, as in item 4.

6. Field abstract sheets, bound in line order. If the abstract is computed directly from computer-recorded observations, submit the final printout; in this case a hand-written abstract need not be submitted.

7. Comparison of new and old elevation differences along the line, showing ties.

8. Project report.

If data have been formatted according to NGS, in addition to the original records for each line, submit two complete data sets for the project on magnetic tape or other media. One set should include files of descriptive data and the other files of leveling results.

If the files are entered directly into the National Geodetic Survey computer system during the project, prepare and transmit a release letter, identifying the files that may be accessed for verification at headquarters, immediately after processing each line as described in this chapter. Release computer-recorded observations (HG files), a description file (HAFF), and a file of leveling results (HGZAB). Arrange for obtaining printouts of these records from headquarters. An example is shown in figure 5-2.

Forward all records to the Quality Control Branch (C174) of the National Geodetic Survey. Prepare and forward a memorandum of transmittal (fig. 5-3) itemizing the records sent and the files released. Include a copy of the letter in each package. Send records by registered mail, requesting a return receipt. At the project office retain backup tapes or printouts sufficient to reconstruct lost records, until notification is received that the records have been archived.

**Project report.** The project report is a permanent record that summarizes project accomplishments. It describes the specific equipment and procedures employed to meet project conditions and requirements. The report provides information useful for verification and adjustment, including detailed explanations of unusual or special features of the project lines. If a project extends over more than one season, prepare a separate report for each season's work. The report should specifically cover the following items: 1. Title page. List the type of report (season or project), project title including the State, accounting task number or numbers, beginning and ending dates of field work, agency name, all HGZ L numbers that are included in the report, field party name, and the name of the project director.

2. Authority and purpose. State the date and title of the project instructions and the date project instructions were received. State the purpose of the survey.

3. Location. Briefly describe the project area, including the State or States in which it is located. Note the number of lines, their general configuration, and their total distance. Also indicate where the project office was located.

4. Scope. State the beginning and ending dates of the survey and the extent to which the project requirements were satisfied.

5. Personnel. List all personnel who participated, including their major duties, and dates of departure or arrival if they did not participate in the entire project.

6. Statistics. Present progress statistics for mark setting, including the following: total distance of lines routed, total number of control points searched for and set, and totals of recovered control points that will be leveled, bench marks set in bedrock, bench marks set in structures, sleeved rod marks set, and unsleeved rod marks set. Also list the totals of recovered points that will not be leveled, points destroyed, points not recovered, and staff days devoted to reconnaissance and mark setting.



UNITED STATES DEPARTMENT OF COMMERCE National Oceanic and Atmospheric Administration NATIONAL OCEAN SURVEY Rockville, Md 20852

22 March 1982

TO: C174 FROM: Level Party G-35 GMH

SUBJ: Release of files for South Region 4, SAN ANTIOIA-DEL RIO-JUNCTION-BRADY-FREDERICKSBURG TX AREA, L24551.L2-DEL RIO VIA ROCK SPRINGS AND JUNCTION TO BRADY TX, Task Number 83624100

The following files and jobs are completed at the field party processing level and are released for further processing.

JOB CODE	LINE NO.	FILE	JOB NO.	DATE PRINTED
<b>*</b> D4*	337	L24551.L2.HGF1	9314 🗸	3/22/82
		L24551.L2.HGF2	9309 🗸	91.
		L24551.L2.HGF3	9307 🗸	11
		L24551.L2.HGZFFAB	731 🗸	**
*D4*	337	L24551.L2.HAF1	9332 V	3/22/82
CKSPSN	of the HA file	of Job Code *D4* is Job Number 93	17. 🗸	

Note: All above files are on W3M NGS001 DSN cc: Cl71, Cl42

NOAA FORM 61-29 U. S. DEPARTMENT OF COMMERCE (12-71) NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION	REFERENCE NO.
LETTER TRANSMITTING DATA	NGSG36:1981 No 07 sheet1/2 DATA AS LISTED BELOW WERE FORWARDED TO YOU BY (Chock):
	ORDINARY MAIL AIR MAIL
то:	REGISTERED MAIL EXPRESS
Director, National Geodetic Survey	GBL (Give number)
National Ocean Survey, NOAA	CERTIFIED MAIL
NOCKVILLE, MD 20652	NAV 11 1001
	NUMBER OF PACKAGES
	[2 (1 roll & 1 pkg.)
etc. State the number of packages and include an executed copy of the tion the original and one copy of the letter should be sent under sep receipt. This form should not be used for correspondence or transmitti	e transmittal letter in each package. In addi- parate cover. The copy will be returned as a ing accounting documents.
L24562-1 (line 380) GILA BEND VIA SENTINEL	AND YUMA AZ TO EL CENTRO CA
(rool) GILA BEND, ARIZ. SMURR, ARIZ. THEBA, ARIZ. (15') SENTINEL, ARIZ. (15' AZTEC SE, ARIZ. AZTEC HILLS, ARIZ. DATELAND, ARIZ. TEXAS HILL, ARIZ. MOHAWK MTS. NW, ARI MOHAWK, ARIZ. TACNA, ARIZ. WELLTON MESA, ARIZ. WELLTON, ARIZ. LIGURTA, ARIZ.	FORTUNA, ARIZ. YUMA EAST, ARIZCALIF. YUMA WEST, ARIZCALIF. ) ARAZ, CALIF. GRAYS WELL NE, CALIFARIZ GRAYS WELL, CALIF. OGILBY, CALIF. (15') MIDWAY WELL, CALIF. Z. MIDWAY WELL NW. CALIF. GLAMIS SW,CALIF. HOLTVILLE EAST, CALIF. HOLTVILLE WEST, CALIF. EL CENTRO, CALIF.
* All topo maps are 7-1/2 minute quads unle	es otherwise noted.
pkg 2 2 Books NOAA form 76-186 BM Des (book 2-295 sheets) 3 Books T.I. XOFF sheets for HG HGF1 (1174 lines) HGF3 HGF2 (742 lines) HGF4	files (812 lines) HGF5 (776 lines) (718 lines) HGF6 (564 lines)
FROM: (Signature)	RECEIVED THE ABOVE
Chine from 9.7.	
Return receipted copy to:	
r	
Bob W. Martine National Geodetic Survey BXBX POB 2141	
Orofino, ID. 85344	

NOAA FORM 61-20 SUPENSEDES FORM CAGS 413 WHICH MAY HE USED

í

(

Ę

NUAA FORM 61-29 (12-71) NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION	REFERÈNCE NO.
LETTER TRANSMITTING DATA	NGSG36:1981 No 07 sheet 2/2 DATA AS LISTED BELOW WERE FORWARDED TO YOU BY (Chock):
	ORDINARY MAIL AIR MAIL
ТО: Г Г ٦	REGISTERED MAIL EXPRESS
	Gat (Give number)
	CERTIFIED MAIL
ATTN: C174	
ب ا	MAY 14. 1981 NUMBER OF PACKAGES
	2 (1 roll & 1 pkg)
NOTE: A separate transmittal letter is to be used for each type of da etc. State the number of packages and include an executed copy of the tion the original and one copy of the letter should be sent under sep receipt. This form should not be used for correspondence or transmitte	ta, as tidal data, seismology, geomagnetism, e transmittal letter in each package. In addi- parate cover. The copy will be returned as a ng accounting documents.
L24562-1 (line 380) GILA BEND VIA SENTINEL	AND YUMA AZ TO EL CENTRO CA
pkg 2 40 Sheets NOAA form 75-82A 2 Sheets NOAA form 76-186 7 Sheets New - Old Compari 3 books NOAA form 77-82 4	Station Descriptions BM Descriptions sons 0/41 cards book 1 140 sheets
1 sheet File release lett	book 2 176 sheets book 3 106 sheets er
FROM: (Signature)	RECEIVED THE ABOVE
TT	(Name, Division, Date)
Return receipted copy to:	
r "	
L J	

NOAA FORM 61-20

SUPERSEDES FORM CAGS 413 WHICH MAY BE USED

Figure 5-3b.—Sample form (NOAA 61-29) accompanying transmitted data.

For leveling, present the following: total progress along lines, double-run progress, single-run progress, total distance leveled, distance leveled as reruns, number of sections, number of setups, and total staff days devoted to actual leveling.

Also, note the total number of personnel days spent during the project and a breakdown of days spent in the field, office, training, travel, holidays, sick leave, and annual leave. Compute the leveling progress per staff day for the entire project.

7. Liaison. Note major contacts with other agencies and individuals interested in the project. Discuss cooperative efforts and any publicity received.

8. Mark setting. Briefly describe the procedures and equipment used, referring to the manuals followed. If unusual bench marks were set, identify and describe them in detail.

9. Leveling. State the leveling specifications that were in force. Describe the equipment, including a list of instruments, rods, and recording equipment with serial numbers and the dates they were in use.

10. Line narratives. Briefly describe each line, including line number, accession number, beginning and ending dates of mark setting and leveling, topography and climate, features of the routing such as control point spacing and frequency of connections, unusual points leveled, unusual procedures, river crossings, ties established, and results of the comparison of new elevations with those from a previous survey along the same line. Mention specific sections that required additional work as a result of preliminary analysis

11. Loops. Discuss closures of the leveling with both previous and concurrent surveys. Describe areas which may require additional leveling in the future.

12. Recommendations. Report suggestions for improvements in instructions, equipment, and procedures and recommendations for future work in the project area

13. Attachments. Include as an attachment to the report a simple sketch of the project area showing completed lines, junctions, and loops. A section of the State Index Map of Control Leveling is sufficient, with progress marked and lines clearly labeled Also attach the following: A copy of the release letter, a list of errors detected in the synoptic file during processing, sketches showing loop closure computations, and the original copy of the project instructions together with any amendments.

### **5.3 Processing Descriptions**

For each line, the mark setting unit provides three sets of records to the project office: logs, smooth plots, and descriptions. As they are received, the logs and smooth plots should be checked off on the Index to Topographic maps. (See fig.2.6) The logs are forwarded to the leveling unit, which in the course of work checks the information given on them. The smooth plots are used by the project office to check, in part, the information given in the descriptions. The descriptions themselves are checked by the project office. Descriptions for points not leveled should be processed separately, following the same routine.

The ultimate product of these verification procedures is a set of complete and correct descriptions for all the control points leveled The steps in processing are discussed next and summarized in figures 5-4 and 5-5.

#### 5.3.1 Creating the Description File

If the project office possesses facilities for checking descriptions with a computer, the first step in processing is to create a file of descriptions for the line. The data supplied by the mark setter on the description forms must first be keyed onto magnetic tape or other media through a computer interface.

The descriptions for a single line typically number several hundred characters; thus numerous typographical errors may be entered inadvertently during keying. To prevent errors from entering the file, provide prompts for each type of data entered. The prompting program should check the responses keyed whenever possible.

After a batch of descriptions is keyed, transfer the data into a computer file Label the file with the appropriate line number (or accession number) and the letters "HA". For example. "L35 HAF1" or "L23021-L2-HAF1". To simplify processing, all the descriptions for one line need not be put into one file at first As each file is created, assign a file number in line order: "L35 HAF1", "L35 HAF2", and so forth. After proofreading and editing are completed, the files should be merged into one final file for the line: "L35 HAFF".

The project office may be required to process vertical descriptions for control points that are not leveled. These should be keyed and transferred to a separate file for each area that corresponds to a Geodetic Control Diagram Label each file with the area code and the letters "HAFF." For example, "CK HAFF" for the area covered by the diagram NI-11, SAVANNAH

As each file is created, obtain a printout of its contents. The printout may be generated in conjunction with an initial run of the format-checking program, discussed in section 5.3.2. This printout serves as the reference for proofreading and editing the descriptions To keep track of the status of editing, stamp the first page with a checkoff list like that in figure 5-6.

#### 5.3.2 Checking Individual Descriptions

Descriptions must be proofread and edited to ensure that the required information is entered, that it is correct and internally consistent, and that it is properly coded and formatted. Without a computer, these items must be checked by systematically and thoroughly proofreading the descriptions, noting changes directly onto the description forms. With a computer, many items may be automatically checked; however, proofreading still plays an important role.

# **VERTICAL DATA PROCESSING**



Figure 5-4.—Flow chart of project data processing.

Spot checking. As soon as possible after receipt of the descriptions from the mark setting unit, the mark setting foreman or project director should spotcheck several of the descriptions in the field. In addition to examining the construction of new monuments, check the choice of routes and the logic of the instructions given. Verify the plotting, the distances and directions, and the physical descriptions of landmarks and reference objects. Note corrections directly on the description forms so the changes are included when keying the data into a file. If a printout of the file is already available, take the printout to the field and note corrections directly on the printout.

*Proofreading.* After obtaining a printout of the file as described in section 5.3.1, check the heading. Then read through every description thoroughly, looking especially for errors that may not be detected by a checking program in the computer. Check that the designation is consistent with the stamping. Check and correct the spelling of names of counties, cities, towns, companies, agencies, and individuals. Look for and correct typographical errors, particularly in the text.

Check the internal consistency of the data given, including the following: That the driving distance from the nearest city or town agrees with the total of the driving distances given in the instructions to reach the point; that reference objects are sufficiently described; that distances are given with appropriate precision; and, that distances and directions make logical sense (for example, a point cannot be south of the centerline of one road, east of the centerline of another, and also southwest of their junction). Finally, check that the latitudes and longitudes of all points new to the data base are correct by scaling them from the smooth plots. Refer to the original records whenever necessary to resolve discrepancies. Note points which must be checked during a visit to the field. When the proofreading is complete and all changes are noted, indicate this fact on the front of the printout. An example of a proofread description printout is given in figure 5-7.

*Editing.* Next, while on-line to the computer, systematically edit the file. On the printout, check off each change as it is made. On the front of the printout indicate when this preliminary editing is complete. Further editing may be required after running a format-checking program with the computer.

The program should check for the following: all entries are in the correct format, the required information has been given for each type of description, as explained in section 2.4.4, and conversions from metric to English units are correct. After the program is run, the computer should respond with a list of the lines in error and their locations. Correct these systematically, looking up the lines in the printout and noting the changes as they are made to the computer file. Then, rerun the checking program and continue editing until no errors are detected. Note each running of the program on the front of the printout and indicate whether or not a rerun is necessary.

When editing is complete, use the printout as a guide to transcribe all changes directly onto the original description forms. The original records are the only source to resolve discrepancies found in the description file in the future.



Figure 5-5.—Flow chart of data files for a line of leveling.

FILE	L24586		HAFI
PROOFR	EAD:	FIRST	EDIT:
Cu	5 3/26/80	as	3/26/80
FORM	IAT-CHECK	ING PR	OGRAM:
JOB #	DA"	ГЕ	RERUN?
1321	3/26		Y
2018	3/29		y y
2025	3/29		N
CORRECT	: CUE 3/29		

# Figure 5-6.—Sample check-off list for station-description editing.

If more than one file exists for a line, merge the files into a final file after all have been completely proofread and edited. At this stage the individual printouts are no longer useful and the processing progress for the file should be recorded on a status sheet (fig. 5-8).

#### 5.3.3 Checking the Entire File

Once the entire file of descriptions for a line is available, it should be checked for agreement with both the existing data base and the observation file for the line. A file of descriptions for points not leveled must be checked only for agreement with the data base. One or more computer programs may be used to make the checks discussed here. Keep track of their completion on the status sheet.

Ĺ

Synoptic file comparison. The synoptic file is an abbreviated version of the data base, including key items of descriptive information for each control point such as an archival cross-reference number, a designation, aliases, and a position. To ensure that entries for these items in the description file are correct, check them against a listing of the synoptic file. After matching the cross-reference numbers, compare the designations, aliases, and positions. Designations and aliases must satisfy specifications explained in section 2.4.4. Latitudes and longitudes must agree within  $\pm 00^{\circ}00'06''$ .

If the comparison is made by means of a computer program, it should result in a list of records containing discrepancies. Resolve these by listing the appropriate records of the description file and comparing them to both the original records and the published descriptions. Look for typographical ornissions and errors (such as transposition of letters or numerals in the designations or cross-reference numbers), incorrect designations, and errors in the plotting or scaling of the positions.

Correct errors found in the description file, noting any changes directly on the original records as well. Include with the release letter (fig. 5-2) a list of errors found in the synoptic file. Do not consider a position given in the synoptic file to be in error until the plotting of the control point has been verified in the field.

4EM	STATION, DE	S-REC CODE IS D QID-QSN		
	21	000210*10*1861D294459N0903734W	*11* *1	2* *2V*
	22	000220×13×Y 358	*14*LA/LAFOURCHE	*ZV*
	23	000230*17*RACELAND JUNCTION	*18*5.0 KM (3.0 MI) WEST	*ZV#
	24	000240*20*1/NGS	*21*1982LHW *2	2* *2¥*
	25	000250*26*24/STAINLESS STEEL ROD	*27*F NGS	*ZV*
	26	000260*28*Y 358 1982		*ZV*
	27	000270*30*5.0 KILOMETERS (3.1 MILES) WEST AL	ONG THE SOUTHERN PACIFIC RAILR	OAD *ZV*
	28	000280*30*FROM THE STATION IN RACELAND JUNCT	ION, TO A SHELL ROAD CROSSING	AND ¥ZV¥
	29	000290*30*THE MARK ON THE RIGHT, 29.26 METER	S (96.0 FEET) NORTH OF THE NOR	TH *ZV*
	30	DOD300*30*RAIL OF THE TRACK, 3.81 METERS (12	.5 FEET) EAST OF THE CENTER OF	THE *ZV*
	31	000310*30*SHELL ROAD, 1.76 METERS (5.8 FEET)	NORTHEAST OF THE NORTHEAST CO	RNER *ZV*
	32	000320*30*OF A RAILROAD TIE CATTLE GUARD, 0.	45 METER (1.5 FEET) EAST OF A	METAL *ZV*
	33	000330*30*POLE WITH A RAILROAD CROSSING SIGN	۱.	*ZV*
	34	000340*41*M2423.5 F		*ZV*
	35	000350*42* LSHELL ROAD		*ZV*
	36	000360#43#MD.30W		*ZV*

Figure 5-7.—Example of a proofread station description.

## DESCRIPTION FILE STATUS SHEET

LINE L24586. L2

# EXAMPLE

			Keyed on	Original	printed:		Synopt	ic file ch	eck:
	File	Received	Tape #	Date	Јођ 🕴	Merged	Date	Job #	Rerun?
	HAFI	3/11/80	TI-3 A	3/20	6975	b			
	HAFZ	415180	TI-3B	4/15	0105	J HAFF	4/25	5042	y y
						T		5101	ý
			l			1	4/2.8	3012	Ň
			( <u> </u>	[ 		1			1
AMERICAN CONTRACTOR OF CONTRAC									l
							···		
					·····				
					·····	ļ			
						1	····		
						+ + 		· · ·	·
						<u> </u>			
						<u> </u>			ļ
					·····	i			
		······			L				i
									•
						i I			
						*			
······································						1		1	1
						1			
			······						
						1			
······································		[		ŧ		;			
						• ••••••••••••••••••••••••••••••••••••			
						1			
	· · · · · · · · · · · · · · · · · · ·		<u> </u>			1	·····		 
			<u> </u>	<u> </u>	·····				
	 					1			
<u></u>						.			
			1						1 1 1
	ļ	<u> </u>	<u> </u>	<u> </u>		<u> </u>		 	ļ
	ļ	<u> </u>	ļ	ļ					ļ
			ļ	L	ļ	1	l		
	ļ			1			<u> </u>	l	
	l	ļ				j		1	
			L			ļ.,			
<u></u>		I	I	1		1	]		1
·····	1	1	1			1			1
	1	1		1	<u> </u>		<u> </u>		1
			<u> </u>	<u>;</u>	<u> </u>	İ		<u> </u>	
		<u> </u>	İ			1	<u>.</u>		<u> </u>
w	<u> </u>	1	1	<u></u>				! 	
	<b>.</b>			<u>}</u>	<u> </u>	<u>;</u>		<u> </u>	1
	<u> </u>	<b> </b>		<u>.</u>		; }	<u>i</u>		<u>+</u>
	1	1	1	1	1		!	ł	I

Survey-point serial numbers. A line of leveling cannot be correctly abstracted if duplicate or mismatched survey-point serial numbers exist in the line. Run a checking program to detect duplicate serial numbers in the description file, preferably before the line is leveled, so corrections may be made to the logs Check the remarks in the observation records for duplicates or possible mismatches found in the logs by the leveling unit. Mismatches between the description file and the observation file most often occur as a result of incorrect serial numbers entered in the original observation records. Check for and resolve them as described in section 5.4.2.

*Positions* The synoptic file comparison provides a check of the latitudes and longitudes entered for only the control points that are already included in the data base. To check points new to the data base and to check the consistency of all the positions along the line, the positions given in the description file must be compared to the leveling results.

This check is conveniently made when computing the field abstract, as described in section 5.4.2. Compare the straight-line distance between each pair of control points, as computed from their positions, to the leveling distance between them, as computed from stadia observations. The position distance must not be more than 185 m (607 ft) greater than the leveling distance. It should not be less than one third of the leveling dis-

tance, unless the leveling distance is 1 km (0.62 mi) or less. Check the positions of both points in a pair that fails to pass this test.

For each point, scale the latitude and longitude from the smooth plot and compare them to the latitude and longitude given in the description file. They should each agree within  $\pm 00^{\circ}00'06''$ . Correct any errors, noting changes on the original description form. After finding one point with an incorrect position, check the positions of all points adjacent to it. If any one of them is incorrect, check the positions of all points adjacent to it, and so forth, until the positions of all adjacent points prove correct. This procedure is necessary because, even though an entire group of positions may be scaled incorrectly, only the first point encountered may appear incorrect if the same error has been included consistently in the following positions. (See the examples in fig. 5-9).

If scaling from the smooth plot reveals no errors in the position of either point in the pair, check the plot in the field.

Even if the plotted positions prove to be correct, an error may exist in the observation file. If the surveypoint serial numbers assigned to control points in the observation file do not match the serial numbers assigned to the same points in the description file, the leveling distance corresponds to points other than those for which the position distance was computed. Such a case of mismatched serial numbers must be resolved as described in section 5.4.2.



(\* indicates flagged item.)

Figure 5-9.—Examples of apparent position errors.

Junction point positions. Check the positions of junction points that begin and end a line for agreement with the positions given in the descriptions for the same points in other lines of the same project. Verify the scaling and plotting of any junction points that do not agree within  $\pm 00^{\circ}00'06''$ .

Creating the descriptive data set. After all checks and corrections are made, rerun the format-checking program a final time to find and remove any errors that may have been made during processing. The error-free description file is submitted as the descriptive data set for the line.

#### 5.4 Processing Leveling Data

As observations are received from the leveling unit they must be checked promptly to determine if additional leveling is necessary to improve the quality of the results. First, to detect blunders in the information recorded for each section, the observation records must be recomputed and checked. Then, to detect errors that result from blunders not previously discovered, the field abstract must be recomputed and checked. The closure of the leveling line with other lines in the network must be examined. Finally, a statistical summary should be computed and checked. (See figs. 5-4 and 5-5.)

#### 5.4.1 Checking Observation Records

The project office may receive observation records from a leveling unit in person, by mail, or by telephone if they are recorded in a computer. In any case, each package of data should include a single day's observations. Data transmitted by telephone must be followed by a backup tape, disk, or printout sent by mail to the project office. As data packages accumulate daily, note those received on a status sheet maintained for each file (fig 5-10). Indicate the date of the observations, the date they are received, and the status of their processing

When observations are recorded with a computer, they may be stored in a computer file and checked automatically with a recomputing program. Even so, proofread printouts of the original observations and handwritten backup forms, if they are used, to find blunders that may not be detected by the program. When records are exclusively handwritten, check them by systematically proofreading and recomputing the observations as they appear on the recording forms.

Creating the observation file. Chronologically collect observations recorded with a computer into an observation file of manageable size. Collect handwritten records similarly. If more than one unit is working on the line, collect each unit's data in a separate file. Label each file with the line number (or accession number), the letters "HG" and the file number. (For example, "L35 HGF1" or "L24032-HGF2") If only one observation file is necessary for the line, label it with the letters "HGFF." If the unit has not already done so, transmit each day's data to the observation file as soon as possible after receipt. At the same time, obtain a printout of the original observations from the unit or from the tape or disk sent by the unit. The printout provides the only visual backup for restoring items of data lost or altered during transmission. It is also the document on which changes to the observation file must be noted and explained. As printouts accumulate, store them chronologically in a notebook corresponding to the file.

*Proofreading*. Proofread and compare handwritten recording forms and printouts of original observations as soon as possible after they are received, looking especially for errors in the identifying information.

First, compare the survey-point serial numbers to the stampings given for each control point. The serial numbers in the observation records must agree with those assigned to the corresponding points in the description file. (The abstracting program may check this automatically if stampings are included in the elevation difference file, as in section 5.4.2.) Check for incorrect line numbers, data codes, dates, equipment serial numbers and codes, codes for measurement units, observer initials, time zones, times, and setup numbers. Check for illogical entries of temperatures, wind codes, and sun codes. Finally, check comments and computations entered as remarks.

Correct the records as necessary to agree with backup forms, recorded remarks, the description file, equipment inventories, and weekly reports. Compute interpolated corrections for illogical or omitted times, wind codes, and sun codes from records immediately preceding and following the error. Unless the mistake is clearly a format error, remove the unit code from an illogical or omitted temperature and leave the entry blank. Do not alter rod readings. If any readings are omitted or unclear, reject the section.

Initial each sheet as it is checked and corrected. Explain changes made to the observation file directly on the printout and then systematically edit the file before recomputing.

Recomputing computer-recorded observations. Check the observation file once it is stored in the computer by running a recomputing program. The program must read information in chronological order to run properly. It may terminate if any time zones, dates, times, or data codes are illogical or out of sequence. To save time and prevent frustration, check such sequence data carefully during proofreading and be sure to correct erroneous items before running the recomputing program.

Should the program terminate prior to completion, find the error by examining the record in the observation file at the termination point. Often, during transmission of data, erroneous characters may be inserted into a record. Delete them to restore the correct format. If no format or other error can be found, check to see if an error actually occurs earlier in the file, causing subsequent data to appear incorrect.

# OBSERVATION FILE STATUS SHEET EXAMPLE

(

(

Ć

LINE <u>L 2 4586. L 2 HGF Z</u>

UNIT Johnson

Observation		Date	Printed +	First	R	ecomputed		Checked	Merged
Date	Tape #	Received	Proofread	Edit	Date	Job #	Rerun?	Heading	HGZ
4/14/78	D6-1	4/16/78	4/16/78	1					
4/10	DL-2	4/18	4/18						
4//6	Dh-3	4/10	4-119	4/18	418	2/03	Y		
		······································	1110			22.01			,
				······		2201			
	<u> </u>	+	1	<u> </u>	<u> </u>	2210	//		<u> </u>
		<u> </u>			· · · · · · · · · · · · · · · · · · ·				
	-								
		<u> </u>		·····					
			<u> </u>						. <u> </u>
					l 				<u></u>
		ļ							
					ļ				
								ļ	
	. <u></u>					ļ			
	ļ		 						
								1	
		1						1	
						1			
		1	1						
	1					<u> </u>			
	1								1
									<u>+</u>
		}							+
· · · · · · · · · · · · · · · · · · ·	t			······		1		<u> </u>	+
		<u>[</u>					······································		+
		<u> </u>				1			
	<u> </u>								
		<u> </u>	<u> </u>		<u> </u>	<u> </u>			
		<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>			
		<u> </u>							
- <u></u>		<u> </u>		·		<u> </u>		<u></u>	
	ļ	<b> </b>			ļ	<u> </u>			<u> </u>
·		<b>_</b>				ļ		<u></u>	
		L							
	L				•	ļ		<u> </u>	
				<u> </u>	<u> </u>	<u> </u>			
-									
,		1	<u> </u>						
	Ì	1		1		1			1

The recomputing program should check for and flag as rejected records in which required information is omitted, format errors exist, or recomputed results are not within logical or acceptable limits. For example, it should recompute setup imbalances and reading checks, flagging those that do not satisfy the tolerances specified in the project instructions.

Possible blunders should be flagged by the program, but not automatically rejected. Examples include a scale reading of less than 0.5 m (1.6 ft), indicating that the line of sight may have passed too close to the ground surface, or a temperature change during one section running of more than 5° Celsius (9° Fahrenheit), indicating an entry that may be incorrect. The program should compare collimation checks and elevation differences recomputed from the original observations to the field-computed results, flagging discrepancies for further investigation by the project office. (See fig. 5-9.)

In addition, the recomputing program may compute the total curvature correction for the stadia distance imbalance for each section running. When the necessary observations are available, it should compute total scale and refraction corrections as well. A scale correction may be computed for each rod reading, based on the most recent calibration of the leveling rod. A refraction correction may also be computed for each reading, using a model for atmospheric refraction and the measurement of air-temperature differential during the setup. If a temperature measurement is omitted, it may be interpolated for this purpose from records immediately preceding and following the omission. The total corrections should be stored on a record in the file of elevation differences that is created by the program. Corrections should not be applied to the elevation differences until a corrected abstract is computed as described in section 5.5.2.

Finally, the recomputing program should provide a section-by-section listing of the observations, including error flags. Examine the flagged lines of data to determine corrections that must be made to the observation file. Only errors and omissions for which backup records exist may be corrected. For example, if a single rod reading causes a setup to fail to satisfy the reading check, it may be changed only to match the reading shown on the backup printout of the original observations. If it already matches the printout, the reading may not be altered; the record must be flagged as rejected.

After correcting observation records, rerun the recomputing program until all correctable errors are removed and indicate this on the status sheet. Be sure the heading of the file is correct. To further check the data, before the leveling unit advances a great deal farther along the line, compute and check a field abstract. (See sec. 5.4.2.)

Recomputing handwritten observations. When observations are handwritten, use a desk calculator with a printer to recompute reading checks and elevation dif-

ferences directly from the recording forms. If there is disagreement between a difference computed in the office and that in the field, check the observations on the calculator printout against those on the recording form to verify that no blunders were made when recomputing. Explain corrections directly on the recording forms. Obviously, the observations themselves must not be "corrected." Should the observations in a setup fail to satisfy the tolerances specified in project instructions, reject the record.

Rejecting section runnings. A section running must be rejected if it contains a blunder, that is, if it includes observations that do not satisfy either the tolerances or the procedural requirements specified in the project instructions. Most blunders are detected and flagged as rejected during leveling, such as observations that fail to satisfy the reading check, movement of a turning point that is reported to the unit chief, or a running that is incomplete.

Setups that do not satisfy tolerances should be flagged as rejected when observations are recomputed. If a section running includes one or more rejected setups, or setups that are missing or out of order, reject the entire running. The recomputing program should do this automatically.

Sometimes the project office discovers a procedural error or equipment deficiency after the leveling has been completed. In such a case, flag the affected runnings as rejected, explaining the circumstances on the recording forms of handwritten data or on a comment record in the observation file of computer-recorded data.

Rejected sections must be releveled until project requirements are satisfied.

### 5.4.2 Checking the Field Abstract

The preparation of a handwritten abstract by the leveling unit is explained in section 3.8.3. As pages of abstracts are received, store them in line order in the notebook of recording forms or printouts for each observation file. If observations are recorded with a computer and automatically recomputed, the handwritten abstract serves as a backup record that may be discarded after processing is complete. If observations are not automatically recomputed, the handwritten abstract itself must be recomputed and checked. In either case, a field abstract must be computed automatically from the file of elevation differences for the line. Maintain a status sheet while computing and checking the abstract. (See fig. 5-11).

Creating the elevation difference file. To be included in the National Geodetic Vertical Network, the results for each section (records \*40\* through \*43\*) must be entered in the required format into a computer file of elevation differences for the entire line. If an observation file is already available in the computer, the recomputing program should create the difference file. If observations are handwritten or cannot automatically be formatted, key the results from the original recording forms (not the field abstract) onto magnetic tape or other media. Transmit the data to a computer file.

Ć

Ç

Ĺ

ABST LINE L24586.LZ				HGZFF EXAMPLE							
	STARTING H	POINT M 4	<u>+06</u>	ELEVATIO	N 22 24.894	59 ENDING	POINT V	362			
Period Abotract Process											
Complete	Date	act progr .ĭoh ∜	Rerun?	Relevel	сачь не	Ogram: Positions	Edit HA	Heading	N-0		
portial	4/18	2301	-V			X	X				
- Fur 11 sum	4125	0/02	y I		~						
<u> </u>	5/02	6571		<u> </u>		Y	~				
con olete	5/02	4311					~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	~			
<u></u>	5/09	4347	Ń				~~~~		~		
······································											
<u></u>											
<u> </u>											
		·····									
······································											
	<u> </u>										
	}										
<u></u>		<b></b>									
							······				
	1							]			
<u></u>				[		[		<u> </u>	1		
<u></u>		1	<u> </u>				t	<u> </u>			
		1						<u> </u>			
<u></u>	1			[				<u> </u>			
	<u> </u>		<u> </u>	Ì		<u> </u>			<u> </u>		
<u> </u>		-	1			-					

Label the elevation-difference file with the line number (or accession number), the letters "HGZ", and the number of the observation file to which it corresponds. For example, "L35.HGZF1" corresponds to the observation file L35.HGF1. Before computing an abstract, merge the elevation-difference files for the entire line, into one file, labeled with the letters "HGZFF". For example, files L35.HGZF1 and L35.HGZF2 are merged to create a file labeled "L35.HGZFF". If only one difference file is necessary for the line, label it with the letters "HGZFF".

Computing the abstract. Using the automated elevationdifference file and the description file for the line, run a program to automatically compute and print an abstract. Enter the following preliminary information: order and class of the survey, survey-point serial number and elevation of the beginning control point, and surveypoint serial number of the ending control point.

The beginning point is normally one of three junction points, preferably one with a published elevation. Assign the same beginning point and elevation to every line that begins at the junction during the project. The other two points may or may not appear as a spur on the abstract, depending on the order in which they were leveled.

For a line that connects to the National Geodetic Vertical Network, obtain the adjusted elevation of the beginning point from the most recently published description. If the point is new to the national network, temporarily assign to it an elevation of 0.00000 m when computing the abstract. As soon as a connection is made to the network, compute a field elevation for the beginning point by subtracting the preliminary elevation computed for the connecting point from the adjusted elevation published for it.

Example: A line begins at the junction point M 245, for which no adjusted elevation has been published. After six sections of leveling have been completed, a connection is made to the network at bench mark J 11, which has a published, adjusted elevation of +3.5132 m. Using 0.00000 m as the elevation for M 245, the computed elevation for J 11 is -1.26418 m. The adjusted elevation minus the computed elevation equals +4.77738m. This value is then assigned as the field elevation for M 245. Should another line begin on M 245 during the same project, assign the same field elevation to M 245.

If the control-point stampings recorded by the leveling unit are available in the elevation difference file, the abstracting program should compare them to the description file by matching survey-point serial numbers and flagging any resulting discrepancies. The program should, as described in section 3.8.3, assign designations to the control points, sort all properly identified section runnings, check multiple runnings of a single section for closure, and compute a field elevation for each control point leveled. It should not compute elevations beyond the point where a necessary running is rejected or omitted. This occurs in single-run leveling at the first section that has not been satisfactorily leveled. In double-run leveling it occurs at the first section that has not been satisfactorily leveled in both the forward and backward directions and closed. The program should, however, continue to sort and check any remaining runnings in the file. Finally, it should compare leveling distances to the positions given in the description file for the control points, shown in figure 5-7.

The program should provide a printout of the abstract similar in form to the handwritten abstract (fig. 5-12). It should flag runnings that are rejected because of misclosure with the letter "R", and list other errors at the end of the printout.

Checking the computer-produced abstract. Examine the abstract printout to detect errors in the elevationdifference and description files and to determine which sections, if any, must be releveled to improve the quality of the data. Check first for incorrect survey-point serial numbers, as explained here. Then investigate and resolve other errors listed on the printout. Assign releveling as necessary to close rejected sections and resolve discrepancies.

Incorrect survey-point serial numbers may cause inappropriate loops, spurs, misclosures, and running directions to appear in the abstract. Not all such errors are noticeable (fig. 5-13). Investigate suspicious cases by comparing the stampings entered on the original observation records, for every section involved, with the stampings in the description file. If the stampings do not agree, find the descriptions for the points that correspond to the stampings given in the observation records. Correct the serial numbers on each record to agree with the ones assigned in the descriptions. Then correct the observation and elevation difference files before computing a new abstract.

If the abstracting program makes a check between serial numbers and stampings, investigate the discrepancies listed. Each discrepancy indicates the presence of an erroneous stamping in either the leveling or the descriptive data, or an erroneous serial number in the leveling data. First, check for and correct format errors in the stampings. If no such errors are found, resolve the discrepancy as described previously, referring to the original observation records.

If a suspected error in the serial numbers cannot be confidently resolved by an examination of the original records, relevel to close the sections involved, carefully verifying the identity of each control point. If points misidentified in the original records can be identified by closure of original runnings with the check leveling, correct the records, explain the circumstances, and retain the original runnings in the data files.

Checking the handwritten abstract. Compare the handwritten field abstract to the recomputed and checked observation records. After correcting elevation differences that were copied or computed incorrectly, systematically proofread and recompute the abstract. Check spurs with especial care.

(

(

(

-\*- FIELD ABSTRACT -\*-19820121-19820318 \*ZV\* HGZ L24680 11 4.0 MM ORDER 1 CLASS 2 PAGE SOUTHERN LOUISIANA GEOTHERMAL DEVELOPMENT PROJECT RACELAND TO GRAND ISLE TIDE STATION

FROM TO	START F/B	DIST TOTAL (KM)	ELEV DIFF (M)	-(F+B) Total (MM)	MEAN DIFF FLD ELEV (M)
1862 D 192		-			2.34300
1862 D 192 1861 Y 358	1291453 B 1291530 F	0.90	1.13905	-0.78	-1.13866
		0.90		-0.78	1.20434
1861 Y 358 1860 Y 191	1291240 F	2.10	0,23885	0.18	0.23894
		3.00		-0.60	1.44328
1860 Y 191 1859 D 8	1291036 B	1.34	0.64082	2,26	-0.64195
		4.34		1.66	0.80133
1859 0 8 1803 Y 298	1290828 F 1290936 B	1.63	0.95712 - 0.95748	0.36	0.95730
······································		5.97		2.02	1.75863
1803 Y 298 1863 X 298	1290738 F 1290758 B	0.55		1.02	-1.18756
	SL 1	6.52		3.04	0.57107
1803 Y 298 1802 J 165	1210900 F 1211058 B	0.10	-0.73280	-0.27	-0.73294
	SL I	6.07		1.75	1.02570
1802 J 165 1801 C 192	1210907 F 1211010 B	1.79	0.72312 -0.72415	1.03	0.72363
	SL 1	7.86		2.78	1.74933
1803 Y 298 1804 A 358	1211120 F 1220933 B	1.50	0.02728	1.92	0.02824
		7.47		3,94	1.78687
1804 A 358 1806 TT 25 I US6S	1211208 F 1211313 B	1.79	2.26067	0.30	2.26082
		9.26		4.24	4.04769
1806 TT 25 L USGS 1805 A 220	1221015 F 1221028 B	0.24	-0.08483	0.18	-0.08474
	SL 1	9.50		4.42	3.96295
1806 TT 25 L USGS 1807 B 358	1221036 F 1221043 B	0.05	-0.44957	-0.50	-0.44982
	**************************************	9.31		3.74	3.59787
1807 B 358 1808 B 220	1221048 F	1.43		2.25	-1.13634
1000 0 660	12/0070 B	10.73	* * * ^ ~ ~ ~ ~	5.99	2.46152

Figure 5-12.--Sample of computer-produced field abstract.

#### DOUBLE-RUN LEVELING



<u>In t</u>	ne d	lata:
6301	to	6302
6302	to	6301
6302	to	6303
6303	to	<u>6301</u>
<u>6304</u>	to	<u>6303</u>
6303	to	6304



6302 to 6303 appears to be single run. 6301 to 6303 appears as a spur run backward, that nearly equals 6302 to 6303. A loop is formed.

6303 to 6304 appears with the wrong sign, but this error may only be detected by checking the leveling data for mismatched SPSN's.

#### SINGLE-RUN LEVELING



Figure 5-13.—Samples of incorrect survey-point serial numbers. Mismatched survey-point serial numbers are underlined.

When leveling results are keyed and transmitted into an elevation-difference file, typographical errors may exist in the file. To check for these errors, compare the elevations that appear on the computerproduced abstract to those on the handwritten abstract. Refer to the original recording forms to resolve any discrepancies. Correct both the difference file and the handwritten abstract, rerunning the abstracting program until no discrepancies are found between them. Creating the vertical observation data set. Either by means of the abstracting program or by keying elevations directly from the handwritten abstract, create the field abstract records (\*30\* lines) in the format required by NGS. These abstracts, merged with the elevationdifference file, form the "vertical observation data set" for the line. Label it with the line number (or accession number) and the letters "HGZAB". (For example, "L35.HGZAB" or "L24038-HGZAB".)

#### 5.4.3 Preliminary Analysis

The field elevations for a line of leveling must be checked to determine if any blunders or inconsistencies remain in the data set. The new data should be compared to data from previous surveys and to data from other lines of the project. These tasks, termed "preliminary analysis," must be performed in a timely manner, so any necessary releveling may be conducted before the leveling unit has progressed to another line.

Comparing new and old elevation differences. After the abstract has been computed and checked, compare the elevation differences between each pair of recovered control points to the differences from the most recent, previous survey along the same route. This check is normally made by comparing the new field elevations to the elevations for the same points from an unadjusted abstract for the previous survey. It may be performed by means of a computer program if results from the previous survey are stored in the National Geodetic Survey data base and may be accessed by the project office. Otherwise it must be performed as described next. If an unadjusted abstract is not available, use published elevations for the points.

Prepare a table, as shown in figure 5-14, with columns for control point designations, distances along the line, new field elevations (enter the accession or line number and year of the new survey), old elevations (enter the accession number and year of the old survey), and new-minus-old values. If the new leveling applies to more than one previous survey, prepare columns for the old elevations and the new-minus-old values for each additional survey. From the field abstract for the new survey, enter the designation, distance, and elevation for each recovered control point. Then, from the abstract for each old survey, enter the corresponding elevations.

Since the new elevation for the beginning point is rarely the same as the old elevation for the same point, the difference between them will be included in the new-minus-old difference computed for every point that follows. To remove the difference, a constant must be added to each new-minus-old difference. Compute the constant,  $C_1$ , by subtracting the new elevation for the beginning point from the old:

 $C_1$  = old beginning elevation – new beginning elevation.

Compare the new and old elevations by subtracting the old from the new and adding the constant:

Ę

New-minus-old value = new elevation  
- old elevation + 
$$C_1$$
.

To compare the new survey with additional previous surveys, compute and add to each new-minus-old difference a constant that refers the comparison back to the beginning point. For example, at the first point common to two previous surveys, compute the constant for the second,  $C_2$ , by subtracting the first elevation of the point from the second and adding the first constant:

 $C_2$  = second old elevation - first old elevation +  $C_{\rm b}$ 

Then compute the new-minus-old values as before, adding  $C_2$  instead of  $C_1$ . If two previous surveys have no points in common, establish a new beginning point and compute an entirely new constant for the second survey. In addition, prepare a column of the published elevations for the recovered points in the new survey that connect the two previous surveys. To make possible a check of the leveling between these points, compute new-minus-old values using the published elevations.

After computing the new-minus-old values, and carefully checking them, examine the results to find potential blunders in the leveling. If the leveling procedures and the environment were such that no accumulations of systematic error occurred during the surveys and no changes in topography occurred between them, the values would fluctuate randomly about zero and well within the tolerances for geodetic leveling. A blunder in the leveling during either survey would reveal itself between two control points as a jump (exceeding the tolerance). Misidentification of a point during either survey (or significant movement of a point) would reveal itself as a spike, which is a jump exceeding the tolerance followed immediately by another jump, bringing the total back within the tolerance. (See fig. 5-15.)

Since small amounts of systematic error may accumulate and since topographic changes occur between surveys in many parts of the continent, the new-minusold values do not usually fluctuate randomly about zero. A difference in the accumulation of systematic error during the surveys typically reveals itself as a gradual trend toward or away from zero. The trend may be readily predictable when it is solely a function of leveling distance. It may be less so when it is a function of both distance and elevation change, perhaps mimicking the topography. Actual topographic changes between the surveys may appear as either trends or jumps and, depending on their magnitude, may not be distinguishable from errors in the leveling.

Look for jumps and spikes that do not conform to the trends. In general, the new-minus-old values should not change from one point to the next by more than the accepted tolerance for agreement between section
#### NEW-MINUS-OLD COMPARISONS

EXAMPLE

LINE 224448

Control Point	Distance (km)	L 24448 1979	L20229 1965	N-0+C1	L656 1933	N-0+C2	L20669 1966	N-0+C3
				C.=0.0000		C_=0.1734		C_=0.0267
L231	0.00	5.47510	5.4751	0.0000		<b>_</b>		
K 231	) 0.09	2.13644	2.1394	-0,0030				
J 23/	0.59	8.06250	8.0641	-0.0016				
Q 22	59.58	8.73215			8.5588	0.0000	<u> </u>	
R 22	60.24	9.54412			9.4817	-0.1109		
A 23	86.83	10.02366			9.7934	0.0569		
R 23	140.21	25.93518			25.8883	-0.1265		
ARCADIA	/53.35	19.51631			19.4364	-0.0934		
X 23	) 164.40	18.67893			18.6016	-0.0960		
A 24	172.17	12.11115			12.03853	-0.1007		
W 260	186.08	8.11921					8.17961	-0.0871
X 260	191.76	3.95504					4.01583	-0.0875
R 260	191.97	4.12528					4.18551	-0.0869
N 260	194.54	2.29061					2.34633	-0.0824
V 260	198.26	2.89240					2.94812	-0.0824
CLEVELAND	199.88	2.64930					2.70118	-0.0785
CLEVE LAND AZ MK	200.47	1.91607					1.96779	-0.0784
L 260	202.00	1.30915					1.36234	-0.0798
H 260	202.59	2.08215					2.13642	-0.0809
G 260	204.10	2.06877					2.12/48	-0.0794
K 38	) 205.56	1:68352			1.58166	-0.0715	1.72835	-0.0715
<sup>L</sup> 38 <sub>RESET 196</sub>	1 205.85	2.87545					2.92881	-0.0800
Q 38	206.60	3.87255			3,75955	-0.0604	3.92092	-0.0750
COLLIER	206.62	3,76595					3.81455	-0.0753

Figure 5-14.-Example of comparisons between new-minus-old observed elevations.

runnings, computed for whichever survey is of lower order and class. However, if most successive pairs of points exhibit relative change or "noise," this tolerance may be unreasonably restrictive. Experience and good judgment should be applied to spot inconsistencies. Since a jump may herald a set of values that has been computed incorrectly and a spike may represent a single point that has been computed incorrectly, double-check the computations before investigating further.

Do not assume that a jump, inconsistent with the trends of the preceding or following sections, represents a topographic change. If the new leveling is single run, check it by releveling to close the section (or sections) that comprises the jump. If the leveling is double run, check the observation records for reversed identification of the control points or for leveling rods observed consistently in reversed order. Reversed identification is likely if the elevation difference for a section of new leveling is opposite in sign and nearly equal in magnitude to one-half of the amount of the jump. Relevel to verify the identity of such a section. If no error is found, note the jump in the project report.

Since a spike results if a control point moves significantly between the surveys, check the description for the point to see if the mark setter noted any damage to, or apparent relocation of, the monument. If new singlerun sections comprise the spike, relevel them to close the leveling line. If the leveling is double run, check that the point is correctly identified in the observation records; sometimes a nearby bench mark may have been



Figure 5-15.—Sample plot of new-minus-old values for observed elevations.

leveled instead of the point originally assigned. In this case, prepare a description for the newly recovered point and level a spur to the assigned point.

Checking loops. As portions of new lines are completed that connect to previous or concurrent surveys, check the closure errors of the circuits or "loops" thus formed. Compute closure error by totaling the elevation differences around a loop, beginning and ending at the same control point. The result should not differ from zero by more than the loop tolerance for the leveling methods employed in the surveys.

First, make a sketch of the loop, labeling each segment of line or "link" with its order and class, accession number, and year. From the field abstracts and/or summary sheets, determine the connecting points and label them. Where a tie is established, choose the connecting point that appears least likely to have moved since the previous survey. Write the unadjusted elevation for each connecting point for each survey in which it is included. (An example is given in fig. 5-16.)

Second, compute the elevation difference for each link. From the elevation at the end of the link, subtract the elevation at the beginning. Write the result over an "arrow" indicating the direction in which the link difference was computed. Write the leveling distance under the arrow.

Finally, working clockwise, sum the elevation differences. If a link was leveled counterclockwise in the loop, change the sign of the elevation difference before adding it to the total. Write the closure error in the center of the loop. Compute the total distance around the loop and write it under the closure error.

To compute the loop tolerance, total the leveling distances separately for each order and class of survey. If the entire loop consists of leveling of the same order and class, the tolerance is computed, as described in fig. 3-9 (sec. 3.1.3). If more than one order and class are included, compute a tolerance for each using the appropriate distances. Then, square the tolerances, sum them, and take the square root of the result:

#### Tolerance =

$$\pm [(T_1^2 \times \sqrt{K_1}) + (T_2^2 \times \sqrt{K_2}) + (T_3^2 \times \sqrt{K_3}) + \dots]^{\frac{1}{2}} \text{ mm}.$$

Example: The loop in figure 5-16 includes four links, two of first-order, class I leveling, one of firstorder, class II leveling, and one of second-order, class I leveling. The total distances are 208 km, 52 km, and 110 km, respectively. The tolerance for the closure error is:

$$Iolerance = \pm [(4^2 \times \sqrt{208}) + (5^2 \times \sqrt{52}) + (6^2 \times \sqrt{110})]^{1/2} = \pm 92.7 \text{ mm}.$$

Compare the misclosure error to the loop tolerance. When a loop includes leveling from a single season or epoch, failure to close within the tolerance is probably the result of one or more blunders in the leveling. First, double-check the computation of the closure. Then, check and recompute the ties at junctions. Do not use as connecting points monuments that appear to have moved between the levelings. Ensure that the junction points have been identified correctly in the observation records.

If the loop still fails to close within the tolerance, investigate further. Try to isolate the links where blunders might have occurred. For example, look at the closure error of a large loop encompassing the one in question, such as ABCD in figure 5-17. The sum of the closure errors of ABD and BCD is equal to the closure error of ABCD. If ABD and BCD are not within tolerance, but ABCD is, the blunder is most probably in link BD. If ABD closes but BCD does not, links BC or CD should be examined.

Reexamine the new-minus-old values to find jumps. Check the field abstracts for an elevation difference in one section that is opposite in sign to and about onehalf the magnitude of the closure error. If the leveling is single run, relevel to close doubtful sections. Check the identification of double-run sections. Consult the project manager at headquarters before undertaking any extensive releveling to close a loop.



Figure 5-16.—Sample sketch for checking a loop.

When a loop includes leveling from more than one epoch, failure to close within the tolerance may or may not be the result of blunders in the new leveling. If a link of new leveling begins in an area of crustal subsidence and ends in an area of relative stability, the loop formed with leveling from a previous epoch cannot be evaluated without knowledge of the rate of subsidence, even though ties are established to the previous surveys at each end of the link. Similarly, if the connection between new and previous leveling consists only of a single point, potential movement or damage to the point itself may prevent confident evaluation of the loop. Nevertheless, a large misclosure error should be investigated, as described previously, particularly if it is inconsistent with the geology of the area.

#### 5.4.4 Statistics

The final step in processing leveling data at the project office is to obtain a statistical summary of the work accomplished for each line. The summaries permit final checks of the validity of the data and are the source of statistics for the entire project.

Some statistical information, such as the standard error of the reading checks attributed to each observer or instrument, may be obtained only when the observation records are recomputed. Most information, however, may be obtained by computing statistics from the vertical observation and descriptive data sets.

When possible, run a program to compute from the leveling file (HGZAB) such statistics as total progress, progress along the main line, progress along spurs, double-run progress, single-run progress, percent of progress requiring one or more reruns, total distance leveled, distance leveled as reruns, percent of leveling rejected, number of sections, number of setups, number of observing hours and days, average sighting distance and time per setup, and average leveling distance per hour and day. The program should provide a breakdown of the hourly and daily statistics for each combination of observer, instrument, and rods in the line, listing the appropriate initials and serial numbers. Check them to verify once more the identification records.



Figure 5-17.—Combining loops.

From the description file (HAFF), the program should compute the total number of control points and the totals of points newly set, recovered in good condition, recovered in poor condition, and recovered new to the network. Also, it should provide breakdowns by type of the numbers of newly set and recovered points, computing the percentages of monuments of each quality rating. (See sec. 2.4.4.) For a file of descriptions of points not leveled, the program should compute the numbers of points recovered in good condition, recovered in poor condition, destroyed, and not recovered, including a breakdown by type and quality rating.

Check that the total number of leveling sections equals the total number of control points, minus one. If the number of sections is too small, one or more points may have been skipped by the leveling. If the number is too large, one or more loops, correct or incorrect, may exist in the line. Investigate such discrepancies thoroughly and correct them, if possible. Note loops that are correctly included in a line on a comment record in the leveling file.

#### 5.5 Verification and Adjustment

When data from a vertical survey are received by the National Geodetic Survey, they are checked for completeness and, if acceptable, verified and stored in the appropriate computer files of the National Geodetic Survey data base. The data are corrected for known systematic errors. The survey is then adjusted, and elevations are published as part of the National Geodetic Vertical Network. These activities are discussed next, to provide an overview of the process by which geodetic elevations are computed.

#### 5.5.1 Quality Control Review

Before a survey may be incorporated into the national network, the survey must meet strict standards of completeness, consistency, and accuracy. The quality control review is designed to detect oversights, misunderstandings, and deviations from the specifications governing inclusion of data in the network. It is conducted in two steps, preaccessioning and verification.

*Preaccessioning.* To be accepted by the National Geodetic Survey, a survey must be documented completely. The required records are listed in section 5.2.4. As soon as a survey is received, it is checked to ensure that all required records are present, complete, and properly formatted.

After all records are obtained, an accession number is assigned (if it had not previously been assigned in the project instructions.) (See sec. 5.2.3.) The number permanently identifies the survey for purposes of storage and retrieval. Then, documents are placed in archives, and computer files are transferred to a working file for further processing. *Verification.* The Vertical Network Division checks each line of leveling received for compliance with the specifications given in this manual and in the project instructions. Data files are checked for consistency, completeness, and compliance.

During this process new control points are assigned archival cross-reference numbers. Discrepancies are resolved by referring to the original records; questionable data that cannot be corrected from original records are rejected. Occasionally, additional leveling may be required to verify questionable data. Acceptable description records are transferred directly into the data base for publication. Acceptable leveling records are corrected for systematic error and cataloged and stored in the data base for eventual adjustment.

The verifier then prepares a report summarizing the results of the quality control review and states the extent to which the survey has been accepted for inclusion in the data base. Specific discrepancies and deficiencies in the data files and the original records are noted. A copy of the report is forwarded to the contributor to aid in the conduct of future surveys.

#### 5.5.2 Computing Corrections

In a survey network, numerous lines are leveled with various instrumental systems and under varying environmental conditions. Before the lines are adjusted, the data must be "normalized" or "reduced" to provide elevation differences that refer to a standard set of conditions, under which the probable magnitude of random error may be estimated. All known instrumental and environmental effects, causing systematic error to accumulate in one section relative to another, must be removed from the measured elevation differences.

Geodetic leveling, as explained in section 3.1.2, limits systematic error with symmetrical procedures. However, to obtain the most accurate results, particularly for surveys covering large areas or areas of dramatic elevation change, certain types of systemmatic error may be removed sufficiently only by applying corrections. These include the effects of instrument collimation error, scale imperfections, atmospheric refraction, curvature of the reference surface, tidal accelerations on the gravity field, and variations inherent in the gravity field.

A detailed explanation of the corrections applied by NGS to precise leveling observations appears in *NOAA Technical Memorandum* NOS NGS 34, Corrections applied by the National Geodetic Survey to precise leveling observations (Balazs and Young 1982).

Typically, the equation describing an error effect is derived mathematically and tested by experiment. To make the correction worthwhile, the effect must be computed with sufficient accuracy relative to its magnitude. The parameters necessary to compute the error are measured routinely during leveling. To remove it from the leveling results, the error is subtracted. Thus, the correction that is added to the elevation difference of a setup or section is always equal in magnitude and opposite in sign to the effect itself. The corrections discussed here are applied routinely by the National Geodetic Survey when preparing an abstract of leveling (PHASE 1) for adjustment. Unless otherwise specified, all corrections are computed in metric units. Typical correction values are shown in figure 5-18.

Collimation error. If the instrument provides a line of sight sufficiently close to the horizontal, as explained in section 3.3.3, and if setups are balanced within tolerance, a correction for collimation error need not be applied. However, for the most precise results, particularly if leveling extends over a large area and section imbalances are large and consistently positive or negative, the correction should be applied. It should also be applied when either the collimation factor or section imbalance exceeds tolerance (specified in fig. 3-9 of sec. 3.1.3).

To correct for the effect of collimation error, the collimation factor must be measured with sufficient accuracy during the collimation check. The collimation factor, C, measured at the most recent, previous, collimation check, is multiplied by the accumulated section imbalance,  $\Sigma \Delta s$ , to compute the effect. The correction, opposite in sign, is added to the elevation difference for the section:

#### Collimation correction = $-C \times \Sigma \Delta s$ .

The correction could be applied to each setup, but nothing is gained by this. Since the factor is assumed to be constant throughout the section, the sum of the setup corrections equals the correction given above.

Scale imperfections. Imperfections in the scales on the leveling rods create error that is a function of elevation difference and, thus, cannot be limited by restrictions placed on sighting distance. Two corrections must be applied to standardize the observed elevation differences with respect to the National Standard of Length. One, the length-excess correction, removes the effect of differences between the graduations observed at a standard temperature and their calibrated values at the same temperature. The second, the thermal-expansion correction, removes the effect of variations in scale temperature.

The length-excess correction is most accurately computed for each setup when observations are recomputed as described in section 5.4.1. The most recent, previous calibrated values of the graduations at a standard temperature are required. First, the graduation component of each scale reading is converted to its corresponding calibrated value and added to the micrometer readings also converted to meters. From the scale readings thus corrected, the elevation differences for each setup are recomputed and totaled for the section. The correction in this case, then, is the differences between the corrected and uncorrected elevation differences.

If calibrated values for every graduation are not available, values for index error and length excess from the most recent, four-interval calibration (see sec. 3.4.2)

RELEVELI Southern Raceland	NG LOUISIANA TO GRAND I	RUN DATE: Geothermal Dev Sle tide stati	21 APR 82 Velopment proji Ion	TIME: 09:42 ECT		PAGE 1 HGZ L24680 4.0 MM 1ST-	PART 11 Order/class II
LHW/NGS		2	21 JAN 1982	18 MAR 198	32		
INST/ROD DESI	S OBS DIFF GNATION	LEVEL R Corr Co	ROD TEMP JRR CORR	MEAN ASTRO TEMP CORR	REFRAC CORR	CORR DIFF Mean Diff	DIST MIN
1/1 B 1/1 F D 192	1.13905 -1.13827	0.00001 0.00 0.00000-0.00 TO Y 35	0002* -0.00000 0003* 0.00000 58	22.5 0.00003 21.4 -0.00000	. 0,00000 0.00000	1.139088 -1.138303 -1.138696	D.903 0.904 0.903
1/1 F 1/1 B Y 358	0.23885 -0.23903	-0.00001-0.00 0.00000-0.00 TO Y 19	0023× -0.00000 0022× 0.00000 91	22.8 0.00011 23.1 0.00007	0.00001 7 0.00000	0.238719 -0.239171 0.238945	2.102 2.098 2.098
1/1 B 1/1 F Y 191	0,64082 -0,64308	0.00000 0.00 0.00000 0.00 TD 0 8	)002* -0.00000 )003* 0.00000	22.8 0.00005 22.8 0.00006	5 -0.00002 0.00003	0.640866 -0.642957 -0.641912	1.338 1.338 1.338
1/1 F 1/1 B 0 8	0.95712 -0.95748	-0.00001 0.00 -0.00000-0.00 TO Y 29	0006* -0.00001 0009* 0.00000 98	18.3 -0.00001 21.4 0.00003	0.00000 0.00000	0.957157 -0.957541 0.957349	1.632 1.632 1.632
1/1 F 1/1 B Y 298	-1.18807 1.18705	0.00001-0.00 -0.00001-0.00 TO X 29	0.00001 0015* -0.00001 78	15.0 -0.00001 16.1 -0.00001	-0,00004	-1.188289 1.186867 -1.187578	D.549 D.551 D.549
1∕1 F 1∕1 B Y 298	-0.73280 0.73307	0.00000-0.00 0.00000-0.00 TO J 16	0023* 0.00000 0020* -0.00000 55	19.7 -0.00000 23.9 -0.00001	0,00001 0,00004	-0.733045 0.732903 -0.732974	0.104 0.102 0.102
1/1 F 1/1 B J 165	0.72312 -0.72415	0.00000 0.00 0.00000-0.00 TO C 19	0004* -0.00000 0001* 0.00000 72	20.8 -0.00004 22.8 -0.00007	0,00010 -0.00012	0.723215 -0.724341 0.723778	1.789 1.787 1.787
1∕1 F 1∕1 B Y 298	0.02728 -0.02920	-0.00000 0.00 0.00000 0.00 TD A 35	0002* -0.00000 0005* 0.00000 58	24.4 0.00009 22.2 0.00005	0.00001 0.00000	0.027399 -0.029091 0.028245	1.500 1.500 1.500
1/1 F 1/1 B A 358	2.26067 -2.26097	0.00000 0.00 0.00000-0.00 TO TT 2	0009* 0.00000 0006* -0.00000 25 L USGS	25.0 0.00009 25.3 0.00008	0.00013 -0.00005	2.260980 -2.261014 2.260997	1.792 1.795 1.792
1/1 F 1/1 B TT 25 L	-0.08483 0.08465 USGS	-0.00000-0.00 0.00000 0.00 TD A 22	0002* 0.00000 003* -0.00000 20	23.6 -0.00001 23.6 -0.00001	0.00003	D,084829 0.084612 D.084721	0.239 0.239 0.239
1/1 F 1/1 B TT 25 L	-0.44957 0.45007 USGS	0.00000-0.00 0.00000-0.00 TO B 35	0021* 0.00000 0014* -0.00000 58	23.3 0.00001 23.3 0.00001	-0,00000 0,00000	-0.449775 0.449936 -0.449856	0.050 0.050 0.050

Figure 5-18.—Sample corrections applied in phase 1 of office abstract.

must be used to compute the length-excess correction for the section. It is approximated by multiplying the observed elevation difference,  $\Delta H$ , by the mean length excess of the rod pair,  $\tilde{e}$ . The correction for the difference between the index errors, *i*, of the scales observed at the beginning and ending control points is also included:

Length-excess correction =  $\bar{e} \times \Delta H - (i_{\text{begin}} - i_{\text{end}})$ .

The correction for index error may be neglected if every section includes an even number of setups; in other words, if the same scale is observed at the beginning and ending control points.

To compute the thermal-expansion correction, leveling data must include at least a mean scale temperature for each section. In addition, the coefficients of thermal expansion from the most recent standardization of the scales must be available. The correction is approximated from mean coefficient of thermal expansion for the rod pair,  $\vec{k}$ , the mean observed scale temperature,  $\vec{t}$ , the standardization temperature,  $t_s$ , and the elevation difference,  $\Delta H$ :

Thermal-expansion correction  $= \vec{k} \times (\vec{t} - t_s) \times \Delta H$ .

Note that the sign of this correction formula is positive. This is because the values read from an "expanded" scale are *smaller* in magnitude than those read from the same scale at the standardization temperature. The magnitude of the elevation difference, obtained from the observation of "expanded" scales, must be *increased* by the correction.

*Refraction.* Because refraction depends on elevation difference and atmospheric conditions as well as sighting distance, the error due to refraction is only partially limited by restricting setup imbalance and sighting distance. Corrections must be applied, preferably to the elevation difference for each setup, to reduce refraction error. They must be applied consistently to leveling data from different epochs, since the tolerances placed on sighting distance have varied significantly in the past.

Corrections for refraction are best obtained by measuring air-temperature differentials at each setup, as described in section 3.6.2. The approximate refraction correction for the setup may be computed from the mean sighting distance (instrument to rod) for the setup,  $\hat{s}$ , the observed air-temperature differential between 1.3 and 0.3 meters,  $\Delta t$ , and the observed elevation difference,  $\Delta h$ , for the setup. A formula such as Kukkamaki's is used:

Refraction correction =  $-0.00070 \times \alpha \times (s/50)^2 \times \Delta t \times \Delta h$ .

The constant,  $\alpha$ , depends on the units of measurement; for  $\Delta h$  and  $\bar{s}$  in meters and  $\Delta t$  in degrees centigrade,  $\alpha = 200$ . Other models for refraction are under development.

Leveling data that do not include measured temperature differentials may be corrected by means of a similar formula, in which the temperature differentials are predicted from atmospheric data given in the leveling records, and historical climatological data for the time of day, date, and geographic position of the leveling. This correction is added to the elevation difference of each section.

*Curvature.* If setups are balanced and sighting distances are maintained within tolerance, the effect of curvature is so small that a correction need not be applied to leveling extending over a small area. However, as with collimation error, for the most precise results the correction should be applied over large areas.

The curvature correction is approximated from a spherical model of the Earth. It is computed for each setup from the imbalance,  $\Delta s$ , and the mean radius of the Earth, r:

#### Curvature correction = $-\Delta s^2/(2 \times r)$ .

The total correction for the section is most efficiently computed from the sum of the sight distance imbalance for the setups.

Tidal accelerations. Since the effect of tidal accelerations on the gravity field is primarily a function of the time and route of leveling, this effect cannot be minimized by procedure. Although the effect is very small, if a survey extends over a large area, the "astronomic correction" should be applied.

The astronomic correction is computed from the date, time, and position of the section. From the date and time are computed the azimuths of the Sun and Moon,  $\alpha_s$  and  $\alpha_m$ , respectively, and the corresponding horizontal deflections,  $\epsilon_s$  and  $\epsilon_m$  of the gravity field. From the positions of the two control points are computed the length, S, and azimuth,  $\alpha$ , of the section. Azimuths are measured clockwise from south; therefore, the correction is positive when the ending point is north of the beginning point:

# Astronomic correction= $0.7 \times S \times [\tan \epsilon_m \times \cos (\alpha_m - \alpha) + \tan \epsilon_r \times \cos (\alpha_r - \alpha)].$

Ļ

The factor, 0.7, is included to reduce the correction because the Earth's crust is somewhat elastic and the models by which deflections may be computed assume a rigid Earth. The deflections are computed by the National Geodetic Survey from equations developed by Longman (1959) and astronomic parameters taken from *The Astronomical Almanac* (U.S. Naval Observatory/ Her Majesty's Nautical Almanac Office).

Gravity field. Since the level surfaces defined by the Earth's gravity field are not parallel, as explained in subchapter 1.3, the elevation difference measured between two control points depends on the route followed by leveling. A difference independent of the route may be obtained by converting the observed difference in orthometric height between each pair of points to the difference in potential between the level surfaces passing through them. For a section, the potential difference is the corrected elevation difference,  $\Delta H$  in meters, multiplied by the mean of the measured values of gravity at the two control points,  $\tilde{g}$  in kilogals. The potential difference is expressed in kilogal-meters, called geopotential units (gpu):

Potential difference  $= \bar{g} \times \Delta H$ .

The elevations obtained by totaling potential differences are designated geopotential numbers.

If gravity is not measured with sufficient accuracy to compute potential differences, a model for the gravity field must be used to compute a correction to be added to the orthometric-height difference of each section. From Helmert's international formula for normal gravity, the following "orthometric correction" has been derived. It has a function of the mean elevation, H in meters, the mean latitude,  $\phi$ , and the northward change in latitude  $\Delta \phi$ , for the section. Two constants,  $\alpha = 0.002644$ and  $\beta = 0.000007$ , are used:

Orthometric  
correction = 
$$-2\alpha \times \overline{H} \times \sin 2\overline{\phi} \left[1 + \left[\alpha - (2\beta/\alpha)\right] \times \cos 2\overline{\phi}\right] \Delta \overline{\phi}.$$

The elevations obtained by totaling elevation differences corrected in this way are called normal orthometric heights.

#### 5.5.3 Adjustment

After correcting the leveling data, abstracts of reduced elevations are computed for every line or link in the survey network. The expected value of a loop closure computed from the reduced elevation differences is zero



Figure 5-19.—Sample profile of a leveling line.

if all systematic errors have been removed. In other words, if one point is assigned an elevation and all other elevations are computed from it, the same elevation should be obtained for each point no matter which lines are used to compute it. However, since random error remains in the leveling data, misclosures are normally obtained. The elevation differences of the lines must be simultaneously "adjusted" to provide consistent values for the most probable elevations of the points.

Adjustments normally include a combination of new and old leveling lines which form loops with acceptable loop closures. The lines are usually "fitted" into the national network by holding "fixed" elevations of two or more widely spaced points that are common to both the points in the new adjusment and the national network. To determine the most suitable points for this purpose, comparison "profiles" are made for points common to both the new survey and previous surveys (fig. 5-19). The profiles are plots of new-minus-old values, computed from corrected elevations. From the profiles and an analysis of the geology of the area, the fixed points are selected.

In the adjustment, "most probable" corrections are computed by the method of least squares, to eliminate the closure errors of the loops. Misclosure are assumed to be the result of random error; therefore, a correction is apportioned to each line on the basis of the probable magnitude of random error included in it.

For this purpose, the elevation difference of each line is assigned a weight. The weight is based primarily on the distance over which the elevation difference was measured and the techniques and equipment employed to obtain it. Of particular importance is the degree of precision achieved when checking setups, sections, and loops. Other factors include the extent to which the data were reduced, the magnitude of the elevation changes along the line, and the times and temperatures at which observations were made. All lines of leveling are assigned an order and class based on these factors. The tolerances given in figure 3-9 of chapter 3, if satisfied, should yield results of the indicated order and class.

The correction for each line is apportioned among the elevation differences of the sections included in the line. Starting with one of the fixed points, the differences are totaled to obtain adjusted elevations. The adjusted elevations, expressed as normal orthometric heights (see subchapter 3.1), are stored in the data base and published in the Vertical Control Data of the National Geodetic Survey.

#### 5.6 References

- Balazs, E. I. and Young, G. M., 1982: Corrections applied by the National Geodetic Survey to precise leveling observations. NOAA Technical Memorandum NOS NGS 34, National Geodetic Information Center, NOS/NOAA, Rockville, Md. 20852, 14 pp.
- Longman, I. M., 1959: Formulas for computing the tidal accelerations to the moon and the sun. Journal for Geophysical Research 64 (12), 2351-2355.
- National Almanac Office/Her Majesty's Nautical Almanac Office, revised annually: *The Astronomical Almanac*. Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402.
- Greenawalt, C. B. and Floyd, R. P., 1980: National Geodetic Survey Operations Manual. NGS Operations Division, NOAA/NOS, Rockville, Md. 20852 (unpublished manuscript).
- Pfeifer, L. and Morrison, N., 1980: Input Formats and Specifications of the National Geodetic Survey Data Base, volume II: Vertical control data. Federal Geodetic Control Committee. National Geodetic Information Center, NOAA/NOS, Rockville, Md. 20852, 136 pp.

(

# APPENDIX A-LISTS OF EQUIPMENT

### **RECONNAISSANCE AND MARK SETTING**

Mark setting truck. For routine reconnaissance and mark setting, a 1 ton, pickup truck with a utility bed is suitable. It should be fitted with waterproof lockers and boxes for storing tools, supplies, and data. In addition, it should have a cement storage box and a water tank of at least 10-gallon capacity with a hose. The rear should be painted with 6-inch, diagonal, black stripes, so the truck can be easily seen by approaching traffic.

Drilling rig. For setting class A rod marks, a truckmounted drilling rig is necessary. The truck should be fitted for routine reconnaissance and mark setting, including additional storage space for class A rod-mark supplies and a larger, pressurized water tank. The truck must have a sufficiently heavy-duty suspension system to support the drilling rig and its associated augers, specialized tools, and large quantities of supplies.

12-Volt amber safety light, mounted on top of the truck

#### Safety equipment

spare fuses for the safety light tire changing tools tow chain 15-pound ABC-rated fire extinguisher safety flares or reflecting triangles flashlight first-aid kit packet of accident forms 6 to 8 traffic cones 2 brilliant orange safety vests extra safety vests 2 hard hats extra hard hats 2 prs. steel-toed work boots 2 prs. ear plugs or covers 2 prs. safety goggles or glasses 2 prs. work gloves Tools pipe and cable locator rock drill w/tool kit and shipping box rock drill supplies (oil, gasoline, funnel, filters) drill bit 9/16-inch rod driver tip for rod driver (make from a threaded stud and a short piece of stainless steel rod) hydraulic crimper/cutter device with foot pump small shovel standard shovel post-hole digger 2 prs. vice grips

hacksaw and blades wheelbarrow, for mixing cement rivet gun and rivets, for attaching witness-post signs post driver or sledge hammer 2-pound hammer carpenter's hammer 1/4-inch die set, for stamping designations stamping block 30-meter tape measure with both English and metric units 100-meter tape measure with both English and metric units compass large clipboard for maps clipboard for descriptions tool kit for vehicle maintenance Supplies 9/16-inch stainless steel rod threaded, stainless steel studs driving points

stainless steel caps 5-inch PVC pipe, cut in 2-foot lengths PVC glue mortar and cement aluminum logo flanges brass bench-mark disks flagging yellow lumber crayon ("keel") witness posts witness signs notebook black pens strapping tape labels notebook covers mailing tubes and envelopes

#### Additional tools and supplies for drilling rig

ball peen hammer pipe wrenches auger pins and hooks grease gun tool kit for maintenance of drilling rig 1-inch PVC pipe 3/4-inch PVC pipe and female adapters O-rings grease

#### Forms and references

project instructions Bench Mark Description (NOAA Form 76-186) record of designations

- administrative forms (e.g., monthly reports, truck reports, travel vouchers)
- NOAA Manual NOS NGS 3, Geodetic Leveling (this publication)
- NOAA Manual NOS NGS 1, Geodetic Bench Marks
- Input Formats and Specifications of the National Geodetic Survey Data Base, volume II: Vertical control data

operating and maintenance manuals for tools

#### LEVELING

Leveling truck. During conventional leveling a ruggedly constructed "carryall" truck, with standard transmission and good all-around visibility, is suitable for the transportation and storage of leveling equipment. The truck should be equipped with a roof rack on which the rod box and other storage boxes may be mounted. Side steps (for the observer's use) and rear steps (for the rodmen's use) should be mounted at a height that permits handholds on the back of the rack to be reached comfortably. The rear should be painted with 6-inch diagonal black stripes so the truck can be easily seen by approaching traffic. Inside the truck, there should be provision for storing the instrument so it will receive a minimum of shock and vibration. The unit chief should see that the truck is kept orderly and receives routine maintenance and repairs.

#### Safety equipment

 1 12-volt, amber, strobe light, mounted on top of the truck
spare fuses for the safety light

jack and tire-changing tools tow chain 15-pound ABC-rated fire extinguisher safety flares or reflecting triangles 2 "SURVEY CREW AHEAD" warning signs 5 brilliant orange safety vests (one per unit member) extra safety vests

5 hard hats flashlight tool kit for vehicle maintenance first aid kit packet of accident forms

#### When controlling traffic

2 "STOP - SLOW" signs for flagmen2 portable, two-way radios2 or more orange flagsadditional warning signs

#### Technical equipment and supplies

leveling instrument, in a protective case

- instrument tool kit (with adjusting pins, lens brush, small wrench, small screwdriver, cleaning cloth, and manufacturer's manual for the instrument)
- tripod (height compatible with the instrument and the observer)

thermometer, for reading air temperature digital temperature display and 2 shielded sensors, mounted on tripod, for measuring air temperature differentials

spare batteries and sensor for temperature equipment black pen with waterproof ink clipboard or notebook, for recording forms ť

ĺ

chipotata of hotebook, for recording form.

lightweight collapsible, stool

2 leveling rods, in padded storage box

2 turning-pin guides ("flippers"), attached to the rods 2 pairs of top-mounted, brace poles for the rods

plumb bob

1-meter Invar scale in a protective case

pair of calibrated 20-millimeter spacers ("plugs") carpenter's level

3 turning pins with driving caps

3 plastic-headed mallets

spare plastic replacement heads for mallets

3 heavyweight turning plates

supply of 2-foot long, 2-inch x 2-inch stakes

double-headed nails (30-40 penny)

2 carpenter's hammers

railroad spikes, for checkpoints

yellow lumber crayon ("keel")

flagging

machete, for clearing brush

30-meter steel tape, with both English and metric units shovel

hand level

large umbrella

5-gallon water cooler

large briefcase or satchel, for storing forms and small supplies

#### Forms and references

Examples of forms used for geodetic leveling appear in appendix C. NOAA personnel may obtain the forms listed in this manual from:

NOAA Central Logistics Supply Center, 619 Hardesty St. Kansas City, MO 64124. Other users requesting these forms should contact the NGS Vertical Network Division NOS/NOAA, Rockville, MD 20852 (301-443-8567)

Bench Mark Description (NOAA Form 76-186)

- Geodetic Leveling Micrometer Observations (NOAA Form 76-191)
- Geodetic Leveling Three-Wire Observations (NOAA Form 76-189)
- Geodetic Leveling Compensation Check (NOAA Form 77-81)
- Level Line Identification (NOAA Form 76-188)

Geodetic Leveling Field Abstract (NOAA Form 76-187)

- administrative forms (e.g., weekly and monthly reports, travel vouchers, leave requests)
- description of the leveling line ("logs")

project instructions

- NOAA Manual NOS NGS 3, Geodetic Leveling (this publication)
- Input Formats and Specifications of the National Geodetic Survey Data Base, volume II: Vertical control data
- Classification, Standards of Accuracy, and General Specifications of Geodetic Control Surveys, Federal Geodetic Control Committee
- Specifications to Support Classification, Standards of Accuracy, and General Specifications of Geodetic Control Surveys, Federal Geodetic Committee

#### Other supplies

notebook spare pens strapping tape paper clips

#### Computer-recording equipment and supplies

computer-recording system power-supply cord and hookup battery-charging cord and hookup spare batteries spare fuses set of recording programs tapes or disks, in a magnetic-proof storage container backup-recording forms: Geodetic Leveling (NOAA Form 77-82) instructions and program documentation for the computer-recording system colored labels, to identify data tapes and disks aluminum mailing containers padded envelopes pre-addressed mailing labels isopropyl alcohol and cotton for cleaning tape recording heads

,

ĺ

(

# APPENDIX B.—INDEX OF GEODETIC CONTROL DIAGRAMS

Two-letter code	de 1° x 2° area designation Diagram No. Two-letter code		1° x 2° area designation	Diagram No.	
AA	Key West	NG-17-11	СК	Savannah	NI-17-11
AB	Brownsville	NG-14-6.9	CL	Macon	NI-17-10
AC	Miami	NG-17-8	СМ	Phenix City	NI-16-12
AD	West Palm Beach	NG-17-5	CN	Montgomery	NI-16-11
AE	McAllan	NG-14-5	CO	Meridian	NI-16-10
AF	Ft. Pierce	NG-17-2	CP	Jackson	NI-15-12
AG	Tampa	NG-17-1.4	CO	Shreveport	NI-15-11
AH	Corpus Christi	NG-14-3	CR	Tyler	NI-15-10
AJ	Laredo	NG-14-2	CS	Dallas	NI-14-12
AK	Orlando	NH-17-11	СТ	Abilene	NI-14-11
AL	Plant City	NH-17-10	CU	Big Spring	NI-14-10
AM	Bay City	NH-15-10	CV	Hobbs	NI-13-12
AN	Beeville	NH-14-12	CW	Carlsbad	NI-13-11
AO	Crystal City	NH-14-11	СХ	Las Cruces	NI-13-10
AP	Eagle Pass	NH-14-10	CY	Silver City	NI-12-12
AO	Davtona Beach	NH-17-8	CZ	Tucson	NI-12-11
AR	Gainesville	NH-17-7	DA	Aio	NI-12-10
AS	Apalachicola	NH-16-9	DB	El Centro	NI-11-12
AT	Brenton Sound	NH-16-7	DC	San Diego	NI-11-11
AU	New Orleans	NH-15-9	DD	Georgetown	NI-17-9
AV	Port Arthur	NH-15-8	DE	Augusta	NI-17-8
AW	Houston	NH-15-7	DF	Athens	NI-17-7
AX	Sepiin	NH-14-9	DG	Atlanta	NI-16-9
AY	San Antonio	NH-14-8	DH	Birmingham	NI-16-8
AZ	Del Rio	NH-14-7	DJ	West Point	NI-16-7
BA	Emory Peak	NH-13-9	DK	Greenwood	NI-15-9
BB	Presidio	NH-13-8	DL	El Dorado	NI-15-8
BC	Jacksonville	NH-17-5	DM	Texarkansas	NI-15-7
BD	Valdosta	NH-17-4	DN	Sherman	NI-14-9
BE	Tallahassee	NH-16-6	DO	Wichita Falls	NI-14-8
BG	Pensacola	NH-16-5	DP	Lubbock	NI-14-7
BH	Mobile	NH-16-4	DQ	Brownfield	NI-13-9
BJ	Baton Rouge	NH-15-6	DR	Boswell	NI-13-8
BK	Lake Charles	NH-15-5	DS	Tularosa	NI-13-7
BL.	Beaumont	NH-15-4	DT	Clifton	NI-12-9
BM	Austin	NH-14-6	DU	Mesa	NI-12-8
BN	Ilano	NH-14-5	DV	Phoenix	NI-12-7
BO	Sonora	NH-14-4	DW	Salton Sea	NI-11-9
BP	Ft. Stockton	NH-13-6	DX	Santa Ana	N1-11-8
BQ	Marfa	NH-13-5	DY	Long Beach	NI-11-7
BR	Brunswick	NH-17-2	DZ	Santa Maria	NI-10-6,9
BS	Waycross	NH-17-1	EA	Beaufort	NI-18-4
BT	Dothan	NH-16-3	EB	Florence	NI-17-6
BU	Andalusia	NH-16-2	EC	Spartanburg	NI-17-5
BV	Hattiesburg	NH-16-1	ED	Greenville	NI-17-4
BW	Natchez	NH-15-3	EE	Rome	NI-16-6
BX	Alexandria	NH-15-2	EF	Gadsden	NI-16-5
BY	Palestine	NH-15-1	EG	Tupelo	NI-16-4
BZ	Waco	NH-14-3	EH	Helena	NI-15-6
CA	Brownwood	NH-14-2	EJ	Little Rock	NI-15-5
СВ	San Angelo	NH-14-1	EK	McAlester	NI-15-4
CC	Pecos	NH-13-3	EL.	Aramore	NI-14-6
CD	Van Horn	NH-13-2	EM	Lawton	NI-14-5
CE	El Paso	NH-13-1	EN	Plainview	NI-14-4
CF	Douglas	NH-12-3	EO	Clovis	NI-13-6
CG	Nogales	NH-12-2	EP	Ft. Sumner	NI-13-5
СН	Lukeville	NH-12-1	EQ	Socorro	NI-13-4
CJ	James Island	NI-17-12	ER	St. Johns	NI-12-6

## Table B.1-Codes for NGS Geodetic Control Diagrams

## Appendix B—Index of Geodetic Control Diagrams

## Table B.1-Codes for NGS Geodetic Control Diagrams-Continued

Two-letter code	1° x 2° area designation	Diagram No.	Two-letter code	1° x 2° area designation	Diagram No.	(
ES	Holbrook	NI-12-5	НЈ	La Junta	NJ-13-9	<u>у</u> .,
ET	Prescott	NI-12-4	нк	Trinidad	NJ-13-8	
EU	Needles	NI-11-6	HL	Durango	NI-13-7	
EV	San Bernadino	NI-11-5	HM	Cortez	NJ-12-9	
EW	Los Angeles	NI-11-4	HN	Escalante	NI-12-8	
EX	Manteo	NI-18-2	НО	Cedar City	NI-12-7	
EY	Rocky Mount	NI-18-1	HP	Caliente	NI-11-9	
EZ	Raleigh	NI-17-3	НО	Goldfield	NJ-11-8	
FA	Charlotte	NI-17-2	HR	Marinosa	NI-11-7	
FB	Knoxville	NI-17-1	HS	San Jose	NI-10-9	
FC	Chattanoga	NI-16-3	нт	San Francisco	NI-10-8	
FD	Columbia	NL16-2	HU	Salishury	NL18-5	
FF	Blytheville	NI-16-1	HV	Washington	NL18.4	
FF	Memphis	NL15-3	HW	Charlottesville	NL17.6	
FG	Russelville	NL-15-7	нх	Charleston	NL17-5	
FH	Ft Smith	NL15-1	НУ	Huntington	NL17-4	
FI	Oklahoma City	NL14.3	H7	Louisville	NL16.6	
FK	Clinton	NE14.2	14	Vincennes	NJ-16-5	
FI	Amarillo	NI-14-1	IR IR	Pallavilla	NL16.4	
FM	Tucumcari	NI_11_1	10	St. Louiz	NI115.6	
FN	Santa Fe	NIL13.0	JC ID	Jaffarson City	NTE15-5	
FO	Albuquerque	NI-13-2 NII.13.1	םו ענ	Lawranaa	14J-1J-J NT 15 A	
FD	Gallup	NI-13-1	JE	Lawrence	NJ-13-4 NT 14 4	
FF	Ganap	NI-12-3 NI-12-3	JF	Autoninson	INJ-14-0	
FQ	riagstall Williama	INI-12-2	JO	Great Beng	INJ-14-5	
FK	Winams Kinaman	NI-12-1	JK	Pueblo	NJ-13-D	
г <b>э</b> г:т	Kingman	NI-11-3	JL M	Montrose	NJ-13-4	
F 1	I FORA	NI-11-2	JML	Moab	NJ-12-0	
rU FV	Bakersheld	NI-11-1	AL N	Salina	NJ-12-5	
FV	San Luis Obispo	NI-10-3	JO	Richlield	NJ-12-4	
FW	Eastville	NJ-18-8,11	JP JP	Lund	NJ-11-6	(
FX	Notiolk	NJ-18-10	JQ.	lonapah	NJ-11-5	·
FY	Greensboro	NJ-17-12	JR	Walker Lane	NJ-11-4	
FZ	Winston-Salem	NJ-17-11	JS	Sacramento	NJ-10-6	
GA	Johnson City	NJ-17-10	JI	Santa Rosa	NJ-10-5	
GB	Corbin	NJ-16-12	JU	Wilmington	NJ-18-2	
GC	Nashville	NJ-16-11	JV	Baltimore	NJ-18-1	
GD	Dyersburg	NJ-16-10	JW	Cumberland	NJ-17-3	
GE	Poplar Bluff	NJ-15-12	JX	Clarksburg	NJ-17-2	
GF	Harrison	NJ-15-11	JY	Colombus	NJ-17-1	
GG	luisa	NJ-15-10	JZ	Cincinnati	NJ-16-3	
GH	Enid	NJ-14-12	KA	Indianapolis	NJ-16-2	
GJ	Woodward	NJ-14-11	КВ	Decatur	NJ-16-1	
GK	Perryton	NJ-14-10	KC	Quincy	NJ-15-3	
GL	Dalhart	NJ-13-12	KD	Moberly	NJ-15-2	
GM	Raton	NJ-13-11	KE	Kansas City	NJ-15-1	
GN	Aztec	NJ-13-10	KF	Manhattan	NJ-14-3	
GO	Shiprock	NJ-12-12	KG	Beloit	NJ-14-2	
GP	Marble Canyon	NJ-12-11	КН	Goodland	NJ-14-1	
GQ	Grand Canyon	NJ-12-10	KJ	Limon	NJ-13-3	
GR	Las Vegas	NJ-11-12	KK	Denver	NJ-13-2	
GS	Death Valley	NJ-11-11	KL.	Leadville	NJ-13-1	
GT	Fresno	NJ-11-10	KM	Grand Junction	NJ-12-3	
GV	Richmond	NJ-18-7	KN	Price	NJ-12-2	
GW	Roanoke	NJ-17-9	ко	Delta	NJ-12-1	
GX	Bluefield	NJ-17-8	KP	Ely	NJ-11-3	
GY	Jenkins	NJ-17-7	KQ	Millett	NJ-11-2	
GZ	Winchester	NJ-16-9	KR	Reno	NJ-11-1	
HA	Evansville	NJ-16-8	KS	Chico	NJ-10-3	
HB	Paducah	NJ-16-7	KT	Ukiah	NJ-10-2	
HC	Rolla	NJ-15-9	KU	New York	NK-18-12	1
HD	Springfield	NJ-15-8	KV	Newark	NK-18-11	ť
HE	Joplin	NJ-15-7	KW	Harrisburg	NK-18-10	
HF	Wichita	NJ-14-9	KX	Pittsburgh	NK-17-12	
HG	Pratt	NJ-14-8	KY	Canton	NK-17-11	
нн	Dodge City	NJ-14-7	KZ	Marion	NK-17-10	

## Table B.1-Codes for NGS Geodetic Control Diagrams-Continued

Two-letter code	1° x 2° area designation	1° x 2° area designation	Diagram No.			
LA	Muncie	NK-16-12	NQ	Torrington	NK-13-5	
L.B	Danville	NK-16-11	NR	Casper	NK-13-4	
LC	Peoria	NK-16-10	NS	Lander	NK-12-6	
LD	Burlington	NK-15-12	NT	Preston	NK-12-5	
LE	Centerville	NK-15-11	NU	Pocatello	NK-12-4	
LF	Nebraska City	NK-15-10	NV	Twin Falls	NK-11-6	
1 G	Lincoln	NK-14-12	NW	Jordan Valley	NK-11-5	
14	Grand Jeland	NK-14-11	NX	Adel	NK-11-4	
11	MaCook	NK-14-10	NY	Klamath Falls	NK-10-6	
	Starling	NIC 13 13	NZ	Madford	NV 10 5	
	George	NK-13-12		Cose Ber	NV 1014	
	Greeley	NK-13-11		Coos Day	NK-10-1,4	
L.IVI	Craig	NK-13-10	OB OC	Dau	INK-19-2	
LN	vernal	NK-12-12		Portiand	INK-19-1	
LO	Salt Lake City	NK-12-11	OD	Gien Falls	NK-18-3	
LP	Tooele	NK-12-10	OE	Utica	NK-18-2	
LQ	Elko	NK-11-12	OF	Rochester	NK-18-1	
L.R	Winnemucca	NK-11-11	OG	Toronto	NK-17-3	
LS	Lovelock	NK-11-10	OH	Boise	NK-11-2	
LT	Susanville	NK-10-12	OJ	Flint	NK-17-1	
LU	Redding	NK-10-11	ОК	Midland	NK-16-3	
LV	Eureka	NK-10-7,10	OL.	Milwaukee	NK-16-2	
LW	Providence	NK-19-7	ОМ	Madison	NK-16-1	
ĨX	Hartford	NK-18-9	ON	La Crosse	NK-15-3	
	Scranton	NK-18-8	00	Mason City	NK-15-2	
17	Williamanat	NK-10-0	00	Fairmont	NK-15-2	
1.2.	Winnamsport	NK-10-7	OF	Classe Calle	NK-13-1	
MA	warren	INK-17-9	0Q CD	Sloux rails	INK-14-3	
мв	Cleveland	NK-1/-8	OR	Mitchell	NK-14-2	
MC	Ioledo	NK-17-7	OS	Martin	NK-14-1	
MD	Ft. Wayne	NK-16-9	OT	Hot Springs	NK-13-3	
ME	Chicago	NK-16-8	OU	Newcastle	NK-13-2	
MF	Aurora	NK-16-7	ov	Arminto	NK-13-1	
MG	Davenport	NK-15-9	ow	Thermopolis	NK-12-3	
MH	Des Moines	NK-15-8	ox	Driggs	NK-12-2	
MJ	Omaha	NK-15-7	ΟΥ	Idaho Falls	NK-12-1	
МК	Fremont	NK-14-9	OZ	Hailev	NK-11-3	
ML.	Broken Bow	NK-14-8	PA	Burns	NK-11-1	
MM	North Platte	NK-14-7	PR	Crescent	NK-10-3	
NANI	Scottchluff	NK-12-0	PC	Docehuro	NK-10-2	
MO	Chavanaa	NIV 12 9	PD	Eastport	NI -10-12	
MO	Develine	NK-13-0	PD DE	Damoan	NI 10 11	
MP	Rawins Deck Certere	INK-10-7	FE DE	Dangor	NL 10 10	
MQ	Rock Springs	NK-12-9	Pr	Lewiston	INL-19-10	
MR	Ogden	NK-12-8	PG	Lake Champlain	NL-18-12	
MS	Brigham City	NK-12-7	PH	Ogdensburg	NL-18-11	
MT	Wells	NK-11-9	PJ	Kingston	NL-18-10	
MU	McDermitt	NK-11-8	РК	Tawas City	NL-17-10	
MV	Vya	NK-11-7	PL.	Traverse City	NL-16-2	
MW	Alturos	NK-10-9	PM	Manitowoe	NL-16-11	
MX	Weed	NK-10-8	PN	Green Bay	NL-16-10	
MY	Boston	NK-19-4	PO	Eau Claire	NL-15-12	
M7	Albany	NK-18-6	pp	St. Paul	NL-15-11	
NΔ	Binghamton	NK-18-5	PO	New Liim	NL-15-10	
NR	Elmin	NK-18-4	PR	Watertown	NI -14-12	
NC	Duffala	NV 176	PS	Huron	NL-14-11	
ND		NK-17-0	IS PT	Diarra	NE 14 10	
ND	Ene	INK-17-3	E I	Pierte Danid Cim	NL-14-10	
NE	Detroit	INK-17-4	FU		INL-15-12	
NF	Grand Rapids	NK-16-6	۲V	Gliette	NE-13-11	
NG	Racine	NK-16-5	PW	Sheridan	NL-13-10	
NH	Rockford	NK-16-4	PX	Cody	NL-12-12	
NJ	Dubuque	NK-15-6	PY	Ashton	NL-12-11	
NK	Waterloo	NK-15-5	PZ	Dubois	NL-12-10	
NL.	Fort Dodge	NK-15-4	QA	Challis	NL-11-12	
NM	Sioux City	NK-14-6	QB	Baker	NL-11-11	
NN	O'Neill	NK-14-5	òc	Canyon City	NL-11-10	
NO	Valentine	NK-14-4	òp	Bend	NL-10-12	
	* WEWEEEEW	· · • • • · · ·	×			

Two-letter code	<u>1° x 2° area designation</u>	Diagram No.	Two-letter code	1° x 2° area designation	Diagram No.	_ i
QF	Fredericton	NL-19-9	RZ	Pullman	NL-11-5	į
òg	Millinocket	NL-19-8	SA	Walla Walla	NL-11-4	
он	Sherbrooke	NL-19-7	SB	Yakima	NL-10-6	
τÓ	Alpena	NL-17-7	SC	Hoquiam	NL-10-5	
ŎК	Cheboygan	NL-16-9	SD	Copalis Beach	NL-10-1,4	
QL	Escanaba	NL-16-8	SE	Campbellton	NL-19-3	
ОM	Iron Mountain	NL-16-7	SF	Edmundston	NL-19-2	
ÔN	Rice Lake	NL-15-9	SG	Hancock	NL-16-1,2	
<u>oo</u>	Stillwater	NL-15-8	SH	Two Harbors	NL-15-3	
QP	St. Cloud	NL-15-7	SJ	Hibbing	NL-15-2	
QQ	Milbank	NL-14-9	SK	Bemidji	NL-15-1	
QR	Aberdeen	NL-14-8	SL	Grand Forks	NL-14-3	
QS	McIntosh	NL-14-7	SM	New Rockford	NL-14-2	
QT	Lemmon	NL-13-9	SN	McClusky	NL-14-1	
οŪ	Ekalaka	NL-13-8	SO	Watford City	NL-13-3	
QV	Hardin	NL-13-7	SP	Glendive	NL-13-2	
ow.	Billings	NL-12-9	SO	Jordan	NL-13-1	
òx	Bozeman	NL-12-8	SR	Lewistown	NL-12-3	
ÔΥ	Dillon	NL-12-7	SS	Great Falls	NL-12-2	
òz	Elk City	NL-11-9	ST	Choteau	NL-12-1	
RA	Grangeville	NL-11-8	SU	Wallace	NL-11-3	
RB	Pendleton	NL-11-7	SV	Spokane	NL-11-2	
RC	The Dalles	NL-10-9	SW	Ritzville	NL-11-1	
RD	Vancouver	NL-10-8	SX	Wenatchee	NL-10-3	
RE	Woodstock	NL-19-6	SY	Seattle	NL-10-2	
RF	Presque Isle	NL-19-5	SZ	Ft. William	NM-16-10	
RG	Quebec	NL-19-4	TA	Quetico	NM-15-12	
RH	Blind River	NL-17-4	TВ	International	NM-15-11	
RJ	Sault Sainte	NL-16-6	TC	Roseau	NM-15-10	
RK	Marquette	NL-16-5	TD	Thief River Falls	NM-14-12	
RL	Iron River	NL-16-4	TE	Devils Lake	NM-14-11	
RM	Ashland	NL-15-6	TF	Minot	NM-14-10	
RN	Duluth	NL-15-5	TG	Williston	NM-13-12	
RO	Brainerd	NL-15-4	тн	Wolf Point	NM-13-11	
RP	Fargo	NL-14-6	TJ	Glasgow	NM-13-10	
RQ	Jamestown	NL-14-5	ТК	Havre	NM-12-12	
RR	Bismarck	NL-14-4	TL	Shelby	NM-12-11	
RS	Dickinson	NL-13-6	TM	Cut Bank	NM-12-10	
RT	Miles City	NL-13-5	TN	Kalispell	NM-11-12	
RU	Forsyth	NL-13-4	то	Sandpoint	NM-11-11	
RV	Roundup	NL-12-6	TP Okanogan NM-11		NM-11-10	
RW	White Sulphur	NL-12-5	TQ	Concrete	NM-10-12	
RX	Butte	NL-12-4	TR	Victoria	NM-10-11	
RY	Hamilton	NL-11-6	TS	Cape Flattery	NM-10-10	

 $\langle$ 

Ĺ

Table B.1-Codes for NGS Geodetic Control Diagrams-Continued

# APPENDIX C.—STANDARD LEVELING FORMS OF THE NATIONAL GEODETIC SURVEY

This section contains the geodetic leveling forms used by the National Geodetic Survey. Reproductions may be made for training purposes or for field use in the event that the supply is temporarily exhausted. Forms appear in the following order :

NOAA Form 76-60, Report on relocation of bench mark. NOAA Form 76-186, Bench mark description . NOAA Form 76-91, Report on condition of survey mark. NOAA Form 76-159, Weekly report-leveling unit. NOAA Form 76-187, Geodetic leveling field abstract NOAA Form 76-188, Geodetic leveling line identification. NOAA Form 76-189, Geodetic leveling three-wire observations. NOAA Form 76-191, Geodetic leveling micrometer observations. NOAA Form 76-198, Monthly report-bench mark and reconnaissance teams. NOAA Form 76-104, Monthly report-leveling unit NOAA Form 77-82, Geodetic leveling. NOAA Form 77-81, Geodetic leveling compensation check Record of bench mark designations assigned by mark setter. Check-off list for descriptive editing. River crossing, simultaneous reciprocal observations River crossing, Zeiss River Crossing Equipment River crossing series

Ć

ſ

(

NOAA FORM (6-71)	76-60 N	ATIONAL OCEANIC	U. S. DEPART	MENT OF COMMERCE	DESIGNATION	DF MARK
	REPORT	ON RELOCATION	STATE	COUNTY		
	(Se	e instructions on	reverse side)		}	
A DESCRI	PTION OF ORIG	INAL MARK				
PUBLISHE	ON	······································		ST AMPING		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
B. WAS OR	GINAL BENCH	ARK DESTROYED?	YES	HORIZO	NTAL CONTROL	POINT
REMAR	<b>&lt;</b> \$					
				DATE	OF LEVELING	
	1	(Feet or Meters)*	T		r	
POINT	B.S.	н. і.	FS.	ELEVATION		REMARKS
				**************************************		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
		-		······································		
C DESCRI	TION OF NEW H	ARK		······································	I	
				STAMPING	ON NEW MARK	
*Strike a	ut unit NOT us	ed.	SIGNED	·		
			AGENCI			

NOAA FORM 76-186 18-801	BENCH MARK I	) E S C R I P T I O N	U.S. DEPARTMENT OF COMMERCE NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION NATIONAL OCEAN SURVEY
*10*SPSN-7, DRC CODE-7_, APPROX LAT-7	APPROX LON-7	*11*QUAD-;	,QSN,*12*LINE,
*13*DESIGNATION-7		*14*STATE CODE/COUNTY-;'/	'
*15*ALIAS-,	<sup>1</sup>	(PARISH in Loui	siana - State CENSUS DIVISION in Alaska)
*16*AREA-1		DISTANCE	AND
*17*NEAREST CITY OR TOWN-7		' *18*NEAREST	CITY OR TOWN-7
*20*CODE/SETTING by AGENCY-7_1/1		*21*YEAR-1, CHIEF OF PA	RTY-7, *22*OTHER CONTROL-7
*23*CODE/RECOVERY BY AGENCY-7_1/1		*24*YEAR-T, CHIEF OF PA	RTY-7, *25*CONDITION OF MARK-7_,
*26*CODE/SETTING CLASSIFICATION-7/_		·	
*27*MONUMENTATION-; DISK TYPE-; QUALITY O	/ERRIDE-T_ ORIGIN-T_		- <u></u>
*28*STAMPING-7			<sup>1</sup>
**************************************	L OR RECOVER	Y DESCRIPTIVE TE	X T **********
*30*,			
*30*,			
*30*,			
*30*,			
*30*,			
*30*,			
*30*,			
*30*,			
			'
*30";			······································
" <u>20"</u> ,			······································
*30*.			<sup>-</sup> .
*40*(Units, .E-English/M-metric) A DISK SET INTO	THE TOP OF A CONCRETE POS	ST., F-FLUSH WITH THE GROUND/P-F	PROJECTING/R-RECESSED, IN/CM
*41*(Units, , Setting Code, , same as in *26*) R	DD/PIPE DRIVEN TO THE DEP	TH OF, ,FT/M, IN A SLEEVE	EXTENDING TO THE DEPTH OF, ,FT/M
(leave preceding entry blank if no s	sleeve), ENCASED IN A PII	PE,_,F-FLUSH WITH THE GROUND/P-P	ROJECT ING/R-RECESSED,, IN/CM
*42*(Units,_,),,FT/M,_,A-ABOVE/B-BELOW/L-A	BOUT LEVEL WITH (Units an	nd FT/M blank if 'L'),	
*43*(Units,_,),,FT/M,,(Point of Compa	ss) FROM A WITNESS POST -	- else specify object,	······································
, Mi/Km,, of Bench Mark,		(start entr	y with article (a,an,the) if applicable)

SUPERSEDES NOAA FORM 76-185 (3-60) WHICH MAY BE USED.

Appendix C-Standard Leveling Forms of the National Geodetic Survey

	REPORT ON CON	NDITION OF SURVEY MARK	Form Approved: OMB -R1923 Approvel Expires: April 30, 1983
Name or Designation:		Year Established:	
State:	County:	Organization Establis	hed by:
Distance and direction from nearest to	wn:		
Description published in: (Line book	or quadrangle numbe	r)	
Mark searched for or recovered by:	Name		······································
	Organization		
Date of report	_ Address		
Condition of marks: List letters and a	numbers found stamped	in (not cast in) each mark.	
Mark stamped:	Condition:		
		****	**
		******	
<b> </b>			
Marks accessible? Yes Please report on the thoroughness of t repairing or moving the mark, or other	No he search in case a m pertinent facts:	Property owner contacted? ark was not recovered, suggested cha	Yes No nges in description, need for
This report is authorized by law (Title While you are not required to respond, needed to make the results of this sur- accurate, and timely.	33 USC Section 883e), your cooperation is rey comprehensive,	Witness P Vitness P	ost? Yes <u>No</u> nark.

If additional forms are needed, indicate number required

Witness Post set \_\_\_\_ feet \_\_\_\_ of \_\_\_\_\_ mark. Witness Post set \_\_\_\_ feet \_\_\_\_ of \_\_\_\_\_ mark.

(

Ć

NOAA FORM 76-91 US DEPARTMENT OF COMMERCE NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION NATIONAL OCEAN SURVEY (5-78)

NOA (3-8)	A FORM	76-159				a <u>,</u>		WEEI	(1 V P	FDODT	1.57	NATIONAL OCEANIC	U.S. DEPARTMENT OF COMMERCE AND ATMOSPHERIC ADMINISTRATION	
										RECORD	ER		MONTH/YEAR	
													19	
RODMEN, PACER								<b>-</b>	PARTY					
LOCA	ATION (S	)									PROJECT			
LINE	(5)										SURVE	YORDER	CLASS	
		DISTAN	ICE (k	.m )		ним	BER	TI!	ME	COLL				
DAY	DBL RUN Prog	SGL RUN PROG	15 T RE- RUN	ADL RE RUN	TOT ALL RUN	SET	SEC-	IST SET UP	LAST SET UP	AND COMP CKS	STAFF DAY	WEATHER CONDITIONS	REMARKS	
Sun														
Mon														
Tue														
Wed														
Thu														
Fri														
Sat														
Total												km/Staff-Days =		

NDAA FORM (8-75)	76-187				NATION	AL OCEANIC AND	J.S. DEPARTA	AENT OF COMMERCE
		GEODETIC LEVELING FI	ELC	) A	BSTRACI	•		
HGZ L						SHEET	SHEETS	
TITLE								
				( <u> </u>			( (B) (B)	·····
DISTANCE	UNITS: H	M SM D/F-E. UNITS: MT FT	Day	۲ В	$\Delta/\Sigma$	D	$\Delta/\Sigma$	D/F.E.
*30*	FROM	tan tan tan tan tan tan tan tan tan tan				<u>, , , , , , , , , , , , , , , , , , , </u>		······································
	τo							
• 30*	FROM							
	то							
^30*	FROM	handarahandarahanda <u>ili li li li ka</u> ukuakuakuakuakuakuakaakaakaakaakaakaakaa						······································
	то							
•30•	FROM	<u></u>						
	то							
•30*	FROM	landundundundundundundundundundundundundun	<b> </b>					······
	то							
• 30'	FROM	└┨╌┇╺╞╍┟╼┇╌┨╸╢╴╢╴╢╴┨╴┇╶┨╌┧╼╢╸╋╼┨╸┨╼┨╼┨╸	<b>—</b>					
	то	<u>, , , , , , , , , , , , , , , , , , , </u>	<u> </u>					······································
						<u>.                                    </u>		
^30	- FROM	Tankankunkunkunkankankunkunkunkunkunkunkunkunkunkunkunkunkun		<b> </b>				
	то							
• 30'	• FROM	┕╅╼┨╴┨╶┨╼┨╼┨╼┨╼┨╴┨╴┨╶┨╶┨╶┨╶┨╶┨╸┥╍┥╍┥╍┥						
نسلسل	то							
			-	 				
• 30	FROM	┶┺┺┺┺┺┺┺┺┺┺┺┺┺┺	<b> </b>		<u> </u>			
	то							······································
• 30	FROM	<u> </u>		<b> </b>				
	<u> </u>			Ŀ	L			
•30 •30 •30 •30 •30	• FROM TO • FROM TO • FROM TO					E AND		

EXISTING STOCK SHOULD BE DESTROYED UPON RECEIPT OF REVISION.

tr U.S. Government Printing Office: 1977-765-005/1144 Region No. 6

NOAA FORM 76-188 (8-76)	U.S. DEPARTMENT OF COMMERCE NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION
GEODETIC LEVELING	LINE IDENTIFICATION
HGZ NUMBER LINE HGZ NUMBER HGZ NU	LER- NCE CTOR
* 1 3 *	
* 1 4 *	
	COMMENTS
* 1 5 *	
TASK NUMBER	RODS
CHIEF OF PARTY	
OBSERVERS	
RECORDERS	INSTRUMENTS

☆ U. S. Government Printing Office: 1976-765-005/1082, Region No. 6

.....

C-6

.----

~~~

| U. S. DEPARTMENT C<br>(4-77) NATIONAL OCEANIC AND ATMOSPHERIC AD |                                  |   |               |           |     |               |                 |                |               | FMENT OF  | COMMERCE | PAGE |               |             |                |           |          |  |
|------------------------------------------------------------------|----------------------------------|---|---------------|-----------|-----|---------------|-----------------|----------------|---------------|-----------|----------|------|---------------|-------------|----------------|-----------|----------|--|
| GEODETIC LEVELING<br>THREE_WIRE OBSERVATIONSOF                   |                                  |   |               |           |     |               |                 |                |               |           |          |      | 0F            |             |                |           |          |  |
| * 4                                                              | 0                                | × | YR,           | /мо.      | /o# | Y CODE INSTRU | MENT SERIAL NO. | CODE           | ROD SERIAL    | NO.       | CODE     | R    | DD SERIAL NO. |             |                |           | Z TIME   |  |
| * 4                                                              | 1 1                              | * |               |           |     |               |                 |                |               |           |          |      |               |             |                |           |          |  |
| FR                                                               | OM BM DESIGNATION TO BM DESIGNAT |   |               |           |     |               |                 |                | M DESIGNATION |           |          |      | Z TI<br>BEGIN | ME END C E  | TEMPER         | ATURE W   | S SERVER |  |
|                                                                  |                                  | L |               |           |     |               |                 |                |               |           |          |      |               |             |                |           |          |  |
|                                                                  |                                  |   |               |           |     | BACKSI        | GHT             |                |               | FORESIGHT |          |      |               |             |                |           |          |  |
|                                                                  | <b>_</b>                         | U | IPPE          | ER        |     |               | -               |                |               | L         | UPP ER   |      |               |             |                |           |          |  |
| IIP                                                              | ļ                                | M | וססו          | E         |     | MEAN          | BACK CENTER     | s <sub>u</sub> |               |           | MIDDLE   | -    | MEAN          | BACK CENTER | S <sub>U</sub> | 111.4.1.) |          |  |
|                                                                  | ┣                                |   | .0 <b>π</b> ε | ER        |     | 5             | -               | SL.            | (U+L)<br>5    |           | Σ<br>Σ   |      | Σ             |             | <u>ه۲</u>      | Σ         |          |  |
|                                                                  | T                                |   | Ť             |           | 1   |               |                 |                | TĨI!          |           |          | Т    |               |             |                |           |          |  |
|                                                                  |                                  |   |               |           |     |               |                 |                |               |           |          | İ    |               |             |                |           |          |  |
|                                                                  |                                  |   |               |           |     |               |                 |                |               |           |          |      |               |             |                |           |          |  |
|                                                                  |                                  |   |               |           |     |               | _               |                |               |           |          | •    |               |             | <u> </u>       |           |          |  |
|                                                                  | ┢╌┼                              | ┿ | -             |           | +   |               |                 |                |               | ┝┿        |          | ÷    |               |             |                |           |          |  |
| -                                                                | ┢┼┥                              |   |               |           | +   |               |                 |                |               |           |          | ÷    |               |             |                |           |          |  |
|                                                                  |                                  | + |               | ┝┼╴       | t   |               |                 |                |               |           |          | i    |               |             | -t-t-          |           |          |  |
|                                                                  |                                  | - |               |           |     |               | 1               |                |               |           |          | i    |               |             | <b>!</b>       |           |          |  |
|                                                                  |                                  |   |               |           | Ţ   |               |                 |                |               |           |          |      |               |             |                |           |          |  |
|                                                                  |                                  |   |               |           | •   |               |                 |                |               |           |          | 1    |               |             |                |           |          |  |
|                                                                  |                                  | _ | _             |           | +   |               | -               |                |               |           |          | •    |               |             | •              |           |          |  |
|                                                                  | ┢╌┾                              | - |               |           | ÷   |               |                 |                |               |           |          |      |               |             |                |           |          |  |
|                                                                  |                                  |   |               |           | +   |               |                 |                |               |           |          | i    |               |             |                |           |          |  |
| <u>'</u>                                                         |                                  | ╈ | +             | ┝╍┝╸      | 1   |               |                 |                |               |           |          | Ť    |               |             | Ţ              |           |          |  |
|                                                                  |                                  |   |               |           |     |               |                 |                |               |           |          |      |               |             |                |           |          |  |
|                                                                  |                                  | _ |               |           | -   |               |                 |                |               |           |          | •    |               |             |                |           |          |  |
|                                                                  |                                  |   | _             | $\square$ | -   |               |                 |                |               | ┣─┼─      |          | -    |               |             |                |           |          |  |
|                                                                  | ┣─┼                              | - | _             | $\vdash$  | +-  |               | -               |                |               | <b> </b>  | ┝╍┟╍┟╸┟╸ | •    |               |             | <b>-</b>       |           |          |  |
|                                                                  | ┢┼                               |   |               |           |     |               |                 |                | ┝┷┷┷┥         | ┠┼─       |          | i    |               |             |                |           |          |  |
|                                                                  |                                  | 十 |               | $\vdash$  | 1   |               |                 |                |               |           |          | i    |               |             |                |           |          |  |
|                                                                  | П                                |   |               |           | Ţ   |               |                 |                |               |           |          | •    |               |             |                |           |          |  |
|                                                                  |                                  |   |               |           | !   |               |                 |                |               |           |          | 1    |               | 1           |                |           |          |  |

2

|                  |                                     | BACKSIGHT     |                  |            |             | FORE                   | SIGHT                                  |                    |           | COMMENTS |
|------------------|-------------------------------------|---------------|------------------|------------|-------------|------------------------|----------------------------------------|--------------------|-----------|----------|
|                  | UPPER                               |               | ļ                |            | UPP ER      |                        |                                        |                    |           |          |
| SET<br>IIP       | MIDDLE MEAN                         | N BACK CENTER | ຮ <sub>ບ</sub>   |            | MIDDLE      | MEAN                   | BACK CENTER                            | ទប                 |           |          |
|                  | LOWER                               | ····          | SL               | (U+L)      | LOWER       |                        |                                        | Տ <sub>Լ</sub>     | (U+L)     |          |
|                  |                                     |               | <b>_</b>         | <u> </u>   | Σ           |                        |                                        |                    |           |          |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  | ┠┶┽┽┼╋┿╋┯┯┯┑                        |               | <b>_</b>         |            |             |                        |                                        |                    |           |          |
|                  | ┠╍┼╌┼╍┞╍┽╌╋╶┠╼┶╴┙╺┛                 |               | <b>_</b>         |            | ┝╍┽╍┽╼┽╴┥   |                        |                                        |                    |           |          |
|                  | ┝┥┥┼┼┿┊┠╌╌╌┑                        |               |                  |            | ┠─┼╾┼─┼─┼─┼ |                        |                                        | <u> </u>           |           |          |
|                  | ┝┿╪╌┼╌┦╌╿╌╿                         |               | <b></b>          | ┟─┴─┴─┶╼┫  |             |                        |                                        |                    |           |          |
|                  | <mark>┟╶┼┈┼╶┽╶┟╴╅╴┨╼┱╌┲╌</mark> ┎╶┓ |               | <b> </b>         |            |             |                        |                                        |                    |           |          |
|                  | <del>╏╶┨╶┨╶┨╶┨╶┨╶┙</del>            | └╴╸┙┨╶┵┙└╺┵╺╸ |                  | ┠┯┯┯┫      |             | ┟──└──└──└             |                                        | - <del>   </del> - |           |          |
|                  | ┣━┼┼┼┼┾┋╊╼┰┰┯┓                      |               | <u>}</u> −⊥⊸•́−− | ┝╼╁╶╅╍╂╍╀╼ |             |                        |                                        | <b></b>            | ┝╌┼╌┼╌╀╌  | 1        |
|                  |                                     |               | 1                |            |             |                        |                                        |                    |           |          |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  |                                     |               | 1 I              |            |             |                        |                                        | Ţ                  |           |          |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           | 1        |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  |                                     |               | ļ                |            |             |                        |                                        |                    |           | ] ]      |
|                  |                                     |               | ┨╌╍┰╺╍┲╍╍        |            |             |                        |                                        |                    |           |          |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  | ┝╌┽╌┽╌┽╌╋╶┲╌┯╌┯╍                    | ·             | ┝┛┻              | ┝╌┼╼┼╌┽╌┨  |             |                        |                                        |                    | ┠╌┼╌┼╾┿╼╸ |          |
|                  | ╾┼╌┼╌┼╶┼╴┽╴┥╴┙╸┙                    |               | <b>_</b>         |            |             |                        |                                        |                    |           |          |
| $\left  \right $ | ┠╶╁╶┼╶╁╶┽╶┫╶┬╶┰╶┑                   | ┍╌┲╋╼┎╌┲╼┲╌┲╸ | +                |            |             |                        |                                        |                    |           |          |
| <b>-</b>         | ┢┈┟╌╋╍╎╌┼╌┽╌╋╼┸╶┵╌┚╴┚               |               | ┟┤┆              |            | ┣╌┼┼┼┼┿╴    |                        |                                        | ┝┼╺╸┥              |           |          |
|                  |                                     |               | L                | ┝╌┼╍┼╍┾╌┤  |             |                        |                                        | <u> </u>           |           |          |
|                  |                                     |               | 1                |            |             |                        |                                        |                    |           | 1        |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  | ┝┉╀╴┞┉╬╌╂╶┽╼┇┉╂╶┸┉┻═┸┈┙             | ······        | 1 1 I            |            |             | ┠━━┅┸╍═┸╼╍╌┸╼╍╩╍╌┍┠╼╍┥ | ······································ |                    | TTT       | 1 1      |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  |                                     |               | T                |            |             | <u> </u>               |                                        |                    | <u> </u>  | 1        |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           |          |
|                  |                                     |               |                  |            |             |                        | ······································ |                    |           |          |
|                  |                                     |               |                  |            |             |                        |                                        |                    |           | 1        |
|                  | EOR1 74-140 44. 771                 |               |                  |            |             |                        |                                        |                    |           |          |

~~

NOAA FORM 76-189

C-8

<sup>☆</sup> U.S. GOV, PRINTING OFFICE: 1977---768-934

| NOAA<br>(8-77 | FORM 76-         | 191         |                |              |                        |                          |               | NATIONAL          | U.S.        | DE PARTME                              | NT OF COMMERCE PAGE            |
|---------------|------------------|-------------|----------------|--------------|------------------------|--------------------------|---------------|-------------------|-------------|----------------------------------------|--------------------------------|
|               |                  |             |                |              | GEODET<br>MICROMETER C | IC LEVELIN<br>DESERVATIO | NG<br>NS (∆h) |                   |             |                                        | OF                             |
| ••            | 0 • YR           | /MO. DAY C  | ODE INSTR      | UMENT SERIAL | NO M CODE ROD          | SERIAL NO.               | CODE ROL      | D SERIAL NO.      |             | ······································ | ZTIME                          |
| ĿĿ            | <u>,   ·    </u> |             |                |              |                        |                          |               |                   |             |                                        |                                |
| FR            | OM BM            | DESIGNATION |                |              | TO BM DESIGN           | ATION                    |               | ZBEG              |             | C BEGIN                                | PERATURE OB-<br>END W S SERVER |
| SET<br>UP#    | STADIA<br>BACK   | Se<br>L     | STADIA<br>FORE | sf<br>D      | LOW SCALE<br>BACK/FORE | Δhr.<br>Σ                | HIGH          | I SCALE<br>K/FORE | Δh H≠d<br>Σ | ∆н∟-<br>∆ћн                            | REMARKS                        |
|               |                  |             |                |              | B                      |                          | B<br>F        |                   |             |                                        |                                |
|               |                  |             |                |              | B                      |                          | B<br>F        |                   |             |                                        |                                |
|               |                  |             |                |              | B                      | -                        | B             |                   |             |                                        |                                |
|               |                  |             | 5              |              | B F F                  |                          | B<br>F        | 3                 |             |                                        |                                |
|               |                  |             |                |              | B F                    |                          | B<br>F        |                   |             | s.<br>X                                |                                |
|               |                  |             |                |              | B F                    |                          | B<br>F        |                   |             |                                        |                                |
|               |                  |             |                |              | B F                    |                          | B<br>F        |                   |             |                                        |                                |
|               |                  |             |                |              | B                      |                          | B F           | 5                 |             |                                        |                                |
|               |                  |             |                |              | B<br>F                 |                          | B<br>F        |                   | 2           |                                        |                                |
|               |                  |             |                | -            | B .                    |                          | B F           |                   |             |                                        |                                |

| SET<br>UP# | STADIA<br>BACK | SB<br>Z | STADIA<br>FORE | sr<br>S | LOW SCALE<br>BACK/FORE                  | ΔhL<br>Σ | HIGH SCALE<br>BACK/FORE | ΔhH±d<br>Σ | ∆н<br>Д <sup>ћн</sup> | REMARKS |
|------------|----------------|---------|----------------|---------|-----------------------------------------|----------|-------------------------|------------|-----------------------|---------|
|            |                |         |                |         | B F                                     |          | B F                     |            |                       |         |
|            |                |         |                |         | B                                       |          | B                       |            |                       |         |
|            |                |         |                |         | F                                       |          | F                       |            | !                     |         |
|            |                |         |                |         | B · · · · · · · · · · · · · · · · · · · |          | B F                     |            |                       |         |
|            |                |         |                | ÷       | B F                                     |          | B                       |            |                       |         |
|            |                |         |                |         | B F                                     |          | B .                     |            |                       |         |
|            |                |         |                |         | B F I                                   |          | B · ·                   |            |                       |         |
|            |                |         |                |         | B F F                                   |          | 8                       |            |                       |         |
|            |                |         |                |         | B F                                     |          | B F                     |            |                       |         |
|            |                |         |                |         | B                                       |          | B F                     |            | 2<br>2<br>7           |         |
|            |                |         |                |         | B                                       |          | 8                       |            | ,<br>,<br>,           |         |
|            |                |         |                |         | B F                                     |          | B F                     |            |                       |         |
|            |                |         |                |         | B F                                     |          | B F                     |            |                       |         |

..----

C-10

| NOA/<br>(3 8 | DAA FORM 76-198<br>3-823<br>MONTHLY REPORT - BENCH MARK AND RECONNAISSANCE TEAMS<br>DROJECT DIRECTOR |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
|--------------|------------------------------------------------------------------------------------------------------|------------------------|------------------|-----------------------------|----------|--------------------|----------|-------|------------------|---------------|------------------------------|--------------------------------|-------------------|--------------------|----------------|--------------|------------|----------------------------------------------|
|              |                                                                                                      |                        |                  | М                           | ONT      | HLY                | REP      | ORT   | <u>– B</u>       | ENCH          | MA                           | RK AI                          | ND R              | ECO                | NNAI           | SSAN         | CE TE      | AMS                                          |
| TËA          | MLEA                                                                                                 | DER                    |                  |                             |          |                    |          |       |                  |               |                              | PROJ                           | ECT D             | DIRECT             | OR             |              |            |                                              |
| TEAL         | M ASSI                                                                                               | STANT                  |                  |                             |          |                    |          |       |                  |               |                              | PART                           | Y                 |                    | ·····          |              |            | MONTH/YEAR                                   |
|              |                                                                                                      |                        |                  | RKS SI                      | EARC     | HEDF               | OR       |       |                  |               |                              |                                |                   |                    | TASI           |              | ER:        | <u>                                     </u> |
|              |                                                                                                      |                        | CONT             | ROL                         | н        |                    |          |       | 201              | MA            | RKS                          | ËSTAB                          | ∃LISH             | ED                 | STA            | г <b>Е</b> : |            |                                              |
|              |                                                                                                      |                        |                  |                             |          |                    |          |       | Γ                |               |                              |                                | [                 |                    |                | [            | Ţ          | · · · · · · · · · · · · · · · · · · ·        |
| Date         | Recovered<br>Sood Conc                                                                               | Recovered<br>Door Cond | 40T<br>Recovered | Destroyed<br>or<br>Unusable | ștațions | Reference<br>Aarks | 4Z Morks | Other | 40T<br>Recovered | Rock<br>Marks | Quality A<br>Rods<br>Sleeved | Quality A<br>Rods<br>Insleeved | Quality B<br>Rods | Structure<br>Marks | Line<br>Number | (i lometer:  | Staff-Days | Remarks                                      |
| 1            |                                                                                                      | <u> </u>               |                  |                             |          |                    | <u> </u> |       |                  | <u> </u>      |                              | 04                             |                   |                    |                |              |            |                                              |
| 2            |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              | <u> </u>   |                                              |
| 3            | <u> </u>                                                                                             |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 4            |                                                                                                      |                        |                  |                             |          |                    |          |       | <b> </b>         |               |                              |                                |                   |                    |                |              | <b> </b>   |                                              |
| 5            |                                                                                                      |                        |                  |                             |          |                    |          |       |                  | <b>-</b>      | <u> </u>                     |                                |                   |                    | <u> </u>       |              | <u> </u>   |                                              |
| 6            |                                                                                                      |                        |                  | 1                           |          |                    |          |       | <b> </b>         |               | 1                            |                                |                   | <b> </b>           |                |              |            |                                              |
| 7            |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              | t          |                                              |
| 8            |                                                                                                      |                        |                  |                             |          |                    |          |       |                  | <u> </u>      |                              |                                |                   |                    | İ              |              |            |                                              |
| 9            |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              | [          |                                              |
| 10           |                                                                                                      |                        |                  | 1                           |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 11           |                                                                                                      |                        |                  | 1                           |          | 1                  |          |       |                  |               |                              |                                |                   |                    | 1              |              |            |                                              |
| 12           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 13           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 14           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   | [                  | ļ              |              |            |                                              |
| 15           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              | <u> </u>                       |                   |                    |                |              |            |                                              |
| 16           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 17           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 18           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 19           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 20           |                                                                                                      |                        |                  |                             |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |
| 21           |                                                                                                      |                        |                  |                             | ļ        |                    |          |       | ļ                |               |                              | L                              |                   | L                  |                |              |            |                                              |
| 22           | ļ                                                                                                    |                        |                  | L                           |          |                    |          |       | ļ                |               |                              | ļ                              |                   | ļ                  |                | ļ            |            |                                              |
| 23           | <u> </u>                                                                                             |                        |                  |                             |          | <u> </u>           | <u> </u> | ļ     | ļ                | ļ             |                              | Ļ                              | [                 | ļ                  | L              |              | ļ          |                                              |
| 24           |                                                                                                      | <u> </u>               | <u> </u>         |                             | <b> </b> |                    |          |       |                  | L             |                              |                                | <br>              |                    | <b></b>        |              | L          |                                              |
| 25           |                                                                                                      |                        | ļ                | L                           | L        | <u> </u>           |          | L     | ļ                | <u> </u>      | L                            | L                              |                   | L                  | L              |              |            |                                              |
| 26           |                                                                                                      |                        |                  | ļ                           |          | <br>               |          |       |                  | L             | ļ                            | <u> </u>                       |                   | L                  |                |              | ļ          |                                              |
| 27           |                                                                                                      |                        | <u> </u>         | <u> </u>                    |          | <u> </u>           | <u> </u> | ļ     | ļ                | ļ             |                              | ļ                              |                   |                    | <b> </b>       |              |            |                                              |
| 28           |                                                                                                      | <b> </b>               | <u> </u>         | <u> </u>                    | ļ        |                    |          | ļ     | ļ                | L             | ļ                            |                                | ļ                 |                    | ļ              |              | L          |                                              |
| 29           |                                                                                                      | L                      | ļ                | ļ                           | L        | <u> </u>           | L        | ļ     | ļ                | <b> </b>      |                              | L                              |                   | ļ                  | ļ              |              | L          |                                              |
| 30           |                                                                                                      | L                      | ļ                |                             | ļ        |                    |          | L     | <b></b>          |               |                              |                                | [                 | [                  | [              | ļ            | [          |                                              |
| 31           |                                                                                                      | <u> </u>               |                  |                             |          |                    |          |       |                  |               |                              |                                |                   | L                  |                |              |            |                                              |
| Total        |                                                                                                      |                        |                  | [                           |          |                    |          |       |                  |               |                              |                                |                   |                    |                |              |            |                                              |

NOAA FORM 76-198 3-82

|                            | ######################################                                                                         | INVENTORY OF                           | MARK SUPPLIES                                  |                                        | L ANNAL & A COMMUNICATION AND AND AND AND AND AND AND AND AND AN                                                |
|----------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------|------------------------------------------------|----------------------------------------|-----------------------------------------------------------------------------------------------------------------|
| PARTY                      |                                                                                                                | TRUCK NUMBER                           | MINING AND AND AND AND AND AND AND AND AND AND | MONTH                                  | /year<br>19                                                                                                     |
| INVENTORIED BY             |                                                                                                                | 1                                      | CONDITION OF TRUCK                             | I                                      | 4 mm / 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - 1947 - |
|                            | BEGINNING<br>INVENTORY                                                                                         | SUPPLIES<br>ADDED                      | ENDING<br>INVENTORY                            | SUPPLIES                               | SUPPLIES<br>NEEDED                                                                                              |
| Rods                       |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Studs                      |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Rod Entry<br>pts. (fluted) | and a second second second second second second second second second second second second second second second |                                        |                                                |                                        |                                                                                                                 |
| Rod Entry<br>pts. (smooth) |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Datum pts.                 |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Logo Caps                  |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| 5'' PVC                    |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| 1'' PVC                    |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Disk (stem)                |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Disk (rod)                 |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Witness Post               |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Witness Sign               |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Grease                     |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| ''O'' Ring                 |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| PVC Reducer                |                                                                                                                | ······································ |                                                | ······································ |                                                                                                                 |
| PVC Cement                 |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| PVC Schedule               |                                                                                                                |                                        |                                                |                                        | ***************************************                                                                         |
| Pop Rivets                 |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| BM Description             |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Flagging                   |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
| Crayon                     |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
|                            |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
|                            |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
|                            |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
|                            |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
|                            |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |
|                            |                                                                                                                |                                        |                                                |                                        |                                                                                                                 |

NOAA FORM 76-198 3-821

☆U\_S\_GOVERNMENT PRINTING OFFICE\_1982-564-008/1146

í

Ĺ

É

| NOAA<br>(3-82 | A FORM             | 76-104             |                   |                   |                   |          |          |                  |                   |                    |              |            | NATIONAL OCEANIC | U.S. DEPARTMENT OF COMMERCE<br>AND ATMOSPHERIC ADMINISTRATION |
|---------------|--------------------|--------------------|-------------------|-------------------|-------------------|----------|----------|------------------|-------------------|--------------------|--------------|------------|------------------|---------------------------------------------------------------|
| OBSE          | RVER               |                    |                   |                   |                   |          | MC       | DNTH             | LYR               | RECORE             |              | VELIN      | <u>G UNIT</u>    | MONTH/YEAR                                                    |
| RODA          | AEN. PA            | CER                |                   |                   |                   |          |          |                  |                   | [                  | PARTY        |            |                  | 19                                                            |
|               |                    | 1                  |                   |                   | ******            |          | ······   |                  |                   |                    | PROJE        | <u>.</u> T |                  |                                                               |
|               |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            | ,                | <u> </u>                                                      |
| LINE          | (5)                |                    |                   |                   |                   |          |          |                  |                   |                    | SURVE        | Y ORDE     | R                | CLASS                                                         |
|               |                    | DISTAN             | ICE (             | km}               |                   | ним      | BER      | τII              | ME                | COLL               |              | [ ]        |                  |                                                               |
| DAY           | DBL<br>RUN<br>PROG | SGL<br>RUN<br>PROG | IST<br>RE-<br>RUN | ADL<br>RE-<br>RUN | TOT<br>ALL<br>RUN | SET      | SEC-     | IST<br>SET<br>UP | LAST<br>SET<br>UP | AND<br>Comp<br>Cks | STAFF<br>DAY | DOWN       | WEATHER          | REMARKS                                                       |
| 1             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 2             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 3             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 4             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 5             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 6             |                    |                    |                   |                   | [                 |          |          |                  |                   |                    |              |            |                  |                                                               |
| 7             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            | [                |                                                               |
| 8             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 9             |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 10            |                    |                    |                   |                   |                   |          |          |                  | -                 |                    |              |            |                  |                                                               |
| 11            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 12            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 13            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 14            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 15            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 16            |                    |                    |                   |                   | <u> </u>          |          |          |                  |                   |                    |              |            |                  |                                                               |
| 17            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 18            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 19            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 20            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 21            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 22            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 23            | L                  |                    |                   | <u> </u>          | L                 | <u> </u> | <u> </u> |                  | ļ                 | <u> </u>           |              |            | ·                | ·····                                                         |
| 24            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 25            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 26            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 27            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 28            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 29            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 30            |                    |                    |                   |                   |                   |          |          |                  |                   |                    |              |            |                  |                                                               |
| 31            | Γ                  |                    | Ι                 |                   | Τ                 |          |          |                  |                   |                    |              |            |                  |                                                               |
| Total         | Γ                  |                    |                   | I                 |                   |          | 1        | ļ                |                   |                    |              | km/Sta     | ff-Doys =        |                                                               |

NOAA FORM 76-104 3-621

CUS GOVERNMENT PRINTING OFFICE 1982-564-008-1148

| GEODETIC | LEVELING |
|----------|----------|
|----------|----------|

ί

(

(

| LINE    |              |    |     |      |       |          |         |     |            |     | "1        | AP  | EN   | ο.      |      |         |     |     |    |     |      |     |
|---------|--------------|----|-----|------|-------|----------|---------|-----|------------|-----|-----------|-----|------|---------|------|---------|-----|-----|----|-----|------|-----|
| PROJ.   | NO.          |    |     |      |       |          |         |     |            |     | F         | PAG | E    |         |      |         | c   | F   |    |     |      |     |
| ST 0    |              |    | ST  | 1    |       |          |         |     | ѕт         | 2   | ST        | 3   |      |         |      |         |     |     |    |     |      |     |
| DATA    |              |    | Y   | R.   | м     | ο.       | D       | Y., | zo         | NE  | INS       | аτ. | NO.  |         |      |         |     |     |    | col | DE   |     |
| # 4     | 0            | ¥  |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     | ٠  |     |      |     |
| ST 4    |              |    |     |      |       |          |         |     |            |     | ST        | 5   |      |         |      |         |     |     |    |     |      |     |
| ROD NO  | <b>5.</b> 1  |    |     |      |       |          | (       | 00  | E          |     | RO        | DN  | 0. 2 |         |      |         |     |     |    | co  | ΣE   |     |
|         |              |    |     |      |       |          | •       |     |            |     |           |     |      |         |      |         |     |     | ٠  |     |      |     |
| ST'6    |              |    |     |      | ST    | 7        |         |     | SТ         | 8   |           |     |      |         | ST   | 9       |     |     |    |     |      | с   |
| OBSER   | VER          |    |     |      | TIN   | 1E       |         |     | LE         | VEL | . co      | NST | _    |         | c /  | F       | UN  | TS  | ST | AD  |      | F   |
|         |              |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      | ₿   |
| 0 - WRI | TEC          | -N | TAF | ۶Ę - | - 9   |          |         |     |            |     |           | BL  | oci  | <       |      |         |     | 1   |    |     |      | R   |
| ST O    |              |    | ST  | 1    |       |          | DE      | SIG | TAR        | ION |           |     |      |         |      |         |     |     |    |     |      |     |
| DATA    |              |    | FR  | OM:  |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
| # 4     | 1            | *  |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    | SТ  | 2    |       |          | DE      | SIG | T A I      | ION |           |     |      |         |      |         |     |     |    |     |      |     |
|         | то:          |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    |     |      |       |          | l       |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    | ST  | 3    |       |          | SΤ      | 4   |            | ST  | 5         | ST  | 6    | ST      | ST 7 |         |     |     |    |     | .ocr | < ) |
|         |              |    | TIN | Æ    |       | ĺ        | тε      | MP  |            | wi  | ND        | รบ  | N    | DA      | ΤE   |         |     |     |    |     |      | -   |
|         |              |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
| 0 - WR  | ITE          | ON | ΤА  | PΕ   | - 7   |          | ••••••• | ÷   | - C        | LR  | q         | =   |      | •       |      |         |     |     |    |     |      |     |
| ST: ''o | 1'' IN       |    | HG  | SIGN | 4     |          |         |     |            | (N  | OTE       | SIG | N O  | F '     | q'') |         | L   |     |    | J.S | 5.S  |     |
| ELEV.   | DIFF         | •  |     |      |       | RC       | L 7     |     |            |     | ĸм        | . F | CL   | •       | NO   | . R     | cı. | 0   | 1  |     |      |     |
| ΠΠ      | Т            |    |     |      | •     |          |         |     |            |     | ÷         |     |      |         |      |         |     |     |    |     |      |     |
| ST 0    |              | 1  | ST  | 1    | ····· | <b>A</b> | SТ      | 2   |            | ST. | 3         | ST  | 4    |         |      | <b></b> |     |     |    | BL  | оск  | :   |
| DATA    |              |    | TIN | AE   |       |          | τε      | MP  |            | wii | D         | รบ  | N    |         | BS.  |         |     | REC |    |     |      | -   |
| # 4     | 1            | *  |     |      | F     |          |         |     |            |     |           |     |      |         | Γ    | ľ       |     |     |    |     |      |     |
| 0 - WR  | ITE C        | οN | TAF | ÞΕ.  | - 4   |          |         | 1   | <b>a</b> m |     | <b></b> , |     |      | <b></b> | 4    |         |     |     |    |     |      |     |
|         |              |    |     | RE   | MAF   | RKS      |         |     |            | Γ   |           |     |      |         |      | RE      | MAR | ĸs  |    |     |      |     |
|         | Rod 1 (init: |    |     |      |       |          | .ti     | als | s)         |     | -         |     | Roo  | 1 2     | (    | in      | Lti | als | s) |     |      |     |
|         |              |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    |     |      |       |          |         |     |            | l   |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |
|         |              |    |     |      |       |          |         |     |            |     |           |     | ,    |         |      |         |     |     |    |     |      |     |
|         |              |    |     |      |       |          |         |     |            |     |           |     |      |         |      |         |     |     |    |     |      |     |

NOAA FORM 77-82 (11-79) U.S. DEPARTMENT OF COMMERCE - NOAA SUPERSEDES NOAA FORM 77-82 (8-79) WHICH SHOULD BE DESTROYED.

.

NOAA FORM 77-81 U. S. DEPARTMENT OF COMMERCE LEVEL ROD NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION (2 - 78)GEODETIC LEVELING COMPENSATION CHECK b INSTRUMENT: Zeiss Ni 1/ С a. WIND SUN TIME DATE SERIAL NUMBER A % If tan a or tan  $\beta < 0.0000097$  the compensator is in good adjustment. BUBBLE POSITION POSITION OF BUBBLE HALF STADIA POSITION OF HALF ROD LEFT ROD LEFT ROD RIGHT INTERCEPT ROD RIGHT STADIA BUBBLE HALF cm HALF cm HALF cm HALF cm (HSI) 0 C α c a - c Mean (a-c)<sub>cm</sub> =  $\frac{(a-c)_{left} + (a-c)_{right}}{4}$ REMARKS (OBSERVER/RECORDER) S<sub>cm</sub> = HSI x Stadia Constant =  $\tan \alpha = \frac{a - c}{S_{cm}} = \frac{a - c}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} = \frac{cm}{cm} =$ b ٥ Ь d b . d REMARKS (OBSERVER/RECORDER) Mean (b-d)<sub>cm</sub> =  $\frac{(b-d)_{left}+(b-d)_{right}}{4}$ =  $\tan \beta = \frac{b-d_{cm}}{S_{cm}} =$ 

# RECORD OF BENCH MARK DESIGNATIONS ASSIGNED BY MARK SETTER



| FILE    |           |           |        |  |  |  |  |  |  |
|---------|-----------|-----------|--------|--|--|--|--|--|--|
| PROOFR  | EAD:      | FIRST E   | DIT:   |  |  |  |  |  |  |
|         |           |           |        |  |  |  |  |  |  |
| FORM    | IAT-CHECH | (ING PROG | GRAM:  |  |  |  |  |  |  |
| JOB#    | DA        | TE        | RERUN? |  |  |  |  |  |  |
|         |           |           |        |  |  |  |  |  |  |
|         |           |           |        |  |  |  |  |  |  |
|         |           |           |        |  |  |  |  |  |  |
|         |           |           |        |  |  |  |  |  |  |
|         |           |           |        |  |  |  |  |  |  |
| CORRECT | ED ON OR  | IGINALS.  |        |  |  |  |  |  |  |

C-16



#### RIVER CROSSING Simultaneous Reciprocal Observations

Appendix C-Standard Leveling Forms of the National Geodetic Survey



RIVER CROSSING Zeiss River Crossing Equipment

~~~.

C-18

| RIVER CROSSING SERIES, SIDE                                 |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   | E       |  |
|---|--|--------|------------------|----------|-----------------|-------------|-------------------|---------------------------------------|------------|-----------------|-------------|-------------------|---|---------|--|
| OF  |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
|   |  |        | YR. MO. DAY CODI |          | SERIAL NO       | ODE INSTRUM | ENT SERIAL NO CO  | DE ROD SERIAL                         | . NO. T    | ARGET POST      | TIONS, THIS | SIDE              |   | RUN NO. |  |
| Ŀ   | 4 2  | ŀ      |                  |          |                 |             |                   |                                       | ⊤          | 09              | 8071        | om                |   |         |  |
| FROM DESIGNATION TO DESIGNATION Z BEGIN END C BEGIN END W 5 |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
| Ш   |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
|   | INSTRUMENT No. 1 INSTRUMENT No. 2 COMPUTATIONS |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
| GRADUATIONS: - *  |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
| SX5   | •8   | R<br>O | LOWER            |          |                 | LOWER       |                   |                                       |            |                 |             | ······            | + |         |  |
|   |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
| TARGETS OPPOSITE:   |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
|   |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
| ĺ   | Ī  | GE     |                  |          | 10.00           |             |                   |                                       |            |                 |             |                   |   |         |  |
|   | 1  | T      |                  |          |                 |             | 1 1<br>1 1<br>1 1 |                                       |            |                 |             |                   |   |         |  |
|   |  | -+     | RECIPROCA        | _ COLLI  | MATION          |             |                   |                                       |            |                 |             |                   |   |         |  |
|   | S  | T      |                  |          |                 |             | ++                | +                                     |            |                 |             |                   |   |         |  |
|   | L<br>T   | n<br>G |                  |          | 10.00           |             |                   | <br>                                  |            |                 |             |                   |   |         |  |
| Ę   | 2  | 6<br>T |                  |          |                 |             | ++                | +                                     |            |                 |             |                   |   |         |  |
| ESIG  |  | +      | RECIPROCA        | LCOLL    |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
| Pop   | S  | T.     |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |
|   | Ē  | A<br>ŋ |                  |          | 10.00           |             |                   | +                                     |            |                 |             | <i>UIIIII</i> III | + |         |  |
|   | <br>  7  | E      |                  |          |                 |             | +                 | <b>+</b>                              | -[         | 2000/00/00      | 7777777777  |                   |   |         |  |
|   | Ľ  | 1      |                  |          |                 |             |                   | UPPER                                 |            |                 |             |                   |   |         |  |
| ŀ   |  | +      |                  |          | LOWE            |             |                   | <u>. 0 w C u</u>                      |            |                 |             |                   |   |         |  |
| ļ   | E<br>E   | T<br>A |                  |          |                 |             |                   | ╆╺╺╸┙╺┨                               |            |                 |             |                   |   |         |  |
|   | Т  | G      | <u> </u>         | ·····    |                 |             |                   |                                       | l          |                 |             |                   |   |         |  |
|   | 4  | Ť      |                  |          | <u>11 4 4 U</u> | ·           |                   |                                       |            |                 |             |                   |   |         |  |
| 6.47  |  | 1      | RECIPROCA        | <u> </u> | MATION          | <u> </u>    |                   | · · · · · · · · · · · · · · · · · · · |            | <i>!!!!!!!!</i> | 91111111    |                   |   |         |  |
| KKS!  | 8  | n<br>0 | LOWER            |          |                 | LOWER       |                   |                                       | <u> </u>   |                 |             |                   |   |         |  |
|   |  |        |                  | i        | il              | Urren       | KI                | .ii                                   | <u> </u> ] |                 |             | <u> </u>          |   |         |  |
| 1 "   |  |        |                  |          |                 |             |                   |                                       |            |                 |             |                   |   |         |  |

(

-

ĺ
## INDEX

abstract, field: adjustments, 5-26 computing, 5-17 description, 3-69, 3-70, 3-71, 3-72, 515 double-run leveling, 3-71 example, computer produced, 5-18 office, 5-25 proofing, 5-15, 5-17 accuracy, criteria for: surveys and maps, 1-5, 2-21, 3-11 Federal Geodetic Control Committee, 2-25, 2-34, 2-45, 3-11 accessioning, 5-23 adjustment: data, 5-23, 5-26 geodetic level: circular level, 3-26, 3-46 collimation, 3-25 computing corrections, 3-29, 3-31 main bubble, 3-25 parallax, 3-26 quality control, 5-23 wedge reticle, 3-27 rod levels, 3-42 airports, leveling to, 2-22 archival cross-reference number, 2-38 area leveling, 2-22, 2-23 atmosphere, effect on leveling, 3-3 air temperature differences, measurement, 3-49, 3-50 B+F: also, see divergence of runnings, allowable accumulation, 3-6, 3-7, 3-65 control, 3-52 computation, 3-65, 3-66 backsights and foresights: length, 3-7 difference, 3-7 equalization, 3-7 maximum, 3-7 battery care, 3-68 bench mark form: description, 2-34, 2-35, 2-36, 2-38, 2-46 illustrations, 2-35, 2-36, 2-37, 2-40, 2-41, 2-46, appendix C bench marks: airport, 2-21, 2-22 concrete post: construction, 2-41 specifications, 2-26 designations, 2-26, 2-39 destroyed, 2-25 locations, 2-20, 2-26, 2-27, 2-45 National Ocean Survey, 2-43

other organizations: connection, 2-21 description, 2-39, 2-40 designation, 2-39 permanent: description, 2-26, 2-27, 2-28, 2-34, 2-35, 2-36, 5-10, 5-11 checking, 5-7 data files. 5-9 data processing, 5-7, 5-8 designation, 2-26 establishment, 2-26, 2-27, 2-28 horizontal control and three dimensional stations, 2-21, 2-22, 2-40 junctions, 2-38 spacing, 2-20 stamping, 2-39, 2-43 designations, 2-39 recovery, 2-25 recovery notes, 2-36, 2-37, 2-38, 2-47 quality classification, 2-42 tidal: connections, 2-21 designation, 2-16, 2-38, 2-39 traverse stations, 2-21, 2-22 bench mark setting unit: assistant mark setter, 2-1 equipment, 2-2 foreman, 2-1 mark setter, 2-1 organization, 2-1 blunders: also, see errors guarding against, 3-5, 3-6 mistakes, 3-5, 3-6 brace poles, 3-42 bridges, 2-27, 2-41 "C": computation, 5-24 determination, 3-29, 3-30, 3-31, 3-32, 3-33 record, 3-32 calibration with laser interferometer, 3-43 check: see B+F and divergence of runnings, allowable check abstract:3-69, 3-70 computerized, 5-15, 5-16, 5-17, 5-18, 5-19 handwritten, 3-69, 3-70 check loops, 5-22 check points, 3-60 circular level, adjustment: 3-26 closing the section, 3-65 rejection procedure, 3-66 collimation, 3-3, 3-4, 3-29, 3-30, 3-31 error factor, 3-22, 5-24

Kukkamaki's method, 3-32, 3-33, 3-34, 3-35 10-40 method, 3-33, 3-36, 3-37 compensation check, 3-34, 3-39, 3-40 computation: B+F, 3-65, 3-66 "C", 3-29, 3-30, 3-31, 3-32, 3-32 stadia constant, 3-27, 3-28, 3-29 river-crossing observations, 4-16, 4-17 computation check, 3-34, 3-39, 3-40 computer: equipment, 3-68 connections, 2-21, 2-22, 2-23 control points, 2-8, 2-20, 2-40 descriptions, 2-8, 2-40, 2-41 destroyed, 2-25, 2-45 geographic positions, 2-25, 2-40 new, 2-45 recovery, 2-25 setting new monument, 2-44, 2-45 corrections: curvature, 5-26 refraction, 5-26 data processing: adjustment, 5-23 equipment, 5-1 project report, 5-4 proofreading, 5-7, 5-8, 5-9, 5-10, 5-13, 5-15 records, 5-2 submission, 5-2, 5-3, 5-23 sets, 5-13 verification, 5-23, 5-24 datums, 1-4 descriptions, 2-34, 2-38, 2-39, 2-40, 2-41, 2-42, 2-43, 2-44 other organizations, 2-39, 2-40 processing, 2-23, 5-7, 5-8, 5-9, 5-10, 5-11, 5-12, 5-13 divergence of runnings, allowable: also, see B+F, 3-6, 3-7 not to be exceeded, 3-7 tables, 3-7 editing records: 5-9 elevations: computing, 5-20, 5-21 differences, 5-20 equipment: bench mark setting unit, 2-1, appendix A computer-recording, 3-68 leveling, 3-8, appendix A supplies, appendix A truck, appendix A errors: also, see blunders collimation, 3-2, 3-3, 5-24 curvature, 3-4 leveling, 3-1, 3-2, 3-3, 3-4, 3-5, 3-6 refraction, 3-3 rod plumb, 3-4

scale imperfections, 5-24 shimmer, 3-4 steep grades, 2-20 tidal deflection, 3-4 field records, 3-66 flagman, 3-8 foresights and backsights: see backsights and foresights forms, blank, appendix C mark setting, appendix A, appendix C leveling, appendix A, appendix C formula: "C", 3-29, 3-31 stadia constant, 3-27, 3-28 General Adjustment of 1929, 1-4 Geodetic Control Diagram, 2-7 codes, appendix B geopotential number, 1-4 graduations: rods: feet, 3-41 meters, 3-41 scarred, 3-44 gravity field, 2-22 hand level, 3-48 height systems, 1-3, 1-4 highways: leveling along, 2-2, 2-9 history of geodetic leveling, 1-1, 1-2, 1-3, 1-4 Index, area lines, 2-7 Index to Topographic Maps, 2-8 instructions: general, 3-51 project, 2-4 recovery and stamping, 2-25 river crossings, 4-1 instruments: carrying the level, 3-25, 3-44 classification, 3-11 compensator type: general description, 3-15, 3-18 Jenoptik NI 002, 3-15, 3-20, 3-21 MOM NiA31, 3-14, 3-15 Zeiss Ni1, 3-14, 3-15, 3-19, 3-20, 3-39 Zeiss Ni2, 3-14, 3-15, 3-17, 3-18, 3-19, 4-5 Wild NA2, 3-15 maintenance, 3-27 micrometer, 3-22, 3-23 spirit level: Breithaupt NABON, 3-15 Fischer, 3-13, 3-15, 4-20 Jenoptik Ni-004, 3-13, 3-15 Wild N3, 3-15 telescopic system, 3-13, 3-16 Zeiss River Crossing Equipment, 4-5, 4-6, 4-9, 4-10, 4-11, 4-12, 4-28

ĺ

junction point positions, 2-19 junctions: beginning of lines, 2-19 code, 2-38 ending of lines, 2-19, 2-20 Kukkamaki's collimation method, 3-31 length of sections, 2-20 length of sights: see backsights and forsights, length of level: also, see instruments bubble, 3-25 geodetic: adjustment, 1-4 description, 1-4 shading, 3-2 shipment, NI 002, 3-27 leveling: along highways, 2-2, 2-9 along railroads, 2-2 area, 2-21 records, 2-23 classification, 2-26, 2-27, 2-42 micrometer, 3-53, 3-54, 3-55, 3-56, 3-57 procedure, 3-1, 3-2 party: organization, 3-6, 3-7 personnel, 3-6, 3-7, 3-8 rod: see rod safety, 3-9 three-wire leveling, 3-56, 3-57, 3-58, 3-59 tolerances, 3-7 logs: see maps for organizing surveys loops, 5-22 magnetic disks, 5-2 magnetic tape, 5-2 maps: for organizing surveys, 2-4, 2-23 for position plots, 2-9, 2-29 mark setting unit, 2-1 equipment, 2-2 personnel, 2-1 reports, 2-2, 2-3 safety, 2-2 micrometer, 3-22, 3-23, 3-53 foot screws, 3-24 reversing point, 3-53, 3-54, 3-55 tribrach, 3-24 use. 3-23 river crossings, 4-20 mistakes: see errors motorized leveling, 3-8 National Geodetic Vertical Network, 1-4 network classifications, 1-5 notes: for use in studying errors, 3-59 in remarks column, 3-59

observation file, 5-3, 5-4 vertical observational data set, 5-3, 5-4 observations: creating file, 5-13, 5-14 recomputing computer recorded data, 5-13 recording, 3-66 rejection, 3-66, 3-67 Zeiss River Crossing Equipment, 4-10, 4-11, 4-12, 4-13, 4-14, 4-15, 4-16 observer: duties, 3-8 qualifications, 3-8 observing: routine, 3-51, 3-52 routine for three-wire leveling, 4-18 program: see program, observing unit: discipline, 3-6, 3-7 management, 3-6, 3-7 organization, 3-6, 3-7, 3-48 organization, bench mark setting unit, 2-1 observing unit, 3-6, 3-7, 3-48 orthometric height, 1-3 outfit: see equipment parallax, 3-3 party: see: bench mark setting unit, leveling, party, observing unit, river crossing, personnel: also, see party computer, 5-1 assistant computer, 5-1 flagman, 3-8 foreman, 2-1 mark setter, 2-1 assistant mark setter, 2-1 observer, 3-6, 3-8 pacer, 3-6 project director, 2-1, 5-1 assistant project director, 5-1 recorder, 3-6, 3-8, 3-9 rodman, 3-6, 3-8, 3-9 umbrella, person holding, 3-8 unit chief, 3-7 plate turning, 5-47 plots, smooth, 2-29, 2-33 plots, working, 2-24, 2-38 plug: see spacer, 3-60 points, control: awkward, 3-62 check, 2-25 description, 2-8 identifying, 3-60 leveling to, 3-61, 3-62, 3-63 precautions to leveling, 3-59

short scale, 3-63, 3-64 upside down scale, 3-65 program: observing, 3-51 river crossings, 4-10 proofreading: see data processing reconnaissance, 2-1, 2-4 recorder: duties, 3-8 qualifications, 3-8 training, 3-7 records, 2-23, 3-66 editing, 5-9 file status sheet, 5-11 recording observations, 3-66 transmittal, 2-23 submission, 3-72 synoptic file, 5-12 recovery and stamping of bench marks: bench marks, recovery, 2-25 control points, 2-25 code, 2-41 refraction: also, see curvature and refraction effect on leveling, 3-3 effect on river crossings, 4-2 elimination of effect, 3-3 rejections, 3-66, 3-67 releveling, 3-65, 3-67 relocating vertical control points, 2-44 reports: condition of bench mark, 2-42 leveling, weekly, 3-9, 3-10, 3-11 leveling, miscellaneous, 2-4, 3-11 leveling, monthly, 3-9, 3-10, 3-11, 3-12 leveling, vehicle, 3-11 mark setting, miscellaneous, 2-4 mark setting, monthly, 2-2, 2-3, 2-4 mark setting, truck, 2-4 river or valley crossing auxiliary scale, 4-7, 4-8 computation, 4-16, 4-17 environmental conditions, 4-5 equipment, 4-5 Fischer level, 4-20 instructions, 4-1, 4-2, 4-3, 4-7 instrument station, 4-7 leveling rod, 4-8 observations, 4-10, 4-11, 4-13 ordinary leveling instruments, 4-17, 4-18, 4-19, 4-21, 4-22 personnel, 4-4 reconnaissance and mark setting, 4-4 records of observations, 4-2 target stations, 4-8 Zeiss River Crossing Equipment, 4-5, 4-9, 4-10, 4-11, 4-12, 4-13, 4-14, 4-15, 4-16, 4-18 rod:

double run, 3-71 leveling, 3-4, 3-5 marks, 2-27 plumb, 3-45 pole, 3-43 scale, 3-41, 3-42 setup, 3-45 single run, 3-71 routes, 2-19 routing level line, 2-4 junctions, 2-19 data, 2-4 liaison, 2-1, 2-9 logs, 2-23 specifications, 2-19 runnings: also, see B+F safety, 2-2 scale auxiliary, 4-7, 4-8 scaling positions, 2-30, 2-31, 2-32 shimmer, 3-3 sights: see backsights and foresights <sup>•</sup> site selection, 2-26, 2-27 spurs, 2-19 spacer (plug), 3-60, 3-62 stability, 2-26, 2-27 stadia: constant, 3-27, 3-28 interval, 3-7 method, determination, 3-28 stamping of bench marks, 2-26 Sun: direction, 3-50 sunshade, 3-2 supplies: appendix A survey point serial number, 2-34, 5-12, 5-19 synoptic file comparison, 2-14, 2-15, 5-10, 5-12 tables: allowable divergence between runnings, 3-7 curvature and refraction, 3-31 target for river crossings, 4-7 column, 4-7 temperatures: reading, 3-49, 3-50 recording, 3-49 leveling in extreme conditions, 3-50 thermometer, on rod, 3-49 three-wire leveling, 3-56, 4-18 tide stations, 2-9, 2-16, 2-21, 2-38, 2-39 tidal accelerations, 3-4, 5-26 time zone, U.S. Navy, 3-68 tolerances, 3-7 tools: leveling, 3-8 mark setting, 2-2 traffic, 3-10 tribrach, 3-24, 4-7

(

Ĺ

tripod, 3-23 turning pin, 3-46 turning plate, 3-47, 3-48 turning point, 3-46 umbrella, duties of person holding umbrella, 3-8 unit: four-person, 3-48 chief, 3-7, 3-8 U.S. Geolgical Survey, 1-2, 2-40 valley crossing: see river crossing verification data, 5-24 vertical control data, 2-10, 2-11, 2-12, 2-13, 2-14, 2-15 water level station, 2-21, 2-39 wind: eliminating effect of, 3-50 strength, 3-50, 3-51 witness post: establishment, 2-25 ( (

;

i

Ć

-



National Geodetic Survey motorized leveling system.



ſ

(

C